· _ ·

i

SHATIBI'S PHILOSOPHY OF

ISLAMIC LAW

SHATIBI'S PHILOSOPHY OF ISLAMIC LAW

a :

An Analytical Study of Shātibi's Concept of <u>maşlaha</u> in Relation to His Doctrine of <u>maqāşid al-sharīca</u> with Particular Reference to the Problem of the Adaptability of Islamic Legal Theory to Social Change

> A Dissertation Presented to the Faculty of Graduate Studies and Research McGill University

In Partial Fulfilment of the Requirements for the Degree of Doctor of Philosophy

by

Muhammad Khālid Mas<ūd

Institute of Islamic Studies McGill University Montreal March 1973

ABSTRACT

Author	Muḥammad Khālid Mas<ūd
Title	Shāṭibī's Philosophy of Islamic Law: An analytical study of Shāṭibī's concept of <u>maşlaḥa</u> in relation to his doctrine of <u>maqāṣid al-Sharī'a</u> with particular reference to the problem of the adaptability of Islamic legal theory to social change.
Department	Institute of Islamic Studies
Degree	Ph.D.

This thesis studies Shāţibī's (d. 790/1388) frequently quoted yet little explored and often misunderstood concept of <u>maşlaha</u>. The thesis argues that Shāţibī's doctrine, that the protection of the <u>maşlaha</u> of men is the main objective of Islamic law, was a product of the grave need of his time to adapt Islamic legal theory to new social conditions. Certain theological and moral considerations had limited the validity of <u>maşlaha</u> as a principle of legal reasoning. After an analysis of such considerations, Shāţibī proposed <u>maşlaha</u> as the most fundamental source of Islamic law. Shāţibī was, however, reluctant to accept the logical conclusions of his argument and let his definition of <u>ta^cabbud</u> be ambiguous.

The study suggests that this doctrine could have led Islamic legal philosophy to a positive outlook in separating legal obligation from theological and moral ones if the analysis were carried on further to refine the ambiguities remaining in the doctrine.

Ŧ

ACKNOWLEDGEMENTS

I would like, initially, to thank, in connection with this work, two institutes. Firstly, the Institute of Islamic Research, Islamabad, Pakistan, where I received my initial training in research on Islamic law, and which granted me six years leave to complete my studies in this field at McGill University, Montreal. Secondly, the Institute of Islamic Studies, here, where I received the academic training as well as the financial aid which has enabled me to produce this work. I am indebted to the Institute and in particular to its director, Professor Charles J. Adams, who, in many ways, made the pursuit of these studies as well as the completion of this work possible.

During the initial stage of the formulation of the problem of the dissertation, on various occasions, I was in touch with Professors J. Schacht, H. Toledano, F. Rahman and J. Van Ess whose valuable comments and suggestions were very helpful in clarifying some of the issues under consideration. At the Institute of Islamic Studies, I have had the benefit of constant consultation and guidance from Professors H. Landolt, M. Muhaqqiq, D. P. Little, N. Shehaby, N. Berkes and T. Izutsu whose comments gave me new insights into the problems of the dissertation. The painstaking supervision, constructive criticism and constant encouragement of Professor Adams considerably facilitated the early completion of this work.

I must also express my gratitude to Mr. Muzaffar ^CAli and his staff at the Institute's

Library, especially Miss Salwa Ferahian, who have procured all the necessary material from the libraries of the Universities of New York, Princeton, Harvard, California, Ottawa, Colorado, Congress and elsewhere. I am particularly grate-ful to the Yale University Library for providing me with a microfilm of the manuscript of [Rāzī's <u>Al-Maḥşūl</u>.

To many colleagues in the Institute whose comments and suggestions have helped me in many ways, I am truly grateful. I should like to mention particularly ^CAzīm Nanji, Ella N. Hann, Linda S. Northrup and Murtadā al-Naqīb in this connection; and also Gerhard Bowering, Manuel Ruiz and Rusen Sezer for helping to translate some of the materials from German, Spanish, Italian and Turkish.

Finally, my thanks are also due to Mrs. Norma Babikian who undertook the tedious task of typing the first draft, and to Mrs. Ivy Ward for the final typing.

TABLE OF CONTENTS

4.15

-5

. ,

Acknowledgemen	ts	īv		
Transliteration Table vii				
CHAPTER 1	THE PROBLEM Recent studies on Shāṭibi on <u>maṣlaḥa</u>	۱		
11	ISLAMIC LEGAL THEORY AND SOCIAL CHANGE The Concept of Law The Historical Nature of Islamic Law Islamic Law in Practice The Question of Method Key Terms The Framework of Discussion	43		
111	SOCIAL CHANGES IN FOURTEENTH CENTURY GRANADA 1) Political Developments 2) Socio-religious Developments 3) Economic Developments 4) Legal Developments	81		
IV	SHĀŢIBĪ: HIS LIFE AND WORKS His Works Sources Shāțibī's Life Disputations His Works	165		
V	SOCIAL CHANGE AND LEGAL THEORY Shāṭibi's Fatāwā: Exgesis Theological Matters Rituals and Worship Family Property Taxes Contracts Procedure	209		
VI	THE CONCEPT OF MASLAHA BEFORE SHĀŢIBĪ Early Period — Juwaynī — Baṣrī — Ghazālī — Rāzī — Ibn Abd al-Salām — Ibn Taymiyya — Ibn Qayyim — Tawfī	246		
VII	SHĀŢIBĪ'S DOCTRINE OF MAQĀṢID AL-SHARĨ'A 1) <u>Maşlaha</u> the First maqşid of Sharī'a 2) An analysis of the terms dalāla and ma'nā 3) An analysis of the term <u>taklīf</u> : Legal Obligation and Physical capability of Mukallaf 4) An analysis of the term ta'abbud: Distinction between legal and moral and religious obligation 5) <u>Mukallaf's maqşid</u> in legal obligation: An analysis of the term <u>niyya</u>	280		
VII	I CONTINUITY AND CHANGE	364		
IX	Summary and Conclusion	403		
APPENDICES	A. A Summary-view of the Contents of <u>al-Muwāfaqāt</u> Bal-l ^c tişām	419 429		
BIBLIOGRAP	ΗΥ	431		

Page

N.

TRANSLITERATION TABLE

• • •

-

1

>	\$	ġ	حث			
b	ب	ţ	d.			
t	ت	Z •	ظر			
th	ث	¢	٤			
i	<u>ی</u>	gh	e e			
ķ	ε	f	ف			
kh	ż	q	ق			
d	د	k	لع			
dh	٤	l	J			
r	ر	m	٢			
z	;	n	Ċ			
s	س	w	و			
sh	in	h	t			
s.	مى	У	ڍ			
al- and 'l) ال Omitted in front of proper namesa (-at in costruct state) -						
VOWELS						
ې	أى تار قار آ،					
	short 二 a L u — i					
doubled						

iyy (final form بَيْہِ تو ق uww (final form ت dipthongs aw i ay

CHAPTER I

THE PROBLEM

In recent years, a number of Muslim countries' governments have adopted policies to initiate the process of social change and modernization of various institutions, perhaps the most basic of which is law. Such attempts have been supported by Muslim modernism -a movement born out of the contact of the world of Islam with Western Civilization -- which aimed at adapting Islam to modern conditions by renovating various medieval institutions. Various segments of the Muslim people opposed modernization of Islam claiming that the teachings of Islamic law did not allow any change. The modernists, on the other hand, have consistently maintained that Islamic law is adaptable to social change.

The above controversy has brought to the fore the problem of the adaptability of Islamic law which has been so widely discussed, yet remains debatable. The problem has been generally formulated in the form of the following question: Is Islamic law immutable or is it adaptable to the extent that the change and modernization sought can be pursued under its aegis? Broadly speaking, there have been two points of view in answer to this question. One view, which is shared by a large number of Islamicists such as C. S. Hurgronje and J. Schacht, and by most of the traditional Muslim jurists, maintains that in its concept, and according to the nature of its development and methodology, Islamic law is immutable and hence not adaptable to social changes. A second view, which is upheld by a few experts on Islamic law such as Linant de Bellefonds and by the majority of Muslim reformists and jurists such as Subhī Maḥmaṣānī, contends that such legal principles as the consideration of <u>maṣlaḥa</u> (roughly translated, human good), the flexibility of Islamic law in practice and the emphasis on <u>ijtihād</u> (independent legal reasoning) sufficiently demonstrate that Islamic law is adaptable to social change.

Recent studies have touched almost all the aspects of the problem of the adaptability of Islamic law. Nevertheless, the matter still remains confused, if not unattended. A clear analysis should not aim to settle the debate in favour of one or the other view. In fact, the continuation of the debate points out, at least, the fact that elements of both adaptability and immutability exist in Islamic law. An attempt to decide in favour of one side, especially if it be motivated by dogmatic, political, or apologetic considerations, may only further the confusion.

Before any general conclusions be drawn regarding the adaptability of Islamic law, the problem requires a great deal of spade work. For a clear analysis to be achieved, the primary task is to study the various aspects and levels of this problem which should be distinguished sharply from one another and yet be studied in con-

junction with one another. Furthermore, since Islamic theory has developed through the writings of various jurists having different historical backgrounds, the problem of adaptability requires to be studied in specific reference to individual jurists in their historical settings.

In a general sense, recently the problem of adaptability has been studied in reference to the following three aspects: the concept of the nature of Islamic law, its history and its methodology. Generally no distinction has been maintained among these three aspects; conclusions obtained from the analysis of the history of Islamic law, for instance, have been read into the concept of its nature and <u>vice versa</u>. It is not possible to deal with all three of these aspects in one dissertation. The present treatise will study only the concept of the nature of Islamic law in reference to the problem of the adaptability. This choice is valid because a discussion of the adaptability of Islamic law, even in relation to the aspects of history and methodology, leads back to the concept of the nature of Islamic law.

An analysis of the concept of the nature of Islamic law in the above context requires a detailed study of the essential ideas in Islamic legal theory, especially those pertinent to the question of adaptability. <u>Maslaha</u> is one such idea. This concept is of fundamental significance to the proponents of the adaptability view. They argue that Islamic law aims at the <u>masalih</u> (plural of <u>maslaha</u>) of man, hence logically, it should welcome any social change that serves this purpose. Furthermore, with such an objective in view Islamic law cannot be rigid and inert in regard to social change. Among the very few jurists who treated the concept of <u>maslaha</u> as an independent principle of legal theory, Abū Ishāq Ibrāhīm b. Mūsā Shātibī (d. 790/1388) made one of the more significant contributions. In his <u>al-Muwāfaqāt</u>, Shātibī presented a doctrine of <u>maqāsid al-sharī a</u> (the purpose or ends of law) which comprises an exposition of the various aspects of the concept of <u>maslaha</u> as a principle of legal theory. Shātibī is therefore a valid choice for a study the requirements of which we have discussed above.

The choice of Shāṭibī is further prompted by the fact that in their support of the adaptability-view, it is largely Shāṭibī upon whose arguments the modern reformists have relied.

In fact, Shāţibī is one of the jurists to whom modern writers on <u>uşūl al-fiqh</u> (Islamic legal theory) owe their greatest debts. His books <u>al-Muwāfaqāt</u> and <u>al-1^ctişām</u> are so extensively used by modern authors on Islamic law that one cannot doubt the significance of Shāţibī's contribution to the modernists' conception of Islamic law. In particular, the concept of <u>maşlaha</u>, which is one of the essential elements of the modernist conception¹, is derived from Shāṯibī to a great extent.

In Egypt, Muḥammad 'Abduh used to advise his students and scholars to study <u>al-Muwäfaqāt</u> in order to understand the real nature of "Islamic law making" (<u>al-tashrīć al-Islāmī</u>)². In Pakistan, Abu'l A'lā Mawdūdī, in his programme to introduce Islamic law in Pakistan, recommends the translation of <u>al-Muwāfaqāt</u>, among other books on the philosophy of law, into national languages, "so that cur legal experts may acquire a deep insight into and gain a correct understanding of the spirit of Islamic <u>Fiqh</u>".³

Since its first publication in 1884 in Tunis, five editions of <u>al-Muwāfaqāt</u> have so far appeared⁴, all edited and annotated by well-known scholars such as Mūsā Jār Allāh⁵, Muḥammad al-Khiḍr Ḥusayn and ^CAbd Allāh Darāz.

Evidence for the merit of Shāṭibī's lengthy work may be drawn not merely from the number of editions it has undergone but, more importantly, from the rank which <u>al-Muwāfaqāt</u> soon attained among Muslim works on law. It came to transcend even the limits of the Sunnī schools of law. With few exceptions, modern Muslim authors on legal matters or theories invariably refer to Shāṭibī as an authority; often they draw heavily upon his doctrines. The works of the following eminent authors adequately substantiate this point: Abū Zahra, Maʿrūf ADawālībī, Muḥammad Iqbāl, Muḥammad Khuḍrī, Yūsuf Mūsā, Muṣṭafā Zarqā', Abū Sima and Abū ʿAbd Allāh ʿUmar.⁶

Furthermore, some modern authors grant to Shāţibī a rank as high as that of a <u>mujaddid</u> (religious reformer believed to appear at each turn of a century). Rashīd Ridā counts him among the <u>mujaddids</u> of the 8th/14th century and regards his contribution as equal to that of Ibn Khaldūn⁷. Fādil Ibn 'Āshūr⁸ and 'Abd al-Muta al al-Ṣa 'īdī⁹ also express the same opinion, but Ṣa 'īdī adds that Shāţibī ranks alongside^{Al}_JShāfi 'ī in significance, because his exposition of the goal and spirit of Islamic law made it possible for Islamic law to escape the impasse into which the strict adherence to the limits defined by Shāfi 'ī in <u>uşūl al-fiqh</u> had led. The present dissertation, therefore, proposes to seek an answer to the following question: What are Shātibī's views on the adaptability of Islamic legal theory to social changes? To answer this question it undertakes to study Shātibī's doctrine of maqāsid al-sharī'a which emerged as an exposition of his concept of maslaha.

Before an analysis of Shāṭibī's views can be launched, the question requires a proper understanding of the following terms: 'adaptability', 'Islamic legal theory', 'social changes', and '<u>maşlaḥa</u>'. A separate chapter (Chapter Two) is set apart to develop a full analysis of these terms.

The question also requires a proper appreciation of the present status of studies on the question for which the answer is sought. Shāṭibī's concept of <u>maşlaḥa</u> as a principle of adaptability in Islamic law has not been yet directly investigated. Scholars have, however, sometimes casually, and sometimes specifically, expressed their views regarding Shāṭibī's thought and the concept of <u>maşlaḥa</u>. Although the discussions on these two matters are not necessarily connected with each other, yet since they sometimes bear upon one another, a survey of previous studies on both matters is necessary.

1

It is curious to note, however, that despite the prominence and the wide acknowledgement of Shātibi's contribution, no exclusive study is yet known to have been made either on the life and works of Shātibi or on his legal thought.

ଙ୍କ ମୁନ Two reasons can, perhaps, be suggested for the absence of such studies. One, as ^CAbd Allāh Darāz, the commentator on <u>al-Muwāfaqāt</u>, remarks is the fact that Shāṭibā's thought is too difficult and too complex to be easily penetrated. ¹⁰ Margoliouth also referred to confusion and subtlety in Shāṭibā's views. ¹¹ This complexity is not due to any abstractness of thought or to any bizarreness in his style or in his choice of words. His style is lucid, and his discussion is systematic and clear. The difficulty in understanding Shāṭibā lies, rather, in the fact that a study of his thought demands not only a sufficient knowledge of the development of <u>usūl al-fiqh</u> in prior times, but also a fair acquaintance with the development of the doctrines of <u>fiqh</u>, theology, philosophy and mysticism and more importantly, there is required a knowledge of the political, economic and social developments in Shāṭibā's time as well. Without this background knowledge his views appear to be contradictory, vague or abstract, and hence difficult to follow.

The second reason has to do with a generally skeptical attitude of Islamicists towards studies of Islamic doctrines on the formal level. Gibb, for example, warns against studying theological doctrines arguing that since Islamic theology is always forced into extreme positions, it exhibits a predilection for words and form.Islamic doctrine thus presents an outer formulation rather than an inner function or reality. Hence islamic doctrines, taken literally, are not of much help in understanding the inner religious attitudes of Muslims.¹²

Such ware ings discouraged any study of Islamic doctrines per se, including legal

theory. In his discussion of Islamic legal theory, S. Hurgronje dismissed a discussion of the question of whether all acts are forbidden by nature and only those specified by the divine law may be allowed...saying that "these and similar questions may be of importance to the Imām al-Haramayn, but they do not help us to a correct understanding of Islam".¹³ Chehata maintains that us<u>ul al-fiqh</u> was born independently of <u>fiqh</u> and developed without in-fluencing the science of law or being influenced by it.¹⁴ Schacht concludes that the theory of us<u>ul al-fiqh</u> is of little direct importance for the positive doctrines of the schools of law.¹⁵ Why, if a study of <u>usul al-fiqh</u> has no relevance to the understanding of <u>fiqh</u> and is merely a consideration of words and forms if studied per se, should it be studied at all?

The first printing of <u>al-Muwāfaqāt</u> in 1884, though diligently edited, did not contain any commentary or analysis of the work. In 1909 the second printing appeared with an introduction in Turkish by Mūsā Jār Allāh. In 1913 some extracts from another of Shāṭibī's work - <u>Al-I'tişām</u>, appeared in the Cairo journal Al-Manār. These extracts stirred the interest of scholars in Shāṭibī.

In 1916, Ignaz Goldziher, in his translation and critical study of Ghazāli's work <u>Fadā'ih al-Bātiniyya</u> made use of these extracts to compare Shātibi with Ghazāli. Although Goldziher's knowledge about Shātibi was limited (only the above-mentioned extracts and Turkish introduction were available to him), and although he confused <u>al-I'tişām</u> with <u>al-Muwāfaqāt</u> (as he insisted on identifying these extracts as part of <u>al-Muwāfaqāt</u>), yet he is the first scholar

who tried to place Shāṭibi's thought into a historical perspective. While comparing similarities in the treatment of the Bāṭinīs by Ghazālī and Shāṭibī, he found them identical. He, therefore, drew a general conclusion that "in many ways Shāṭibī is through and through penetrated with the ideas of Ghazālī".¹⁶

Rashid Ridā, himself a warrior against <u>bid^ca</u>, was largely responsible for creating the image of Shāţibi as a crusader against <u>bid^ca</u>. After publishing the abovementioned extracts from Shāṯibi on <u>bid^ca</u> in <u>Al-Manār</u>, he edited and published Shātibi's al-l^ctişām in 1913/1914.

This theme was further stressed by Rashid Ridā in the biography of Muḥammad Abduh which was published in 1931.¹⁷

<u>Al-I'tişām</u> was reviewed by D. S. Margoliouth in <u>The Journal of the Royal</u> <u>Asiatic Society</u> in 1916. In his very brief review Margolicuth described the work as "occupied with juristic subtleties and distinctions which become more and more confused towards the end of the book".¹⁸ Thus implicitly he rejected the work as not worthy of further scholarly attention.

It was about the same time that, on the suggestion of Goldziher, a notice on Shātibī was included in Brockelmann's Supplement. This notice was based entirely on the information provided by Goldziher. Some of the factual mistakes by Goldziher were also included without correction.¹⁹

About the same time, Muḥammad Khuḍrī (d. 1927) a teacher at Gordon Law College in the Sudan at that time, published his <u>Usul al-fiqh</u>, for which, in many ways, he drew heavily upon Shātibī's al-Muwāfaqāt. He also disclosed in the preface that it was on the suggestion of Muhammad ⁽Abduh that he had turned to <u>AI-Muwāfaqāt</u> for understanding the nature of Islamic legislation (<u>asrār</u> aI-tashrī⁽ aI-Islāmī).²⁰

While Rashid Ridā's interpretation of Shātibi depended solely upon <u>Al-I^ctişām</u>, that of Khudri was entirely shaped by <u>al-Muwāfaqāt</u>. In the former he appears as a crusader against <u>bid^ca</u>, while in the latter as a philosopher-jurist. Khudri argued that Shātibi's teachings present the real spirit of Islamic law which had been forgotten by medieval jurists.

Muḥammad Ḥasan al-Ḥajawī, in his lectures on the history of Islamic Jurisprudence, given in 1918, did not differ greatly from Ridā and Khudrī in presenting Shāţibī's image as a reformer.²¹ But believing in this image he misread Shāṯibī's concept of obedience (<u>Ta'abbud</u>). Ḥajawī, in his lectures, maintained that the flexibility of Islamic law was lost in later Islamic history as the jurist extended <u>ta'abbud</u> even to those acts which fell under the category of <u>mu'āmalā</u>t. A certain correspondent, in order to refute Ḫajawī's argument, quoted Shāṯibī on the point that the consideration of <u>Ta'abbud</u> is inevitable in <u>mu'āmalā</u>t as well. To reject this argument, Ḫajawī referred to^cIzz al-Dīn ^cAbd al-Salām in his support and judged the quotation from Shāṯibī in this light as he said:

"This (statement of ^Clzz al-Din) is opposite to your quotation from the author of <u>al-Muwāfaqāt</u> where he narrowed (the application of <u>maslaha</u>) by imposing <u>ta'abbud</u> on all categories of acts. But he (Shāţibi) did not support his contention with any proof."²²

We shall deal with this point in detail later in the course of our discussion. It must, however, be pointed out at the moment that such an interpretation of Shāṭibī's view of ta `abbud is quite misleading. Shāṭibī certainly differentiated between two kinds of obligations, those which are absolute and not subject to changes, consisting of <u>`ibādāt</u>, and those which are relative and subject to changes, consisting of <u>`adāt</u> which include <u>muʿāmalāt</u>. The former are <u>taʿabbudī</u> and the latter <u>maşlaḥī</u>. This distinction is <u>maintained</u> on the first level, i.e. that of <u>shāri</u>, though both may become <u>taʿabbudī</u> on the second level, i.e. that of mukallaf.

In 1941 Lopez-Ortiz published his invaluable detailed study of certain fatāwā (responsa) given by Granadian jurists of the fourteenth century.²³ Among these Shāţibī's fatāwā were also included. This study provides us with the actual historical context against which Shāţibī's doctrine can be studied. Although Ortiz's study is not concerned with the philosophical questions of a legal theory and thus does not include <u>al-Muwāfaqāt</u>, yet he confirms that in his fatāwā, Shāţibī relied on the notions of tashīl (facilitation) and <u>lstişlāh</u>. Shāţibī defended custom against the rules of figh. It is also significant to note that Ortiz was impressed by the deep insight that Shāţibī showed into the economics of the society.

Since Ortiz was concerned with Shāṭibī's <u>fatāwā</u> and not with his philosophy of law, one might be misled by his remarks to conclude that Shāṭibī's reference to tashīl and Istişlāḥ was a measure of expediency. Such an understanding of

7

L F

Shāţibī is misleading because the principle of <u>maşlaha</u> in Shāţibī's legal philosophy is a basic concept; not an expedient method of legal reasoning. Lopez-Ortiz's remarks may, however, be best understood in reference to Shāţibī's doctrine of the Ends of the law.

In 1916, in his study on Mālik b. Anas, Abū Zahra observed that on the problem of <u>Umum</u> and <u>Khuşuş</u> (the general and specific use of words/expressions in general or specific meanings), Shāţibī forsook the Mālikī stand in favour of that of the Hanafīs.²⁴

We need not go into the details of Abū Zahra's explanation. It is sufficient to note that Hanafis and Mālikis disagree on the definition as well as on the legal value of ' \overline{amm} and \underline{khass} . According to Abū Zahra, for Hanafis, the ' \overline{amm} is rated as definite or absolute ($\underline{qat'i}$); while for Mālikis it is only probable (\underline{zanni}). Both schools, however, agree that a $\underline{qat'i}$ can be particularized ($\underline{takhsis}$) only by another $\underline{qat'i}$; consequently, Hanafis reject particularization of the Qur'āni commands by those $\underline{ahadith}$ which have only probable (\underline{zanni}) authenticity. Mālikis, on the other hand, accept such particularizations, because, for them it is only the \underline{khass} in the Qur'ān, which is $\underline{qat'i}$, and which cannot be particularized by a probable \underline{hadith} .

In 1951 ^cAbd al-Muta^cāl al-Ṣa^cīdī observed that in matters of dogma, Shāṭibī was rigid like other jurists such as Ibn Taymiyya and Ibn Qayyim. Ṣa^cīdī refers to Shāṭibī's view of <u>ribāṭ</u> to uphold his point. He states that Shāṭibī declared that to dwell in a <u>ribāṭ</u> for the sake of <u>cibāda</u> only, constitutes bid^ca.²⁵ Fādil ibn ^cĀshūr credited Shāţibī with providing an escape from the impasse that Islamic jurisprudence faced in the fourteenth century. Furthermore, according to lbn ^cĀshūr, Shāţibī rejected the differentiation between theoretical and practical religion - a distinction which was maintained by a number of theologians and philosophers.²⁶ Shāţibī insisted on inity in the essence of religion. That is why he also opposed the practice of classification of bid^ca into praise-worthy and condemnable.

Ibn ^CAshūr argues that Shāţibī and Ibn Lubb had fundamental differences on the legally binding nature of certain acts. By binding nature Ibn ^CAshūr means the process of acts being or becoming <u>Cibādāt</u> or religious obligations. Ibn ^CAshūr concludes that Shāţibī's concept of religion was more comprehensive than most other jurists because he considered the payment of taxes to government to be a religious duty, thus regarding them as <u>Cibādāt</u>.

In a study of transactions in the <u>Sharīća</u>, made in 1955, Şubhī Mahmaṣānī was struck by the modern subjective approach adapted by Shāṭibī in torts.²⁷ Shāṭibī maintained that if an act which is legal in itself is committed with the sole intent of inflicting injury upon others, it is legally prohibited and must be prevented. Maḥmaṣānī observed that this subjective approach is quite modern as it directs itself to the intent of the person exercising the right. This approach also stands in contrast to the traditional objective approach as formulated in the Majalla.

It was, perhaps, this finding that led Mahmasani to a further study of Shātibi.

In his lectures in 1962 he was more enthusiastic and admiring of Shāṭibī. Maḥmaṣānī believes that the foundations of the modern renaissance in Islamic legal thought were laid in the fourteenth century by the Muslim jurists who wrote on the methodology and the ends of Islamic law. In these writings they were the precursors of western legal philosophers such as Montesquieu who taught that the evolution of law takes place conditioned by local, temporal and situational changes. Maḥmaṣānī recalls Shihāb al-Dīn, Garāfī, ^c lzz al Dīn ʿAbd al-Salām, Ibn Qayyim and Shāṭibī as such philosophers of law. Among them, however, he singles out Shāṭibī for the finest exposition of Islamic jurisprudence and philosophy of law.²⁸

Since 1960 references to Shāṭibī have become so frequent in almost every work on Islamic law that a complete account of them is quite impossible. Further, such an account would not be relevant to our purposes because few of these works aim to study Shāṭibī's philosophy. We will, however, take note of some of the more important recent studies.

In his <u>Islamic Methodology in History</u> published in 1965, Fazlur Rahman discusses Shāṭibī's views in detail to a far greater extent than earlier scholars. Rahman, in his <u>Islam</u>, considered <u>al-Muwāfaqāt</u> as a work on the philosophy of law and jurisprudence.²⁹ Rahman has observed Shāṭibī's views on the following points: his concept of knowledge, his views about the role of human reason in acquiring knowledge, and his views on <u>ijtihād</u> and <u>taqlīd</u>. Since these points have been studied mainly in reference to Shāṭibī's epistemology, Rahman finds Shāṭibī little different from other Muslim thinkers in whose

4

т

arguments Rahman sees a "patent denial of faith in the intellectual and moral powers of man".³⁰

Rahman, however, is reluctant to carry the above conclusion to Shāţibī's legal thinking. He observes that although Shāţibī "categorically denies that reason has any primary role in law-making or even in the formulation of the moral imperatives, yet he (Shāţibī) himself has exercised a great deal of rational power in fixing the "goals of <u>Sharī'</u>⁴."³¹

He also finds an implicit confusion in Shāṭibī's statement about <u>iļtihād</u> that it "is the necessary duty of a Muslim" along with the stipulation that the <u>iļtihād</u> should not contradict the objectives of <u>Sharīća</u>. Rahman finds this stipulation inconsistent because the objectives of the lawgiver cannot be formulated without the operation of <u>iļtihād</u>.³²

The above observations have significant implications for our question. Goldziher's suggestion of Ghazālī's thorough influence on Shāţibī may mean Shāţibī's acceptance of Ghazālī's view on <u>maşlaha</u>. Ghazālī is known to have rejected <u>maşlaha</u> <u>mursala</u>. His influence on Shāţibī would thus amount to the rejection of the adaptability of Islamic legal theory to social changes. Shāţibī's opposition of <u>bid'a</u> (innovation), as presented by Rashīd Ridā and others, signifies that he believed in the immutability of Islamic law.

^{Al}Hajawi, ^{Al}Sa^cidi and Rahman conclude that Shāṭibi was rigid, conservative and opposed to rational interpretation of legal matters. In other words, they are suggesting that Shāṭibi would oppose the accommodation of Islamic law to social changes.

15

On the other hand, Khudri, Mahmasāni and Lopez-Ortiz have observed that his views in legal matters were flexible and that he preferred the consideration of human need to the hardship incurred in following the legal texts to the very letter.

Ibn ^cĀshūr's interpretation of Shāţibī's concept of <u>dīn</u> (religion) and Hajawī's conclusion about Shāţibī's conception of <u>ta</u> abbud (obedience) have very serious implications for Shāţibī's view of the adaptability of Islamic law. An allcomprehensive concept of religion and an all-inclusive conception of obedience suggest that Shāţibī views every legal and social change from the angle of "religion" and "obedience" which only imposes limits on the adaptability of Islamic legal theory to social changes.

Abū Zahra's comment has obvious methodological implications. It suggests that Qur'ān and Ḥadīth, being $\underline{qat}^{c}i$ (definitive), cannot be particularized by what is \underline{zanni} (probable). In the light of this view, if the concept of <u>maslaha</u> is employed to particularize the Qur'ān and <u>Hadīth</u>, it must either be invalid, or the concept of <u>maslaha</u> must be proven to be as definitive as the Qur'ān and Hadīth.

To conclude, the scholars are disagreed as to the assessment of Shātibī's contribution to Islamic jurisprudence. Their disagreement stems from their differences of understanding and interpretation of Shātibī's basic terms such as <u>bid'a</u>, <u>ta'abbud</u>, <u>dīn</u>, etc. As is shown in the following chapters, the above terms are related to Shātibī's conception of <u>maşlaha</u> which is the basis of his doctrine of <u>maqāşid al-sharī'a</u>, and they cannot be properly understood in isolation from this conception. In fact, the confusion and sometimes the misinterpretation of these terms is caused by disregarding their relationship to Shāţibī's conception of maslaḥa.

Recently there have been a few significant studies of the concept of <u>maşlaha</u>, but they have not paid due attention to Shāţibī. The present study contends that by failing to take into account Shāţibī's conception of <u>maşlaha</u>, recent studies have fallen short in bringing out the real significance of the concept of <u>maşlaha</u> as a principle of adaptability of Islamic law.

Critical remarks and studies of the concept of <u>maşlaha</u> in western scholarship started to appear in the nineteenth century. This was the period when Muslim reformists of Islamic law had revived the interest in the concept of <u>maslaha</u> as a principle of change. Before going into the details of the criticism of the concept of <u>maslaha</u> by modern scholars, some remarks about the emergence of the concept of maslaha among modern Muslim scholars must be made.

In 1857 the <u>'Ahd al-Amān</u>, a document of reforms in Tunisian law, was issued. This document later became the fundamental legal instrument in the 1860 Constitution - "the first Constitution to be issued in any Muslim country in modern times". ³³ In its preamble, <u>maslaha</u> was referred to as the principle of interpretation of law: "God ... who has given justice as a guarantee of the preservation of order in this world, and has given the revelation of law in accordance with human interests $/ masalih_2 / ... ³⁴$ The document then expounded the following three principles as the components of the concept of <u>maslaha</u>: "liberty, security, equality".³⁵ In 1867 Khayr al-Din Pasha, in his <u>Aqwam al-masālik</u>, reaffirmed that the principle of <u>maşlaha</u> must be the supreme guide of the government.³⁶ He found this principle extremely significant as it could be used to justify a change of institutions in the interest of the public as well as to condemn a change when it opposed public interest.³⁷

In 1899, in his speech on the reforms in the court systems in Egypt and Sudan, Muḥammad ^CAbduh also stressed the use of <u>maslaha</u> as a guiding principle in law making.³⁸ J. Schacht has argued that the principle of <u>maslaha</u>, according to ^CAbduh, was preferable to the literal application of Islamic law.³⁹ Henry Laoust has also observed that the principle of <u>maslaha</u> was one of the two ideas on the basis of which ^CAbduh considered Islam to be superior to Christianity. It is because of this principle that Islam has a sense of reality more developed than Christianity.⁴⁰

It is to be noted that Khayr al-Din and ^CAbduh both referred to <u>maşlaha</u> as a principle of interpretation of law – and as such a principle of change, dynamism and adaptability.

The same theme, in varying versions, has been repeated by a large number of modern Muslim scholars of Islamic law. Among them the following are notable illustrations: Rashīd Ridā, Şubhī Mahmasānī, [°]Abd al-Razzāq[°], [°]Sanhūrī, Ma[°]rūf [°]Dawālibī, Mustafā, [°]Shalabī, [°]Abd al-Wahhāb, [°]Khallāf, Muḥammad, [°]Khuḍrī and Mustafā Abū Zayd.⁴¹

In 1906, <u>Al-Manār</u> published Najm al-Dīn Tawfi's treatise on <u>masāli</u>h. Tawfi, α

Hanbali jurist, sometimes also considered a Shi^ci, represented radical views on <u>maşlaha</u>. For example, he held that the principle of <u>maşlaha</u> could even restrict (takhsis) the application of <u>ijmā</u> as well as that of the Qur² an and <u>Sunna</u> if the latter were harmful to human interests. This publication raised a strong reaction among the conservative group of scholars in Egypt. Consequently Jawfi as well as the concept of <u>maşlaha</u> was bitterly opposed. Only to illustrate this opposition, we quote Zāhid al-Kawtharī as follows:

"One of their spurious methods in attempting to change the <u>Shar</u> in accordance with their desires is to state that 'the basic principle of legislation in such matters as relating to transactions among men is the principle of <u>maslaha</u>; if the text (<u>nass</u>) opposes this <u>maslaha</u>, the text should be abandoned and <u>maslaha</u> should be followed'. What an evil to utter such statements, and to make it a basis for the construction of a new <u>Shar</u>. This is nothing but an attempt to violate divine law (<u>al-Shar</u> al-llāhī) in order to permit in the name of <u>maslaha</u>, what the <u>Shar</u> has forbidden. Ask this libertine (<u>al-fājir</u>) what is this <u>maslaha</u> on which you want to construct your law?... The first person to open this gate of evil...was Najm Tawfi Hanbalī... No Muslim has ever uttered such a statement...This is a naked heresy. Who-ever listens to such talk, he partakes of nothing of knowledge or religion." 42

Kawtharī did not deny that the <u>Shar</u> took into consideration the interests and good of the people, but what is good and what is bad can only be known through revelation. <u>Maslaha</u> as an independent principle for the interpretation of law has, therefore, no validity whatsoever.

Kawthari's criticism of <u>maşlaha</u> is typical of the traditional view of the concept. To him <u>maşlaha</u> is arbitrary and merely personal. In fact this fear of arbitrariness arising from regard for human interests, and resulting in violation of divine law is a familiar feature in the history of the development of Islamic legal theory. <u>Maslaha</u> and similar legal principles which were employed in favour of the adaptability of Islamic law, were opposed on the same grounds. In this sense the concept of <u>maslaha</u> has always been connected with the question of adaptability.

As a problem of legal theory the question of adaptability to social change has been a controversial one in the history of <u>usul al-fiqh</u>. The <u>qadis</u> in the early courts of law, particularly in the Umawi period, relied mostly on <u>ra²y</u> (considered opinion). The use of <u>ra²y</u> generally amounted to a general consideration of human needs. The <u>ra²y</u> was, thus, a method that kept the then institution of law adaptable to social change.

There, however, existed an opposition to ra'y among the scholars who specialized in <u>hadith</u> and in local practice. These scholars considered the use of ra'y as an arbitrary and therefore unreliable method of making a decision. The diversity of laws that resulted from the exercise of ra'y by the <u>gadis</u> in various cities increased the number of opponents to the use of ra'y.

The general attitude of the <u>Hadith</u> group was to adhere strictly to the Qur'ān and <u>Sunna</u> (of the Prophet as well as that of his companions), and thus to reject any idea of the adaptability of Islamic law. This attitude was motivated by the religious apprehension of distortion of Islamic tradition by the use of $ra^{2}y$. This attitude was, however, impossible to be maintained in view of the enormous degree of social changes that had taken place in Islamic society by the end of the eighth century.

The literal provisions of the Qur'an and <u>Sunna</u> were insufficient to accommodate the growing number of social changes. Even the method of extending these provisions by accepting the <u>ijmā</u> (consensus) of the past generation of scholars on certain matters failed to meet the demand of accommodation. The need to accommodate the changes could not be denied, but how to extend the limited legal provisions to adapt to these changes.

The method of <u>qiyās</u> (analogy) developed as an answer to the need of the adaptability of Islamic law. Even among the <u>Hadīth</u> group, a large number of scholars recognized this need and accepted the validity of the method of <u>qiyās</u> for this purpose. The religious and theological implications of the attitude of the <u>Hadīth</u> group, however, spelled out the same fear of arbitrariness for the method of <u>qiyās</u> as it had done for <u>ra'y</u>. Consequently, the <u>Zāhirīs</u> who still adhered to the older trend of rejecting anything beyond the literal provisions, opposed the use of <u>qiyās</u> and departed from the mainstream of the <u>Hadīth</u> group.

Although initially a method of adaptability, yet in reaction to the Zāhirī and similar criticism, <u>qiyās</u> was soon ushered into the protection of strict formality. It was sought as a foolproof corrective of the method of <u>ra'y</u>. To remove the fear of arbitrariness, <u>qiyās</u> was connected with the "sources" -- the Qur'ān and Hadīth. The appeal of this method was so strong that it overshadowed its opposition as well as any other methodological developments in Islamic legal theory.

Nevertheless, the method of <u>ra'y</u> was not completely swept away by <u>qiyās</u>. Trends similar to the use of <u>ra'y</u> survived in the form of principles such as <u>istihsān</u>, <u>istişlāḥ</u>, <u>darūra</u>, <u>munāsaba</u>, etc. Incidentally</u>, rules derived from these principles constitute the basis of a considerable part of Islamic law (<u>fiqh</u>) -probably even more than those based on <u>qiyās</u>.

The <u>qiyās</u> which was the basis of a number of other methods in extending or adapting legal doctrines to social changes, was itself hampered by at least two limitations. One was the attitude of formalism which required that in order to be conclusive, the analogy must be derived explicitly from the original sources $\int_{\Lambda}^{he} Qur^3 \bar{a}n$, Sunna or $\int_{\Lambda}^{he} qur generations$. In other words, the basis of analogy must be explicitly expressed as a "cause" or "reason" for the original ruling. This attitude discouraged the use of implicit cause in the original ruling as a basis of analogy. Also this attitude required reference to specific original rulings rather than encouraging the search for, and the application of, general principles or the intent and "spirit" of the law in original rulings.

The second limitation, which further strengthened the attitude of formalism, stemmed from the theological view of the problem of causality in reference to the attributes of God. The Ash aris opposed the idea of there being any causality behind God's actions and speech. Thus, since the command of God, being one of His acts, cannot have any cause or motive, the entire method of <u>aiyās</u> came to be suspected as wrongly or arbitrarily seeking to appoint causes for the commands of God. One of the major consequences of the above limitations - i.e. formalism and me denial of causality - was that the discussion on the problem of social change and legal theory became essentially a question of "sources of law".

To escape this dilemma, the Zāhiris rejected giyas altogether. The Shaficis, who did not entirely reject qiyas, imposed limitations on its application. They rejected any method of reasoning or any form of qiyās which was not linked with certain specific rulings in the Qur'ān or Sunna. Nevertheless, they could not deny the occurrence of social changes, nor could they refuse to accept these changes in practice. They had, therefore, to adopt methods such as istishab (presumption of continuity of a legal evidence) to justify these changes. Hanafis and Malikis employed certain methods which did not strictly adhere to the requirements of the theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of law, principally $\int_{\Lambda}^{\pi e}$ theory of the sources of methods are istihsan (to decide in favour of something which is considered hasan, good, by the jurist, over against the conclusion that may have been reached by giyās), attributed to Hanafis, and istislāh (to decide in favour of something because it is considered maslaha, more beneficial, than any alternative rule decided on another basis.) These methods were not accepted by all the schools. Yet the concepts of istihsan and istislah have in common the consideration of human good. Invariably the underlying principle in the reasoning of these schools was to favour the adaptability of Islamic law.

In order to render the concept of <u>maslaha</u> suited to their legal philosophy, the Shāfi^cī jurists imposed upon this concept the approach of the "sources of law".

They divided <u>maşlaha</u> into categories according to its basis in the sources. If <u>maşlaha</u> accorded with the sources, it was not disputable, since it was somehow justifiable as a method of <u>qiyās</u>, when it was literally derived from the sources. The only category which was questionable was that which was not based on the sources. This category was called <u>maşlaha</u> <u>mursala</u>. Naturally for the Shāfi^ci jurists the only discussion of <u>maşlaha</u> that mattered was discussion of <u>maşlaha</u> <u>mursala</u>. This view predominated in other schools, and even Mālikis eventually accepted it.

The significant consequence of the above categorization of <u>maşlaha</u> was that the original idea of <u>maşlaha</u> as a principal independent source came to be disregarded, and <u>istişlāh</u> came to be equated with <u>maşlaha</u> <u>mursala</u>. Recent studies related to maşlaha also betray this traditional outlook.

A brief survey of the significant observations on the concept of <u>maslaha</u> made in recent studies, to which we now turn, illustrates the above comments.

Ignaz Goldziher compared <u>Istişlāņ</u> with <u>istiņsān</u> saying that the latter is'a Hanafī principle according to which a decision reached by analogy can be dismissed when the legislator finds that this decision opposes a certain matter which he believes is useful', which is to say that <u>Istiņsān</u> removes the rigidity of law depending upon the discretion of an individual jurist. <u>Istişlāņ</u>, on the other hand, depends upon a rather objective method; it removes the rigidity of law in consideration of general human "interests" (<u>maslaņa</u>). He also suggests that <u>istişlāņ</u> partially resembles the Roman legal principle of <u>utilitum publicum</u> as well as Rabbinic law.⁴³

∠4

Recent studies on <u>maşlaha</u> can be generally divided into two groups. First, there are studies dealing with <u>maşlaha</u> <u>mursala</u> or <u>istişlāh</u> and, second, those dealing with <u>maşlaha</u> as such. The focus in the first group of studies is not on <u>maşlaha</u> proper but on <u>maşlaha</u> <u>mursala</u>, yet it is significant to note that for them <u>lstişlāh</u> is in no way different from maşlaha <u>mursala</u>.

N. P. Aghnides and G. H. Bousqet also refer to <u>Istişlāh</u> in the same sense. Aghnides defines it as a principle that consists in prohibiting or permitting a thing because it serves a useful purpose, although there is no express evidence in the revealed sources to support such action.⁴⁴ Bousqet's definition is as follows: "<u>Istislāh</u> consists of discarding by exceptional disposition the rules deduced by <u>qiyās</u> in cases where the application of general rules would lead to illogical, unjust and undesirable results.⁴⁵

J. Schacht's treatment of <u>maşlaha</u> is not much different from that of the above scholars. He described <u>lstişlāh</u> as a special form of analogy, or rather a type of <u>istihsān</u> used by early Mālikī scholars and which later came to be called <u>istişlāh</u>.⁴⁶ Schacht re-emphasized that <u>istişlāh</u> is identical with the Roman legal principles of <u>utilitas publica</u> which characterises <u>jus</u> <u>honorarium</u>.⁴⁷

R. Paret also finds <u>istişlāh</u> to be connected with <u>istihsān</u>, but the latter is more limited and definite as it replaces a general principle such as "finding good", by a rather specific principle, such as "according to the demand of human welfare (<u>maşlaha</u>)". <u>Maşlaha</u> thus is the material principle underlying <u>istişlāh</u> which is a method of reasoning. In actual details where Paret traces the history of <u>istişlāh</u>, he specifically refers to <u>maşlaha</u> <u>mursala</u>, rather than <u>maşlaha</u> as such. This is why he finds nothing of much importance after Ghazālī had theorized about <u>istişlāh</u>. His references to <u>uşūl</u> works are confined to the discussions of <u>maşlaha</u> <u>mursala</u>.⁴⁸

Analysing the treatment of <u>maslaha</u> by modern Muslim scholars such as ^cAbduh and others, A. Hourani criticised their use of <u>maslaha</u> in a utilitarian sense. He argued that such an interpretation of <u>maslaha</u> was not justified; "for the traditional thought, <u>maslaha</u> had been a subordinate principle, a guide in the process of reasoning by analogy rather than a substitute for it."⁴⁹

Von Grunebaum, in his study of the concept of reason in Muslim ethics, concluded that <u>istişlāḥ</u> (the public interest) is unmistakably one point at which human "reason" is permitted to impinge on traditional or systematic considerations that would normally be viewed as the determining factors of <u>Shar^cī</u> developments. ⁵⁰

Although all of the above opinions agree in regarding <u>maşlaha</u> as a principle that removes rigidity and suggests adaptability to changes based on human needs, yet according to the same writers, its function is restricted to exceptional cases or to use a special form of analogy. The reasons for such a limited view of <u>maşlaha</u> in these studies is either that they have studied only <u>maşlaha</u> <u>mursala</u> to the exclusion of other aspects of <u>maşlaha</u> or that they have equated <u>maşlaha</u> <u>mursala</u> with maşlaha.

There are, however, a few studies which evince an integral approach to the problem of maslaha or which study the concept of maslaha as such. Among such studies, the following four are relevent to our point. G. F. Hourani has examined <u>maşlaha</u> as an ethical concept. M. H. Kerr and Sa^cid Ramadān^{*l*}_ABūţī have analyzed it in particular reference to legal theory. E. Tyan has studied it as a principle of methodology.

Tyan describes <u>maşlaha</u> as 'general interest', 'social utility' and 'good' and has defined <u>istişlāh</u> as "to recognize a rule as useful".⁵¹ He distinguishes two conceptions of <u>istişlāh</u>. In the original conception of <u>istişlāh</u>, the interests (<u>maşālih</u>) were divided into three categories according to its recognition by the law, the last category being <u>maşālih</u> <u>mursala</u>. The directing principles in this kind of research consisted essentially in considering the elements of social utility (<u>maşlaha</u>) and of convenience (<u>munāsaba</u>). The speculation according to this conception of istişlāḥ remains within the limits of law.

The other conception of <u>istişlāh</u> is more extensive.⁵² According to this conception of <u>istişlāh</u> "it may be admitted that this method can be employed not only in relation to matters which are not regulated by the precise texts of law, but also in those matters which have been subjects to such regulations, so much so that it be legitimate to make it prevail over precise rules or over conflicting or contradicting regulations, provided that, in the final analysis, they (the rules derived from this method of reasoning) remain in conformity with the objectives of law, i.e. they accord with the above-mentioned five major interests (religion, physicial integrity, descendance, patrimony and mental faculty)".⁵³

Tyan, thus, concluded that istislah "is a method of interpreting already existing

rules by disengaging the spirit of these rules from the letter; exceptions and extensions are reached which command practical utility and correspond to the fundamental goals of the law". 54

G. F. Hourani has studied <u>maslaha</u> as an ethical concept in medi eval Islam.⁵⁵

He observes that there were two theories of value in medieval Islam: one, that of objectivism, i.e. that the value has real existence; the second theory of value was that of theistic subjectivism, that the values are determined by the will of God. The theory of objectivism was expounded by the Mu[<]tazila; the idea of rational good was called by them hasan or maslaha. The theory of theistic subjectivism was maintained by the Ashraris. The opposition of these two theories manifested itself in the field of figh also. Jurists in the early period used certain methods which did not correspond with "theistic subjectivism". Principles such as istiķsān and istislāķ tended rather towards "objectivism". The ethical basis of these principles, however, remained unarticulated. The Mustazili theory of rational good that there is an objective good including a real public interest (mașlața) and a real justice (cadl), and that they could be recognized by human reason] could have provided a basis to support the above principles. But the theory of objectivism was superseded by theistic subjectivism. Why? Hourani suggests that, apart from religious and political factors that prevented objectivism from being adopted by the lawyers, the Mu^ctazili theory of objectivism had its own deficiencies. First it could not show how moral judgment operates. Second, it could not fill up the theoretical gap between means (moral and legal acts)

and the end (the eternal happiness, which is the happiness in the world hereafter for Muslims).

On the other hand, the theory of theistic subjectivism corresponded with Shāfi^cī and Zāhirī views on legal reasoning, which opposed the use of ra'y and any judgment independent of the revelation. Shāfi^cīs denied the objective value of idle fancy, <u>zann</u> and <u>hawā</u>. Theologically also the theory of objectivism appeared to curtail the omnipotence and omniscience of God, which the theory of theistic subjectivism promoted.

Hourani's study of <u>maslaha</u>, in reference to history, is confined to the early period of Islamic tradition. Because of this limitation he could not take into consideration the development in the treatment of <u>maslaha</u> by later <u>usuliyyin</u> such as Shāţibi. In fact, Hourani's criticism of objectivism is mainly ethical. The three deficiencies that he ascribed to <u>maslaha</u> as an objective value are not found in Shāṯibī's conception of <u>maslaha</u> as a legal value.

Muḥammad Saʿid Ramaḍān Būṭi presented his doctorial dissertation, Dawābit al-Maslaḥa fī al-Sharīʿa al-Islāmiyya, at Azhar University in 1965. In his introduction to the published edition of this dissertation Būṭi explains that the Orientalists, whom he regards as new crusaders against Islam, have adopted a new measure to destroy Islam. They are urging Muslims to open the gate of ijtihād, and to accomplish this end they refer to the concept of <u>maslaḥa</u> as the fundamental principle of <u>Sharīʿa</u>. He is, however, convinced that the real motive behind this proposal for <u>ijtihād</u> is the destruction of Islam. He admits
that the gate of ijtihad has never been closed and that the lawgiver has given full consideration to the principle of maslaha, but this principle has always been restricted with a number of qualifications.⁵⁶ After a detailed analysis of etymology and the theory of the concept of maslaha, he deduces the qualifications which the traditional jurists have suggested in the application of this principle. He also compares this concept with the concept of 'utility' and 'pleasure' in the philosophies of Stuart Mill and J. Bentham. He concludes that Maslaha in its unqualified sense is identical with the above concepts which he considers as purely hedonistic. The qualified concept of maslaha, however, contradistinguishes itself from utility and pleasure as it takes into consideration the following three characteristics. First, it is not limited to this world only but equally includes the hereafter. Second, the Islamic value of good is not material. Third, the consideration of religion dominates other considerations.⁵⁷ He has thus concluded that if these and other qualifications are disregarded "and the term maslaha alone is held up as a light post and a criterion, then upon my life ! an ijtihad such as that will descend upon Muslims from all sides. (To prove such terrifying results after opening the gate of ijtihad) it suffices to observe the evil that brings the laws of Shari'a out of the fortress of texts into the open, exposed to desires and arbitrary opinions that deceive (us) behind the name of maslaha and manfa^ca."⁵⁸

In fact Būti's view of maslaha is no different from that of Zāhid^{Al}Kawthari. If Al-Būti's expositions of maslaha and its qualifications are accepted, maslaha, as a matter of fact, becomes superfluous as a legal concept. The consideration of maslaha by the <u>Shāri</u>^c, then only means that <u>maslaha</u> is what the <u>Shāri</u>^c commands. In other words, <u>maşlaha</u> has no objective value. This is a logical conclusion from ${}_{A}^{B\bar{u}}\bar{t}\bar{i}$'s view of Islamic law according to which he rejects a distinction between this world and the hereafter. He does not separate <u>mu'āmalāt</u> from <u>cibādāt</u> but rather considers the former part of the latter. He does not distinguish between <u>huqūq Allāh</u> and <u>huqūq al-cibād</u>. In fact, his conception of Islamic law is that of <u>tacabbud</u> (mere obedience). On all these points he is in disagreement even with the jurists who employ the concept of <u>maşlaha</u> in reference to human needs. His disagreement becomes particularly evident if his conclusions are compared with Shāțibī's conception of <u>maşlaha</u>.

Ale-Būți has frequently referred to Shāţibi in his dissertation, but these references are selective and often out of the context. ^{Ale}_ABūţi's study fails to bring out the real significance of the concept of <u>maşlaḥa</u> mainly because he has not given full consideration to the proponents of this concept such as Shāṯibi.

The same deficiency is found in M. Kerr's study of <u>maşlaha</u>, which also offers a detailed analysis of the concept. Examining Rashid Ridā's legal doctrines, Kerr observed that the logical conclusion of Ridā's arguments for the use of <u>maşlaha</u> would be that it is something equal to natural law and that <u>istişlāh</u> does not depend on the texts and <u>qiyās</u>. Such conclusions, however, are not spelled out by Ridā himself.⁵⁹ Why? According to Kerr, the failure to spell out the full implications of the argument has to do with the theological nature of Islamic law which influences even <u>maşlaha</u>, theoretically the most liberal principle of legal interpretation in Islamic jurisprudence. The theological foundations of Islamic law insist on minimizing the part of human reason in the formulation of law.

Before he goes into a detailed analysis of the concept of <u>maslaha</u> in traditional jurisprudence, Kerr clarifies two general aspects of Islamic law which, in turn, affect the function of <u>maslaha</u>. Firstly, Islamic law has its basis in revelation and thus is an expression of the will of God. Kerr refers to the theological differences between Ash^carīs and Mu^ctazilīs about the will of God. In contrast to the Mu^ctazila, Ash^carī denied freedom in man's acts. Consequently, the intellectual spirit and methods of Islamic jurisprudence "could not entirely escape the influence of the law's theological underpinnings, which proclaimed that reason is essentially irrelevant to the substance, determination and obligatory character of moral principies."⁶¹

N,

The second aspect that affected <u>maslaha</u> was the emphasis on <u>qiyās</u>. According to Kerr, the method of <u>qiyās</u> itself is a means of protecting the authority of revelation.⁶² In fact, the term <u>cilla</u> in jurisprudence is not applied in the usual sense of cause and effect. <u>cilla</u> is not a value judgment, but only the attribute or the characteristics of the matter under consideration that gives rise to the judgment.⁶³ Further, the limitations of the means to identify <u>cilla</u> are also confined to the use of indication within the text. <u>Munāsaba</u> (suitability) is the only means that goes beyond the indication of the texts. Kerr finds even <u>munāsaba</u> to be a conservative, circumscribed and timid acknowledgement of the place of social utility (<u>maslaḥa</u>) in God's commands. In fact, he concludes, in final analysis even <u>munāsaba</u> is subordinate to the indications of the text..⁶⁴

4

J۲

Kerr, thus treats <u>maşlaha</u> as one of the aspects of <u>munāsaba</u>. He also divided <u>maşlaha</u> on the basis of the conformity to sources, and thus it is only <u>maşlaha</u> <u>mursala</u> which really needs to be discussed. According to him <u>maşlaha</u> <u>mursala</u> is a form of <u>qiyās</u>, because whereas <u>qiyās</u> looks for <u>cilla</u>, <u>maşlaha</u> <u>mursala</u> seeks <u>hikma</u>, a more general <u>cilla</u>. Kerr concludes that because it is not based on a specific <u>cilla</u>, <u>istişlāh</u> has been a subsidiary and occasional technique of disputed validity.⁶⁵

In a final analysis Kerr comes to equate maslaha with maslaha mursala.

"The maslaha is therefore a more specific term for hikma and since it is known in each case not by direct indication in the textual source but by the jurist's own judgment, it is a maslaha mursala." 66

To sum up, Kerr also confines <u>maşlaha</u> to its correspondence with the textual sources. It is noteworthy that Kerr, in his discussion, refers to such jurists as \mathcal{H} -Ghazālī and $\bigwedge_{\Lambda}^{a\ell}$ addities and $\bigwedge_{\Lambda}^{a\ell}$ darāfī who viewed <u>maşlaha</u> in the above terms. He also discusses the views of Ibn Taymiyya, Ibn Qayyim and Tawfī whom he chose as proponents of the validity of <u>maşlaha</u> as a principle of legal interpretation, but these jurists, too, regarded <u>maşlaha</u> as subordinate to the textual sources and <u>qiyās</u>. The consideration of <u>maşlaha</u>, according to them, would prevail over the texts and qiyās only when the latter are harmful to obey.

Kerr has not taken into account jurists, such as Shāṭibī, who favour <u>maṣlaḥa</u> as an independent legal principle. The significance of studying Shāṭibī's views is evident from Tyan's analysis of <u>istişlāḥ</u> which gives a more integral picture of maṣlaḥa. The absence of Shāṭibī from Kerr's analysis of <u>maşlaḥa</u> is regrettable. According to Kerr, Rashīd Riḍā, whose views led Kerr to study the concept of <u>maşlaḥa</u> in detail, characterizes Shāṭibī "as exceptionally outspoken in his defence of <u>istişlā</u>ḥ."⁶⁷

> It comes as a further surprise that Shāţibi was not only disregarded but also suffered a sort of indifference when Kerr, probably following Paret, ⁶⁸ confused him with Abū'l Qāsim^{AL}_NShāţibi.⁶⁹

To sum up, the present studies on <u>maşlaha</u> generally present an unbalanced analysis of this concept. They have failed to see the real significance of this principle as it was conceived and employed by those jurists who viewed it as an independent principle. A study of Shāṭibī's concept of <u>maşlaha</u>, as already indicated by Tyan, can fill this gap.

The present study, therefore, aims to investigate Shāṭibī's concept of <u>maşlaḥa</u> as a principle permitting the adaptability of Islamic law. The enquiry is concerned mainly with the theoretical aspect of the question of adaptability. Nevertheless, Shāṭibī did not conceive <u>maşlaḥa</u> in isolation from the social realities of his time, and his doctrine of the <u>maqāşid</u> was actually an attempt to answer the questions that arose in relation to <u>maşlaḥa</u>. The various developments in the society in which Shāṭibī lived and the actual legal problems with which Shāṭibī was faced must be studied, as they not only explain the cause of Shāṭibī's interest in this problem, but also clarify the nature of the answer that Shāṭibī was seeking in the concept of maşlaḥa. A brief outline of the dissertation may clarify its scope, method and its limitations. The first two chapters, the present and the following, introduce the problem and explain the basic terms of the enquiry. The next three chapters deal with the social milieu in which Shāţibī expounded this doctrine. Chapter three outlines the social developments in fourteenth century Granadian society in general. Chapter four deals with the available information about Shāţibī's life and his academic disputations with other scholars, and reviews his works. Chapter five analyses his <u>fatāwā</u> to point out the actual legal problems which he faced. It also investigates whether or not Shāţibī showed willingness to adapt to social changes. The following four chapters deal with Shāţibī's concept of <u>maşlaḥa</u> itself. Chapter six outlines the major problems that arose in traditional Muslim jurisprudence regarding the concept of <u>maşlaḥa</u>. Chapter seven analyses Shāţibī's doctrine of the <u>maqāşid</u> and reconstructs an understanding of his concept of <u>maşlaḥa</u>. Chapter eight examines Shāţibī's views on social and legal change, and attempts to define his basic terms in regard to the problem of adaptability.

For the purpose of limiting the work the focus of the study falls on Shāṭibī's doctrine of <u>maqāşid al-sharī'a</u>. The main sources of Shāṭibī's thought for this dissertation are thus <u>Al-Muwāfaqāt</u>, a part of which is devoted to the exposition of the above doctrine, and <u>Al-I'tişām</u>. Among Shāṭibī's works, only these two are relevant to our study. Detail of the publication of these two works are noted in Chapter four.

The sources of information on the history of the period and on Shātibi's life have been reviewed in the beginning of the relevant chapter or in the first footnote. Regarding transliteration and translation, a transliteration table is attached. The Arabic affix <u>al-</u> with proper names is omitted. The exact translations of the terms are not attempted. An explanatory English translation is given in parenthesis when the Arabic term is used first. At later points the Arabic term itself is normally used without repeating the translation. English equivalent of Arabic terms are used only when they are usually so accepted; in case of doubt, the Arabic term is supplied in parenthesis.

References to sources in footnotes are usually short. Fuller bibliographical information can be found in the section on Bibliography.

NOTES: CHAPTER I

- Malcolm H. Kerr, Islamic Reform, The Political and Legal Theories of Muhammad Abduh and Rashid Ridā. (California: University of California, 1960), p.55 ("The element in their jurisprudence which the modernists have particularly seized upon as the basis for dynamism and humanism is the notion of maşlaha (welfare, benefit, utility".)
- 2. Muḥammad Khuḍrī, Uṣūl al-Fiqh, (Cairo: Maṭbaʿ al-Istiqāma, 1938), p.11, relates that when he was appointed to teach Islamic law in Gordon Law College in Sudan, he planned to write a book on uṣūl. He discussed with ʿAbduh, when the latter was visiting Sudan, who then recommended al-Muwāfaqāt to be used as a basis for the studies on the asrār al-tashrīć al-Islāmī.

Also Muḥammad ʿAbd Allāh Darāz, in his introduction to al-Muwāfaqāt, Vol. I (Maṭbaʿ Tijāriya, N.D.) relates that "often we listened to the recommendation (waṣiyya) of the late (Shaykh ʿAbduh) to the students to obtain this book, and I was ever anxious to fulfil his will." (p.12-13).

- 3. Abū'l A^clā Mawdūdī, Islamic Law and Constitution (Lahore: Islamic publications, 1960), p.113–114.
- 4. See below Chapter IV, p. 197ff.

بر ار

- 5. Musa JarAllah's edition, in spite of patient search, is not available to the writer of this dissertation.
- 6. Abū Zahra, Mālik (Cairo: Maţba'Aḥmad ^CAlī, First edition published in 1946); ^ADawālibī, <u>Al-Madkhal ilā ^cilm usul al-Fiqh</u> (Beyrouth: Dār al-'ilm l'il mala' in 1965), especially pp.433-41.
 Iqbāl, <u>Reconstruction of Religious Thought in Islam</u> (Lahore: Ashraf, 1965), pp.169-174.
 Yūsuf Mūsā, <u>Al-Madkhal li dirāsāt Fiqh al-Islāmī</u> (Cairo: 1961), pp.196-202.
 Muştafā Zarqā, <u>Al-Madkhal l'il-Fiqh al-Islāmī</u>, Vol. 1 (Dimasha: Maţba' Jāmi'a, Dimasha, 1961), pp. 62 ff., particularly p.68, no. 1.
 Abū Sima: and others, <u>Madkhal al-Fiqh al-Islāmī</u> (Cairo: Jāmi'a, Azhar, 1965), pp.97-100, 119-131 and 163-165.
 ^CAbd Allāh ^cUmar, <u>Sullam al-Wuşūl li ^cilm al-Uşūl</u> (Cairo: Dār al-Ma'ārif, 1956), pp.73-76, 233-239.
- Ridā, Ta'rīkh al-Ustādh al-Imām al-Shaykh Muhammad ^CAbduh, Vol. I, (Cairo: Dār al-Manār, 1350/1931), p. Jīm and his introduction to al-I^ctiṣām, (Cairo: Tijāriya, circ. 1332/1913) p. Jīm.

- Fādil ibn ʿĀshūr, Aʿlām al-Fikr al-Islāmī fī Tārīkh al-Maghrib al-ʿArabī (Tūnis: Maktabat al-Najāḥ, n.d.), pp.71-72.
- Abd al-Muta al al-Sa^cidi, Al-Mujaddidun fi'l Islam, (Cairo: Maktabat al-Adab, n.d.), pp.294-296, 309.
- 10. ^cAbd Allah Darāz, <u>op</u>. <u>cit</u>. p.11-12.
- 11. D. S. Margoliouth, "Recent Arabic Literature", in Journal of Royal Asiatic Society (London: 1916), pp.397–98.
- 12. H. A. R. Gibb, Studies on the Civilization of Islam (Boston: Beacon, 1968), pp. 205-206.

٨

- 13. Hurgronje, Selected Works of C. Snouck Hurgronje, (Ed.) G. H. Bousquet and J. Schacht, (Leiden: Brill, 1957), p.287.
- 14. Chafique Chehata, "Logique Juridique et droit musulman" in <u>Studia Islamica</u> Vol. XXIII (1965), p.16.
- 15. J. Schacht, "Fikh" in <u>El²</u>, Vol. 11, p.890.
- Ignaz Goldziher, <u>Streitschrift des Gazali gegen die Batinijja-Sekte</u> (Leiden: 1916), pp.32–34.
- 17. See above note 7, and below n.19.
- 18. See note: II.
- 19. Goldziher op.cit. p.33. He says, "Meine Kenntnis von den Beziehungen dieses Sātibī auf das Mustazhirī grunden sich auf Auszüge, die aus dem Kapitel (الاستسام) der <u>Muwāfakāt</u> in der arabischen Zeitschrift <u>al-Manār</u> unlängst".

Goldziher wrote thusly despite the fact that the title of the book in this issue was specifically mentioned as "Kitāb al-I^ctiṣām" cf. Al-Manār, Vol. XVII (1913-14) pp.54-63, 273-293.

Elsewhere as well, on the basis of this conclusion, Goldziher commenting on "Alī al-Qārī's mention of a book on "al-Hawādith al-Bida[<]" by Shātibī, again suggested, "Es ist jedoch möglich, dass damit ein Kapitel der Muwāfakāt gemeint sei." (op.cit. note: 1).

This confusion was further carried on by Brockelmann, <u>Geschichte der</u> Arabischen Litteratur, S II, p.375, where he, probably basing his information on Goldziher's remarks, wrongly describes these excerpts in <u>Al-Manār</u> as "Auszüge" of al-Muwāfaqāt.

- See above note: 2. Khudri's Uşūl...is available to us in its third edition, 1938 and second edition, 1933. The date of the first edition could not be known.
- Muḥammad B. Ḥasan al-Ḥajawi, <u>Al-Fikr al-Sāmī fi Tārīkh al-Fiqh al-Islāmī</u> (Ribāț, 1345 A.H.), Vol. IV, p.82.
- 22. Ibid., p.17. For the complete discussion see pp.304-17.
- 23. Lopez-Ortiz, J. "Fatwas granadinas de los siglos XIV y XV", <u>Al-Andalus</u>, Vol. VI (1941), pp. 73-127.
- 24. Abū Zahra, op. cit. p.267
- 25. Al Şa^cidi, op. cit. p.311. Şa^cidi's views on Shāţibi's rigidity are mostly unfounded. For instance his claim that Shāţibi believed in only one saved sect, Salaf Ahl al-Sunna, and none else, is just opposite to what Shāţibi maintained. cf. Al-I^ctişām, Vol. II, pp.217ff.
- 26. Ibn Áshūr, op. cit. pp.73-74.
- Subhi Mahmasāni, "Transactions in the Shari^ca", in Law and the Middle East, Vol. I, ed. M. Khadduri and H. J. Liebesny, (Washington, 1955), p.187.
- 28. Şubhī Mahmaşānī, Muqaddima fī Ihyā^{, C}Ulūm al-Sharī<u>ča</u> (Bayrūt: Dār al-^cilm li'l Mala[,] in, 1962), pp. 22, 63, 65-67.
- 29. Rahman, Islam (London: Weidenfeld, 1966), p. 108.
- 30. Rahman, Islamic Methodology in History (Karachi: Central Institute of Islamic Research, 1965), p.154.
- 31. Ibid., pp.133-4.
- 32. Ibid., p.160.
- 33. Albert Hourani, Arabic Thought in the Liberal Age, 1798–1939 (London: Oxford, 1962), p.65.
- Muḥammad Bayrām, Safwat al-I^ctibār, Vol. II, p.11, vide Hourani, op.cit. p.64.
- 35. Ibid.
- 36. Ibid, p.92.

- 37. Ibid, p.93.
- Mufti Muhammad Abduh, "Taqrir Mufti al-Diyar al-Misriyya fi Islāh al-Mahākim al-Shar'iyya", Al-Manār, Vol. 11, 1899, p.761.
- Schacht, "Muḥammad ʿAbduh", Shorter Encyclopaedia of Islam, (Leiden: Brill, 1961), p.406.
- 40. Henry Laoust, <u>Le Califat dans la doctrine de Rašid Ridâ</u>, (Beyrouth, 1938), p.273, n.61.
- The following works by the authors enumerated below stress the dynamism 41. of maslaha as a principle of interpretation of Islamic law: Rashid Ridā, Yusr al-Islām (Cairo: Nahda, 1956), pp.72-75. Şubhī Mahmaşānī, Falsafat al-tashrī^cfī al-Islām, transl. by F. J. Ziyadah (Leiden: Brill, 1961). (Arabic Ed. Bayrut, 1952), pp.130-133, 205. 'Abd al-Razzāq Šanhūri, "Wujūb Tanqih al-Qānūn al-Madani al-Misri", in Majalla al-Qānūn wa I-lqtisād, Vol. 6 (1963) 3-144. vide Kerr, op. cit. Marrūf, Dawālībī, Al-Madkhal ilā 'ilm usul al-fiqh (Bayrūt: Dār al-'ilm l'il-Mala'in, 1965), pp.442.450. Mustafā,Šhalabi, Taʻlil al-Aḥkām, ʿArḍ wa Taḥlil li Ṭariqat al-taʻlil wa tatawwurātihā fi (usūr al-ijtihād wa al-taqlīd (Cairo: Azhar, 1949), pp.278-384. Abd al-Wahhabakkhallaf, Masadir al-Tashric al-Islami fi ma la nassa fihi, (Cairo: Dār al-Kutub al- Arabi, 1955), pp.70-122. Mustafā Zayd, Al-Maslaha fi al-Tashrić al-Islāmi wa Najm al-Din Tawfi (Cairo: Dar al-Fikr al-Arabi, 1954). Muḥammad Khuḍrī, Usul al-fiqh, (Cairo: Mustafā Muḥammad, 1933).
 - Zāhid al-Kawthari, <u>"Maqālāt al-Kawthari</u>, posthumous publication in 1372 A.H., as quoted by Mustafā Zayd, <u>op</u>. <u>cit</u>. pp.164-66.
 - 43. Goldziher, "Das Prinzip des Istishāb in der Muhammedanischen Gesetzwissenschaft", Wiener Zeitschrift fur die Kunde des Morgenlandes. Vol. I (1887), pp. 128-236; vide Summary in French by G. H. Bousquet, in Arabica, VII (1960), pp. 12-15. The points of resemblance with Roman and Rabbinic law are not elaborated but most probably Goldziher refers to certain areas of flexibility in contrast to the strict application of law.
 - 44. N.P. Aghnides, <u>Mohammedan Theories of Finance</u> (London: Longman, 1916), p.102.

- 45. G. H. Bousquet, <u>Précis de droit musulman</u> (Alger: Maison du Livre, 1947), p.37.
- 46. J. Schacht, The Origins of Muhammadan Jurisprudence, (Oxford: Clarendon, 1959), p. 11, n. 1.
- 47. <u>Ibid</u>, "Classicisme, traditionalisme et ankylose dans la loi religieuse de l'islam", in <u>Classicisme et déclin culturel dans l'histoire de l'Islam</u>, (ed.) Brunschvig et als (Paris: 1957), p.158, note no. 4.
- 48. Rudi Paret, "Istihsān and Istislāh", in Shorter Encyclopaedia of Islam, p. 185.
- 49. Albert Hourani, op. cit. p.234.
- 50. G. E. Von Grunebaum, "Concept and Function of Reason in Islamic Ethics", Oriens, Vol. 15 (1962), p.15.
- E. Tyan, "Méthodologie et sources du droit en Islam", <u>Studia Islamica</u>, Vol. X (1959), p.97.
- 52. Ibid.
- 53. Ibid.
- 54. Ibid., p.98.
- 55. G. F. Hourani, "Two Theories of Value in Medieval Islam", <u>Muslim World</u>, Vol. L (1960), pp.269–278.
- Sa^cid Ramadān al-Būţi, <u>Dawābiţ al-Maşlaha fi al-Shari^ca al-Islāmiyya</u>, (Damascus: Umawiya, 1966-67), pp.12-14.
- 57. Ibid, pp.23-60.
- 58. lbid., p.414.
- 59. M. Kerr, "Rashid Rida and Legal Reform", Muslim World, 1960, p.108.
- 60. Ibid., Islamic Reform, p.56.
- 61. lbid., p.60.
- 62. lbid., p.77.
- 63. lbid., p.67.

- 64. lbid., p.73.
- 65. lbid., p.76.
- 66. Ibid., p.81.

- 67. Ibid. Muslim World, op. cit. p. 107.
- 68. R. Paret, op. cit. mentions 1194 as the year of Shāţibī's (d. 1390) death, which is in fact the year in which Abū'l Qāsim Muḥammad b. Fīra al-Shāţibī died. See Brockelmann, G.A.L.(Brill, 1943), p.520, and F. Krenkow, "Shāţibī", in E.I.¹, Vol. IV, p.337-388. Unfortunately, the same error is repeated in the heavily edited and revised version of Paret's article in the Urdu Encyclopaedia of Islam. Urdū dāira maʿārif Islāmiyya (Lahore: Danishgāh Panjāb), Vol. II, (pp.586-591), p.589.
- 69. Kerr, op.cit. p.194, referring to Ridā's mention of Shāţibī rightly quotes his year of death, but in his index on p.247 identifies him as Abū'l Qāsim b. Firra ash-Shātibī.

CHAPTER 1

ISLAMIC LEGAL THEORY AND SOCIAL CHANGE

The relationship between legal theory and social change is one of the basic problems of the philosophy of law.¹ Law, which by its nature, tends to be unchanging, always faces the challenge of social changes which demand adaptability from law. Most often the impact of social change is so profound that it affects legal concepts as well as institutions and thus creates a need for a fresh philosophy of law. The problem of social change and legal theory is of particular significance in case of Islamic law. Islamic law is generally defined as religious, sacred and hence immutable. How does such a law face the challenge of change?

Shāţibī sought an answer to such a challenge in the principle of <u>maslaha</u>. A discussion of Shāţibī's answer is, however, unwarranted unless we first explain what the immutability of Islamic law means. The present chapter attempts to examine the arguments of recent studies on the immutability of Islamic law. From this examination will be drawn definitions of the key terms in the problem of the present study.

Presuming that the interaction between social change and legal theory must have been at work in Islamic law before Shātibī as well, it may be rightfully suggested that to evaluate Shātibī's contribution to the philosophy of Islamic law his views must be studied in comparison with those of his predecessors. Unfortunately, fulfilment of this task is not possible in view of the present state of scholarship on the philosophy of Islamic law; not only because a general history of Islamic legal philosophy does not exist, but also because very few studies have been made on individual usul works.

Ĭ.

On the other hand, an attempt to establish the views of Shāṭibī's predecessors by surveying the original sources is also beyond the scope of this study. The literature available on <u>uşūl al-fiqh</u>, belonging to the pre-Shāṭibī period is enormous and there is no way to estimate how much more material was lost or not yet discovered. There is, in addition, the problem of the differences in the legal doctrines among various schools of law due to the various theological and philosophical predilections of the <u>uşūl</u> writers. Such extra-legal considerations are reflected in the treatment of legal theory. A survey of the philosophy of law, therefore, would demand an investigation of all these aspects which is impossible within the limited scope of this dissertation.

It is with these limitations in view that in attempting to formulate an understanding of the key terms of the problem of adaptability of Islamic legal theory to social change, this chapter proposes to make an analysis of the findings of recent scholarship on this problem. This choice is made mainly in consideration of the fact that in the modern period (since the beginning of the nineteenth century) the question of the adaptability of legal theory to social change has been asked more pointedly than ever before. Hence, the formulation of the problem can be expected to be clearer than in earlier periods of the history of Islamic law.

It must be stressed, however, at the very outset, that the following is not a report on the present state of scholarship on this problem, and, as such, it does not aim to be exhaustive. What we intend to do in the remaining pages is to establish the prominent land marks of the problem in reference to which Shāţibī's views may be analysed.

In the nineteenth century when most of the Muslim peoples, directly or indirectly, came to be ruled by Western powers, a number of attempts were made to reform the laws of the Muslim peoples. Whether they were attempts to codify or to modify the Muslim laws, the strong religious reaction among the Muslim peoples against such legislative attempts made the reformists aware of the complexities of the problem of change in the Islamic law.

The early colonial policy of non-interference in personal and religious matters, particularly in India, in fact, tended to support the conservatives' view of the immutability of Islamic law.² One of the solutions to avoid interference in personal laws was sought in establishing separate courts for personal and religious matters. This solution required either that these courts should be entrusted entirely to the traditional jurists or that the judges should be assisted by specialists trained in the traditional Muslim laws. The situation led to a series of translations of the traditional texts and their codification along Western patterns. This was the beginning of legislative modernism in Islamic law.

The early legislative modernism, however, added a new dimension to the problem. Most of the translators and jurists were lawyers such as Van Den Berg and M. Morand and their attempts at translations and codifications were meant for judges in modern courts. More significantly, most of them were foreigners and non-Muslims. Perhaps naturally they tended to treat the whole body of Islamic law as though it were Western law. At the extreme of their reform efforts, they excluded from the body of Islamic law what they considered as not belonging to Law. The underlying conception in these attempts was that Islamic law, like other laws, could be changed, reformed and codified by government legislation according to social needs. Confronted with orthodox conservative opposition, these men spelled out their views more explicitly by questionning the idea of the immutability of Islamic law.³

Ņ

This view of Islamic law was strongly criticised by Islamicists, especially by Snouck Hurgronje⁴ and G. Bergsträsser.⁵ Hurgronje pointed out that it was a mistake to treat Islamic law like Western law and that Islamic law was a 'doctrine of duties'. By its nature it was religious law, and as such it was immutable.⁶

Consequently, from that time, as J. Schacht also reported in his lecture on the status of scholarship on Islamic law,⁷ there appeared two approaches to the study of Islamic law: one, that of the lawyers, the other, that of the Islamicists. An implicit controversy between these two approach es continues even today on the problem of legal theory and social change.

In a very broad sense this problem has been formulated by recent scholarship thus:

Is Islamic law immutable, or is it adaptable to social change? Whereas the lawyers have been inclined to regard Islamic law as adaptable to social change, the Islamicists have stressed the immutable character of Islamic law.

The arguments of the advocates of the immutability of Islamic law can be summed up in the following three general statements:

- Islamic law is immutable because the authoritative, divine and absolute concept of law in Islam does not allow change in legal concepts and institutions. As a corollary to this concept, its sanction is divine and hence cannot change.
- Islamic law is immutable because the nature of its origin and its development in its formulative period isolated it from the institutions of legal and social change – the courts and the state.
- Islamic law is immutable as it did not develop an adequate methodology of legal change.

The advocates of the adaptability-view disagree with the above conclusions, yet their arguments also turn around these three aspects of Islamic law: concept, history and methodolgy.

It is, therefore, possible to accept these three aspects as general landmarks in surveying the problem of social change and legal theory. The following discussion is, therefore, arranged according to these three aspects.

The argument that the Islamic concept of law is absolute and authoritative and hence immutable, has been advanced from two points of view. First, with regard to the source of Islamic law, it is contended that the source of Islamic law is the will of God, which is absolute and unchangeable. The second point of view springs from the definition of Islamic law; there it is demonstrated that Islamic law cannot be identified as law in the proper sense, rather it is an ethical or moral system of rules. The first view, thus, treats the problem of the concept of law in terms of the distinction between reason and revelation. The second view deals with it in terms of the distinction between law and morality.

The arguments in regard to the first view take into account two subject matters: i) law and theology and ii) law and epistemology.

J. Schacht has very forcefully argued in his article, "Theology and Law in Islam", that there has always been a close connection between Islamic law and theology; and that certain isolated instances of separatist trends are only accidental. He has demonstrated this connection by the fact that the schools of law and their eponyms showed their interest both in law and in theology.⁸ Further, a certain symbiosis of the schools of law and the schools of theology existed throughout the history of Islamic law.⁹

Malcolm H. Kerr also observes that the concept of Islamic law is very firmly grounded in theology.¹⁰

The connection between law and theology, however, must not be understood in the sense that law was theological so as to be a counterpart of "Divine Law" or "Canon law" as in Christian teachings. "Islam", as Schacht put it, "is a religion of action rather than of belief".¹¹ Hence a "Theology" in the Christian sense could not be conceived of in Islam. The argument asserting the theological foundations of the concept of Islamic law is advanced simply to stress that the law's source is Divine will, and not human reason.

١,

C. H. Toy has put this idea more neatly by comparing the Greek and the Semitic concepts of law. He found that Semites conceived law as absolute, revealed by God; whereas the Greeks worked out the idea of natural law. The absolute law of the Semites is external, imposed on man from without, by God, while the Greek conception is of an inward law which is part of man's nature.¹²

It appears that arguments holding Islamic law to be theologically grounded are advanced in the sense in which Toy speaks about the Semitic concept of absolute law. The evidences that the advocates of the immutability view present to prove their point confirm our observation.

The first evidence they advance concerns the divinity of the sources of Islamic law. It is argued that Islamic law seeks its basis in Divine Revelation through the Prophet; it is embedded in the Qur'ān and <u>Hadīth</u>. Being divine, or divinely inspired these sources are believed to be sacred, final, eternal and hence immutable. It is in this sense that some scholars have understood Islamic law as divine law. Among them N. J. Coulson, ¹³ H. A. R. Gibb, ¹⁴ H. J. Liebesny, ¹⁵ M. Khadduri, ¹⁶ H. Lammens, ¹⁷ G. Makdisi, ¹⁸ and particularly J. N. D. Anderson¹⁹ have expressed this view.

Leon Ostrorog, ²⁰ S. G. V. Fitzgerald²¹ and some others have disagreed with the view of the scholars mentioned above. They argue that the strictly legal materials in these 'revealed' sources are limited and, indeed, negligible. Furthermore, this material is more concerned with the religious and moral teachings than with matters strictly pertinent to law. The whole body of Islamic law, cannot, there-fore, be called revealed and sacred when the amount of legal material existing in the revealed sources is very little.

The second evidence advanced by the advocates of the immutability view takes the question of the sources of law in a more abstract sense. It contends that Islamic law has its source in the Will of God. Since Gibb has expressed this view more succinctly, we quote him as follows:

"The conception of law in Islam is thus authoritarian to the last degree. 'The law, which is the constitution of the Community, cannot be other than the Will of God, revealed through the Prophet'. This is a Semitic form of the principle that 'The will of the soverign is law', since God is the sole Head of the Community and therefore sole Legislator".

The concept of the Will of God has theological implications, which render it entirely absolute and immutable. The reason for this situation Gibb finds in the nature of the development of Muslim theology. Because of its stress on monotheism, Islamic theology refused to admit any limitations whatsoever upon the Power and the Will of God. But the frame of reference of these theological discussions was Aristotelian logic rather than metaphysics. Consequently, the theology was forced into extreme positions; one such position is that there could be no agent of any kind in the universe except God, since the existence of an agent implies the possibility of an action independent of God, and, therefore, a theoretical

na.

limitation upon the absolute power of God.²³ This conclusion was extended even to 'human acts'; man was not considered the free agent of his acts. This, apparently, would also imply a denial of moral and legal responsibility on the part of man. It would also imply that nothing can be qualified as good or bad except in relation to His will, because the Creation would have no intrinsic value. The knowledge of this value can only be had through revelation and not through human reason; leading to the other subject matter of the concept of law, its epistemology.

The arguments in respect to epistemology of Islamic law have referred to two aspects of the problem, a) the possibility and method of knowing the law, and b) the role of human reason.

Gibb has brought these points clearly to the fore. He argues that Islamic law is thought of, not as a product of human intelligence and adaptation to social needs and ideals, but of divine inspiration and hence immutable. The Qur³ an and <u>Hadith</u> are not the basis of Islamic legal speculation but only its sources. The real foundation of the law is to be sought in the attitude of mind which determined the methods of utilizing these sources. The ultimate reason of such a mental attitude is metaphysical; an <u>a priori</u> conviction of the imperfection of human reason and its inability to apprehend by its sole powers the real nature of the good, or indeed, of any reality whatsoever.²⁴

As a corollary of the above concept of the epistemology of law, no primary role is allowed to independent human reason in law making.

-3

Schacht has pointed out that as a consequence of such an epistemological attitude a number of irrational elements have survived in the Islamic law.²⁵ R. Brunschvig also speaks of the irrationality of Islamic law in this special sense.²⁶ G. F. Hourani's distinction of two theories of values in Islam is also concerned with the point we are discussing.²⁷

The concept in Islamic legal theory that implies the employment of reason in knowing and interpreting law is <u>maslaha</u>. In fact, both Grunebaum²⁸ and Hourani²⁹ have classified it as a rational principle. This classification has been, however, disputed by scholars like Schacht.³⁰ Ŋ

The second view, in regard to the concept of Islamic law – dealing with it in terms of the opposition between law and morality – is concerned with its definition. Since law and morality or ethics have a great deal in common, they are often liable to be confused. Hence, any attempt to define law necessarily starts by distinguishing one from the other, law from morality. In defining Islamic law, Islamicists conclude that it is a system of ethical or moral rules. This conclusion must be understood in reference to the separation of law from morality. By defining Islamic law as 'ethics' it is certainly never implied that it is a branch of philosophy; nor is it 'morality' in the sense of having its source in social customs only.

The main aim of the argument in describing Islamic law as ethical law was to refute the modern lawyers' approach to Islamic law as being law in the modern sense. The second aim was to maintain the position that, being a system of ethics,

Islamic law is not capable of change through legislation. Snouck Hurgronje was the first scholar to advance this argument. He defined, in very clear terms, 31 Islamic law as a 'Doctrine of Duties'. Th. W. Juynboll³² and others agreed with Hurgronje. G. H. Bousquet carried this argument to the extent of affirming that Islamic law is idealistic and casuistic, based on imaginative, non-discursive and often rationally absurd hypotheses.³³

Gibb's elaboration on this point is very succinct. To maintain that Islamic law was a system of ethics would naturally imply that it was a system based on human reason; Gibb explained that it was an ethical system in contradistinction to a legal system; yet it was not a rational or philosophical system as it sought its basis in revelation. The main points in his argument that distinguish Islamic law as an ethical system in contrast to a legal system were the following:

a) The classification and categories of actions in Islamic law are moral,

not juridical. The five categories of obligatory, recommended, indifferent, reprehensible and forbidden which are to cover all human actions, are moral and ethical.³⁴

Schacht, however, made it clear that the ethical nature of the categories of action does not mean that there did not exist any legal subject-matter in Islamic law. As a matter of fact, Schacht maintained, the legal subject-matter can be distinguished from other subjects but what is meant by the all-inclusiveness of these five moral categories is that even legal subject-matter is classified as an ethical and religious duty.³⁵

JJ

- b) Islamic law speaks of "duties", not of "rights". In other words there is much more emphasis on what one ought to do rather than upon what one is entitled to claim as a right. The term <u>huqūq</u> even though it means "rights" in a sense, nonetheless, does not contradict the point. In Islamic law, <u>huqūq</u> are divided into those belonging to God and those belonging to men. Subsequently, the latter are subordinated to the former, and this, in fact, renders them into religious and ethical duties rather than rights in the strictest meaning.
- c) Penalties and sanctions in Islamic law are religious and moral, not civil and legal. The term used for a penalty, even in matters belonging to penal law, is <u>hudūd Allāh</u> (the limits of God) which stresses the fact that a certain offence has been committed against God and that it is His right to impose penalty.³⁶

Schacht explains further that the other category of penalty called $\underline{ta^{c}zir}$, according to which a $q\bar{a}d\bar{i}$ (judge) may punish at his discretion any act which, in his opinion, calls for punishment, in fact, did not belong to the Islamic legislation which appears in the Qur'ān and in the tradition of the Prophet.³⁷ What is implied in this explanation is that the concept of civil penalty which the term $\underline{ta^{c}zi}$, might convey, originally did not belong to the concept of Islamic law.

2) THE HISTORICAL NATURE OF ISLAMIC LAW

In the above section on the concept of law we dealt with explanations of how the idea of law is conceived in Islamic legal thought. The present section treats the explanation of the characteristics of Islamic law as it developed historically. This section, however, does not include questions regarding its application in practice; the matter of practice is treated separately in the section following.

Ŋ

In general, those who took a historical approach for understanding the nature of Islamic law have pointed out the following as its characteristics: 1) its idealistic nature, 2) its religious nature, 3) its rigidity, and 4) its casuistic nature. All four characteristics are related to one another and are presented as the reasons for the law's immutability.

The arguments about the nature as revealed in its history of Islamic law, concern the analysis of the following areas: i) the origins of Islamic law; ii) Islamic law and state legislation; iii) the role of the institution of the <u>qādi</u>; and iv) the establishment of the schools of Islamic law.

The Origins of Islamic Law:

The traditional Muslim point of view, later accepted by a number of modern scholars, maintained that Islamic law began with the Divine Revelation in the Qur'ān and with Muḥammad's decisions. These decisions as preserved in the large corpus of Ḥadīth literature were believed to be the foundation of Islamic law.

1. Goldziher's study of the Hadith literature from the point of view of its historicity

exposed the authenticity of the larger part of this literature to serious criticism.³⁸ J. Schacht³⁹ and R. Brunschvig⁴⁰ brought this criticism to bear upon that part of the <u>Hadith</u> literature that concerned Islamic law. Schacht argued that a large number of legal <u>ahādīth</u> were, in fact, legal doctrines of the early scholars of Islamic law which were projected back to the Prophet in the form of <u>hadīth</u>, <u>hadīth</u> being the most acceptable method of establishing a point.

Some scholars also found that there existed in Islamic law a considerable foreign element coming especially from Roman law.⁴¹ As the ancient schools of law developed in areas where Roman law had been applied before the advent of Islam, these scholars concluded that the origins of Islamic law must be sought in Roman law. This view has been a point of controversy among a number of scholars. Schacht connected the existence of the foreign elements to the <u>Sunna</u>. He argued that the ancient scholars, in fact, had assimilated local administrative practices and foreign legal elements into a series of doctrines which they had Islamicized by incorporating them into the <u>Sunna</u>.⁴²

The need for projection backward to the Prophet was not felt until Shāfi^cī very forcefully presented the thesis of the traditionists and established the sole authority of the Prophet in opposition to the authority of "living tradition".

The bearing of these studies on the origins of the law and upon the problem of the law's immutability lies in their conclusion that Islamic law originated from a pious and religious motivation. This motivation became stronger as the religious element in the law was threatened by the attempts of government in the early

^cAbbāsī period to control Islamic law. To save Islamic law from government control, Muslim jurists stressed its religious and divine nature so as to raise it above any human tampering.

Goldziher's and Schacht's criticism on the authenticity of the <u>Hadith</u> literature have been questioned in a number of recent studies, but since most of these studies are not directly relevant to the question of the law's origins, they do not concern us here. Two studies are, however, relevant to our discussion.

Fazlur Rahman⁴³ disagreed with Schacht's conclusion that the <u>Sunna</u> of the
Prophet was a late concept that emerged in consequence of the development of the
<u>Hadīth</u> movement. Using literary, philological and historical evidence,
F. Rahman showed that, contrary to Schacht's argument, the <u>Sunna</u> of the Prophet
could not have been a late concept. If Rahman's conclusion is accepted, it
would mean that the origin of Islamic law is to be sought in the early period of
Islam.

S. D. Goitein,⁴⁴ although he has insisted that his conclusions do not differ from those of Schacht, suggested that the origins of Islamic law may be dated to the year 5/627. Goitein draws his conclusions from a Qur'ānic verse which, he says, establishes Muḥammad's role as law-giver. From the verse he concluded that the idea of Islamic law was not the result of post-Qur'ānic developments but was formulated by Muḥammad himself.

Besides these differences in determining the historic beginnings of Islamic law, all of the above arguments agree upon the religious nature of its origins.

ii) Islamic Law and State Legislation:

Gibb observed that in Islam the law preceded the state, both logically and in terms of time, and that the state existed for the sole purpose of maintaining and enforcing the law.⁴⁵ Gibb argued that in the Umawī period the formulation of the Revealed Law was left in the hands of theologians. The advent of the ^cAbbāsī Caliphs brought this scholastic law, for the first time, to the test of practice.⁴⁶ Schacht's investigation of the early development of Islamic law explains the above observation historically.⁴⁷ As was mentioned above, Schacht concluded that Islamic law began with the activities of the jurists due to religious motives; it was not created by state legislation. This phenomenon resulted in the jurists' conviction of the independence of Islamic law from state control. Certain historical events in the eighth century solidified this attitude further.

In the early 'Abbāsi period the administration of justice was in chaotic condition because of the lack of unity in juridical doctrines. Ibn al-Muqaffa', a secretary in the 'Abbāsi government, strongly recommended that the caliph control this diversity of opinions by state legislation.⁴⁸ The jurists reacted to this suggestion by insisting that the law was superior to the state, and hence not subject to state legislation.⁴⁹

Whether Islamic law maintained this independence in actual practice is a matter dealt with in the next section. What concerns us here is the conclusion that many scholars have drawn from observations on the nature of the law in relation to the state. H. Lammens and others have argued that, being severed from state legislation, Islamic law became divorced from social realities.⁵⁰ G. H. Bousquet concluded that the idea of successive adaptations to changing circumstances was strange to its system.⁵¹

Claude Cahen, however, has disagreed with such conclusions. He argues that the problem for the early jurists was not to derive the ideal of Muslim government but rather to institute a very loose filtering which would reserve to them the bestowal upon the régime as a whole of its certificate of 'good Muslim'. He concluded that "it would be supremely unjust...to regard the work of the Abbāsī jurists as abstract and turning the back on reality".⁵²

iii) Role of the Institution of the . di:

The institution of the <u>qādī</u> evolved out of the pre-Islamic institution of the <u>Hakam</u> (arbitrator). Like the <u>hakam</u>, the early <u>qādī</u> was bound by the precedents of local tradition and decided cases, not through some formal methods of reasoning, but according to his own discretion.⁵³ As Schacht has shown, the decisions of the Umawī <u>qād</u>īs incorporated local elements. In the later development of Islamic law these decisions were assimilated into the body of Islamic law.⁵⁴ Yet the role of <u>qādī</u> was not recognized to be that of making or interpreting the law, but, essentially, only of applying it.

In the 'Abbāsi' period the office of <u>qādi</u> was connected with Islamic law, thus separating it from the general state administration and making it subject to Islamic

υ,

law only.⁵⁵ Later when the schools of law were established, the role of the $\underline{q\bar{a}d\bar{l}}$ was reduced to the application of the teachings of one of these schools. This limitation caused the complete stagnation of the law.

N. J. Coulson, in analysing the causes of the widespread dislike of the office of <u>qādī</u> among the jurists, concluded that the rejection of the office could not be fully explained by such factors as the fear of sudden political disfavour or as pious motives, such as 1. Goldziher, Amerdoz and E. Tyan had suggested. According to Coulson the real cause of dislike of the office was its impracticable and idealistic nature.⁵⁶

Coulson observed a significant distinction in the attitude of the jurists toward the institution; the distinction between the attitude of the practical lawyers and the He stresses that this distinction was real attitude of the idealist traditionists. For lawyers, Islamic law consisted of and vital in the history of Islamic law.⁵⁷ enforceable legal rules; for traditionists it was a code of moral and religious The former regarded the office of \overline{qadi} as essential and honourable; duties. the latter wished to avoid it at all costs. The attitude of the lawyers was a continuation of the outlook of the early Umawi qadis who, as legal secretaries, were responsible to the governor. The other attitude was the result of the growing influence of the religious concept of law in the eighth century, extending to the The morally-inclined qādīs began to feel that their allegiance office of the qadi. lay to religion rather than to the interests of the governor.

As a result of this dichotomy there developed two trends of law; the 'religious law'

as expounded by the jurists and the 'positive law' as administered by the courts. An example of the latter is the development of <u>camal</u> (juridical practice), as court law in the Mālikī school.⁵⁸

H. Toledano has observed that <u>camal</u> became "an instrument for modifying and adapting the <u>sharica</u> to meet the practical needs of the society, and the judges in Morocco were filling the same role as their predecessors in the first two centuries of Islam".⁵⁹

iv) The Establishment of the Schools of Law:

× -

As a result of the rapid legal activity from the late Umawi period until the end of the second century, there emerged certain schools of law which were consolidated to the extent that adherence to one of these schools was common and also necessary. This adherence was required not only of the layman but also of the qāqī and the jurist. This requirement was called taqlīd.

The effects of <u>taqlid</u> on the growth of Islamic law were fateful. It reduced legal activities to the confines of particular schools. On the one hand, the procedure of legal reasoning became mechanical and, on the other hand, the whole body of Islamic law was cast into a rigid mold, not allowing further independent growth.⁶⁰

The phenomenon of <u>taqlid</u> has been considered by a number of scholars as a factor responsible for the belief in the immutability of Islamic law.

Most studies on Islamic law lay stress on the gap between theory and practice. This gap has been so striking that some scholars such as J. Kramers even suggested the distinction between two systems of law in Islam: 'droit de l'Islam', the laws in practice, and 'droit islamique', the law in theory.⁶¹

The cleavage between theory and practice has been observed under three aspects: i) between Islamic law and the customs of the Muslim people; ii) between Islamic law as elaborated in Texts and as practiced in the courts; iii) between different kinds of subject matters in reference to their application.

Although custom was not recognized, theoretically, as a source of Islamic law, ⁶² yet scholars have observed that custom not only played an important role in the growth of Islamic law but also that it always co-existed with the law. ⁶³

As for the cleavage between the jurists' law and the court law, scholars have observed that the administration of justice was not completely subject to Islamic law. An evident example of this was the introduction of the courts of <u>nazar fi almazālim</u> where decisions were reached through individual discretion and <u>Siyāsa</u>. The jurisdiction of the <u>qāqi</u> was limited, and even there interference by the governor and other government officials in the <u>qāqi</u>'s decision, and restricting his competence in legal matters, was so frequent that, in fact, the applicability of <u>sharī'a</u> law in courts was more and more restricted. Consequently, the 'positive law' applied in the courts grew separately from the religious law.

As mentioned above the <u>camal</u> tradition is an example of the positive law. As a

matter of fact, it was assimilated into Mālikī law as a doctrine that had a regulative force. The judges were required to follow it even when it ran contrary to the dominant opinion of the school.⁶⁴

Customary law and the law of the courts that responded to social needs and that were adaptable to social changes ought to have influenced Islamic law. This influence, as observed by the scholars, did operate, but it varied according to the various subject matters of Islamic law.

Ν

A. L. Udovitch believes that Bergsträsser was the first scholar who pointed out this influence.⁶⁵ He distinguished three broad categories of the subject matters of Islamic law:

- Ritual, family and inheritance laws, which though they accepted certain changes based on custom, yet remained as a whole closest to Islamic law.
- 2. Constitutional, criminal and fiscal laws an area where Cahen believed the jurists to be very flexible⁶⁶ – which was constantly being adapted to social changes. In fact, Bergsträsser observed that this category of Islamic law diverged farthest and in some cases completely from the classifcal formulation of Islamic law.⁶⁷
- 3. Commercial laws, or to use Schacht's terminology, the laws of contract and obligation, fell somewhere between the two extremes. Schacht,⁶⁸ in one of his early statements agreed with Hurgronje⁶⁹ who maintained that Islamic commercial law remained for the most part a dead letter.

4) THE QUESTION OF METHOD

The significance of the question of method in reference to the immutability of Islamic law has been generally recognized by all scholars. Every system of law tends to be perfect and permanent; hence a sense of immutability has gathered around the concept of law. But changing social needs challenge such an attitude. Various systems of law have devised certain methods to meet such challenges. For instance, Roman law resolved this problem by distinguishing between jus civile which was strict and jus honorarium which was elastic.⁷¹ In Common law the flexibility was achieved through Equity.⁷²

The question of method in Islamic law has generally been discussed in reference to the classical theory of the 'four sources of Islamic law'. 73

Modern scholarship also discusses the question of the method in reference to the sources of Islamic law. E. Tyan ⁷⁴ and Ch. Chehata⁷⁵ observed that Islamic law did evolve methods to adapt legal theory to changes. Chehata spoke about the principle of <u>Istihsān</u> as being the counterpart of Equity in Common law.⁷⁶ Tyan pointed out three such methods: <u>Istihsān</u>, <u>Istişlāh</u>, and <u>Siyāsa Shar^ciyya</u> (administration of justice according to Islamic law). All of these methods were devices to incorporate social changes into Islamic law where the strict requirements of Islamic law would not allow this.

Schacht contended that Islamic law did not and could not evolve such methods, 77 mainly because by its very nature Islamic law was not in need of them.

Islamic law was not an official law like other laws. Official law came to be by the authority of secular legislators, but Islamic law did not recognize it. Hence Islamic law was a 'sacred law' par excellence; perfect, immutable, and not in need of change. Schacht maintained that principles such as <u>(urf, istihsān, istişlāh</u> and <u>amal</u> were not used as principles of change but rather to interpret and justify the already existing rules of Islamic law. Moreover, if ever they were used to adopt certain changes they were meant to build a protective zone around that particular change lest it affect the whole of the theory.⁷⁸

Malcolm H. Kerr, in his study of Islamic reforms in the nineteenth century, has confirmed Schacht's conclusions. Kerr chose to study the principle of <u>maslaha</u> ("welfare, benefit, utility")⁷⁹ because it was considered by the upholders of the dynamism in Islamic law as a principle of adaptability.⁸⁰ He concluded that although theoretically a liberal principle, the <u>maslaha</u> in actual application succumbed to the theological and idealistic limitations imposed upon it by the Islamic legal theory.

Conclusion

As suggested at the beginning of this chapter, for the subsequent discussion of the problem, this chapter provides us with a conceptual framework in two respects. First, it helps us develop definitions of the key terms of the problem. Second, it gives us the basic assumptions and premises of the argument. The conclusion of this chapter, therefore, consists of two parts. First, it deals
with definitions of the concepts and terms in reference to the above discussion. In the second part it defines the assumptions and the manner of argument to be followed in the rest of the thesis.

Before proceeding to definitions, a general conclusion of the above debate upon the problem must be given.

The above discussion shows that the scholars are divided on the question of the adaptability of legal theory to social changes.

The immutability view maintains that the main reason for affirming the unchangeability of the law is that, by its very concept, Islamic law is not adaptable to social changes. In the actual history of the law, because of its self-concept, Islamic legal theory has been divorced from social realities. It has been separated from those institutions which are adaptable to social needs and for that reason could not develop a method of adaptation of its own.

The adaptability view does not differ from the immutability view on the concept of law but they do not give so much significance to this matter; they rather argue from the nature of the law's development. In practice Islamic law accommodated to social changes. The origin of the law came about in response to social needs, and in its subject matter and methodology it showed adaptability to social change.

Both positions, however, admit the view of the opposite group on some points. For instance, the immutability view submits to the opposite position in maintaining that Islamic law was adaptable in its formative period. The adaptability view admits that after the "closing of the gate of <u>ijtihād</u>", Islamic law showed less and less adaptability.

KEY TERMS

A closer look at the above debate shows that it is the different understanding of the key terms that have caused the controversy. What follows is an attempt to redefine the basic terms of the problem.

N

"Adaptability" and "immutability"

It is clear that the above views have the following questions as a starting point: does Islamic Iaw in fact change? Further, is Islamic Iaw changeable? The two views provide different answers to these questions. The immutability-view claims that Islamic Iaw does not change, adding that in fact it cannot change. The "immutability", to them, therefore, means that the rulings pronounced by Islamic Iaw are static, final, eternal, absolute and unalterable. The adaptability-view, on the contrary, maintains that Islamic Iaw changes and that, in fact, it has changed, and moreover, can be changed further. This view also stresses that it can be changed and modified to fit new social conditions. In other words, "adaptability", in the specific context of the above controversy, is not simply a contrary term to "immutability", but it consists of an additional meaning, i.e. a distinct implication of modifying to meet new conditions.

"Social Change"

The term "adaptability" is, thus, immediately concerned with social changes.

Social change, here, is obviously not a technical term which implies "transformation of society" or "social control".⁸¹ This term is rather used as a general term to signify that the change in question has happened in society in response to social needs. A legal change that interacts with such social changes or recognizes the social needs, demonstrates the adaptability of a particular legal system.

"Islamic law"

In the above controversy neither of the views dispute that social changes occurred in Islamic history and that legal changes did take place accordingly, but whereas the adaptability-view connects these changes to the nature of Islamic law, the other view does not. The immutability-view asserts that these changes took place only in practice but were not recognized by the theory of Islamic law. The question is then obviously not about the historicity of legal changes, but about the theory of Islamic law regarding these changes. The difference of the two views is confined, therefore, to the theoretical aspect of the question. Since the two hold opposite views on this point, it is worth investigating whether they mean the same thing when they say "Islamic law", or not.

The adaptability-view refers to figh as Islamic law, and even $\underline{shan}^{\bar{s}a}$ is understood as figh. The immutability-view is not so monolithic. In reference to the concept of law, Islamic law is identified with $\underline{shar}^{\bar{i}s}a$, but even here the arguments about its ethical and moral nature are made in reference to figh. In the arguments contending that the law is divine and the will of God, obviously it is not the figh which is meant. In discussions of the nature of the law and practice what is implied by Islamic law is figh. The contrast between theory and practice is made in reference to figh. The reason for this apparent inconsistency and ambiguity is that the immutability view believes that <u>shanica</u> and <u>figh</u> are inseparably connected, <u>shanica</u> being the law, and <u>figh</u> the science of knowing the law. This explanation, however, does not remove the ambiguity.

To explain this ambiguity we may borrow Kerr's formula of the levels of meaning. He observed the following four levels of meaning implicit in the discussion of juristic theory: (1) Divine Will, the sole metaphysical reality; (2) the spiritual relationship between man and God; (3) the normative relationship between man and man, and (4) the non-normative relationship of man and nature.⁸²

In reference to these four levels we may say that $\underline{shari}^{c}a$ belongs to the first level, and <u>figh</u> covers both the second and third levels. The third and fourth levels concern social changes. Now social changes would usually have immediate effects on the third level; its effects on the second level are not immediate, however. In respect of the question of adaptability therefore, the <u>figh</u> at level three is more significant than at level two.

In view of this explanation, both positions involve ambiguity in some sense. The adaptability view confuses the first and third levels by equating <u>figh</u> with <u>sharifa</u>. The immutability view also confuses the two levels. A distinction in these levels can help in demarcating <u>sharifa</u> from <u>figh</u> and also in distinguishing among various subject matters of figh.

The question whether <u>shari</u>éa or <u>fiqh</u> can be called law is another source of ambiguity. The question stems from the fact that the English term "law" has a special sense which is not conveyed by the Islamic terms. The adaptability view believes that figh may be called "law". This position is taken by Linant de Bellefonds.⁸³ He has argued that the theocratic and religious nature of Islamic law has been stressed in an exaggerated manner, by referring to its teachings on <u>clbādāt</u> (rituals, worship) and by comparing it with Western concepts of law. He maintained that even if the theocratic nature of its origins be admitted, it was not prevented from becoming a juridic system so long as its precepts were sanctioned by a secular authority. Implicit in his argument is the view that <u>figh</u> became law as much as and whenever it was sanctioned by governors and administrators.

The opposite view contends that <u>sharī'a</u>, though not law in the proper sense, is the law of Islam. <u>Figh</u> is a science that deduces rules of law from the <u>shaīt'a</u>. Accordingly <u>sharī'a</u> is known through the <u>figh</u>. Does there exist <u>shaīt'a</u> outside the <u>figh</u>? Although the answer should be in the affirmative, yet there are different answers to the question of its location. In the abstract sense the <u>sharīt'a</u> is a metaphysical reality known through the Qur'ān and the sayings of the Prophet. The question whether everything contained in the Qur'ān and <u>Hadīth</u> is law takes us back to <u>figh</u>, as that is where the law is spelled out. Hence for practical purposes, even in this position, <u>figh</u> comes to stand for Islamic law.

"Islamic legal theory"

Now, coming to the question of the legal theory and social change, can we consider <u>figh</u> to be the legal theory? Most probably not. In the preceding discussion, to consider <u>figh</u> as legal theory is possible but only in a limited sense. <u>Figh</u> cannot stand for legal theory in the sense of principles and methods, because the branch of the Islamic legal sciences that concerns such matters is U<u>s</u><u>u</u> al-figh.

<u>Uşūl al-fiqh</u> is the formal science in which Muslim jurists have dealt with legal theories, the principles of interpretations of legal texts, methods of reasoning and of deduction of rules and other such matters. Thus, this thesis proposes to mean uşūl al-fiqh, when it speaks of 'Islamic legal theory'.

Having defined our terms of analysis, we now come to the second part of the conclusion.

Our framework of discussion in this thesis consists of two sets of arguments. One part of the argument is that Shātibā's concept of <u>maşlaha</u> in relation to his doctrine of <u>maqāşid</u> <u>al-sharā'a</u> was the product of the need of his time to adapt Islamic law to the new social conditions. For this part the argument comprises the following steps: 1) A broad picture of the social changes in fourteenth-century Granada (Chapter 3) will be drawn to see the extent to which the political, religious and economic developments in this period brought a basic change in Granadian society. 2) We will also see how the legal system may have been affected by these social changes. These observations will then be substantiated with an analysis of the actual <u>fatāwā</u> in this period. (Chapter 5). Since these <u>fatāwā</u> are answers to the actual guestions arising out of new social conditions, we will be able to assess

~5

to what extent the need for legal change was connected with the developments as already observed. 3) This will also enable us to observe how Islamic legal theory dealt with the problem, i.e. what legal concepts and methods were used and whether, in this respect, there was a departure from the legal tradition.

. . .

> This will lead us to the second part of the argument in our thesis – that Shājibī's concept of <u>maşlaha</u> is an attempt to justify the adaptability of Islamic legal theory to social needs. In this regard, the assumptions and premises of our argument are drawn from the above discussion. These assumptions are as follows: First, to determine the adaptability of Islamic law, one must examine whether a certain method or concept, proposed as a theoretical justification of the adaptability, succeeds in freeing the concept of legal obligation from the theological determinism that it has received from having its origin in the absolute Will of God. To verify this hypothesis, Shājibī's concept of <u>maşlaha</u> will be examined in this frame of reference. The analysis is undertaken in reference to the development of this concept in <u>uşūl al-fiqh</u> (Chapter 6). The purpose of this analysis is to assess the direction in which Shājibī wanted this concept to lead.

This comparison also helps us in defining the meaning of theological determinism and its consequences for the concept of <u>maslaha</u> and then to assess whether Shātibi succeeded in freeing the concept from this determinism. (Chapter 7).

In examining Shātibi's attempt to free legal obligation from theological determinism, we will interpret such an outlook as a positive element in his legal philosophy. Since our use of the term "positivism" may create some misunderstanding, we must explain that our use of the term is related to, but not identical with, "legal positivism".

In this thesis "positivism" refers to the well-known doctrine which explains the evolution of human thought in three stages: theological – metaphysical – positive. The third stage, positive, seeks to separate philosophical thinking from theological and metaphysical modes of thought, and stresses observable phenomena. Historians of legal philosophy, such as Huntington Cairns, attribute this development to the tendency of jurisprudence towards complete independence. ⁸⁴ Jurisprudence has shown this tendency by breaking with theology in the sixteenth century and culminating in the recent trend which is called "legal positivism" or the analytical school of jurisprudence. ⁸⁵

Recent exponents of 'legal positivism', such as H. L. A. Hart, have excluded considerations of morality and justice from the concept and definition of legal obligation.⁸⁶ Hart has, however, made a significant observation at this point. He admits that the origin of the rules of law may be found in the ideas of morality a nd justice but that this does not prevent legal obligation from separating itself from morality in actual enforcement of law.⁸⁷

This observation will help us in understanding the distinction that Shātibī suggests in defining legal obligation in reference to tas abbud.

The suggestion that there was an element of positivism in Shāțibi's legal thinking

is advanced as described above. His attempt to free legal theory from theology and morality will be interpreted as a step towards positivism. Shātibi's distinction between <u>'ādāt</u> and <u>'ibādāt</u> on the basis of observability of <u>maṣālih</u> in the former is understood as an attempt to separate positive law from religious elements.

NOTES: CHAPTER II

- See W. Friedmann, <u>Law in a Changing Society</u> (London: Pelican, 1964), 19. In Chapter One, Friedmann surveys the history of this problem in Western legal philosophy.
- 2. In 1897, in a case (Calcutta, 25:18) before the Privy Council the appellants sought a revision of the decision of a lower court of law. The appellants contended that the decision which was based on the Hidāya was not in accordance with the Qu'ān on this point. The Privy Council confirmed the decision of the lower court and declined to go beyond Hidaya. Their Lordships explained: "It would be wrong for the Court on a point of this kind, to attempt to put their own construction on the Koran in opposition to the express ruling of commentators of such great antiquity and high authority". Quoted by A. J. Robertson, The Principles of Mahomedan Law (Rangoon, 1911), 5.
- 3. See, for instance, M. Morand, Introduction a l'étude du droit musulman algérien (Alger, 1921), 108–112.
- For example, M. L. W. C. Van den Berg's translation <u>Minhâdj at-tâlibîn</u>, texte arabe publié avec traduction et notations (Batavia, 1862-1884), was strongly criticized by Hurgronje in his review article in <u>Revue de l'histoire</u> des religions, XXXVII (1898), 1-22, 174-203; available to us in G. H. Bousquet and J. Schacht (editors), <u>Selected Works of C. Snouck Hurgronje</u> (Leiden, 1957), 214-255.
- G. Bergsträsser's work, Grundzüge des islamischen Rechts, published by J. Schacht in Berlin, 1935. Cf. J. H. Kramers, "Droit de l'Islam et droit islamique" in Analecta Orientalia Posthumous Writings and Selected Minor Works of J. H. Kramers, Vol. II (Leiden: Brill, 1956), 67.
- 6. Selected Works, op. cit., 256.
- 7. J. Schacht, "The Present State of the Studies in Islamic law", in Atti del Terzo Congresso di Studi Arabi e Islamici (Naples, 1967), 621-622.
- 8. J. Schacht, "Theology and Law in Islam", in G. E. von Grunebaum (ed.), Theology and Law in Islam (Wiesbaden, 1971), 4 ff.
- 9. Ibid., 7.
- 10. M. H. Kerr, Islamic Reform (California, 1966), 56.

11. Schacht, "Theology and Law...", op. cit., 10.

м. х,)

- C. H. Toy, "The Semitic Conception of Absolute Law", in Carl Bezold (ed.), <u>Orientalische Studien Theodor Nöldeke</u>, Vol. II (Gieszen, 1906), 804. Toy's attribution of natural law to Indo-European people (p. 798) and of external absolute law to Semites has been confirmed in a recent article by Hajime Nakamura in "The Indian and Buddhist Concept of law", published in E. J. Jurji (ed.), <u>Religious Pluralism and World Community</u> (Leiden: Brill, 1969), 131-174.
- N. J. Coulson, A History of Islamic Law (Edinburgh: Edinburgh University Press, 1964), 1,2.
- H. A. R. Gibb, "Constitutional Organization", in M. Khadduri and H. J. Liebesny (eds.), <u>Law in the Middle East</u>, Vol. 1 (Washington, 1955), 4. Also in Gibb and H. Bowen, <u>Islamic Society and the West</u>, Vol. 1, part 2 (Toronto: Oxford University Press, 1957), 114.
- 15. H. J. Liebesny, "Religious law and Westernization in the Moslem Near East", in The American Journal of Comparative Law, Vol. 11 (no. 4, 1953), 492.
- M. Khadduri, "From Religion to National Law", in J. H. Thompson and R. D. Reischauer, (editors), <u>Modernization of the Arab World</u> (Princeton: Nostrand, 1966 38.
- 17. H. Lammens, Islam, Belief and Institutions (London, 1929), 82.
- G. Makdisi, "Remarks on Traditionalism in Islamic Religious History" in Carl Leiden (ed.) <u>The Conflict of Traditionalism and Modernism in the</u> Muslim Middle East (Austin: University of Texas, 1966), 77.
- J. N. D. Anderson, Islamic Law in the Modern World (New York: New York University Press, 1959), 17.
- 20. Léon Ostrorog, The Angora Reform (London: University of London, 1927), 19.
- 21. S. G. Vesey-Fitzgerald, "Nature and Sources of Shari a", in Law and the Middle East, op.cit., 87.
- 22. Gibb, Mohammedanism (New York: Oxford, 1962), 99.
- 23. Gibb, Studies in the Civilization of Islam (Boston, 1968), 204-206.
- 24. Mohammedanism, 90f.

- 25. Schacht, Introduction to Islamic Law (Oxford, 1966), 202-203.
- 26. R. Brunschvig, "Logic and Law in Classical Islam", in G. E. Von Grunebaum, (ed.), Logic in Classical Islamic Culture (Wiesbaden, 1970), 9f.
- 27. See Chapter I. p. 29f.
- 28. Von Grunebaum, "Concept and function of Reason in Islamic Ethics" in Oriens, Vol. XV (1962), 15.
- 29. G. F. Hourani, "Two theories of Value in Medieval Islam", Muslim World, L (1960) 273, and Islamic Rationalism (Oxford: Clarendon, 1971), 8-14.
- 30. See Chapter I. p.25, Ch. II, p. 65.
- 31. Hurgronje, op.cit., note 6.
- 32. Th. W. Juynboll, Handbuch des Islamischen Gesetzes (Leiden: Brill, 1910), 22f.
- 33. G. H. Bousquet, Précis de droit musulman (Alger, 1947), 44.
- 34. Gibb, Islamic Society and the West, 118.
- 35. Schacht, Introduction, 200-201.
- 36. Ibid., 176f.
- 37. Ibid., 207.
- 38. "Closer acquaintance with the vast stock of <u>hadiths</u> induces sceptical caution rather than optimistic trust regarding the material brought together in the carefully compiled collections. We are unlikely to have even as much confidence as Dozy regarding a large part of the <u>hadith</u>, but will probably consider by far the greater part of it as the result of the religious, historical and social development of Islam during the first two centuries". Goldziher, <u>Muslim Studies</u>, translated by C. R. Barber and S. M. Stern, Vol. II (London: Goerge Allen and Unwin, 1971), p.19.
- 39. Schacht, Origins of Muhammadan Jurisprudence (Oxford, 1954), p.4.
- 40. R. Brunschvig, "Polemique medievales autour du rite de Mālik", <u>Al-Andalus</u>, XV (1950), 377-435.
- 41. This point has been a subject of controversy among scholars of Islamic law. The following studies on this subject are noted by S. D. Goitein in The Birth-Hour of Islamic Law?, Muslim World, L (1960), 23:

J. Schacht, "Foreign elements in Ancient Islamic law", Journal of Comparative Law, 1950, 3-4, 9-16; A. D'Emilia, "Roman Law and Muslim Law, a Comparative Outline", in East and West, Vol. IV/2 (1953); Fitzgerald, "The Alleged Debt of Islamic Law to Roman Law", Law Quarterly Review, Jan. 1951, 81-102. C. A. Nallino, "The Impossibility of the Influence of Roman Law on Muslim Law" (trans. Hamidullah, Voice of Islam, Karachi, 1963, 1/2, 63-67).

- 42. Schacht, Introduction, 19ff.
- 43. F. Rahman, Islamic Methodology in History (Karachi, 1965), 5-21.
- 44. S. D. Goitein, op.cit. 27ff.
- 45. Gibb, "Constitutional Organization" op.cit., 3.
- 46. Mohammedanism, op.cit., 102.
- 47. Schacht says that "during the greater part of the first/seventh century, Islamic law, in the technical meaning of the term, and therefore, Islamic jurisprudence, did not yet exist", article "Fikh" in E.I. (2nd edition) Vol. 11, 887-8.
- 48. Schacht, Introduction, 55.
- 49. Ibid., 56.
- 50. H. Lammens, Islam..., op.cit., 92.
- 51. G. H. Bousquet, Précis..., op.cit., 50.
- 52. Claude Cahen, "Body Politic" in Grunebaum, Unity and Variety in Muslim Civilization (Chicago: University of Chicago, 1955), 139.
- 53. Schacht, Introduction, 24f.
- 54. Ibid., 25.
- 55. Ibid., 4.
- 56. N. J. Coulson, "Doctrine and Practice in Islamic Law", BSOAS, Vol. XIII/2 (1956), 211.
- 57. Ibid., 226.

- 58. Cf. J. Berque, "'Amal" in E.I. (2nd edition) Vol. 1, and H. Toledano, "The Chapter on Marriage from Sijilmāsī's Al-'Amal al-Mutlaq: A Study in Moroccan Judicial Practice (Ph.D. Thesis, unpublished, presented to Columbia University, 1969), Introduction.
- 59. Toledano, op.cit., p.ii.
- 60. Coulson, History of Islamic Law, 80f.
- 61. J. Kramers, "Droit de l'Islam et droit islamique", op.cit., 63-64.
- 62. Schacht, Introduction, 62.
- 63. Ibid., 76ff.

. .

÷

- 64. For details, see Toledano, op.cit., ii-vii.
- 65. Abraham L. Udovitch, <u>Partnership and Profit in Medieval Islam</u> (Princeton: Princeton University Press, 1970), 7.
- 66. Cahen, op.cit.
- 67. Udovitch, op.cit.
- 68. Schacht, "Islamic Law", Encyclopedia of Social Sciences (Old Edition), Vol. VIII, 348.
- 69. Hurgronje, Selected Works, 260.
- 70. Udovitch, op.cit., 10.
- 71. Schacht, "Classicisme, traditionalisme et ankylose dans la loi religieuse de l'Islam", in R. Brunschvig and others, (editors) <u>Classicisme et déclin</u> culturel dans l'histoire de l'Islam (Paris, 1957), 141.
- 72. Y. Linant de Bellefonds, "Immutabilité du droit musulman et reformes legislatives en Egypte", <u>Revue internationale de droit comparé</u>, Vol.VII (1955), 10f.
- 73. See above Chapter I. p. 20ff.
- 74. E. Tyan, "Methodologie et sources du droit en Islam", in <u>Studia Islamica</u>, Vol. X (1959), 84f.
- 75. Chafik Chehata, "Logique juridique et droit musulman", in <u>Studia Islamica</u>, Vol. XXIII (1965), 5-26.

- Chehata, "L'équité en tant que source du droit hanafite", <u>Studia Islamica</u>, Vol. XXV (1966), 123–138.
- 77. Schacht, "Classicisme...", op.cit., 142.
- 78. Ibid., Introduction..., 61ff.
- 79. M. H. Kerr, op.cit., 55; also see above Chapter I. p. 31f.
- 80. Ibid., 56.

1

- 81. In maintaining this distinction, we are relying on Amitai Etzioni and Eva Etzioni, Social Change, Sources, Patterns and Consequences (New York: Basic Books, 1964), Introduction, p.7f.
- 82. M. H. Kerr, op.cit., 21ff.
- 83. Y. Linant de Bellefonds, <u>Traité de droit musulman comparé</u> (Paris: Mouton, 1965), Vol. 1, 46-48.
- 84. Huntington Cairns, Legal Philosophy from Plato to Hegel (Baltimore: John Hopkins, 1967), p.2f.
- 85. The exact definition of 'legal positivism' has been a subject of debate among recent philosophers. Lone L. Fuller and others contrast "legal positivism" with "natural law" philosophies. H. L. A. Hart regards such a use of the term as a cause of the confusion in modern discussions of separation of law from morals. See the following:

H. L. A. Hart, The Concept of Law (Oxford: Clarendon, 1965), pp.253-54. Law, Liberty and Morality (New York: Vintage, 1963), p.2. Lone L. Fuller, The Morality of Law (New Haven: Yale, 1970), the chapter: "A Reply to Critic".

- 86. Hart, The Concept..., Ch. 8, section 2; Ch. 9, section 3.
- 87. Hart, Law, Liberty and Morality, Ch. 1.

CHAPTER III

SOCIAL CHANGES IN FOURTEENTH CENTURY GRANADA

For a better understanding of Shāţibī's views on the adaptability of Islamic legal theory to social changes, a general study of the changes that occurred in Shāţibī's period is necessary. Shāţibī flourished in Granada in the reign of the Naşrī ruler Muḥammad V al-Ghanī Billāh (755-760/1354-1359) and 763-793/1362-1391).

The present chapter, therefore, attempts to present a broad picture of the social changes in fourteenth century Granada. It must be made clear, however, that the present chapter does not aim to give a complete historical account of this period. This chapter serves only the purpose of providing a general context by indicating the significant factors of social change in the political, religious, economic and legal areas of Granadian society.

SECTION ONE

POLITICAL DEVELOPMENTS

Fourteenth Century

The fourteenth century was a period of rest for the Muslim world after the turmoils of the thirteenth century. Two major Mongol dynasties, the Ilkhānīs and the Golden Horde had been converted to Islam. The Mamlūks who had withstood the Mongol invasion had stabilized their rule in the fertile crescent and in Egypt. In North Africa as well, conditions were rather stable. ^{The} Banū Marīn had emerged as powerful successors to the Muwaḥḥidūn. In Spain, *me* Banū Naṣr had succeeded the Muwaḥḥidūn. They maintained their rule by keeping a delicate balance of alliance with the Christian kingdoms in Spain and with the Banū Marīn in Africa.

This political stability provided the much-needed peace for the intellectual activities essential to re-evaluating the tradition in the light of the multitudinous changes brought about by the turmoils of the thirteenth century caused by the Mongol invasion in the Muslim East and by the rapid Christian advances in the Muslim West. These changes affected the political, financial, commercial, social and religious domains. A number of social changes that had taken place needed somehow to be accommodated within the tradition.

The intellectuals of the community who had either personally experienced these

changes or been affected through the experience of others received a lasting impact on their minds. This is no doubt the reason why we find that a number of distinguished works dedicated to the re-evaluation, systematization and readjustment of the tradition appeared in this period. In North Africa, Ib n Khaldūn (784/1382) worked on a philosophy of history.¹ In Syria, Ibn Taymiyya (728/1328) reviewed the entire tradition of political and legal theory.² In Persia, Al-Ijī (756/1355) resystematized Sunnī theology.³ In Spain, Shāṭibī was occupied with the philosophy of Islamic law. All of these efforts imply some breakdown in the community's sence of itself, and are acknowledgements by their very existence, of the need for new and more satisfactory formulation of certain basic values and standpoints.

Muslim Spain⁴

To help build an appreciation of political conditions in fourteenth century Muslim Spain, a brief survey of events in the reign of Muḥammad V is in order. Such a survey, however, in its turn, requires a review of the rise of Banū Naṣr dynasty for a better understanding of the nature of the political structure that Sultān Muḥammad V inherited from his predecessors.

Al-Ghālib Billāh

With the decline of the Muwaḥḥidün, the political situation in Andalus (Muslim Spain) fell into a chaotic condition.⁵ Two warlords appeared in this period: Ibn Hūd in Murcia and Ibn al-Aḥmar in Arjona. Ibn Hūd revolted against the Muwaḥḥidūn in 625/1228 in Murcia.⁶ He received investiture from the ʿAbbāsī caliph Al-Mustanṣir Billāh (623-640/1226-1242) in Baghdad. Once established as Sultān, he assumed the title Al-Mutawakkil Billāh. Important cities such as Almeria, Malaga, Granada, Sevilla and the greater part of Eastern Andalus fell to him.

Ibn al-Ahmar declared his independence in 629 A.H.⁷ when he captured Jaen. Ibn al-Ahmar (Muhammad b. Yüsuf ... b. Naşr b. Qays al-Khazraji al-Anşāri) was a soldier who fought on the borders of Andalus. He earned his fame mainly by his campaigns against the Christians. After Jaen, he quickly captured Sevilla and Cordova from Ibn Hūd.

At the instigation of Ibn Abī Khālid, the people of Granada proclaimed Ibn al-Aḥmar their king. In 634 A.H. Ibn al-Aḥmar moved to Granda and after inflicting a heavy defeat upon Ibn Hūd, captured Granada and declared himself the Sulṭān of Andalus, and assumed the title of al-Ghālib Billāh. Thus was founded in Granada the dynasty of the Banū Naṣr, also called Banū Aḥmar. Ibn al-Aḥmar's only rival was Ibn Hūd who died in 635 leaving Ibn al-Aḥmar the sole Sultān of Andalus.

Toward his neighbouring states Ibn al-Ahmar pursued a policy of truce. He professed submission to his African neighbours and ordered that the name of the Hafşī ruler, Abū Zakariyya Ibn Hafş (625-647/1228-1249), be recited in the <u>khutba</u> -- a sign of allegiance. This gesture was meant to acquire Hafşī help. He even included the name of the ^cAbbāsī ruler in the <u>khutba</u> to 'elevate his prestige among his subjects;' later, however, he discontinued the practice.⁸ He concluded peace with Ferdinand III, the king of Castille, in 643 A.H. but this truce cost him the surrender of Jaen. The conditions of the truce made Ibn al-Ahmar repent his decision.⁹ In 662, however, he signed another peace treaty with the Christians, but also issued an appeal to the African tribes for Jihād.

5. 14. 14

> After the decline of the Muwaḥḥidūn there emerged three dynasties among the African rulers; the Hafsis in Tunis; the Zayyānis in Tlemcen; and the Banū Marin in Morocco. Among them the last proved themselves most powerful. lt was, therefore, the Banū Marin who crossed over to Spain in answer to the Nasri appeal for help. The relations between the Banū Marin and the Banū Nasr, however, became a source of trouble internally as well as externally; internally because they headed the African mercenaries and thus held a major source of power in their hands. They were often in conflict with the wazirs The balance of power often oscillated between who tried to control them. Externally, being related to the Banū Marin they these two major offices. constituted a threat to both Banū Marin and Banū Nasr rulers -- to the Banū Marin as claimants to the throne, to the Banū Nasr as a pretext for interference by the Banū Marīn in their affairs.

This delicate balance of power continued to be critical for the successors of al-Ghālib Billāh, until this situation changed in the reign of Muḥammad V al-Ghanī Billāh, the eighth ruler in the line of his dynasty.

Al-Ghani Billäh

At the age of sixtzen, ¹⁰ Muḥammad V al-Ghanī Billāh succeeded his father in 755/1354 when the latter was assassinated. The affairs of the state were completely in the hands of his chamberlain (<u>hājib</u>), the Qā'id Abū Naīm Ridwān.¹¹

Other important offices of the kingdom were the following: the office of <u>Shaykh</u> <u>al-Ghuzāt</u> was given to Yaḥyā b.⁶Umar; <u>Qādī al-Jamā^ca</u> to Abū'l-Qāsim Sharīf al-Sabtī; <u>Kātib al-Sirr</u> to Ibn al-Khațīb. Since these offices played a significant role in the political structure as well as the political development of this period, a detailed analysis of them is attempted in the following lines:

Within a month Ibn al-Khaṭīb and Abū'l Qāsim al-Sabtī were sent on a mission to seek help from the Marīnī ruler, Abū^cInān, against the Christians. The Castillian king, Pedro, was occupied with dynastic troubles. He confirmed his truce of 751 made with Muḥammad V's father.¹²

This peaceful situation, however, did not last long. In 760 A.H. a revolt broke out against Muḥammad V. He had two brothers whom Ridwān had imprisoned in Alḥamrā'. Their mother sought the help of Ra'is Muḥammad, the head of a contingent of soldiers.¹³ A²Ra'is killed Ridwān and proclaimed Muḥammad V's brother, Ismā'il, as Sultān and himself as his regent. Ibn al-Khatib and other supporters of Ridwān were imprisoned.

Sulțăn Muhammad V, however, escaped to Guadix. There he received a visit from Abū'l Qāsim al-Sabtī, his former qādi who had joined the Marīnī

court. Abū'l Qāsim was sent by Abū Sālim, the Marini, who invited Sultān Muḥammad V to Fez to express his gratitude for the refuge he had received at the Naṣrī court when he fled from his brother Abū^cInān. Muḥammad V accepted the invitation. Abū'l Qāsim then proceeded to Granada to negotiate the safe conduct of the Sultān to Fez as well as the release of other prisoners, including Ibn al-Khatīb.¹⁴ The mission succeeded, and Muḥammad V along with his supporters arrived in Fez in 761 A.H. He received a warm welcome from the Marīnī Sultān.

In the meantime Ra' is Muḥammad, after assassinating Ismā'il, had assumed power. King Pedro defeated him in a battle. Shaykh al-Ghuzāt deserted to the Christian king, to escape the consequences that he feared would follow if he returned to Granada.

At the same time in Fez, during a revolt, Abū Sālim lost his life.¹⁵ Muḥammad V left for Andalus. To regain his throne Muḥammad V depended very much on the help of the amirs of Ronda and Malaga. The castle of Ronda which belonged to Andalus had been taken by the Marīnī regent ^CUmar b. ^CAbd Allāh. Muḥammad V, however, succeeded in regaining it.¹⁶ From there he proceeded to Malaga. Alliance of Ronda and Malaga in favour of the Sulțān assured his capture of Granada. ^{Al}Ra² īs Muḥammad, who saw himself pressed from both sides, decided to surrender himself to P edro. There he was treacherously killed. Thus, ground was prepared for the recapture of Granada; Muḥammad finally remounted the throne in 763 A.H. Muhammad V was no longer a youth, and the incident of deposition had been an instructive experience. In his second reign he seemed determined to make himself independent of internal as well as external powers.¹⁷

j L

> He decided to undermine the office of <u>wazir</u>. He succeeded in routing his <u>wazir</u>, ^CAli b. Yūsuf b. Kumātha, whom he had been obliged to accept as his <u>wazir</u> during his stay in Ronda. He sent Ibn Kumātha on a mission to the Marīnī court to get rid of him. On his way Ibn Kumātha heard the news of Sulṭān Muḥammad's successes. He tried, in vain, to instigate the rulers of Castille, Barcelona, and of Tunis against Muḥammad V, but he was finally captured in Castille and sent to prison in Fez.¹⁸

> After a while, Ibn al-Khaţīb joined Muḥammad V. Following lengthy secret talks and promises, Muḥammad V accepted him as <u>wazīr</u>.¹⁹ Ibn al-Khaţīb soon prevailed upon the Sulţān who charged Ibn al-Khaţīb with the responsibility of almost all the affairs of the government.

The office of <u>Shaykh al-Ghuzāt</u> was confirmed for ^CUthmān ^CAlī who had deserted ^{Al-}_Ra² is Muḥammad. Muḥammad V, however, had the same apprehensions regarding this office as he had had concerning that of <u>wazīr</u>. In 764 he suddenly took captive all the members of <u>Shaykh al-Ghuzāt</u>'s family and expelled them from the political domain. He appointed successively Abū'l Ḥasan ^CAlī b. Badr al-Dīn and ^CAbd al-Raḥmān b. Abī Sa^Cīd, both from Banū Marīn, to the office of <u>Shaykh al-Ghuzāt</u>, but reduced their powers drastically. As a matter of fact, almost all of the military campaigns against the Christians, which justified the title of ghāzī, were led by the Sultan himself.²⁰ The office of Qādī al-Jamā^ca was given to Qādī Abū'l Hasan al-Nubāhī, and the office of Kātib al-Sirr to the faqīh and poet, Ibn Zumruk.

In 767 A.H. Muhammad V decided to lead a series of campaigns against the Christians to establish himself as the defender of Islam.²¹ Some fortresses close to Malaga and Ronda were taken back from the Christians. Jaen was recaptured. The campaigns in the years 770 and 771 A.H. were carried out as deep into Christian territory as the neighbourhood of Sevilla. These campaigns brought a huge amount of booty to the Muslims.

The success of these campaigns was partially due to internal troubles in the Christian kingdoms, which did not allow them to attend to the defence of their borders. In this way Muhammad V's period remained generally safe from Christian attacks. In fact, we can say that the situation had reversed itself in the fact that the Christian kingdoms were in a defensive position against the attacks of Granada.

The same was true for his neighbours in Africa. Muḥammad V was no longer threatened by powerful neighbours. But this situation was partially accidental and partially, as we shall note below,²² due to his skillful manoeuvres respecting the political affairs of the African rulers.

From the above survey it can be noted that the strength of the Banū Naşr depended on two things: first, maintaining a balance between their neighbours by alternating peace treaties and court intrigues; second, by controlling the internal sources of power. We will review these two aspects of the Naşrī political structure on the following pages.

Political Structure

Foreign Relations

The Banū Naşr had Christian neighbours to the north and Muslim Berbers to the south. Among the Christians the more powerful kingdoms were those of Castille and Aragon. In Africa there were three kingdoms as mentioned before. In short, Banū Naşr had to deal with the Castille and Aragon on one side and with the Banū Marīn on the other.

From 643 A.H. onwards the Banū Nasr were vassals of the king of Castille. According to the conditions of the treaty, the Banū Nasr, among other things, were to pay an annual tribute whose amount fluctuated from 150,000 to 259,000 <u>Doblas</u>. In return, they were entitled to attend the Castilian court like Christian chiefs. Both parties agreed to supply troops to each other during wartime.²³

This status was humiliating both politically and financially, but the Banu Naşr were forced to accept and confirm it continuously, first to keep peace with the Christians and second as a check against the Banu Marin designs lest they repeat the role of the Murābițun and Muwaḥhidun.

This state of affairs had a social as well as an intellectual impact upon Granadian society. These treaties allowed an exchange of scholars and mystics on both sides.²⁴ The social impact of this situation is evident from the fact that the Granadian Muslims generally came to accept the Christian dress.²⁵

These changes must have been a challenge to the Māliki fuqahā'who were known for strict adherence to their tradition.

Relations With The Banū Marīn

 $\ln_{\Lambda}^{h\nu}$ early seventh century $\Lambda^{h\nu}$ Banū Naşr had depended more on the Banū Hafş but later, when the Banū Marīn grew stronger, they leaned towards the Banū Marīn. In 634 A.H. the Marīnī Sultan Manşūr crossed over to Spain in answer to the Naşrī appeal and defeated Sancho of Castille. On his return, he left behind several Marīnī clans to defend Andalus against the Christians.²⁶ These clans played a very active role in Naşrī politics because the office of <u>Shaykh al-Ghuzāt</u> remained in their hands.

The Banū Naşr needed Marīnī help against the Christians, but their relations were not always friendly. Each conspired constantly to weaken the other. Both provided political refuge to defecting princes, <u>wazīrs</u> and scholars from the other's camp. The Banū Marīn could dictate their terms²⁷ as long as they were strong, but the situation was reversed during the reign of Muḥammad V.

The Banū Marīn were heavily defeated by the Christians in 741/1340²⁸ and from that time onward were not in a position to stand in aid of the Banū Naṣr. The regular internal quarrels among the Banū Marīn during 759-774/1358-1373 weakened them still further. The following incident worsened terribly the relations between the Banū Marīn and the Banū Naṣr. The Marīnī <u>wazīr</u> ^CUmar b. ^CAbdAllāh, who was responsible for a series of dethronements and bloodshed during the period of 762-767, expelled the Marīnī prince ^CAlī b. Badr al-Dīn and his <u>wazīr</u> Mas^cūd b. Masā'i. They were welcomed in Granada by Sultān Muḥammad V; he even appointed Alī as Shaykh al-Ghuzāt. In the meanwhile Sultān Abd al-Azīz had taken all powers into his hands after killing ^CUmar b. Abd Allāh. He was apprehensive of prince Alī. He requested the Naṣrī Sultān to send the prince and his <u>wazīr</u> back to Fez. The Sultān refused, but Ibn al-Khatīb whom Sultān Abd al Azīz had taken into his confidence, prevailed upon Muḥammad V; yet the latter only agreed to imprison them. The Marīnī Sultān accepted but did not like this move. Ibn al-Khatīb, apprehensive of the intrigues against him in the Naṣrī court, was planning to escape. Sultān 'Abd al Azīz welcomed him in the Marīnī court. Now Sultān Muḥammad V requested Sultān ʿAbd al Ázīz to send Ibn al-Khatīb back again to Granada, but he refused. This disagreement soured their relations to the extent that from that point on both the Banū Marīn and the Banū Naṣr spent their efforts in staging intrigues against one another.

Muhammad V released the Marini prince and his <u>wazir</u> and sent the prince as pretender to the Marini throne. He even marched toward the Marini borders and captured Ceuta to stress his support for the pretender. He succeeded finally in staging a revolt and establishing his own choice on the Marini throne.²⁹ Thus Sultan Muhammad V succeeded in solving an almost centuryold problem. His successes against the Marinis brought further security to his rule as well as to the Granadian society in general.

...

Internal Political Structure

The Granadian political structure consisted of three major offices directly responsible to the sultan: Shaykh al-Ghuzāt, Wazīr and Qādī al-Jamāʿa.

<u>Shaykh al-Ghuzāt</u> The chief of the <u>ghāzī</u>s (warriors for the faith) actually was the office of the head of the armies, both regular armies and mercenaries. This office, according to Gaudefroy Demombynes, was "comparable to that of the <u>amīr al-'umarā'</u>in the late Abbāsī period."³⁰ The peculiar tribal structure and allegiance to the chief provided the <u>Shaykh al-Ghuzāt</u> with absolute power.

The office of <u>Shaykh al-Ghuzāt</u> was introduced to replace the power of the Banū Ashqīlūla who had been responsible for the establishment of the Naṣrī dynasty but who had soon fallen into the custom of revolting against the Banū Naṣr on frequent occasions. To counterbalance their power the Banū Naṣr welcomed the Marīnī clans left behind in Andalus by the Marīnī Sulṭān Manṣūr. The first Shaykh al-Ghuzāt was appointed from among these Marīnīs.

The <u>Shaykh al-Ghuzāt</u> was given vast powers as is evident from a <u>zahīr</u> (investiture)³¹ conferred upon Yaḥyā b. ^cUmar by Sultān Abū'l Ḥajjāj Yūsuf (733-755/1334-54). The titles mentioned in the investiture include: ^{the} Lillar of Power', ^{the} Sword of <u>Jihād</u>', 'The Supermost Head of Defence', 'The Bond of the Kingdom' etc. The part on the description of his authority reads as follows: "...He is the chief of the ghāzīs in spite of the differences of their tribes and the diversity in their manner of living. The promotions in their grades of acceptance will be determined by his approval... Their salaries will be determined by his assessment. Further allowances will be made to them by his confirmation and recommendation. In all, may God support him, he is the <u>gibla</u> (turning point) of their hopes, the balance of their deeds...and it is he with whom the kingness of the administration of their food and prosperity is sought".

^cUthmān b. Abī[·]1 ^cUlā' was the most powerful and illustrious <u>Shaykh al Ghuzāt</u> in Naṣrī history. ^cUthmān was the chief of the Banū[·]Ulā' clan of the Marīnī tribes in West Africa. He had been gathering forces against the Marīnī ruler Abū Yūsuf Ya^cqūb (685-706 A.H.) After a few gains ^cUthmān was heavily defeated in 707 A.H. and fled to Andalus with his contingents.³³ He was warmly welcomed in Granada. Despite the threats and the pleas of the Marīnī sultān to send ^cUthmān back to Africa for punishment, the Banū Naṣr bestowed upon him the office of <u>Shaykh al-Ghuzāt</u>.

^cUthmān soon came into conflict with the <u>wazīr</u> Ibn Maḥrūq. Ibn Maḥrūq succeeded in suppressing him temporarily. Soon, however, the situation reversed itself. ^cUthmān gathered his troops and besieged Granada. Alfonso, seizing the opportunity, captured a few border towns. Sultān Muḥammad IV (725-733 A.H.) was forced to be reconciled with ^cUthmān. To do that he had his <u>wazīr</u>, Ibn Maḥrūq, murdered. Muḥammad IV himself, however, met the same fate at a later point when, dissatisfied with the Sultān, ^cUthmān's ghāzīs assassinated Muḥammad IV in 733-34 A.H.

Muḥammad IV's son Yūsuf's attempt to replace Banū 'Ulā'with Banū Raḥū, another

sub-clan of the Banū Marīn, did not bring about much change. The <u>Shaykh al-</u> <u>Ghuzāt still enjoyed the same powers</u>. Shaykh ^CUthmān b. Yaḥyā of^{hc}_ABanū Raḥū participated in the plot against Muḥammad V, and supported the Sulṭān's rival. He was, however, defeated in a battle against the Castillians and took refuge with them. The Castillian king Pedro was an ally of the deposed Sulṭān Muḥammad V. He delivered Shaykh ^CUthmān to Muḥammad V who reinstated him in his post when the latter remounted the throne.

Muhammad, however, had decided to break the power of the <u>Shaykh al-Ghuzāt</u>. Consequently within a year he struck out at Shaykh ^CUthmān and banished the entire family from the political scene.³⁵ He appointed other individuals from Banū Marīn to perform the necessary functions, but he reduced their powers by taking two steps: first, he led most of the campaigns against the Christians himself, thus taking the credit of <u>Jihād</u> away from the <u>Shaykh al-Ghuzāt</u>. Second, he sent the <u>Shaykh al-Ghuzāt</u> on campaigns against the Banū Marīn,³⁶ thus discrediting them as ghāzīs since they fought against Muslims and their own kith and kin.

Wazir The wizāra was the second most powerful office in the Nașri political

structure. Ibn Sa^cid observed that the institution of <u>wizāra</u> in Um^aawi Andalus consisted of a group of notables who assisted the caliph by counsels and aided in the administration. One of them whom the caliph appointed his deputy was called <u>hājib</u>. This office became hereditary and continued within certain families. The designation of <u>wazīr</u> was lower than that of <u>hājib</u>.³⁷

During the Nasri period the emergence of the institution of Shaykh al-Ghuzat had

overshadowed the powers of the <u>hājib</u>. Moreover, the offices of <u>hājib</u> and <u>wazīr</u> were often combined. Some <u>wazīrs</u> even claimed to be regents of the minor sultāns whom they succeeded in bringing to the throne. Such <u>wazīrs</u> enjoyed the highest powers. Instances of such <u>wazīrs</u> are Ibn al-Hakīm al-Lakhmī during the reign of Muḥammad al-Makhlū[<]; Ibn Maḥrūq in the period of Muḥammad IV, Qā^jid ^A_Ridwān in the time of Abū'I Hajjāj Yūsuf and Ibn al-Khatīb during the reign of Muḥammad V.

: . •

÷,

Under the <u>wazīr</u> were <u>kuttāb</u> (secretaries) who held the various offices of civil administration. The <u>wazīr</u> also commanded the <u>shurta</u> or the city police.³⁸

Early Naşrī wazīrs such as Abū Marwān b. Şanādīd, who was the ruler of Jaen, and the Qā'id Abū ^{(Abd Allāh al-Ramīmī, who was the son of the ruler of Almeria, both wazīrs of ^{Ab-}_bGhālib Billāh, had powerful family connections. The later wazīrs were, however, men of learning, having no such powerful connections. This is why the wazīrs depended for their support on diplomatic influence. Their powers were often temporary. Whenever their plots failed, it proved easy to break their power. The wazīrs were invariably imprisoned, expelled or assassinated.}

Qādī al-Jamāʿa This was the most respected office in the plitical structure.

The Qādī al-Jamā^ca was responsible for the administration of justice, the inspection of markets and for regulating commercial contracts. The Qādī al-Jamā^ca also sometimes was the chief khatīb of Granada.³⁹

No executive powers such as the command of soldiers, police, etc., belonged to the Qaadi. It was rather supposed to be the duty of the Sultan to support a qadi's

judgment with his executive powers.⁴⁰ The Sultān and often the <u>wazīrs</u>, as well, interfered in the administration of justice; yet symbolically, the Qādī enjoyed the highest prestige in the political structure.

In spite of the absence of executive powers, the chief Qādī had vast influence in the affairs of the state as he was responsible for the appointments of a significant number of functionaries in the administration of judicial and religious affairs.

The real basis of the Qādī's power, as we shall see later in detail,⁴¹ lay in his being part of a sort of 'religious élite' which had grown in strength in the Umawī period and proven itself indispensable ever since.

Qādī al-Nubāhī's success in prosecuting the powerful wazīr Ibn al-Khatīb, is one of the recurrent examples of the powers of <u>qād</u>īs in the political structure of Muslim Spain.

As stated earlier, Sulțăn Muḥammad V was enraged by Ibn al-Khațīb's defection to Morocco. From certain accounts it appears that there existed a rivalry between <u>al</u>-Nubāhī and Ibn al-Khaţīb. Ibn al-Khaţīb, as Ibn Khaldūn has noted, enjoyed the highest powers after the collapse of the office of <u>Shaykh al-Ghuzāt</u>.⁴² He interfered with Qādī al-Nubāhī in many cases. It is evident from certain stories recounted by Ibn al-Khaţīb in his <u>Armāl al-Arlām</u>⁴³ and <u>Al-Katība al-Kāmina</u>, that Ibn al-Khaţīb went beyond the limits of politeness in ridiculing Nubāhī in the court. Publicly insulting the <u>Qādī al-Jamāra</u> must have undermined the office of <u>qadā</u>. This derision was not without the Sulţān's approval.⁴⁴ The Sulţān must have encouraged the <u>wazīr</u> for such derision to weaken the office of Qādī al-Jamāća. It was only after Ibn al-Khaṭīb had left for Morocco that Nubāhī could accuse Ibn al-Khaṭīb, in public, of heresy and burn his books. In this accusation, of course, he, too, was encouraged by the Sulṭān. The Naṣrī Sulṭān sent Nubāhī to Morocco to bring Ibn al-Khaṭīb back once more to Spain. The Sulṭān failed to bring Ibn al-Khaṭīb back to Spain but he finally succeeded in having him killed in Morocco.

This is how the Sultān eventually succeeded in removing a <u>wazir</u> who had become too powerful and, by using the <u>qadi</u> to his advantage, also achieved his designs to make himself independent of the offices of the Qādi and Wazir both.

Conclusion:

At the conclusion of this section we may say that the reign of Sultān Muḥammad V was relatively speaking a peaceful and politically stable period. This stability was gained by the skillful management of relations with the Christian neighbours and the Marīnī rulers, but more significantly, by the consolidation of the absolute rulership of the Sultān. The Sultān succeeded in achieving this goal by weakening and reducing the powers of the <u>Shaykh ai-Ghuzāt</u>, the <u>Wazīr</u> and the <u>Qādī</u> <u>al-Jamār</u>, which were the major offices of political significance. He used the influence of each office against the others to weaken them all.

Politically, the Qādā al-Jamā'a had been a very influential office, yet the Sultān was able to use it to consolidate his own power. This was possible because the religous authority of the <u>fuqahā</u>' on which the power of the <u>Qâdā</u> depended had been already weakened. This phenomenon is discussed in the next section.

SECTION TWO

SOCIO-RELIGIOUS DEVELOPMENTS

We noted earlier¹ how important a role was played by the jurists in political events during the early days of the Sultān Muḥammad V, especially by Qādī al-Sabtī and Ibn Marzūq. The significance of the jurists in the political affairs in this period was, in fact, a continuation of the role they had fulfilled from approximately the ninth century.

The Political Power of the Fuqahā'

The historians of Muslim Spain do not forget to point out the role of the <u>fuqahā</u>, in political affairs as a trait of Muslim history in Spain.² The various opinions about the significance of <u>fuqahā</u>, are not immediately relevant to our purpose. Nevertheless, in general, we learn that scholars have suggested three reasons for the political significance of the <u>fuqahā</u>.

First, some scholars such as Ibn Khaldūn³ and Goldziher⁴ argue that it was the conservatism of the Spanish Arabs that encouraged the spread of Mālikism and that eventually conferred significance on the <u>fuqahā</u>' since they were the bastions of this tradition.

The second reason, as suggested by Lopez Ortiz⁵, Hussain Monés⁶ and others⁷, was the need for the legitimization of their rule as was always felt by the Muslim

rulers of Spain. Lopez Ortiz argues that because of their breakaway from the Abbāsī caliphs, the Banū Umayya in Spain needed the support of religion to justify their caliphate. Mālik b. Anas, being an antagonist of the 'Abbāsī's, was the ideal choice for them.⁸ Monés furthers this argument more strongly in the case of Hishām I and Hakam I.

Contrary to the claim made by historians of Hishām I's piety, Monés argues that Hishām, in fact, chose religion for rather political reasons. Hishām felt himself weak in the struggle against the rightful claimant to the throne, Sulaymān, who enjoyed the support of the Syrian contingents of the army.⁹ Thus from political necessity on the part of the Umawi rulers there arose a class of <u>Ulamā</u> and Fugahā' who played a continuously important role in political affairs.

The third factor, suggested by L. E. Provençal ¹⁰ and Roger Idris¹¹, was the establishment of a kind of 'religious aristocracy' -- composed of <u>fuqahā</u>' and <u>'ulamā'</u> -- who comprised of the intellectual as well as the social élite in the capital by the time of Hakam I (180-206/796-822). When Hakam tried to reduce their influence, they staged two insurrections in Cordova¹². In these revolts the <u>fuqahā</u>' had the support of a number of aristocrats in the court as well as the social stage the power of the fuqahā'.

We cannot agree that any one of these factors alone can sufficiently explain the influence of the <u>fuqahā</u>', particularly in the Nasrī period. Nevertheless, the third factor probably clarifies the phenomenon better than the others.

It is difficult to classify the generality of the Spanish people in the manner of Ibn Khaldun as having been primitive and conservative. There is much evidence to the contrary. Especially in the Naşrı period the Spanish people were quite flexible in accepting their Christian neighbours' way of life, particularly with regard to dress and recreational activities¹³. Conservatism was also absent from their ever-changing practices in trade and commerce¹⁴. There was conservatism, of course, in the intellectual attitude and academic activities of the élite. The latter, however, were probably the result, not the cause, of the conservatism of the fugaha'.

Similarly it is hard to maintain that the rulers' alliance with the <u>fuqahā</u>, was based on the former's need for religious legitimacy. In a society where the rule of a usurper can be justified in the political theory by equating <u>de facto</u> with de jure¹⁵, the need of a religious institution for that purpose is not very great.

In the case of the Naṣrī claim to the legitimacy of their rule, stress fell upon their Arabness rather than on any religious doctrine. The tension between Arabs and Berbers had been a salient feature of Muslim Spanish history. The two Berber dynasties, the Murābiţūn and the Muwaḥḥidūn, had pushed the Arabs aside. With the decline of the Muwaḥḥidūn, the Arab element rose again, as the rise of Banū Hūd and Banū Naṣr manifested. These Arab tribes were supported by the local Spanish element and by the Arab aristocrats. They did not trust the Berbers; they sought the Berbers' help only temporarily. While the Berbers were
inclined more toward religion and piety¹⁶, as expressed by their zeal for <u>Jihād</u> and <u>Taşawwuf</u>, the Banū Naşr laid stress on genealogical nobility¹⁷. The founder of the Naşrī dynasty was called <u>Marwānī</u> by a contemporary historian.¹⁸ This shows that the Banū Naşr in all probability wanted to present themselves as a continuation of the Banū Umayya. Later, however, they linked themselves with the Khazraj tribe of Madīna. Ibn al-Khaṭīb established the proof of their genealogy from earlier sources¹⁹ and Ibn Zumruk recited eulogies narrating the merits of the Khazrajīs in the days of the Prophet.²⁰ This very fact that they stressed their descent from the <u>anşār</u> would have been detrimental to their cause if they had been seeking religious support for their legitimacy, in view of the Anṣār. The nature of the argument shows how much significance the Banū Naṣr gave to the religious aspect of the legitimacy of their rulership.

The foregoing discussion was necessary to show that the need of legitimacy existed but that it was not sought necessarily from the <u>fuqahā</u>'. The rulers needed the support of the <u>fuqahā</u>' because the latter, through strong family relations and land holdings, had established themselves in Spain as a political power. We need not go into these details; what interests us here is their strength as a political group. We will briefly review the factors of their strength.

The Factors in the Political Strength of the Fuqahā'

The high status that the <u>fuqahā</u>' enjoyed in Andalus is evident from the fact that the appelation "faqih" had acquired a sense of nobility. Ibn Sa'id points out in his narrative of the Andalusian society that:

Į.

The appelation of 'faqih' is most honourable for them, so much so that if they want to make an honourable mention of their grand amir (sultān), they call him 'Faqih'. At present a faqih in the West is what a qādi is in the East. They even sometimes call the kātib (secretary), a grammarian and a linguist faqih because it is the highest appellation for them.²¹

The factors that contributed to the sustenance of <u>fuqahā</u>'s political power were mainly three:

- 1) The control of a number of important lucrative offices in the political system;
- 2) The control of the institutions of learning;
- 3) The control of the movement of free thought,

It was through the operation of these factors that the <u>fuqahā</u>' could preserve the Mālikī tradition in its conservative mold and hence maintain their power. When they lost control of these factors in the fourteenth century they could no more maintain their religious authority and hence their political power. We will briefly review these elements in the power of the <u>fuqahā</u>' in the following pages. The highest religious and judicial office was that of $Q\bar{a}d\bar{d}al - Jam\bar{a}a$, the appellation of the chief $q\bar{a}d\bar{d}a$ in Granada. The historians stress that it was the noblest office in the political structure. The evidence for the truth of the claim is to be found in the generous salaries, the ceremonial investitures, and the lengthy formal decrees of appointment given to $q\bar{a}d\bar{d}s$. The $Q\bar{a}d\bar{a}al - Jam\bar{a}a$ also enjoyed a wide range of prerogatives.²²

Beside the administration of justice, the <u>fuqahā</u>'were officially attached to courts as <u>muft</u>is (jurisconsults), <u>mushāwirs</u> (consultants) and <u>wuththā</u>gs (formularies and notaries).²³ The administration of religious and trust properties was also in their hands. Whenever a ruler made a donation for a special purpose, he appointed a <u>faqih</u> to supervise it. The appointments of Abū Abd Allāh al-Haffār (811 A.H.)²⁴ and Ibn al-Qabbāb (779/1378)²⁵ in Granada were of such nature.

The inspection of trade and commerce was also the domain of the <u>fuqahā</u>. They were responsible for fixing prices and for the quality and weight of commodities in the market. The particular teachings of Islamic law²⁶ against <u>Ribā</u> (usury) and <u>Qimār</u> (speculation) prohibited a number of transactions which thus required the supervision of experts in the law, i.e. the <u>fuqahā</u>?²⁷

These prohibitions also extended to transactions involving money exchange and minting. The <u>fuqahā</u>' were therefore required also to supervise the minting of coins. The important offices in the mint such as <u>Nāzir al-Sikka</u> were held by <u>fuqahā</u>: ²⁸

The office of $\overset{hc}{L}$ Chief <u>Khatib</u> was next to that of $\overset{hc}{Q}$ <u>adi al-Jamā'a</u> in importance in the capital city of Granada as well as in other cities and towns in the kingdom. Often both offices of <u>adi</u> and <u>khatib</u> were held by one person.²⁹ Since in Islamic history the <u>sikka</u> (coining) and <u>khutba</u> (Friday sermon) which became the vehicle of the announcement of the ruler's name, had become the formal signs of a claim to rulership or to allegiance of one ruler to another ruler, the <u>khatib</u> had also become a kind of political office.³⁰ Attached to the office of <u>khatib</u> were a number of other religious offices such as that of the mu'adhdhin, etc.

All the above-mentioned offices were lucrative, and often tracts of land, commensurate to their rank, were attached to these offices.³¹ This land ownership also contributed to the political power of these office holders.

Intellectual Control

The status of the <u>fuqahā</u>' established by their function in the political administration was sustained by their control of intellectual life. This was achieved mainly in three ways: 1) The control of the institutions of learning,

- 2) The suppression of any movements of pure rationalism,
- 3) Opposition to Tasawwuf and Tariqas as a threat to the political, as well as the economic, system.

We will explain these measures briefly in the following paragraphs:

1) The Institutions of Learning

Ibn Sa^cid, who visited Andalus in the early Nasri period, depicts the conditions of learning in the following words: As to the conditions of the Andalusians in respect of the art of sciences, the truth of the matter is that they are most eager people in this regard. ... The scholars enjoy the noblest rank among the élite as well as the common people... Despite the fact that the Andalusians do not have school (<u>madāris</u>) to help them in seeking knowledge, they rather study (learn) all the sciences in mosques on paying fees. Thus they read in order to learn, not in order to earn a stipend.³²

Ibn Sa^cid praised the Andalusian system as leading to learning in contrast to the system of the <u>madrasa</u> in the Muslim East where the interest of the student was monetary rather than learning.

This conclusion of Ibn Sa^cid stands in contrast to that of Ibn Khaldun who praised the system in the East saying that the system of $\frac{\hbar}{\hbar}$ encouraged learning and made it possible to study even for those students who could not afford to pay fees to individual teachers. On the other hand, the system in the West limited the spread of learning and eventually resulted in the decline of the sciences.³³

Neither Ibn Sa'id nor Ibn Khaldun mentions one significant fact – that learning itself could not have been the sole aim of all the students. The majority of them graduated, thereupon to be given various offices in the administration. The factor that must be emphasized here is that the teachers who were mostly fuqahā' had more influence in the system of the East in comparison to the West.

The institutions of learning in Andalus was completely in the hands of the <u>fuqahā</u>. They were absolutely independent in choosing the materials of teaching in the manner of teaching and the assessment of achievement of the pupils. Shāṭibī, dealing with the question of learning, in fact, discouraged the method of learning from the books without a teacher.³⁴ This system was advantageous to the <u>fuqahā</u>' in two ways. First, it established their influence and supremacy over the people. The <u>fuqahā</u>' could not have had this advantage in the <u>madrasa</u> system, because in that system they could not be as independent as they were without the <u>madrasa</u>. Because of the absence of an institutionalized system of higher learning the pupils had to depend on the teachers if they were to get diplomas of graduation.

Second, the Andalusian system made possible the preservation of tradition and strict adherence to it, as well as the control of any ideas or movements that might change the tradition.

The <u>fuqahā</u>' in the West were certainly aware of their advantages when they opposed the establishment of <u>madāris</u>.³⁵

The institution of the official <u>madrasa</u> was introduced quite late in Spain. Provençal mentions that the first <u>madrasa</u> was established by the Qā'id^A_Ridwān (d. 760/1359) the <u>Hājib</u> of Abū Yūsuf al-Hajjāj (733-755/1333-1354).³⁶ This move was strongly opposed by a number of scholars. Two main arguments were advanced in this respect. First, that it was a <u>bid</u>^ca (innovation), hence prohibited; second, that it suppressed the freedom of the <u>culamā</u> and hence the independence of cilm (scholarship).³⁷

After the establishment of <u>madrasas</u>, the '<u>ulamā</u>' and <u>fuqahā</u>' gradually lost their independence. The change did not immediately, however, affect their aristocratic status; but their control over intellectual movements and their resistance to Taşawwuf certainly relaxed. It was after the establishment of madaris that Taşawwuf and <u>Sufi Tariqas</u> gained a wider following in Granadian society.

2) Control of Intellectual Movements

Again the same Ibn Sa'id says that:

"They (the Andalusians) take part in every science with the exception of philosophy and astronomy. These are specially enjoyed by the élite, but they do not show this (interest) in public for the fear of the common people. Because as soon as it is stated that 'so and so studies philosophy' or 'practices astronomy', at once he is declared Zindia (heretic), and his days are numbered (qayyadat 'alayhi anfāsuhū). If someone showed skepticism (zāla fi shubhatin) the people would stone him to death or would burn him alive long before his case was brought to the Sultān." 28

Ibn Sa^cid's observation is supported by stories that frequently refer to an aversion of philosophy. Ibn Khaldūn narrates that his teacher 'Ābili used to teach philosophy to Ibn 'Abd al-Salām in secret.³⁹ The condemnation of the study of philosophy was a very common theme in the literature written by fuqahā'.⁴⁰ This antagonism had grown to such an extent that even ^{al-}_AGhazālī's works were counted as being philosophical.⁴¹ One of Shāțibī's teachers, Sharīf^d_ATilimasānī, on one occasion was forced by his students to use a certain book by ^{al-}_AGhazālī. He dreamt the same night that he was soiling his books in filth.⁴²

The outstanding case in Shāṭibī's lifetime was the condemnation of the Wazir Ibn al-Khaṭīb. We need not recount the event which has been mentioned earlier. Qāḍī *2*-Nubāhī was asked to bring charges against Ibn al-Khaṭīb.⁴³ He declared the latter as heretic because of his indulgence in philosophy and other such matters. Qāḍī *2*-Nubāhī's attitude to philosophy can be learnt from the following passage in his book pertaining to the administration of justice and the biographies of <u>Qādī</u>s:

If something relating to the philosophical schools contradicting $\frac{Sharisa}{S}$ or something similar to that is found in someone's handwriting, then the practice (<u>hukm</u>) in this respect is to study the written material. If it is clear that it is the opinion of the writter and (something) to which

4

- - j he agrees, even though he may deny it verbally, the case will be decided on the basis of the written material. ... If this writing is found only to quote these philosophical schools without relating the statement to the writer... who could be worse than the man who possesses such books... such books must be burnt and the man must be punished..." 44

Towards the middle of 773, Qādī al-Nubāhī announced his <u>fatwā</u> about the books composed by Muḥammad b. al-Khaṭīb, relating to beliefs and morals. These books were burned in the presen**ce** of the <u>fuqahā</u>' and <u>mudarrisīn</u> (teachers) and others from the same class as the <u>fuqahā</u>'. "This happened because the aforementioned books contained articles that necessitated this action." ⁴⁵

Sulțān Muḥammad, assisted by Qāḍī Nubāhī and Ibn Zumruk finally succeeded after a few years struggle to have Ibn al-Khaţīb charged in the Marīnī court as a heretic. He was treacherously killed in prison and then burnt.⁴⁶

Ibn al-Khaţīb's tragic death illustrates the extent to which the <u>fuqahā</u>' could go in their opposition to philosophy. The case of Ibn al-Khaţīb also provides evidence to the fact that the reason for opposing philosophy and such trends was to preserve the supremacy of the <u>Sharī</u> which was the religious authority. These facts are to be found in al-Nubāhī's letter to Ibn al-Khaţīb which has been preserved in Nafḥ al-Ţīb. Al-Nubāhī charged Ibn al-Khaṯīb saying: "I had spoken to you a number of times about your pamphlets (manuscripts) in which you invoked innovation (heresy) and made fun of the Sharīća. I urged you to tear them up and burn them.

This unfortunate office tenure (Qāḍi, Nubāhī's period of qaḍā' during lbn Khaţīb 's wizāra) endured the nonsense resulting from your ridiculing the rules of <u>Sharī'a</u>, and your scorn at matters of religion...some of such cases are the following: one of them was the case of lbn al-Zubayr who, after payment of his dues, was sentenced to death on account of heresy (<u>Zandaqa</u>) despite your disdaining such a decision.

Ņ

Another case was that of Ibn Abi'l 'Aysh, detained (<u>muthaqqaf</u>) in prison on account of his heterodox statement, one of such heterodoxies was that he cohabited with his wife after pronouncing the formula of triple divorces, because he claimed that the prophet himself commanded him to mate with her. You sent one of your men to secure the escape of Ibn Abi'l 'Aysh from the prison with no consideration of others. Another of such cases was that one young man related to you was prosecuted on the charge of murder. I could not do anything but imprison him according to the requirements of religion and the decision of <u>Sunna</u>. You detested this judgment. You imprisoned the plaintiff and immediately released the above-mentioned young man." 47

For a better understanding of the contents of this letter it must be pointed out that Ibn al-Khaţīb was very much distrustful of the <u>fuqahā</u>². His reasons for this attitude were the <u>fuqahā</u>'s general ignorance of the Arabic language, the absence of piety and too much concern for the mundane matters. He wrote a few treatises combining satire and criticism on the practices of the <u>fuqahā</u>².⁴⁸ The main targets of his writings were the qādī Ibn al-Hasan al-Nubāhī and qādī Ibn al-Qabbāh. It is evident from Nubāhi's letter that Ibn al-Khaţīb did not agree with the <u>fuqahā</u>² in condemning heretics to death. His interference in the implementation of court decisions was considered as ridicule of the <u>Sharī¢a</u>. Ibn al-Khaṭīb's boldly favourable attitude towards philosophy and pure thought was made possible among other factors, by the introduction of Rāzism into Western Mālikism in the thirteenth century.

Faktral-Din Razi was responsible for raising the status of <u>Kalām</u> to bring it closer to philosophy, ⁴⁹ but his influence also meant the revival of an interest in philosophy - a forbidden science among the conservative orthodox. Rāzism was introduced to Mālikism mainly through <u>Usūl al-fiqh</u>. This made the acceptance of Rāzism easier, and the resistance to pure philosophy, though it continued, but grew weaker and weaker.

In Eastern Mālikism this impact manifested itself in two works on Usūl al-fiqh which were in many ways based on Rāzi's work on Usūl al-fiqh, Al-Mahsūl. One of these works was by Ibn Hājib (d. 646 A.H., Alexandria), <u>Muntahā al-Su'. I wa'l-'Amal fi 'Ilmay al-Uşūl wa'l-Jadal</u>. The second was the work by Ibn Hājib's pupil Shihāb alDīn al-Qarāfī (d. 684 A.H) 50 <u>Tanqīḥ al-fusūl</u>. Both soon became very popular Usūl texts of the Mālikī School. Ibn Hājib's work had gained currency even in his life time. Consequently he had to prepare an abridged version of it.⁵¹ This abridged work on Usūl along with another short work on furū' were called <u>Mukhtaşar aşlī</u> and Mukhtaşar far'ī since they were used as texts in <u>madāris</u>.

Ibn Hājib's <u>Mukhtaşars</u> were introduced into the Muslim West by one of his well known disciples, Nāşir al-Dīn al-Mishdhālī (d. 731 A.H.)⁵² He was one of the three Western scholars who travelled to the East and who served as an agent

 for the influence of Rāzism on Māliki thought. The other two were Ibn Zaytūn and al-Haṣkūni.⁵³ In the West more attention had been paid to the study of fiqh and Arabic grammar, but under the influence of these scholars <u>Kalām</u> began to be given equal attention.

Philosophy was also making inroads, but it was still tabu. Some stories, as told by the biographers of this period, indicate that philosophy and other rational sciences were eagerly sought after by certain individuals, but in secret. Such secretly perused texts included those by Ibn Sinā and al-Fārabi.⁵⁴

Among the above-mentioned scholars Mishdhālī seems to be critical of Rāzism, although he retained his interest in philosophy. His son, Abū Manşūr al-Zawāwī's and al-Sharīf al-Tilimsānī who were in Mishdhālī's circle of influence both show this critical attitude towards Rāzī and exhibit favour towards the prepatatic school of Islamic philosophy.

Such trends were encouraging freedom of thought and general intellectual activities. Yet, what probably accelerated the spread of movements of free thought the most was the rise of Şūfī Țarīqas. Even the Mālikī <u>fuqahā</u>'seem to have failed in their resistance to <u>Taşawwuf</u> which encouraged a relaxed attitude towards the strict legalistic tradition of Mālikism. The reasons for the rise of this phenomenon are dealt in the following pages.

3) Tasawwuf

120

The absolute supremacy of $\underline{Shari}^{c}a$, the palladium of the power of the religious authority of the fuqahä', was threatened by philosophy as well as

<u>Kalām</u> insofar as these two sciences undermined the authority of <u>Shari'a</u> as the only guide to life. <u>Taṣawwuf</u>, however, probably presented a more direct threat to <u>Shari'a</u> than any other movement of thought. The emphasis on piety, religiosity and moral commitment appealed to intellectuals as well as to the common people. The rise of <u>Taṣawwuf</u> in their midst was, therefore, naturally considered a threat by the Mālikis in the West.

In addition to this consideration, certain events heightened this feeling of danger. In the twelfth century when Mālikism had been re-established by the Murābiţūn, the <u>fuqahā</u>' had begun the purge of <u>Taṣawwuf</u> from Andalus. Among the <u>sūfī</u>s denounced by the <u>fuqahā</u>', the following three were prominent: Abū Bakr Muḥammad from Cordova, Ibn al-^cĀrif from Almeria and Ibn Barrajān from Sevilla. They were persecuted, and all three died in prison. Ibn Barrajān had criticized the Mālikī <u>fuqahā</u>' very severely for their neglect of <u>Hadīth</u>. He succeded in gathering enough supporters in Almeria to form an opposition that was directed primarily against the <u>fuqahā</u>'.⁵⁵

Another such uprising against the ruling class and the <u>fuqahā</u>'was led by another <u>sūfī</u>, Abū'l Qāsim Ibn al-Qasiyy, a disciple of Ibn al-ʿĀrif (1088-1141). This insurrection took place in Algraves region (Southern Portugal) in 1141. Ibn Qasiyy was killed in 546/1151).⁵⁶

Viewing <u>Taşawwuf</u> in the perspective of these uprisings, the <u>fuqahā</u>'naturally considered <u>Taşawwuf</u> a threat against Mālikism and hence against themselves as a class. One significant victim of this opposition to Taṣawwuf was al-Ghazāli's book, <u>Iḥyā 'Ulūm al-Dīn</u>. One of the earliest reactions to ${}^{a\ell}_{A}$ Ghazālī's <u>Iḥyā</u> was that of Abū Bakr ${}^{a\ell}_{A}$ Jurtūshī (d. 520 A.H.) who wrote a treatise <u>Al-Tibr al-Masbūk</u> refuting ${}^{a\ell}_{A}$ Ghazālī's <u>Iḥyā</u>. ⁵⁷ The aforementioned <u>şūfī</u>, Ibn al-ʿĀrif, was the first to interpret ${}^{a\ell}_{A}$ Ghazālī's <u>Iḥyā</u> in the West. ⁵⁸ Along with the persecution directed against him came the suppression of <u>Ibyā</u>. In 537 A.H. ^cAlī b. Yūsuf b. Tāshufīn, who also persecuted Ibn Barrajān and other <u>şūfī</u>s, ordered that all copies of <u>Iḥyā</u> be burnt in public. ⁵⁹ Qādī ^cIyād (d. 544 A.H.) also issued a <u>fatwā</u> in favour of burning <u>Iḥyā</u>. Abū'I Ḥasan ibn Ḥirzihim prohibited the study of the <u>Iḥyā</u> and ordered that all copies of it be burnt. ⁶⁰

1.

Like other movements of free thought Taşawwuf continued to be considered dangerous both by rulers and fuqahā' until Muwaḥḥidūn toppled this alliance. Although the religious views of the Muwaḥḥidūn, because of their stress on the Qur'ān and <u>Sunna</u> did not allow absolute freedom to pure thought, yet Mālikism definitely lost its supremacy. Especially in Ya'qūb al-Manṣūr's (580-590 A.H.) reign, a sort of war was declared on Mālikism⁶¹

The Muwaḥḥidun could not, however, destroy the power of the <u>fuqahā</u>' in Spain. It grew stronger. The best illustration is the fact that Muwaḥḥid Sulṭān Manṣūr under the pressure of Mālikī <u>fuqahā</u>', was forced to expel Ibn Rushd, for the

During this period another movement was gaining force. It grew much stronger

in the period of the decline of the Muwaḥḥidūn. We refer to the establishment of <u>sūfī ribā</u>ts. As G. Marçais⁶³ has pointed out, originally the <u>ribā</u>t was a military institution, but the mystic movements which began in eleventh century and bloomed in the thirteenth century in North Africa, changed the nature of the <u>ribā</u>t. The volunteers for Jihād in the <u>ribā</u>ts were also connected with <u>Sūfī</u> <u>Tarīqas</u>. The <u>Ribā</u>t, thus, was no longer a military post but also a place for ascetics and travellers. By the thirteenth century the <u>ribā</u>ts were also transformed into <u>Zāwiyas</u> or centres for certain <u>Sūfī</u> tarīqas.⁶⁴ By that time every <u>ribā</u>t had a resident <u>s</u>ūfī-Shaykh.

This phenomenon had an effect on the fuqahâ' intellectually as well as socially.

Spain had resisted <u>Taşawwuf</u> successfully until in the thirteenth and fourteenth centuries we find travellers and biographers mentioning the emergence of a number of <u>Zāwiyas</u>, notable <u>sūfis</u> and a number of works on <u>Taşawwuf</u>, all in Spain. Ibn Baṭṭūta mentions, among other such centres of sūfism in Muslim Spain, two <u>Zāwiyas</u> in the vicinity of Granada: <u>Zāwiya Maḥrūq</u>, and <u>Rābiṭa al-'Uqāb</u>.⁶⁵ Two of the significant works on <u>Taṣawwuf</u> in the fourteenth century were written by Spanish <u>sūfis</u>; Abū Isḥāq Ibrāhīm b Yaḥyā al-Anṣārī (d. 751 A.H.) of Murcia, <u>Zahrat al-Akmām</u> and Abū ʿAbd Allāh Muḥammad b. Muḥammad al-Anṣārī al-Mālaqī's (d. 754 A.H.) <u>Bughyat al-Sālik fī Ashraf al-Masālik fī marātib</u> <u>al-Şūfiya wa Ṭarā'iq al-Murīdīn</u>.⁶⁶

This phenomenon affected the intellectual as well as the social status of the <u>fuqahā</u>'. The emphasis on piety and simple living in their personal lives by $\frac{\pi}{k}$, $\frac{\pi}{k}$ life style made \int_{5}^{4c} offs more popular than the <u>fuqahā</u> among the common people. The rising influence of \int_{5}^{4c} offs among the people and especially among the Berber mercenary volunteers for <u>Jihād</u> was also recognized by the rulers who, to establish their piety and influence among the warrior tribes, began to give attention to <u>sūfī Shaykhs and ribāts</u>.⁶⁷ The <u>fuqahā</u>' also acknowledged this change, and some of them began to drift towards <u>Taṣawwuf</u>. This trend is evident from a number of <u>fatwā</u>s which mention the popularity of <u>sūfism</u> among the <u>fuqahā</u>'.⁶⁸

4

The impact of <u>Taşawwuf</u> can be seen in two principal ways. First, the <u>suffis</u> did not abolish the <u>Shari'a</u>, but they undermined the status of the <u>fuqahā</u>', by their emphasis on principles of moral commitment (<u>wara'and zuhd</u>) to one's obligations. The <u>fuqahā</u>'s treatment of obligations was rather legalistic. Second, instead of limiting themselves to the figh books, the <u>suffis</u> appealed to the <u>Qur'ān</u> and the Sunna.

Both of these aspects affected the <u>figh</u> tradition. The most obvious influence can be seen in the discussions on <u>usul al-figh</u> in this period. The <u>fugahā</u>' had to make concessions to both principles. Qarāfī discussed <u>zuhd</u> and <u>wara</u>'as one of the bases of <u>figh</u>.⁶⁹

Ibn ^cAbd al-Salām's legal theory is more illustrative of this accommodation.⁷⁰

The influence of <u>Tasawwuf</u> had grown very strong by the thirteenth century. At the same time with the passing away of the Muwaḥḥidūn, Mālikism was also rising again. But this rise of Mālikism could no more be a continuation of the past tradition. Mālikism now faced many challenges; social as well as theoretical. Hence in this period <u>fiqh</u> and <u>Taşawwuf</u> both are actively present on the scene, and both are alive with a rejuvenating spirit. The Banū Marin and Banū Hafş who had succeeded the Muwaḥhidūn, realizing the force of both movements, made steps toward combining the two.⁷¹ They encouraged the <u>fuqahā</u>' to concede to Taşawwuf. They also began to endow the ribāţs with large trusts.

The <u>fuqahā</u>', realizing the situation, soon became themselves involved in <u>Taşawwuf</u>, but they still held to the supremacy of <u>Shari'a</u>. A typical example of this rapprochement was the formation of a new <u>sil_sila</u> (chain of a <u>tariqa</u>) whose connection with the Shādhiliyya <u>Tariqa</u> is discussed below, which combined the <u>sūfis</u> and the <u>fuqahā</u>'. Abū 'Abd Allāh₁^{Al} Maqqari (d. 758 A.H.), a famous jurist, is also noted for his work on <u>Taşawwuf</u>, <u>Al-ḥaqā'iq wa al-raqā'iq</u>.⁷² Ibn 'Abbād₁^R undī (d. 792 A.H.), the famous Shādhilī sūfī, was one of <u>al-</u> Maqqari's disciples of whom he was very proud.⁷³

Ae-Maqqarī, along with his lectures on figh, also initiated his pupils into his <u>silsila</u> of <u>Taşawwuf</u>. The initiation was done with a symbolic act in which the <u>shaykh</u> placed a morsel of food into the mouth of the disciple. A most significant indicator of the new conjunction between <u>fuqahā</u> and <u>sūfī</u>s is to be seen in the names comprising this silsila.

Maqqarī -- Abū 'Abd Allāh al-Musfir -- Abū Zakariya al-Maḥyāwī -- Abū Muḥammad Şāliḥ -- Shaykh Abū Madyan -- Abū 'I-Ḥasan b. Ḥirzihim -- Ibn al-'Arabī -- Ghazālī --Abū' I Ma'ālī -- Abū Ṭālib Makkī -- Abū Muḥammad al-Ḥarīrī (sic) -- Junayd -al- Saqaṭī-- Ma'rūf Karkhī -- Dā'ūd Ṭā'ī -- Ḥabīb Ajamī -- Ḥasan Baṣrī -- Alī b. Ṭālib -- Rasūl Allāh.⁷⁴ This chain has been subjected to criticism by some authors mainly because of gaps in the chain between Abū'l Ma'āli and Makkī. Paul Nwiya, after comparing the presentation of this chain as given by Shāţibī with those given by others, maintains that it belongs to the Shādhilī Order which became better known after Ibn 'Abbād.⁷⁵ The chain comprises four parts: the first part consists of Maqqarī and Musfir both primarily <u>faqīh</u>s; it is connected with the second part comprising a chain from Maḥyāwī to Abū Madyan – primarily <u>sūfi</u>s. They are connected again with the third part consisting of mainly <u>fuqahā</u>; starting with Ibn Ḥirzihim to Abū'l Ma'ālī. They are then connected with the traditional chain of early <u>sūfi</u>s, through Abū Ṭālib Makkī.⁷⁶

Y.

Nwiya's suggestion about the connection of this chain with the Shādhiliya, together with his conclusion that Ibn 'Abbād's reanimation of the Shādhiliyya was a revival of the early sūfism of Muḥāsibī, also partly explains the compromise of the sūfīs with the fuqahā' in order to exclude the more comprehensive and radical type of sūfism, such as that of Ibn 'Arabī which the fuqahā' considered a threat to the supremacy of Sharī'a.

Having found this compromise possible, the <u>fuqahā</u>' eased their opposition to <u>Taşawwuf</u> as such. There was, yet, another aspect of <u>Taşawwuf</u> which continued to threaten their status. This threat can be seen in three ways. First, <u>Tarīqa</u>-<u>Taşawwuf</u> required total submission to the Shaykh. This submission undermined the religious authority of the <u>fuqahā</u>'. One event (probably an anecdote) illustrates this tension: Qadi Abu'l Qasim al-Sabti had two sons. One, Abu'l Abbas Ahmad became qādī, the other, Abū'l Maʿālī chose the path of "qawm" (sūfīs). He never used or ate anything at his brother's house. After many years he visited Zāwiya Mahrūq in the outskirts of Granada. He saw Shaykh Abū Ja far Ahmad al-Mahdud and asked him if he could explain a mystery that had been worrying him. The mystery was that he had a torch that always showed him light, but suddenly he lost it. The Shaykh asked the first person entering the Zāwiya to answer that question. This person who appeared to be an illiterate villager answered that Abū'l Ma'ālilost this torch as punishment for some of his actions. After a number of questions it was revealed that Abū'l Ma'āli had taught someone the Divine name of al-Latif which he was not permitted to do. A curse fell on him as a consequence. He became Qad al-Jama'a and died a worldly man.

The second aspect of the threat to the <u>fuqahā</u>^{*} was that a number of <u>sūfī</u> practices such as <u>dhikr</u> and <u>samā</u>^{*} virtually substituted for the rituals prescribed by <u>fiqh</u>. This could not be tolerated by the <u>fuqahā</u>^{*}. Shāțibī goes as far as to declare insistence on such practices in defiance of <u>Sharī^{*}</u>(a, to be <u>Kufr</u>, and condemns the practitioners to death.⁷⁸ To add to the offence caused by these practices, which were considered bid^{*}a by the <u>fuqahā</u>^{*}, another important development took place.

In the thirteenth century the celebration of the Prophet's birthday was introduced into the Muslim West. This celebration took place in mosques. The poets wrote and recited for the occasion. Various forms of <u>dhikr</u> and <u>samā</u>^c were also part of the celebration. A significant factor in this development was the patronage that rulers provided for this celebration.⁷⁹ The <u>fuqahā</u>² could scarcely afford to offer strong resistance to these ceremonies in view of the wide popularity of this "innovation" among all groups of people. The situation forced them, therefore, to revise their stand on bid^ca.

The third aspect of the threat was economic. As we mentioned earlier, generous donations and trust properties were given to zāwiyas and <u>ribāt</u>s.⁸⁰ This wealth attracted a number of devotees as well as travellers. Ibn Bațtūța came across <u>sūfi</u>s in these centres from almost all corners of the Muslim world.⁸¹ <u>Fuqahā</u>'were appointed for the supervision of the expenses of such donations, although the supervision and maintenance of such properties was left to the <u>shaykh</u> of the zāwiya and his associates.

Some <u>fuqahā</u>, resisted the temptations of <u>Sūfī</u> <u>tarīqas</u>. According to these <u>fuqahā</u>, the <u>Sūfī</u> centres were attracting and encouraging idleness in the society. For many devotees asceticism meant to forsake all worldly occupations and spend one's life in some <u>Zāwiya</u>. The finances of the <u>Zāwiya</u> made it possible to live in such a manner. This practice, however, was creating a large number of unproductive elements in the society who were living on the labours of others. For the already stringent economy of Granadian society this was a very heavy burden.

This economic burden becomes very significant as we shall see that the Granadian economy was in process of changing from an agricultural to a commercial and mercantilistic economy. Even the rural areas could no more support the maintenance of such a burdensome institution as the <u>Sūfī</u> Zāwiya or <u>ribāt</u> had become. The problem became acute in the days of Shātibī. A distinct economic view of the matter, in contradistinction to the older political and theological view that had motivated the fuqahā'to oppose Şūfism now came to be.

The inhabitants of a small town Qanālish, ⁸² an agricultural town on the borders of Aragon, sought a <u>fatwā</u> concerning the <u>Shar'i</u> attitude towards a <u>zāwiya</u> αl -<u>Ghurabā</u>'in their vicinity. The Chief Qādī Al-Balfīqī answered vaguely, justifying the existence of such an institution. The Chief Muftī Ibn Lubb countersigned the <u>fatwā</u>. The people of Qanālish, however, mounted a protest against the <u>fatwā</u> accusing both <u>muftī</u>s of subjecting the people to an unnecessary burden.⁸³

The same <u>lstifta</u> was then sent to Shāțibi and Abū 'Abdallāh al-Ḥaffār. Ḥaffār's <u>fatwā</u> spelled out the economic aspect in more detail. A few excerpts from this fatwā are worthy of notice:

"This band of people who claim their connection with Taşawwuf, has caused the severest harm to religion in this period and in this part of the world. Their evils have spread throughout the Muslim world and especially in the fortresses and towns and villages which are farther from the capital...They are more dangerous for Islam than the infidels...

They have no virtue...None of them knows how to clean himself or to make ablution...In the name of religion they only know how to sing, to utter nonsensical statements and to encroach upon others' property unlawfully...

What made this band of people to adopt this way of life which is so dangerous for the existence of religion? Was it that they needed things basic for the human being, food, drink, clothing and such things, and they did not know any trade or craft to live from? Or if they knew a trade, did they find it hard to toil to earn their livelihood?...The devil seduced them and suggested to them this path which was full of fun and pleasure. They confuse the ignorant with the practice of dhikr... wearing patched clothes...as these were the signs of the virtuous people of this path... A certain scholar said that the people in a city must be like the parts of the body. As every part of the body has a particular use and none of them is futile...so are the inhabitants of a city. The soldiers guard the city, the <u>fuqahā</u>' and judges protect the law (<u>Sharī'a</u>) and also teach it...Therefore one who is of no use in a city whereas he is capable of being so...must be expelled from the city...

A philosopher (hakim) taught his disciples to be like bees in a beehive...they do not let any idle member stay there. They would drive it out of the hive, because it would cramp their space, would use their honey and would spread idleness, and abandonment of trades...

It is incumbent upon whoever can do so to restrain these people who are like a gangrenous sore in the side of religion. He must obstruct the way to this group for those who are inclined towards it. He must expel them from these places. (If he does so) he is a warrior of faith (<u>mujāhid</u>) in this respect."⁸⁴

To conclude, the political power of the <u>fuqahā</u>' declined in the reign of Ghanī Billāh, because the factors that strengthened their religious authority, were no more controlled by the <u>fuqahā</u>'. The introduction of <u>madrasas</u> deprived them from the control of institutions of learning which were, until then, a private business of the <u>fuqahā</u>'. Since the <u>madrasas</u> were now controlled by the Sultān, the <u>fuqahā</u>' lost their independence. Consequently, they could no more enjoy the influence on the important administrative offices which were previously filled by their privately-taught pupils. Nor could they resist the penetration of <u>Taṣawwuf</u> into the Granadian society. Rather, as a general trend, they eagerly joined Ṣūfī Ṭarī**q**as.

There were only a few jurists who, nevertheless, opposed <u>Sufi</u> <u>Tariqas</u>. Unlike their predecessors who condemned Taşawwuf mainly because of political reasons,

-7

these jurists rejected <u>Suff</u> institutions largely due to economic considerations. The following section examines the economic conditions and developments which shaped the opinions of such jurists. In fact, the change in Granadian economy was also a very important factor in the decline of the religious authority of the fuqahā'.

1

SECTION THREE

ECONOMIC DEVELOPMENTS

Geography

Muslim Spain under Banū Naṣr was reduced to the Southern part of Spain. The Naṣrī kingdom extended in the South to the shores of the Mediterranean Sea and the Strait of Gibraltar. For a certain period even the African seaport of Ceuta came within Naṣrī territories. In the North were the principalities of Jaen, Cordova and Sevilla. In the East it extended to the principality of Murcia and its Mediterranean shores. In the West lay the principality of Cadiz and La Frontera.

The kingdom was divided into three provinces (Kūrāt): Gharnāta (Granada), Al-Marīya (Almeria) and Mālaga (Malaga).

The kingdom was crossed in the middle by the lofty mountains of the Sierra Nevada and the steep hills of Basharrat. The depressed areas were traversed by the river Genil (Shanil), a tributary of Guadalquivir and by the rivers Andrex and Mansūra. The land was a combination of plains and valleys with thick forests.

The difference between the present geographical conditions and those described by historians is confusing. Today this part of Spain is dry and arid,² but the historians vie with each other in praising the fertility and the greenery of this region. The following description by Ibn al-Khatib is typical of other historical geographical descriptions:

> God Almighty has distinguished this our country by endowing it with gentle hills and fertile plains, sweet and wholesome food, a great number of useful animals, plenty of fruits, abundance of waters, comfortable dwellings, good clothing... a slow succession of the seasons of the year".³

The city of Granada, situated **sou**th of the Sierra Nevada, was the capital of the kingdom. By the city flowed the rivers Genil and El Derro. In the Southwest were the meadows of La Vega. Granada was surrounded by approximately 300 small towns (qurā).⁴

Population

Granada in this period attracted a great number of immigrants. Fleeing from the various Spanish territories which had been conquered by Christians, or having been persecuted by Christians, the Muslims came to Granada. In addition,

a large number of Berbers kept coming constantly from Africa: they came as Ṣūfīs, mercenaries, students or simply fortune seekers. We have no way of knowing the exact number of the population as the sources generally do not mention it.

Nevertheless, the growing burden of the population in this small kingdom can be seen in the educated guesses in the secondary sources. According to Imamuddin, in the days of al-Ghālib Billāh the population in the city of Granada was 150,000.⁵ Seybold estimated the figure in the later period as approximately 500,000.⁶ Over and above the rising numbers, the ethnic diversity of the population also affected the economy of the kingdom.

The bulk of the population in Granada and other cities was composed of Berbers and Arabs, both usually soldiers and hence fief holders.^{The}Spaniards who were mostly cultivators thus worked for both. The Berbers were hated by the Arabs, who considered themselves more culturally advanced than Berbers, as well as by The Spaniard Muslims who inhabited most of the rural areas.

Economy

Generally speaking prior to the eventful impact of the change in Mediterranean trade, the economy of the kingdom had two aspects: urban and rural. The economic activity in rural areas consisted of agricultural and pastoral occupations. In the urban areas the crafts and commerce were the main productive economic activities. Urban economic activity was largely concentrated on luxury goods, hence the actual burden of production fell upon the rural economy. Village life was severe. This situation forced a number of villagers to go to the cities, which were already few in number. This meant the availability of cheap labour, but since the production of luxury goods had a limited number of consumers, city life also was becoming highly expensive. The impact of Mediterranean trade, however, as we shall see below, shifted the burden of production from rural to urban economy.

Prosperity

There was a marked difference in the standards of living even among urban dwellers. The aristocrats who also owned the sources of production lived a luxurious life. Their wealth was distinctly evident in the ornaments and jewellery worn by the women of this class.⁷ Their jewellery consisted of such precious stones as emeralds and rubies, and their dresses were embroidered with gold and silver.

It was, in fact, the prosperity of this section of the population which so much impressed travellers such as Ibn B**er**tūta who described Granada as the most prosperous kingdom in the West.⁸

Financial Conditions

The revenue of the kingdom consisted mainly of taxes collected from lands. According to one secondary source, the yearly income of the kingdom was 1,200,000 ducats.⁹ The permanent deposits in the treasury consisted, of course, of precious stones and diamonds, ¹⁰ but the expenses of the kingdom were, however, met by the revenue.

The major source of revenue was land tax, called <u>Kharāj</u>. It was usually 1/9 or 1/10 of the produce, but another 1/5 was also levied as rent of land.¹¹ Since land was scarce and irrigation facilities were not commonly available, the most fertile lands around Granada were procured by the Sultān. These lands were called <u>Mukhtass</u> and were leased at very high rent prices. Because of the nature of the lands they were eagerly sought by the people. In addition to <u>Kharāj</u>, the other sources of revenue consisted of the customs duties collected from in-coming and in-transit commercial ships in the ports of the kingdom of Granada. Another occasional but frequent source of revenue was the proceeds from raids carried out in enemy territory which brought back prisoners, slaves, movable properties, etc.

: : :

> The taxes were collected in kind, but latterly, more emphasis was given to collection of revenue in cash. There was a complex system of tax collection. The tax collectors, called <u>Musharrifs</u> were responsible to one of the important <u>Kātibs of the Sulțān, called Şāḥib al-Ashghāl</u>.¹² The taxes were collected in the name of the <u>Makhzan</u> which applied to both Islamic and non-Islamic taxes. Even the trust properties belonging to mosques were not exempted.

The provincial and local administration as well as tax collection was in the hands of a $Q\bar{a}$ 'id in each district.¹³

The expenses of the kingdom were very high. The major expenditure was the tribute paid to the kingdoms of Castille and Aragon. According to Imamuddin ¹⁴ the amount of such tribute in Ghālib Billāh's days was 250,000 ducats. The second major expenditure was the salaries and compensation paid to the soldiers and mercenaries. In addition, large amounts were also paid to the Banū Marīn to recompense the expenses incurred in the preparation for war against the enemies of Granada.

Since in both modes of expenditure the terms were cash, the country had been geared to a money economy.¹⁵

The Nașri currency was similar to that of the Muwahhidun both in type and value. The basic units of money were the Dinar and the Dirham. Dirhams were usually silver currency and varied in value and fineness. The Dinar remained comparatively stable, the quality and quantity of gold helping to stabilize its monetary value.¹⁷ From the legal documents it appears¹⁸ that three types of Dinars were in currency: the golden Dinar, the silver Dinar and the Dinar Ayni (cop-The golden Dinar was usually of 2 grams in weight containing 22 carats per). Its monetary value was equal to 5 to 7 silver Dinārs or 75 silver Dirhams. gold. The Banu Nasr struck silver Dinars in square shape in contradistinction to the round shape of the golden Dinar and the dobla (the well-known non-Muslim gold Contrary to the conjectures of early scholars of numismatics,¹⁹ the piece). silver and 'ayni' (copper) Dinārs were not debased coins; but as studies of documents of contracts in the Nasri period show, they seem to have been introduced by the Naşri rulers according to fixed monetary values, while the gold piece was accepted in the market according to the current price of gold.

We have here the evidence of a money economy in the form of $\int_{\Lambda}^{\frac{2}{100}} copper Dinar$. The reason for this development was most probably the rapid growth of trade between Granada and foreign principalities. This trade is discussed at a later point in this section. What concerns us here is the plausable explanation of the copper Dinar by the fact that because of the need for gold for trade a kind of currency based on credits to the treasury could have been introduced in the form of the copper Dinar. Such a development could also be interpreted to mean that because of commercial needs the internal money was devaluated.²⁰

Agriculture

Spain had been known for highly developed agricultural methods and ample fertile land,²¹ but in the Nașri period the extent of Muslim Spain was reduced to Southern Andalusia. The nature of the soil and climatic conditions did not allow a scale of cultivation that permitted self-sufficiency in the production of grains. Often it proved necessary to import grains from North Africa.²²

The soil seemed to be conducive, however, for the growth of durable plants. Andalus produced a variety of fruits which were eagerly sought at home and in foreign markets as well. For export purposes, however, the cultivation of olives and mulberry trees became very common in the fourteenth century. Even though manufactured with primitive methods, Andalusia even exported olive oil. Mulberry leaves used in rearing silkworms had also gained a commercial value.

As mentioned earlier the <u>Mukhtass</u> lands, the best lands of Granada, were leased to cultivators who used to pay the dues in kind. In the fourteenth century, it appears, these lands began to be rented to those tenants who would pay the rent in cash. These tenants hired seasonal labour for cultivation.²³

By the fourteenth century land had become critically scarce. Evidence of this fact is found in the <u>fatāwā</u> literature where various forms of ownership and complex methods of the division of the property and produce are noted.²⁴ The ever growing population and the continuous loss of territories to the Christians were also responsible for the fact that extraordinary forms of ownership appeared in the distribution of cultivable land. A small tract of land might be co-owned by a number of persons.²⁵ Not only that, but the division and subdivision of property even extended to a tree and its branches; a tract of land was divided among its owners by the number of trees; or a tree, when it was owned by more than one, was divided by its branches.²⁶

The extraordinarily intense cultivation even forced the people to use or rent the gardens around their houses for agricultural and commercial purposes.²⁷

Besides the seasonal crops, fruit cultivation was a major occupation. A highly developed system of irrigation made higher level lands useful for orchards.²⁸

In general, however, it appears that the pressure toward a cash economy was forcing even the rural agricultural economy to change into a certain type of economy, which for lack of a better term, we may call "mercantile" economy. It must, however, be made clear that our use of the term 'mercantile' should not be confused with its technical use in a special sense which refers to the sixteenth century 'mercantile policies' in certain European countries.²⁹ We are using this term in its simple sense to mean a type of economy that lays stress on trade and commerce, and where money as wealth becomes important in preference to land.

Some of the indicators of the rise of this type of mercantilism are the following:

The use of seasonal labour and contract-workers who received their wages in cash or kind at the end of the contract ³⁰ period was replacing the older system of semi-serfdom for the peasants. Forms of co-operative cultivation where partner and production contributed money in place of land were also current. Evidence for this development is found in the specific cases of rearing of silkworms³¹ and of production of cheese.³²

Industries

The existence of gold, precious stones, amber and metals such as copper and iron³³ in the kingdom of Granada encouraged various industrial activities. These industries had become a major base of the Naşrī economy. Louis Bertrand tells how the Granadians enjoyed "up to a certain point, industrial wealth."³⁴

Among others the major industrial centres were Granada, Malaga and Almeria. The following industries flourished: Weaponry, Silk, Pottery, Leather, Cotton and other textiles.³⁵ The most profitable industry was silk.

The cities that were busy in the silk industry were Jubiles, Granada, Guadix, Fiñana and Almeria. In Almeria there were about 800 looms for brocaded silk and about 1,000 for embroidered silk. Similarly there were looms for other kinds of silk among which the following were well known in the foreign markets: Usqulāțun, Georgian, Isphania, ^cUnābī, Ma^cajir al-Mudhashsh.³⁶

In the fourteenth century, because of the growth of the Italian silk industries,

the Granadian silk industry suffered heavily.³⁷ Nevertheless, the market demands for raw silk material insured that this industry in Granada remained profitable.³⁸

Crafts

.

The crafts were usually connected with luxury commodities. Many artisans coming from other parts of Spain, had settled in the kingdom of Granada. This influx advanced not only the development of crafts but also turned the craftindustries toward the production of luxury goods. These crafts concentrated on jewellery, golden silk embroidery, decorative pottery and fancy leather among other things.³⁹

Trade and Commerce

The most significant phenomenon in the economic history of the Islamic West in the fourteenth century was the quick development of a commercial economy. The coastal cities developed significantly along with the growth of their political influence. This is evident in the case of Ceuta, Malaga, Ronda and Almeria. The Alliance of Ronda and Malaga with Muḥammad V al-Ghanī Billāh meant his remounting the throne of Granada. His capture of Ceuta meant a greatly increased influence in Marīnī political affairs.⁴¹

Besides political influence, these cities also experienced a rapid growth of the textile, metal, leather, dairy, flour-milling, and ceramic industries and other crafts. The produce was largely meant for foreign markets.

The main cause of the quick development of a commercial economy in this area, according to S. M. Bastieva, was the economic upsurge in the Mediterranean countries in the thirteenth and fourteenth centuries. The cause of that development was the sudden growth of manufacturers in Italy.⁴² Trade, however, necessarily required relations with foreign nations. The overseas trade of the Italian cities reached its zenith in the fourteenth century. This trade was carried further by the maritime cities of Catalonia, Provence, Constantinople, Alexandria and others. This activity created in the Mediterranean a wide market which made possible the enormous upsurge of production there and which was conducive to the appearance of mercantilism in Italy.

The emergence of vigorous trade patterns around the Mediterranean made possible a wide sale of agricultural produce in the foreign market. This, in turn, affected Granadian economy by producing a stimulus to commercialize agriculture. Lopez Ortiz, studying the <u>fatāwā</u> literature of this period, concluded that in Granada, the agricultural production was moving towards a mercantilized economy.⁴³

Among the materials that Florence, Naples, Catalonia and Provence imported were raw leather, processed leather, olive oil, cotton, silk, wax, etc. The main importer of Granadian raw silk was Florence.

The main seaports of Granada, Almeria and Malaga, were situated at very significant points on the Mediterranean trade routes. They were, thus, in a favourable position to benefit from the new trade.

Almeria and Malaga were situated on the sea trade route connecting the maritime cities in Western Spain and in West Africa with Naples in Italy. This sea route connected with another sea trade route starting from Sevilla and going through Murcia, Valencia, Barcelona and ending in Marseilles. In terms of land trade Granada was connected with a number of trade routes that spread throughout Spain and which were also connected with the maritime cities. Granada was connected with the land trade routes in Africa through Ceuta which was under her suzeranity at that time. These land routes led to Fez, Tlemcen and Algiers.

The significance given to the safety of these trade routes by the rulers can be seen in the mutual trade pacts between the kingdom and its neighbours in that period. The Banū Naṣr frequently signed trade pacts with their neighbours; or one should perhaps rather say that every treaty included a condition of mutual agreement on the safety of trade routes and merchants. In 684 A.H. in a treaty with Castille the condition read that the Muslim merchants going to Castille would be exempted from taxes.⁴⁵ The treaty with Aragon, signed in 695, provided that the cities in the territories of both partners to the treaty would be open to the merchants from both territories and that their lives and merchandise would be safeguarded.⁴⁶ In 721 A.H. in the renewal of this treaty an additional condition provided that the boats (ships), shores and ports of each partner would be safeguarded.

Such security pacts with Christian neighbours were essential for the Banū Nasr as the major part of their trade consisted of exports to Italy by these trade routes. Naples, Catalonia and Provence were the main importers of such Granadian commodities as raw and processed leather, olive oil, cotton and wax. The city of Florence was the major importer of raw silk from Granada.

Money Lending

A natural result of ther mercantilistic activities was the growth of a strong and widespread money-lending class. This money-lending class operated both in Christian and Muslim territories. Most probably the intermediaries in such trans-actions were Jews. In the literature of that period they were called "transgressors and unjust".⁴⁸

These money lenders controlled the markets where agricultural products were brought for auction. They worked also as intermediaries in auctions.⁴⁹ They were also responsible for the exchange of currencies. There is also an indications in a fatwā that they even determined the values of the currencies.⁵⁰

A peculiar and typical product of this economic and political milieu was <u>al-Fakkāk</u>. The term, originally meaning to separate, disjoin, redeem, ⁵¹ probably under the influence of the Qu'ānic legal term <u>Fakk-u-raqaba</u>⁵² (to liberate someone) came to be used also in commercial legal transactions to mean the redemption of pledges and of debts. ⁵³ Most probably this Andalusian term <u>al-Fakkāk</u> etymologically springs from that usage. In Andalus this term was applied to an intermediary who was paid by the relatives of a prisoner in the enemy territory to buy the liberty of the prisoner by paying the required amount to the enemy.⁵⁴

े २ मः म

> To grasp the situation it must be pointed out that despite the truces, payments of tribute, and promises of protection, the Granadians found themselves often invaded by armed bands which cut down fruit trees, carried off crops and cattle and took prisoners. These events were so common that Muslim fraternities along the lines of the French fraternities such as the <u>Ordre de la Merci</u> were established to ransom Muslim prisoners and slaves in Christian territory.⁵⁵

> Ibn Battüta witnessed such an incident in Spain. He relates the story that on the coast of Marbella four galleys of a Christian band appeared, and after killing a fisherman, captured eleven horseriders who were travelling to Malaga a little distance ahead of Ibn Battūta. When he reached Malaga and arrived at the main mosque, he found the Chief $Q\bar{a}d\bar{i}_{\Lambda}\bar{f}anj\bar{a}l\bar{i}$ already busy talking to a number of jurists and a notable businessman in Malaga. They were collecting a sum to buy back the freedom of the captives.

From the legal literature of this period, it appears that the institution of <u>al-Fakkāk</u> was an already established practice.⁵⁷ Under Muslim influence the Castillians also called such intermediaries <u>Alfaqueques</u>. In Castille they were supposed to be responsible for the administration of the property of prisoners of war.⁵⁸

In Andalus, however, although the institution may have originated from pious and selfless interests, yet by the fourteenth century it had more of a commercial nature than anything else. The Fatāwā indicate that al-Fakkāks used to
contact interested persons on both sides and earned a commission from both parties. One <u>fatwa</u> shows that <u>al-Fakkaks</u> bargained about the prices for ransoms, etc., devaluated the currencies, and earned profits from such transactions.⁵⁹

In the light of this and other descriptions of the institution of <u>al-Fakkāk</u> in the sources, it may be rightly assumed that <u>al-Fakkāk</u> belonged to the money-lending class. The assumption gains weight since the sources indicate that <u>al-Fakkāk</u> also traded in silk, advanced money on anticipated earnings and dealt in debased currencies.⁶⁰

From the above survey of economic developments, especially such matters as $\hbar^{\mu\epsilon}$ the emergence of <u>al-Fakkāk</u>, the growth of Mediterranean trade, the introduction of the devalued copper <u>Dinār</u>, and the transformation of agriculture into commercialized forms of cultivation and other such facts, it can be seen that the economy was rapidly changing towards a type of mercantilism. This would imply, among other things, the disappearance of institutions that were based on an agricultural economy and the emergence of new ones.

This would mean also that the Mālikī <u>fiqh</u> had to face some essential changes. To justify new institutions it would not be sufficient to attempt to accommodate them under some legal fiction or some legal device. The number and nature of these new institutions forced the <u>fuqahā'</u> to push the problems they faced back to fundamental matters of legal methodology and general principles of legal theory.

SECTION FOUR

LEGAL DEVELOPMENTS

The data available for this section is particularly scanty. Since a description of the legal system and legal developments, however cursory it may be, is helpful to complete the picture of social changes which is the objective of this chapter, this section makes such an attempt.

This section deals first with the legal system and second with legal developments in fourteenth century Granada.

Legal System

ŧ

i

Reference has already been made to the institutions of <u>qadā'</u> and <u>futyā</u> and the place of Mālikism in the legal system.¹ Not to repeat what has already been said, we will briefly state the main points relating to the legal system.

- 1. Mālikī fiqh was recognized as the law of the kingdom.
- 2. Mālikī figh was applied on three levels:
 - a) On the level of <u>futyā</u>, strictly religious matters including those of exegesis and theology were referred to <u>muft</u>is, and except for cases of heresy,² such matters were beyond the courts' jurisdiction. The opinion of the <u>Muft</u>is was called <u>fatwā</u>, and its implementation largely depended on the individual conscience.

- b) On the level of the courts (qaqā), the decision of the judge (hākim) was called hukm. Although the judge had no executive powers, yet in contradistinction to fatwā, the hukm was enforced by government agency. The qāqi was assisted by a concilium of fuqahā' called mushāwirūn.³
- c) On the level of the notaries (<u>wuththāq</u>), the Māliki fiqh was applied to register and validate various kinds of contracts and other types of legal documents. These <u>wuththāq</u> were usually faqihs and were often appointed also as <u>muft</u>is and <u>mushawirs</u>.
- 3. In matters of procedure the litigants sought the <u>fatāwā</u> of the <u>muft</u>is in favour of their claims and presented them in the court. The judge reached his decision after consulting the notables in his court. The <u>Qādī</u>'s decision was final in the sense that neither he nor any other judge could revise this decision; in the opinion of some scholars the decision stood as it was even if the witnesses changed their testimony. In certain matters an appeal could be made to the Sultān against the decision of the court.⁴
 - 4. Since Mālikī <u>fiqh</u> covered all matters relating to religion, ethics, family, property, etc., and the <u>muftī</u> could be consulted even on matters which were also in the <u>qādī</u>'s jurisdiction, a confusion between the jurisdictions of <u>muftī</u> and <u>qādī</u> always existed. The function of the

notaries added to the confusion. The notaries were sometimes given limited jurisdictions such as the attestation of a witness or a contract, yet they could not decide the case.

1

In short, the essential problem of the Granadian legal system became one of confusion of the function of <u>fatwā</u> and <u>hukm</u>. The Egyptian Mālikī jurist *al*-Qarāfī (d. 684 A.H.), whose influence, as has been mentioned earlier, was felt deeply in Mālikī <u>fiqh</u> in Andalus, wrote the following treatise on this problem: <u>Al-lhkām fī tamyīz al-fatāwā ^can al-ahkām</u>.⁵

Qarāfī disagreed with the usual distinction made between <u>fatwā</u> and <u>hukm</u> by considering the former as only <u>ikhbār</u> (statement) and the latter as <u>ilzām</u> (binding).⁶ On the contrary, he maintained that both are <u>ikhbār (an hukm Allāh</u> (statement about God's command) and both are "binding". According to him a <u>fatwā</u> is a statement which implies either <u>ilzām</u> or <u>ibāha</u> (permission), and the <u>hukm</u> is a statement which implies either <u>ilzām</u> or <u>inshā</u> (preceptive action).⁷ In respect of subject-matter, the <u>hākim</u> has jurisdiction only in $\frac{al}{b}$ (the matters which were not agreed upon among the Mālikī scholars) and <u>maṣāliḥ dunyawiyya</u> (matters relating to this world); the <u>hukm</u> has no jurisdiction in ^cibādāt (ritual and worship) and ijmā^c.⁸

Qarāfi, however, could not remove the confusion completely as he maintained that both <u>fatwā</u> and <u>hukm</u> form part of the function of the <u>imām</u> (in this case the Sultān)⁹ but whereas he made the <u>muftī</u> responsible to God, he did not define to whom <u>hākim</u> was responsible.

Legal Developments

i

Beside the confusion that existed in the functional aspect of Mālikī <u>fiqh</u> certain new developments had added more to the confusion. We will briefly mention a few of them.

A. The legal status of Andalus: Mālikī <u>fiqh</u>, in certain cases, maintains that the legal status of a territory changes according to its political condition; whether it is on peace terms with another territory or at war. In the fourteenth century Andalus was constantly at war or on peace terms with her Christian neighbours. It even had the status of a vassal state to the principality of Castille. A number of questions in the <u>fatwā</u> literature show the confusion that this situation created in the application of law.¹⁰

B. Diversity of Laws: The diversity of laws had a number of causes. In many cases the diversity came about because of the differences in the local practices which were recognized in $\int_{L}^{M^{2}}$ Andalusian Mālikī Tradition as a source of law.

The diversity of laws was also caused by other factors such as the use of the principle of <u>murā^cāt al-khilāf</u>. These aspects have been discussed elsewhere in detail.¹¹

It seems that Ibn al-Khaţib became painfully aware of the weakness of the legal system and tried to reform them. He criticised Qādi Ibn al-Hasan al-Nubāhi in his treatise Khala' al-rasan. He also wrote the following books on legal theory: <u>Sadd al-Dharīća fī tafqīl al-Sharīća</u>, ¹² <u>Alfiya fī</u> <u>uşūl al-fiqh</u>¹³ and <u>Muthlā al-ţarīqa fī dhamm al-wathīqa</u>.¹⁴ In his <u>Muthlā al-tarīqa</u> he strongly criticised the institution and practice of notaries (<u>wuththāq</u>). He condemned them for their ignorance of the Arabic language and of figh. His essential criticism of this practice was on the basis of <u>wara</u>^c(moral responsibility) that was completely neglected by the legalistic and formalistic trends in the legal practice.¹⁵

The little information we possess on the actual legal developments in Andalus should perhaps be supplemented by comparison with Christian Spain. It is quite probable that developments similar to those in neighbouring areas took place in Andalus, since both countries underwent the same kind of socioeconomic changes. For an understanding of legal developments in Christian Spain it is quite revealing to notice the various stages through which <u>Fuero</u>, an important Spanish legal institution, went.

The institution of <u>Fuero</u> existed before the arrival of the Muslims in Spain. It survived under Muslim rule and later became a stronghold of resistance to the renaissance of Roman law in Spain in the thirteenth and fourteenth centuries.

Fuero, deriving its name from Lex fori, tribunal justice,¹⁶ came to stand for the legal practice of townships and thus took the name of <u>Fuero Juzgo</u>.¹⁷

Fuero Juzgo also called Liber Judiciorum and Lex Barbara Visigothorum, was a compromise between Visigothic and Roman law, developed during the

Į

seventh century.¹⁸ <u>Fuero Juzgo</u> was a medley of legal rules which included, among others, subjects such as the following: rules for visiting the sick, the graves of the dead, heretics, etc.¹⁹ Under Muslim rule these <u>Fueros</u> incorporated some Muslim elements as well.²⁰

In the thirteenth century the administration was faced with the profusion of all kinds of law in Spain. The excessive diversity became threatening to the fabric of the state.²¹ The progress of trade also demanded system of uniform laws.

By the middle of the thirteenth century a movement for the reform of laws emerged. A long contest between the supporters of <u>Fueros</u> and the supporters of legal unity began. Two weapons were used to reform <u>Fueros</u>: (1) Exposing the shortcomings of the <u>Fuero</u> system and (2) the renaissance of Roman law.²² Three Castillian kings Ferdinand III (1199–1252), Alfonso X (1221–84) and Alfonso XI (1311–50) are known as staunch supporters of these legal reforms to bring about the uniformity of law.²³

In the days of Alfonso the Learned another development was also taking place. In Southern France there arose a school which both there and in Bologna supplanted the glossators (medieval commentators on Roman civil law). Instead of seeing in Roman law a multitude of texts to be examined and interpreted, those of the new trend sought to do two things:

a) systematize Roman law in accordance with the rigid method of Aristotle and in the light of Christian doctrine, and

ł

b) to ascertain what reasons could have motivated its rules. The trend thus marked the beginning of a philosophy of law.²⁴

Many scholars from Spain travelled to Bologna to study and teach Canon Law. In Spain, the University of Salamanca became an important centre for the study of Roman and Canon law.²⁵

The purpose of the above description is to indicate that factors such as the diversity of laws and the need for reform of local legal practices to bring about the uniformity of laws led scholars to investigate the motive and purpose of law. The attempts of these scholars had very far-reaching effects on the evolution of law in Europe in later centuries. Although this evolution came about two centuries after Shāṭibī, it is not irrelevant to refer to it briefly as it helps in understanding the direction to which the legal philosophy was led by the legal developments in Shāṭibī's period.

As a result of the continuous concern for the philosophy of law in Spain there emerged a group of prominent legal philosophers who are now known as "Spanish Theologian Jurists". Two of these jurists are usually described in the following manner: Vitoria (Fransisco de Vitoria d. 1546 in Salamanca),"the expounder of the law of nations" and Suarez," the philosopher".²⁶ Francis Suarez was born in Granada in 1548 and died in Lisbon in 1617. His influence on the later development of the philosophy of law is well-known. His legal philosophy had its pivotal point in the exposition of the end of law which, according to Suarez, was the "Common good of the Community".²⁷ Despite the time interval of two centuries between Shāṭibī and Suarez, the similarity in their approach towards law and its end is worth noting. Shāṭibī also investigated the purpose of law and he also found the concept of <u>masālih al-ʿibād</u> (the good of the people) to be the objective of law.

ī

Unfortunately the similarity in the legal developments in Muslim Spain with that in Christian Spain does not go beyond this point. There is similarity in the socio-economic factors that led to an investigation of the philosophy of law in both Muslim and Christian Spain. $\overset{\#e}{L}$ Jurists' conclusions about the objectives of the law were the same. Yet whereas in Christian Spain these investigations continued and were responsible for the shaping of the concept of law in Europe, among the Muslims this attempt seems to have stopped with Shātibī.

NOTES: CHAPTER III

د بر رب

Section One

- 1. See Muhsin Mahdi, Ibn Khaldun's Philosophy of History (Chicago: Phoenix, 1964).
- See Henri Laoust, Contribution a une etude de la Méthodologie Canonique de Țaki-d-Din Ahmad b. Taimiya (Cairo, 1939) and Essai sur les doctrines sociales et polimiques de Țaki-d-Din Ahmad b. Taimiya (Cairo: Imprimerie de l'Institut Français d'Archeologie oriental, 1939).
- See Josef van Ess, Die Erkenntnislehre des Aduddin al-ici: Übersetzung und Kommentar des ersten Buches seiner Mawāqif. (Wiesbaden: Steiner, 1966).
- 4. Not to speak of secondary sources, even the primary sources on the Naşri period are often confusing. The confusion of the secondary sources is partly due to their indiscriminate use of the primary sources which are often conflicting. For a general history of the period a critical study of the primary sources is indispensible. The two contemporary historians on whom the later sources have depended are Ibn al-Khatib and Ibn Khaldūn. Not only did these two men belong to different courts which were often enemies to each other, but Ibn Khaldūn also had a particular philosophy of history that stresses the role of tribes and families. These differences make their narratives of the same period conflict with one another. Furthermore, the attachments of the two historians to these courts also fluctuated. These changing loyalties also affected their narratives. Ibn al-Khatib revised, added and suppressed much information at various stages of writing the history of this period.

Only to avoid confusion, we have chosen Ibn al-Khaţib's al-Ihāţa (Cairo: Maţba' Mawsū'āt, 1319 A.H.), as the basic text mainly because this was written before Ibn al-Khaţib had been prejudiced against Muḥammad Al-Ghanī Billāh. For the events after 771 A.H., mair'y for the story of Ibn al-Khaţib's persecution, we have relied upon the intermation in Al-Maqqarī, Nafh al-Ţib (ed. M.M.A. Hamid, Cairo: Sa'āda, 1949), which derives its information mainly from Ibn Khaldūn's Kitāb al-Ibar (Bayrūt, 1959). We have also used Qādī Nubāhi's Al-Marqaba al-Ulyā, (Cairo, 1948) to supplement Nafh al-Ţib. Secondary sources have been used only complementarily.

5. Ibn al-Sa^cid, an African traveller, who visited Andalus at that period described the capricious political attitude of the populace as follows:

Their attitude towards a sultan can be described by the fact that whenever they find a horserider who distinguishes himself among his peers...they rush to his side and appoint him their king without any consideration for the future...or sometimes there is in the kingdom a soldier of the officer rank (qā'id) who has earned fame for his campaigns against the enemy... They offer him the rulership in one of the fortresses..."

quoted by al-Maqqari, <u>Nafhal-Tib</u> op.cit., Vol. 1, p.201. Another evidence of this political confusion is the story narrated by Ibn al-Khatib saying that the Andalusian territories were in the hands of robbers and warlords whose alliance Ibn Hud sought in order to become the sultan of al-Andalus.

See Al-Ihāta, op.cit., II, p.91

- Ibn Khaţib: <u>Al-Ihâţa</u>, II, p.90-91. Huici Miranda is of the opinion that Ibn Hūd's insurrection personified Spanish Muslims against the Berber Al-Muwahhidūn. See article "Gharnāţa" in <u>Encyclopedia of Islam</u>, (New Edition), Vol. II, p.1014.
- 7. Al-lhāta, II, p.61.
- 8. lbid., p.62.
- 9. <u>Ibid.</u>, p.65. Ibn al-Khaţib, however, does not mention the events that made Ibn al-Ahmar repent his submission to Castille. Ibn Khaldün narrates further that in making truce with Ferdinand, Ibn al-Ahmar was satisfying his anger against Ibn al-Jadd, the ruler of Sevilla. He supported the Christians in every manner. But when Ferdinand was not content with taking Cordova and Sevilla but went on capturing more fortresses and important towns, Ibn al-Ahmar was irritated and repented his decision. See Gaudefroy Demombeynes "Histoire des Benou I Ahmar", (translation from Ibn Khaldūn's Kitāb al-^cIbar in Journal Asiatique, 9th series, Vol. XII (1898), p.325.
- 10. Ibn al-Khațib (Iḥāṭa, II, p.59) mentions that Muḥammad was born on 22 Jumāda al-Ākhira in the year 739 A.H.
- 11. Ibid., p.4.
- 12. Ibid., p.9.
- 13. Ibn al-Khatib attributes this revolt to the negligence of Ridwan. Ibid., p.11.
- 14. Ibn Khaldun, <u>Kitāb al-^cIbar</u>, Vol. VII, p.637.
- 15. Ibn Khaldun provides more details of this event. For him the cause of this revolt was the faqih Ibn Marzuq a scholar and sufi connected with the ribāt of Abu Madyan. Ibn Marzuq was so influential that Abu Sālim had

left almost all his affairs in Ibn Marzūq's hands. This antagonized other officers at the court. Consequently, 'Umar b.'AbdAllāh, the Wazīr at the court conspired with Garcia Antoine, the Andalusian Christian who was at the head of the mercenary soldiers. Abū Sālim was killed. Soon after ^cUmar b. 'AbdAllāh, apprehending the plan of Garcia, succeeded in assassinating him and thus became the virtual ruler. Ibn Khaldūn, Kitab al- ^cIbar, Vol. VII, pp.648ff.

V.

- Ibn al-Khațib gives no detail, and mentions the name of 'Umar b. AbdAllāh 16. in derogatory terms (Khabith: wicked), Ihāta, op. cit., p.14. But Ibn Khaldun, giving the details, takes the credit to himself. He narrates 'Umar b. AbdAllah was his friend, and that he advised 'Umar to that surrender Ronda to Sultan Muḥammad V. Kitāb al-^clbar, op. cit., p.694. Ibn al-Khatīb, in Aʿmāl al-Aʿlām (p.314) however, takes credit for this event by mentioning the same reasons that Ibn Khaldun gives. Interestingly enough in another place Ibn Khaldun mentions that Umar b. 'AbdAllāh, who, after killing Abū Sālim had taken over the Marini throne, was looking for a proper candidate for the throne from the Marini family. He found the Marini prince Muhammad, then in the custody of the Castillians, to be the most proper choice. Since Muhammad V was on good terms with the Castillians, 'Umar b. 'AbdAllah promised him the surrender of Ronda if he would procure the release of the prince from Castille. (Vol. VII, p.659).
- 17. Ibn al-Khaţib mentions this in reference to two offices: in case of the Wizāra, he says: "His prudence demanded to neglect this office entirely, even though it was essential in the political and financial administration. (This resolution was made) to avoid the evils that had come out of it before..." (Ihāta, II, p.15). He repeats the same reference in case of Shaykh al-Ghuzāt (p.20).
- 18. Al-lhāta II, p. 17.
- 19. Ibid.

à ..

- 20. Ibid., p.48 ff.
- 21. The details of these campaigns are provided by lbn al-Khatib, <u>al-lhāta</u>, ll, pp. 48-59.
- 22. See p. 91.
- 23. Imamuddin, Political History of Spain, (Dacca: Najma, 1961), p.284. Also ^cInān, Nihāyat al-Andalus, <u>op. cit</u>. p.36.

- 24. ^CAbbādī, in his comments on Ibn al-Khaţīb's <u>Misyār al-Ikhtibār</u> (<u>Mushāhādāt</u>, p.99 no.2) notes that this exchange was well known. Many Spanish Christians knew Arabic; similarly, many Andalusian Muslims knew the Castillian and Angoinese languages. Frequent debates and disputations on religious and academic subjects took place. Ibn al-Khatīb states that a Muslim scholar Muḥammad b. Lubb al-Kanšānī wandered in Spanish lands debating with priests. Another Muḥammad Kāqūţī went to Murcia to teach Jews and Christians. ^CAbc Jlāh b. Sahl was well known in mathematics. His fame reached as far as Toledo, and many scholars came to Baeza to study with him.
- 25. Inan, Nihāya...op.cit., p.432. Nafh al-Ţīb, Vol. 1, p.207.
- Muhammad Ridwan al-Daya (ed.), Isma^cil Ibn al-Ahmar, <u>Nathir Fara'id</u> al-Juman fi Nazm Fuhul al-Zaman, (Cairo, 1967), Introduction: p.17.
- 27. Lévi-Provençal, "Nașrids" in E.I. (1st edition), III, p.879.
- 28. Inan, Nihāya, p.117.

1.

il Interne

- For the lengthy details of this event see Ibn Khaldūn, <u>Al-⁴Ibar</u>...Vol. VII, pp.695 ff.
- 30. Gaudefroy Demombynes, "Histoire des Benou'l-Ahmar", p.340: n.61.
- 31. This Zahir is mentioned by Ibn al-Khaţib in his al-lhāţa in those parts which still remain in Ms. preserved in the historical section of Gayangos Collection. We are quoting it here from the excerpts by Muhammad Kamāl Shabāna, "Shuyūkh Ghuzāt al-Maghāriba fi I Andalus Kamā arrakha lahum Ib_n al-Khaţib fi Al-lhāţa" in Al-Bahth al-ʿIlmī, (December-January, 1968), pp.134-136.
- 32. Ibid. 135f.
- 33. Ibid., p.125-126.
- The details are given by Ibn al-Khatib, Kitab A'mal al-A'lam fi man Buyi'a qabl al-Ihtilam min Muluk al-Islam, op.cit.,
- 35. Ibn al-Khatib, Al-Ihāta, Vol. 11, p.20.
- For details see the notice of Muhammad al-Ghani Billah in Ibn al-Khatib, al-Ihāta, Vol. 11, pp.48-59.

- 37. As quoted by Maqqari, Nafh al-Tib, Vol. 1., p.202.
- 38. Inan, Nihāyat al-Andalus, op. cit., 426.
- 39 Ibid.

یو ہے۔ اور اور

- 40. Abū'l Hasan al-Nubāhī, Al-Marqabat al-^CUlyā..., p.49.
- 41. See p. 102ff.
- 42. Ibn Khaldūn, op. cit., p.694.
- 43. Ibn al-Khațib, A^cmāl...p.78-79.
- 44. Ibid. Al-Katibat al-Kāmina fī man Laqaynāhu bi al-Andalus min Shuʻarā al-Mi'at al-Thāmina, ed. Ihsān Abbās, (Bayrūt: Dār al-Thaqāfa, 1963) pp.146-152.

NOTES: CHAPTER III

Section Two

- 1. See p. 86 and Sec. 1, n. 15.
- See for instance: R. Dozy, <u>Histoire des musulmans d'Espagne</u> Vol. I (Leiden: 1932), pp.286 ff and Levi-Provençal, <u>Histoire de l'Espagne</u> musulmane, Vol. I (Paris, 1950), pp. 149 ff.
- 3. Ibn Khaldun, Muqaddima (Cairo, 1320 A.H.), p.425.
- 4. I. Goldziher, "The Spanish Arabs and Islam", trans. from Spanish by J. de Semogyi, <u>Muslim World</u>, Vol. LIII (1963), p.13 and passim.
- 5. Quoted by J. T. Monroe, Islam and the Arabs in Spanish Scholarship (Leiden, 1970), p.233.
- Hussain Monès, "Le role des hommes de religion dans l'histoire de l'Espagne musulmane jusqu'a la fin du Califat", <u>Studia Islamica</u>, XX (1964), 49 ff.
- 7. Among many others see the recent thesis in Jamil Abū'l Naṣr, <u>A History</u> of the Maghrib (Cambridge, 1971), p.11.
- 8. Monroe, op. cit.
- 9. Monès, op. cit., 50 ff.
- Provençal (op. cit., p.149) says: "Dès cette époque, en effet, on vit se constituer, principalement dans la capitale..., une sorte d'aristocratie, à la fois religieuse et intellectuelle, composé par les fakihs ou juristestheologiens malikites".
- 11. Roger Idris, "Reflexions sur le Malikism sous les Umayyades d'Espagne," Atti del Terzo Congresso di studi Arabi Islamici (Napoli, 1967) p.399.
- 12. For details see Dozy, op. cit., p.288 ff.
- 13. See above Section 1, n.24.
- 14. See Section 3, and p. 223ff.

- 15. E.I.J. Rosenthal, <u>Political Thought in Medieval Islam</u> (Cambridge, 1962), p.44.
- 16. Salāwī, <u>Al-Istiqşā</u>'(Dār al-Bayḍā', 1954), Vol. III, p.101, recounts that a Granadian envoy was punished by the Marīnī Qāḍī for drinking.
- 17. The genealogical superiority is very often expressed in the eulogies of the court poet Ibn Zumruk. This aspect is officially expressed particularly in fanā'al-aswad in al-Hamrā'of which the following inscription is very indicative:

وبإوارث الافسارلاعن كلالة متراث جلال ليتخف الرواسيا

For details see (Inan, Al-Athar al-Andalusiya (Cairo, 1956), p.170.

18. Ibn Sa'id: Al-Mughrib fi Hulā al-Maghrib, Vol. 1 (Cairo. Dār al-Ma'ārif, 1953, p.57 and Vol. 11, 1955, p. 109 (for the explanatory note on the authors of this work see Shauqi Dayf the editor's introduction).

20. See for instance two of Ibn Zumruk's eulogies preserved by al-Maqqari, Nafh al-Tib, Vol. VII, pp. 96-107. The following verses illustrate our point:

وخصى انصادة الاعلين صغرت واسكنوا من جوارالله اعلاه المار ملته اعلام بيست مناقب شرنت اثنى بعا الله لاغرو ان فقت الملوك سيادة اذكان جدك ستيد الالصاد السانبون الاملوت الى المصدى والمعطفون للمرة المختاب ورثت حددا المغن بإ ملك المصدى من كل آوى المنبى ومن نصر من شام بيرف فضرهم وكما لسعم أسيارهم أبيد لسصر تعسد هم ليستول وحى الله قد النصر

21. Ibn Sa^cid, as quoted in Nafh al-Tib, Vol. 1, p.206.

- 22. See for instance the <u>Zahir</u> of Qādi lbn 'Asim for his appointment as Qādi. This Zahir is preserved by al-Maqqari, <u>Nafh al-Tib</u>, Vol. VIII, pp. 262-268.
- 23. ^cInān, Nihāyat al Andalus, op. cit., pp. 426 f.
- 24. Ahmad Bābā, <u>Nayl al-Ibtihāj</u>, (Cairo; Abbās b. Abd al-Salām, 1351 A.H.) p.282.
- 25. Al-Ihāta, op. cit., Vol. 1, p.71.

- 26. Al-Maqqari, quotes lbn Sa^cid saying that the inspector of the market, on his mount, passed through the market along with his assistants. A balance was carried by one of his assistants in which the bread was weighed, as its weight and price were fixed. A small child or a charming girl was sent to the market to buy the bread, which was then weighed and inspected. The same process was followed for meat and other merchandise. For details see Nafh al-Tib., Vol. 1, p.203 ff.
- 27. The rules of Islamic law in this respect were based on the Qurānic verses prohibiting usury and speculative transactions. For instance the verse: "O ye who believe! Devour not usury, doubling and quadrupling (the sum lent). Observe your duty to Allāh, that ye may be successful" (3:129), and "O ye who believe! Strong drink and games of chance and idols and divining arrows are only an infamy of Satan's handiwork, leave it aside in order that ye may succeed." (5:90).
- 28. See Abū'l Hasan Alī b. Yūsuf al-Hakīm, Al-Dawhat al-Mushtabika fī dawābit dār al-sikka, Ed. Husayn Mūnis (Madrid, 1960), pp.52–53.
- 29. For instance see the notice of Qāḍī Ibn 'Ayyāsh in Nubāhi, <u>Al-marqabat</u> al-'ulyā, op. cit., pp.20-21.
- 30. This point is explained by R. Levy, <u>The Social Structure of Islam</u> (Cambridge, 1957), p.291–293.
- 31. Although it is difficult to find specific evidence on the number and quality of these tracts of land, yet, the large number of fatwas requested about abbas which were attached to mosques and the fuqaha s complaints when these lands were made subject to tax are indicative of the point we are making. For such fatāwā see Wansharīsī, Al-Mi'yār al-Mughrib (Fas, 1314-15 A.H.), Vol. VII, 68ff.
- 32. Vide Nafh al-Ţīb, Vol. l, p.205.
- 33. Ibn Khaldun, Muqaddima, op. cit., p.407f.
- 34. Shāțibi, <u>Al-Muwafaqat</u> (Cairo: Raḥmāniya, n.d.), Vol. 1, p.97.
- 35. Paul Nwiya, Ibn 'Abbad de Ronda (Beyrouth, 1961), p.XXVII, mentions 'Àbili, Ibn 'Abd al-Salām and Qarāfi among them who opposed the establishment of madāris. They regarded such institutions as bid'a.
- 36. Levi-Provençal, Inscriptions Arabes d'Espagne (Leyde: Brill, 1931), p.158 ff, particularly see n.1, where Ibn al-Khatib is quoted calling this madrasa as'bikr al-madāris.

- 37. Nwiya, <u>op</u>. <u>cit</u>.
- 38. Nafh al-Tib, Vol. 1, p.205.
- 39. Muhsin Mahdi, Ibn Khaldun's Philosophy of History, p.35, n.2.

. . .

40. See p.109

1

- 41. Qādī ^clyād (d. 544 A.H.) and Ibn Hirzihim issued a f<u>atwā</u> giving orders to burn Ghazāli's <u>Ihyā^cUlūm al-Dīn</u>. See n. 59,60.
- 42. Nayl, p.261.
- 43. Ih_sān Abbās, Introduction to Ibn al-Khaţīb's <u>Al-Katība al-Kāmina</u>, op. cit., p.10f.
- 44. Nubāhī, op. cit., p.201.
- 45. Ibid.
- 46. lbid., p.202.
- 47. Nafh al-Tib Vol. VII, p. 49 ff.
- 48. Among them the following has been edited and studied by Abdelmagid Turki, in his article "Lisān al-Din Ibn al-Hatib (713-76/1313-74), Juriste d'aprés son oeuvre inédite, Mutha al-Tariqa fi Damm al-Watiqa (<u>Mutha lā al-Ţariqa fi dhamm al-Wathiqa</u>) <u>Arabica</u>, Vol. XVI (1969), pp.155-211 and 280-311.
- 49. E. Sharqāwi, <u>Religion and Philosophy in the Thought of Fakhr al-Din</u> al-Rāzī (Unpublished thesis, McGill University, 1970), p.285 f.
- 50. Brockelmann: G.A.L. 1, p.667 and SI 921.
- 51. Makhlūf, Shajarat al-Nūr al-Zakiyya, Vol. I (Cairo, 1930-31), pp. 167-168.
- 52. lbid., p.218.
- 53. Muhsin Mahdi, <u>op. cit</u>., p.30, n.3.
- 54. lbid., p.35, n.2.
- 55. P. Nwiya, op. cit., p.XVIII.
- 56. J. Spencer Trimingham, The Sufi Orders in Islam, (Oxford, 1971), p.46.

57. Brockelmann, G.A.L. SI, p.830.

Margaret Smith, <u>Al Ghazali, The Mystic</u>, (London: Luzac, 1944), in Chapter XIII mentions other critics of al-Ghazāli and omits <u>Turţūshi</u> whose refutation of <u>lhyā</u> is one of the most significant and voluminous contributions towards a criticism of al-Ghazāli. The reason why she omitted this work is probably her statement about the co-operation between al-Ghazāli and al-Ţurţūshi. She said, "...al-Ghazāli in consultation with Abū Bakr Ţurţūshi, a well-known authority on law and tradition (d.520/1126), addressed letters of advice to Yūsuf (bin Tāshufin), urging him to govern with justice..." (p.21). She made this statement on the authority of De Slane's translation of Ibn Khaldūn's statement in this connection. The original statement by Ibn Khaldūn, however, as we quote below, does not imply a co-operation between al-Ghazālī and al-Ţurtūshī. This misunderstanding probably led her to suppose the above-mentioned fact. Ibn Khaldūn's sentences are as follows:

وشالحب الاعام المغذالى والقاصى الجنك الطوطونتى بجضائ على العددل والتمسلظ بالخلار

(Kitāb al-⁴lbar, op. cit., Vol. VI, p.386).

De Slane's translation (Histoire des Berbères, Vol. II, (Paris, 1927), p.82 is as follows: "L'imam El-Ghazzali et le cadi Abou Bekr-et-Tertouchi lui addresserent aussi des lettres de conseils et l'engagèment de la manière la plus pressante à gouverner avec justice..."

- 58. A. M. M. MacKeen, "The Early History of Süfism in the Maghrib prior to Al-Shādhilī (d. 650/1256)" in the Journal of the American Oriental Society, Vol. 91, 1971, P.402.
- 59. Muḥammad al-Zabīdī al-Murtadā, Ithāf Sādat al-Muttaqīn, commentary on Ihyā'Ulūm al-Dīn, (written in 1193 A.H.), (Cairo, 1893), Introduction, Vol. 1, p.10.
- 60. Ibid., p.27.
- 61. Ambroxio Huici Miranda, "The Iberian Peninsula and Sicily" in The Cambridge History of Islam. Vol. 11 (Cambridge: Cambridge University Press, 1970), p.427.
- 62. Arnaldez, "Ibn Rushd", <u>E.1</u>. new edition, p.910, referring to D.B. Macdonald's view on this point.
- 63. George Marçais, "Note sur les ribâts en Berbérie" in "Mélanges René Basset, Vol. II (Paris, 1925), p.399. Particularly, see his article "Ribât" E.I. (1st ed.), Vol. III, pp. 1150–1152.

- 64. H. A. R. Gibb, Introduction to "Ibn Battūta: Travels in Asia and Africa 1325–1354, (London, 1963), p.34.
- 65. Ibn Baţţūţa, Tuḥfa al-Nuzzār fī Gharā'ib al-Amṣār wa 'Ajā'ib al-Asfār, Vol. II, Ed. 'Awāmirī Bek and Jād al-Mawlā, (Cairo, 1934), p.294.
- 66. ^(Inan, Nihāya, op. cit., p.449.)
- 67. See for instance Ibn Qabbāb's appointment to supervise a trust donated by the Sultān to a Zāwiya, See p. 104.
- 68. See p.121f.

.

- 69. Shihābal-Dīn Qarāfī, AlFurūq, Vol. IV, (Cairo: Dār Iņyā Kutub al-Arabiya, 1346 A.H.), p.210.
- 70. See Ch. VI, p. 263ff.
- 71. S. Trimingham, op. cit., p.50.
- 72. Maqqari, <u>Nafh al-Tib</u>, Vol. VII, <u>op</u>. <u>cit.</u>, pp. 232-249 gives long extracts from this work.
- 73. <u>Ibid.</u>, p.261.
- 74. lbid., p.189-90.
- 75. P. Nwiya, Ibn ^CAbbād de Ronda, op. cit., pp.XXXIX and XLI relates a similar chain from Ibn Qunfudh. He observed certain historical defaults in this chain, and provides a detailed criticism on this point.
- 76. It is curious to note that in the third patch Ibn al-Arabi (Qādī Abū Bakr) and Ibn Hirzihim are juxtaposed with Ghazālī. Ibn Hirzihim, as has been pointed above, is known for his fatwā against Ghazālī's Ihyā, and Qādī Ibn Arabī's commentary bears a general trend of opposition to Sūfism.
- 77. <u>Nafh al-Tib</u>, Vol. VII, op. cit., p. 124–125 quoting from <u>Al-Fath</u> al-Munir.
- 78. Shāṭibī's f<u>atwā</u> issued in 786 A.H., preserved by Wansharīsī, <u>op</u>. <u>cit.</u>, Vol. XI, p.34.

- 79. In the Muslim East the practice of celebrating the Prophet's birthday started earlier, but in the Muslim West, according to Salāwī, it was started by the Marīnī ruler, Abū Yūsuf Ya<qūb in 691 A.H., from whence it came to Andalus, Salāwī, <u>Al-Istiqsā</u>, op. cit., Vol. 3, p.290. Ibn Khaldūn, op. cit., pp. 864,881 and 885 mentions how on three such occasions poems were recited. These celebrations lasted for a few days. Verses from the Qur'ān were recited, and animal sacrifices were offered. Maqqarī, <u>Azhār al-Riyād</u>, (Cairo, 1939), Vol. 1, p.245. Same fuqahā' considered this a bid'a, and opposed it. Shāţibī, in a fatwā, refused the validity of a will which wished to dispose of one third of the property for the purpose of the Prophet's birthday celebrations. Wansharīsī, <u>op. cit</u>. Vol. IX, p.181.
- 80. See p. 117.
- 81. Ibn Bațtūța, op. cit., p.294.
- 82. See for the description of this rural town now called Canales in Ibn al-Khatib, Khatrat al-Tayf fi Rihat al-Shitā'wa'l Şayf in A. M. 'Abbādi, Mushāhadāt Lisān al-Dīn Ibn al-Khatib fi Bilād al Maghrib wa al-Andalus (Iskandariya, 1958), p.33 and 'Abbadi's note no. 4.
- 83. See Wansharisi, Vol. IX, p.31.
- 84. Ibid., pp. 34-36.

NOTES: CHAPTER III

Section Three

- The best contemporary source of information on the economy and geography of this period, in our opinion, is Ibn al-Khatib's following two treatises: Khatrat al-Tayf wa Rihlatal-Shitā'wa al-Şayf and Mi'yār al-Ikhtibār fī dhikr al-Ma'āhid wa'l -Diyār edited and published by A. M. 'Abbādī, in Mushāhadāt Lisān al-Din Ibn al-Khatib, op. cit. The above must be supplemented with the study of fatāwā of this period (i.e. 14th century) by Lopez Oriz, "Fatwas granadianas de los siglos XIV-XV, Al-Andalus, Vol. VII (1941), pp. 73-127.
- 2. Levi-Provençal, "Al-Andalus", E.I.², pp.486-492.
- Quoted in Maqqari, <u>Naf h al-Tib</u>, Vol. I, p.124, translation Pascual de Gayangos, <u>The History of the Mohammedan Dynasties in Spain</u>, Vol. I (Cairo: 1902), p.17.
- 4. Ibn al-Khatib, Al-Ihāta, Vol. 1, p.32.
- 5. Imamuddin, A Political History of Spain, p.294.
- 6. Cf. Seybold, "Granada", in E.I. (1st edition), Vol. II, p. 176.
- 7. Iḥāṭa, op. cit. 1, p.38.
- 8. Ibn Battūta, translation H. A. R. Gibb, Ibn Battūta, Travels in Asia and Africa 1325–1354, (selection), (London, 1963), p.319.
- 9. ^(AbdAllāh) Inan, <u>Nihāyat al-Andalus</u>, p. 430. ^{(Inān does not indicate his source but probably he derives this information from Prescott, <u>History of Ferdinand and Isabella the Catholic</u> on which he relies for the most part of the data of this period.}
- 10. A detailed description of the reserves of the Nasrid treasury are given by 'AbdAllah Muhammad b. al-Haddād al-Wādī 'Ashī as quoted by Muhammad Kamāl Shabāna, "Al-Hāla al-Iqtişādīya bi'l-Andalus Khilāl al-Qarn al-Thāmin al-Hijrī" in Al-Bahth al-'llmī, Ribāţ, Vol. III (August, 1966), p.137. Unfortunately, Shabāna's reference to the original source is not clear.

- 11. For this information we have relied on Lopez Ortiz's above-mentioned study. See Ibid, pp.95-97.
- 12. Nafh al-Ţib, Vol. I, p.202.
- 13. Lopez Ortiz, op. cit. p.96.
- 14. Imamuddin, op. cit., p.294.
- 15. Lopez Ortiz. op. cit., p.95.
- Antonio Vives, "Indicacion del Valor en las Monedas Arabigo Espanolas", in D. E. Saavedra. (Ed.) Homenaje a D. Fransisco Codera, Zaragoza, 1904, p.522. Also, see D. F. Codera Y Zaidin, <u>Tratado de Numismatica Arabi-Espanola</u>, (Madrid, 1879), p.231.

N.

t

- 17. See H. W. Hazard's analysis of the metrology of the coins of North Africa (which partly includes Spain as well) in The Numismatic History of Late Medieval North Africa, American Numismatics Society, New York, 1952, pp.48-49. Hazard says that the Zirid dinār was 4.11 - 4.35 grams in weight and 22-24 milimeters in diameter. The Muwaḥhidūn introduced double dinārs averaging 4.55 grams and 27-32 milimeters, whereas their normal dinār averaged 2.27 grams and 19-22 milimeters. The Nasrid dinar seems to have been better than its predecessors. Two specimens registered in M. H. Lavoix, Catalogue des monnaies musulmanes de la Bibliotheque Nationale, Vol. V (Paris, 1891), pp.328-329, provide the following data:
 - Catalogue no: 780: Yūsuf b. Muḥammad (1333-1354), gold; weight
 4.65 grams, diameter 31 milimeters.
 - 2) Catalogue no: 781: Muḥammad V al-Ghani Billāh (1354-1359 1362-1391), gold; weight 4.70 grams; diameter 32 milimeters.
- 18. Luis Seco de Lucena, in Documentos Arabigo-Granadinos, Instituto des Estudios Islamicos, (Madrid: 1961), studied a number of documents of a judicial nature belonging to fifteenth century Granada. In his analysis he found very interesting data bearing on the social and economic conditions of that period. We have derived our information from Lucena's analysis of currency in these documents as given by him in Ibid., pp.XLVI-XLVIII.
- 19. Most probably Lucena is here referring to Antonio Vives, <u>op</u>. <u>cit</u>., and Fransisco Codera, op. cit.
- 20. See Lopez Ortiz, op. cit. p.94f.

- 21. See such statements by early historians and geographers as quoted extensively by Maqqarī, Nafh al-Tīb, Vol. I, pp.124-194, but particularly pp.136-138, where the description by Razi, the geographer, is quoted at length.
- 22. "Andalus" <u>E.I.²</u>, p.491.
- 23. Lopez Ortiz, op. cit., p.97, 106f.
- 24. lbid., p.106ff.
- 25. Ibid.

 $\xi = \infty$

Ļ

- 26. Ibid., p.109.
- 27. Ibid., p.103.
- In the days of Muhammad V, the irrigation system was further improved.
 See Imamuddin, op. cit. p.294.
- 29. On this point we are relying on E. F. Heckscher, "Mercantilism" in H. W. Spiegel, The Development of Economic Thought (New York: Science Books, 1966), pp.32-41.
- 30. Lopez Or tiz, op. cit. p.114.
- 31. Ibid., pp.114 ff. For details see Ch. V, p.231.
- 32. Ibid., p.110. For details see Ch. V, p.232.
- 33. Ibn al-Khatib, Al-Ihāta, Vol. I, p. 15.
- 34. Louis Bertrand and Charles Petrie, <u>The History of Spain</u>, trans. W. B. Wells (New York, 1934), p.204.
- 35. Ibn al-Khatib, Mufākharāt Mālaga wa Salā, op. cit., pp. 77–90.
- 36. Nafh al-Tib, op. cit., Vol. 1, 154.
- S. M. Bastieva, "Ibn Khaldūn et son milieu social", <u>Atti del Terzo</u> Congresso, op. cit., p.138.
- 38. ^(Inan, op. cit., p.429.)
- 39. Ibid., p.428.

- 40. See p. 87.
- 41. See p. 92.

е 1. ж

- 42. Bastieva, op. cit.
- 43. Lopez Ortiz, op. cit., p.95.
- 44. ^cInān, <u>op</u>. <u>cit.</u>, p. 428.
- 45. Ibid., p.97.
- 46. Ibid., p.100.
- 47. lbid., p.110.
- 48. Lopez Ortiz, op. cit., p.101.
- 49. lbid., p.98.
- 50. lbid., p.94f.
- 51. Dozy, Supplement aux dictionnaires Arabes, Vol. II (Paris, 1967), p.275.
- 52. Al-Quran 90: 12-13: "What would make thee know what is an uphill task? The freeing of a slave or captive."
- 53. Lane, Arabic English Lexicon, (London: 1874), Vol. 6, p.2431.
- 54. E.N., Van Kleffens, Hispanic Law until the End of the Middle Ages (Edinburgh. 1968), p. 105, no.3, refers to the Hispanic legal term "alfaqueques" meaning the administrator of the property of the prisoners of war.
- 55. L. Bertrand, The History of Spain, op. cit., p.200. This observation is supported by Ibn Buttuta's narration of such an attempt by the notables in Malaga (below note:53). Ibn al-Khatib also praised the people of Malaga for the purchase of freedom of such prisoners. See <u>Miyar al-Ikhtibar</u>, op. cit., p.78.
- 56. Ibn Battūta, Tuhfat al-Nuzzār...op. cit., pp.290-292.
- 57. See al-Wansharisi, op. cit., Vol. II, p.127.
- 58. See above note 54.
- 59. Lopez Ortiz, op. cit., p.95.
- 60. lbid., p.95.

NOTES: CHAPTER III

Section Four

- 1. See p. 102ff.
- 2. See p. 211ff.
- 3. See p. 104.
- 4. See p. 220. The story of the litigant in case of habs.
- Shihāb al-Din Qarāfi, Al-lhkām fi tamyiz al-fatāwā 'an al-Ahkām. (Ed.)
 ^SAbd al-Fattāh Abū Ghadda. (Halab: Maţbū'at Islāmiyya, 1967).

Ŋ.

- 6. Ibid., p. 18.
- 7. lbid., p.20
- 8. Ibid., pp. 22-23
- 9. Ibid., p.32.
- This conclusion is drawn from the following sources: Lopez Ortiz, <u>Fatawa Granadinas</u>...op. cit. p.91; Wansharisi, <u>Al-Mi^cyār</u>...<u>op</u>. cit. Vol. II, p.166; Vol. V, 186f.
- 11. See p.184f.
- 12. Nayl, p.265.
- 13. Shajarat, p.230.
- 14. This treatise was studied by Abdelmagid Turki in "Lisan al-Din Ibn al-Hatib (713-761/1313-74), Juriste d'après son oeuvre inédite: <u>Mutla al-Tariqa...</u>, <u>Arabica</u>, XVI (1969), pp.155-211; 280-312.
- 15. Ibid., pp.280, 284 and passim.
- E. N. Van Kleffens, <u>Hispanic Law until the end of the Middle Ages</u>, (Edinburgh: University Press, 1968), p.124.
- 17. Ibid., p.154.
- 18. Ibid., p.74.
- 19. lbid., p.77.

20. lbid., p.79.

- 21. lbid., p.145.
- 22. Ibid., p.146.
- 23. Ibid., p.147.
- 24. Ibid., pp.176-78.
- 25. lbid.
- 26. Henry Lacerte, The Nature of Canon Law According to Suarez (Ottawa: University of Ottawa Press, 1964), p.3.

١.

27. Ibid., pp.20ff.

CHAPTER IV

4 - 1

a ->

SHATIBI: HIS LIFE AND WORKS

This chapter attempts to construct a sketch of certain significant events in Shāţibī's life which, as we shall see, in the absence of sufficient data about his life, are very helpful in an understanding of the reasons for Shāţibī's interest in the philosophy of Islamic law.

When writing a biography of Shāṭibī, one's attention is drawn first of all to the scarcity of data about his life, although he was one of the most prominent among Mālikī jurists. An answer to the question of why there should be so little information on so important a man is attempted. This is followed by a discussion of the information available about his life, his career, his disputations with other scholars, and his works.

SOURCES

To my knowledge Ahmad Bābā¹s (d. 1036/1626) <u>Nayl al-Ibtihāj²</u> contains the first biographical notice on Shātibī.

Among his contemporaries Lisān al-Dīn Ibn al-Khaţib (d.776/1374) and Ibn Khaldūn (d.784/1382) wrote at length about Granada and scholars living there in this period. Although it would be a reasonable assumption that both Ibn al-Khaţib and Ibn Khaldūn would have known Shāţibi, he goes unnoticed in their accounts. Ibn al-Khaţib and Shāţibi had common teachers³ (and one of the sources even describes Ibn al-Khaţib as a pupil of Shāţibi)⁴ and common friends.⁵ Ibn Khaldūn wrote a treatise,⁶ in response to Shāţibi's query addressed to the scholars in the West. Nevertheless, neither of these important writers makes mention of Shāţibi.

A possible explanation for this omission might be that Shāṭibī had not yet written his controversial work, <u>al-Muwāfaqāt</u>, when the other two composed their works. This is quite possible because Shāṭibī refers to Ibn al-Khaṭīb's <u>Al-Iḥāṭa</u> in his work (though without mentioning his name).⁷ This reference means that Shāṭibī's work must have been written after the completion of <u>Al-Iḥāṭa</u>, as we believe after 771/1369.⁸ This fact also explains Ibn Khaldūn's omission of Shāṭibī's name. Ibn Khaldūn visited Granada in 764-65/1362-63⁹ while Shāṭibī had not yet become a sufficiently controversial figure to attract notice at that time.

Among the authors of the <u>Tabaqat</u> of the Mālikis,¹⁰ Ibn Farhūn (d.799/1396), author of Al-Dibāj al-Mudhahhab was Shātibi's contemporary, but did not mention him. Since it cannot be established whether Ibn Farḥūn was acquainted with Shāṭibī we cannot be certain that this exclusion of Shāṭibī from <u>al-Dībāj</u> was deliberate. One possible explanation could, however, be suggested.

Ibn Farhūn was born in Medina and, except for a few journeys to the West, ¹¹ he passed most of his life in the East of the Muslim world. His knowledge of the Muslim West, though generally thorough, was based on secondary sources.¹² Besides, he had already completed <u>al-Dībāj</u> in 761 A.H. ¹³ As was previously suggested, it is most probable that Shāţibī had not yet written his <u>al-Muwāfaqāt</u>. Otherwise, Ibn Farḥūn could not possibly have overlooked him. The basis of our conjecture is Ibn Farḥūn's insistence on including in his <u>al-Dībāj</u> only the names of those who had been authors of some books.¹⁴

Badr al-Dīn⁶Qarāfī (d. 1008/1599) is known to be the next writer of Ţabaqāt after Ibn Farḥūn.¹⁵ His <u>Tawshīḥ al-Dibāj</u>¹⁶ is the complement of <u>al-Dībāj</u>. He too does not mention Shāṭibī. His reasons seem to be the same as those we suggested in the case of Ibn Farḥūn. In a number of places, as Aḥmad Bābā points out in strong language, ¹⁷ Qarāfī, lacking sufficient knowledge of the West, confuses the names and kunyas of many well-known scholars.

Aḥmad Bābā is not only the first biographer but also an original authority in this respect. Almost all of the later scholars who have taken notice of Shāṭibī belong to the twentieth century, and they depend largely on Aḥmad Bābā's notice.¹⁸ Aḥmad Bābā treats of Shāṭibī in <u>Nayl al-Ibtihā</u>j as well as in Kifāyat al-Muḥtāj¹⁹

÷

which supplemented the former. <u>Nayl</u> was written during Aḥmad Bābā's internment period in Morocco, where he was taken as a prisoner after the invasion of the Sudan by the Sultān of Morocco in 1591. There, Aḥmad Bābā, though he was without his personal collection of sources, was able to use the books in the possession of Moroccan scholars and in the libraries.²⁰

The reasons why Ahmad Baba mentioned Shātibi while his predecessors did not, could be the following:

First, as a general reason, the <u>Nayl</u> was meant to be a supplement to <u>al-Dibāj;</u> "complementing what was missing in it and supplementing it with (the mention of) those eminent <u>a'imma</u> who came after him".²¹

Second, he was certainly better informed about the learned tradition in the Muslim West²² than Qarāfī or Ibn Farḥūn, and hence he was capable of making up the deficiencies of al-Dībāj.

Third, he felt this deficiency more strongly because for a long time there was no other work on the subject but that of Farhūn,²³ and this too suffered from grave faults.

Apart from such general considerations, Aḥmad Bābā's high regard for Shāṭibī may be suggested as a specific reason why Aḥmad Bābā mentioned Shāṭibī. This esteem is reflected in the honorific titles with which he mentions Shāṭibī.²⁴ His regard for Shāṭibī further manifests itself when he disputes Abū Ḥāmid Makkī's claim for his master Ibn ^cArafa (d.803 A.H.)²⁵ as "being peerless in taḥqīq (the skill of applying general principles of (Māliki) school to particular cases)".²⁶ Aḥmad Bābā mentions Shāṭibī as one example of scholars who were in no way Iesser than Ibn ^cArafa.²⁷ Elsewhere he says,

"Among the people of the ninth century (sixteenth) there are those who assert their attainment of the status of <u>ijtihād</u>, while Imāmal-Shāṭibī and Ḥafīd Ibn Marzūq (d.842/1438) declined it for themselves. It is certain that both of them had more profound knowledge (of <u>Sharī'a</u>) and thus (were) more deserving of this status than those who claimed it.28

We have dwelt long on the question of why Ahmad Bābā first took notice of Shāțibī while others did not. Let us now discuss Ahmad Bābā's sources for his biography of Shātibī.

Beside the sources mentioned towards the end of <u>Nayl</u>, the most significant among them being Wansharīsī, ²⁹ Aḥmad Bābā used Shāṭibī's own work <u>Al-Ifādāt wa'l</u> <u>Inshādāt</u>. ³⁰ This work seems to consist of Shāṭibī's class notes and of anecdotes narrated by his teachers. The extracts from this work, as quoted by al-Maqqarī³¹ in his <u>Nafḥ al-Ṭib</u> and by Aḥmad Bābā in <u>Nayl</u>, indicate that the '<u>Ifādāt</u> must contain considerable information about Shāṭibī's teachers and himself. If that be so, Aḥmad Bābā's information about Shāṭibī may be taken as first hand.

As to our information in the following pages, it is based mainly on <u>Nay1</u>. We have used the extracts of <u>'Ifādāt</u> as quoted in <u>Nay1</u> and <u>Nafh</u>. We have also used Shāţibi's <u>al-Muwāfaqāt</u> and <u>al-Istişām</u>. The preface of <u>al-Istişām</u> explains the circumstances that led to Shāţibi's thought on <u>shari'a</u> passing through various stages and how he was accused of "heresy". <u>Al-Muwāfaqāt</u> refers to the discussions³³ in which Shāţibi became involved with other scholars.

Ĩ

To sum up, we may say that the information which follows has been compiled from Nayl and from Shāțibi's own works.

•

SHATIBI'S LIFE

1.4

His full name is reported as Abū Ishāq Ibrāhīm b. Mūsā b. Muḥammad al-Lakhmī al-Shāṭibī. We know virtually nothing about his family or his early life. The most that we can learn by deduction from his <u>nisbas</u>, is that he belonged to the Lakhmī Arab tribe. We know also that his immediate family came from Shāṭiba (Xativa or Jativa). This latter nisba has misled some scholars to maintain that Shāṭibī was born or lived in Shāṭiba before coming to Granada.³⁴ This is not possible because Shāṭiba was taken by the Christians a few decades aga, and, according to the chronicles, the last Muslims were driven out of Shāṭiba in 645/1247.³⁵

Shāṭibī grew up in Granada and acquired his entire training in this city which was the capital of the Naṣrī kingdom. Shāṭibī's youth coincided with the reign of Sulṭān Muḥammad V al-Ghanī Billāh, a glorious period for Granada.³⁶ The city had become a centre of attraction for scholars from all parts of North Africa. It is not necessary to list here all the scholars who visited Granada or who were attached to the Naṣrī court, names such as Ibn Khaldūn and Ibn Khaṭīb being sufficient to illustrate our point.

Training

--- - - We do not know when and what subjects Shāṭibī studied for his training. What follows is the account of some of his teachers, from which an idea of his training may be drawn. It appears that, according to normal practice, Shāṭibī started his training with studies in Arabic language, grammar and literature. In these subjects, he benefited from two masters. He began his studies with Abū ^fAbd Allāh Muḥammad b. ^fAlī al-Fakhkhār al-Bīrī³⁷ who was known as the master of grammarians (Shaykh al-Nuḥāt) in Andalus. Shāţibī stayed with him until the latter's death in 754/1353. Shāţibī's notes about al-Fakhkhār in ^{/he}/lfādāt illustrate clearly that he received a thorough training in matters pertaining to the Arabic language.³⁸

His second teacher in the Arabic language was Abū'l Qāsim al-Sharīf al-Sabtī (760/1358), author of the well-known commentary on Maqsūra of Hāzim.³⁹ He was called "The Bearer of the Standard of Rhetoric".⁴⁰ He was chief Qādī in Granada in 760/1358.

The famous Andalusian fagih Abū Sa^cid Ibn Lubb began his lectures in the Madrasa Naşriya in 754/1353.⁴¹ Most probably he succeeded al-Fakhkhār on the latter's death. Ibn Lubb was well versed in figh and was recognized for his "rank of <u>ikhtiyār</u> (decision by preference) in respect to fatwā".⁴² Shāṭibī's training in figh was almost entirely completed with Ibn Lubb. Shāṭibī owes much to this man, but he also entered into controversy with Ibn Lubb on a number of issues.⁴³

We need not recount the names of all of Shājibi's teachers;⁴⁴ it seems he benefited from all well-known scholars in Granada as well as those who visited Granada on diplomatic missions. Among such scholars mention must be made of Abū 'Abd Allāh al-Maqqari⁴⁵ who came to Granada in 757/1356 on a diplomatic mission sent by the Marīni Sultān Abū 'Inan.⁴⁶ Maq**q**arī had an eventful career. Sultān Abū 'Inān chose him as his chief Qādī, but soon Qādī Abū 'Abd Allāh al-Fishtālī succeeded in having him deposed. Maqqarī was sent to Granada from whence he refused to return to Fez. The Nastri Sultan arrested him and sent him back. Abū'l Qāsim al-Sabtī and Abū'l Barakāt Ibn al-Hājj al-Balfīqī, qādīs of Granada, followed him to Fez to secure his release. Nevertheless, Maqqarī was tried by al-Fishtālī and was convicted.⁴⁷

43.

Maqqarī's academic tastes were versatile. He is the author of a book on Arabic grammar. He was known as holding the rank of "<u>muhaqqiq</u>"⁴⁸ (expert on the application of general principles of the *M*aliki/school to particular cases).

Maqqarī seems to have acquainted Shātibī with Rāzism in <u>uşūl al-fiqh</u>. He started to compose an abridgement of Fakhr al-Dīn Razī's (606/1209) <u>al-Muhaşşal</u>.⁴⁹ He is also the author of a commentary on <u>Mukhtaşar</u> of Ibn Hājib who introduced Razism into Mālikī uşūl al-fiqh.

Maqqarī is also responsible for initiating Shāţibī into sūfism – a special <u>Silsila</u> of which we have spoken elsewhere.⁵⁰ Maqqarī is known for his book <u>al-Haqā'iq</u> wa'l-raqā'iq fi al-taşawwuf.⁵¹

Mention must also be made of two of Shāțibi's teachers who introduced him to <u>falsafa</u> and <u>kalām</u> and other sciences which are known in the Islamic classification of the sciences as the rational sciences (al-culum al-caqliya) as opposed to the traditional sciences (al-culum al-naqliya).

Abū ⁶Alī Manşūr al-Zawāwi⁵² came to Granada in 753/1352. Ibn al-Khaţib praises him highly for his scholarship in traditional as well as rational sciences. He appears to have run into frequent controversy with the jurists in Granada. He was accused of various things. Finally in 765/1363, he was expelled from the Andalus.⁵³
Shāţibi mentions Zawāwi quoting his teacher, Ibn Musfir, saying that in his commentary on the Qur'ān, Rāzi relied on four books, all written by the Mu^ctazilis; in <u>uşūl al-din</u> Abū'l Husayn's <u>Kitāb al-Dalā'il</u>, in <u>uşūl al-fiqh</u> his <u>al-Mu^ctamad</u>, in <u>uşūl al-tafsir</u> on Qādi ^cAbd al-Jabbār's <u>Kitāb al-Tafsir</u>(?), in <u>Uşūl al-^cArabiya</u> and <u>bayān</u> on Zamakhshari's <u>Kashshāf</u>.⁵⁴ This comment seems to imply that Zawāwi and his teacher saw in Rāzi a continuation of Mu^ctazili <u>kalām</u>.

Al-Sharif al-Tilimsäni (d. 771/1369) also seems to have been critical of Rāzism. He studied with 'Ābili and specialized in the rational sciences. Ibn Khaldun mentions that Tilimsäni secretly taught Ibn 'Abd al-Salām the books of Ibn Sina and Ibn Rushd.⁵⁵ Tilimsäni was well-versed in both the traditional and the rational sciences. Contemporary scholars laid stress on his attainment of the rank of <u>Mujtahid</u>.⁵⁶ Ibn 'Arafa lamented Tilimsäni's death as the death of the rational sciences.⁵⁷

From the above account of his notable teachers it may be concluded that Shāṭibī's training must have been quite thorough in both the traditional and the rational sciences. His main interests, however, as we shall see from the list of his works, were concentrated upon the Arabic language and <u>usul al-fiqh</u>, particularly the latter.

Shātibi's Interest in Uşūl al-Fiqh

Figh was a very profitable and hence popular subject, but interest in <u>uşūl al-figh</u> was rare in the Andalus.⁵⁸ What induced Shāțibī to interest himself in <u>uşūl al-figh</u> was his feeling that the weakness of <u>fiqh</u> in meeting the challenge of social change was due largely to its methodological and philosophical inadequacy. This weakness struck Shāțibī very early in his training years. He says:

Ĩ

"Ever since the unfolding of my intelligence for understanding (things) and ever since my anxiety was directed towards knowledge, I always looked into its (the <u>sharica</u>'s) reasons and legalities; its principles and its branches. As far as the time and my capacity permitted I did not fall short of any science among the sciences, nor did I single one out of the others.

l exploited my natural capacity or rather plunged into this tumultuous sea...so much so that I feared to destroy myself in its depths...until God showed His kindness to me and clarified for me the meanings of Sharī^ca which had been beyond my reckoning...

From here I felt strong enough to walk on the path as long as God made it easier for me. I started with the principles of religion (<u>usūl al-dīn</u>) in theory and in practice and the branches, based on these problems. (It was) during this period (that) it became clear to me what were the bida[<] and what was lawful and what was not. Comparing and collating this with the principles of religion and law (figh), I urged myself to accompany the group whom the Prophet had called <u>sawād al-a[<]zam</u> (the majority)." ⁵⁹

One of the most perplexing problems for Shāţibī was the diversity of opinion among scholars on various matters. Use of the principle of <u>murā<āt al-khilāf</u> made the problem even more complex. This principle, as we shall see below, ⁶⁰ was employed to honour differences of opinion by treating them all as equally valid. Because of this attitude, diversity of opinions was proudly preserved even from the earliest days of Mālikī figh. Shāţibī himself recalled that the diversity in the statements of Mālik and his companions used to occupy his mind frequently. ⁶¹

Studying with Abū Sa^cid b. Lubb, Shāṭibī faced such perplexities very often. He states:

22

"I once visited our master, Abū Sa^cid b. Lubb, the <u>mushāwir</u>, along with my friends...He said, "I wish to inform you about some of the basic principles on which I relied in such and such a fatwā, and (to explain) why I intended for leniency in that". We knew about his <u>fatwā</u>...we disputed with him on his answer...He said, "I want to tell you a useful rule in issuing a <u>fatwā</u>. This rule is authentically known (as practiced) by the scholars. The rule is not to be hard on the one who came asking for a <u>fatwā</u>." Before this meeting various aspects in the statements of Mālik and his companions used to confuse me. But now God cleared my mind with the light of this discourse." ⁶²

This satisfaction, however, did not last long. His indulgence in the problem of <u>murātā al-khilāf</u> shows that Ibn Lubb's clarification was not satisfactory. Shāṭibī felt that the body of the law was without spirit, its formalism will remain devoid of reality unless the real nature of the legal theory was investigated.⁶³ Shāṭibī's works were dedicated to such an investigation.

Shātibi's Career

We do not find any allusion to Shāṭibī's career or to his profession. Three conjectures, however, can be made. First, in Shāṭibī's account of the accusations brought by people against himself, on one occasion it can be deduced that he was an imām and also a khaṭīb in a certain mosque. During his period of trial, it can be assumed, he was dismissed from these posts.⁶⁴

The second conjecture can be made on the basis of the <u>fatwas</u> asked from him, that he was a <u>muff</u>. Since he is never called <u>al-mushawir</u>, it may be assumed that he was not officially appointed to this office.

He, however, had a number of disciples. From this, a third conjecture can be made, that he taught in the madrasa of Gharnāta.

Among his disciples, Ibn ^CÁşim is noteworthy. He became the chief qāḍī of Granada. He is known for his <u>Tuḥfat al-Ḫukkām</u>, a compendium of <u>figh</u>ī rules compiled for <u>qāḍī</u>s. He also wrote an abridgement of Shāṭibī's <u>al-Muwāfaqāt</u>.⁶⁵

. . .

His Death

Shāțibi died in 790/1388.66

Shāțibi Accused of Heresy

Sometime during his career Shātibī was accused of introducing innovations (bida<). The exact date of this period of trial is not known. The inquisitive mind of Shātibī led to discussions and controversies with other <u>fuqahā</u>'. Most probably the period of trial occurred during the time he was writing his book <u>al-Muwāfaqāt</u>, when he corresponded with scholars about a number of subjects.

Shāțibi's verses in reference to this trial indicate how he felt about these accusations. He says:

O my people you put me to the ordeal (balayta) whereas an ordeal shakes violently, The one who whirls with it, until it seems to destroy him, (You condemn me) to prevent wrong, rather than to attain any good (maşlaba). May God suffice me in my reason and religion.

Shāțibi recounts the story of this ordeal in Al-Istişām in the following words:

"I had entered into some of the common professions (<u>khutat</u>) such as <u>khutāba</u> (preaching) and <u>imāma</u> (leading the prayers). When I decided to straighten my path, I found myself a stranger among the majority of my contemporaries. The custom and practice had dominated their profession; the stains of the additional innovations had covered the original tradition (<u>sunna</u>).... I wavered between two choices; one to follow the sunna in opposition to what people had adopted in practice. In that case I would inevitably get what an opponent to the *f*social practices would get, especially when the upholders of this practice claimed that theirs was exclusively the sunna....The other choice was to follow the practice in defiance of the sunna and the pious ancients. That would get me into deviation *f*from the true path ... I decided that I would rather perish while following the sunna to find salvation...

I started acting in accordance with this decision gradually in certain matters. Soon the havoc fell upon me; blame was hurled upon me... I was accused of innovation and heresy." ⁶⁹

Shāțibī, at this point, enumerates the following charges that were laid against ⁷⁰ him:

- Sometimes I was accused of saying that invocations(du^cā) serve no purpose...that was because I did not adhere to the practice of invocations in congregational form after the ritual prayer (salāt).
- (2) I was accused of raid (extreme shicism) and of hatred against the companions...that was because I did not adhere to the practice of mentioning the names of the pious Caliphs in the <u>khutba</u> (Friday sermon)...
- (3) I was accused of saying that I favoured rising against the <u>a rimma</u> (the ruler)...that was because I did not mention their names in the <u>khutba</u>.
- (4) I was accused of affirming hardship in religion...that was because I adhered to the well-established tradition in duties and <u>fatwā</u>s, while they ignored it and issued <u>fatwā</u>s in accordance with what was convenient to the enquirer...
- (5) I was accused of enmity against the <u>awliyā</u>² of Allāh (friends of God)... that was because I opposed some of the innovating <u>sūfis</u> who opposed sunna...."

Shāṭibī was accused of <u>bid<a</u> (heresy) mainly because he opposed the practices of the <u>fuqahā</u>'. Particularly, as we shall see later, ⁷¹ one of the controversial problems was that of mentioning the name of the Sulṭān in the <u>khuṭba</u> and praying for him towards the end of the ritual prayers. Shāṭibī called this practice a <u>bid<a</u>. His action shook the foundations of the political power of the religious élite. On this issue, it is interesting to note that he was opposed by all the qādīs in Spain and North Africa as well as by some dignitaries holding government offices.⁷²

· · · · ·

1

Shāţibī's account of his trial for <u>bid<a</u>, refers to the controversies that brought him into conflict with other scholars. What follows are the details of his main disputations. Here we have limited ourselves to theoretical problems.

Ŋ

SHATIBI'S DISPUTATIONS

Taşawwuf and Fiqh

Shāțibī was much worried not only by the fact that <u>taṣawwuf</u> comprised a number of rituals which he considered as <u>bida</u>, but also by the fact that <u>taṣawwuf</u> was having an adverse effect upon <u>fiqh</u> and <u>uṣūl</u>. He did not oppose the <u>sūfī</u>s on certain matters if they followed their peculiar practices individually or as a requirement of <u>taṣawwuf</u>. What he opposed was that certain <u>sūfī</u>s or certain <u>fuqahā</u>' under the influence of <u>taṣawwuf</u> should suggest that these things were obligatory in a fiqhī sense. The following two issues became very prominent in this concern.

Ņ

1) The obligation of freeing one's inner self (sirr)

A certain scholar sent an epistle to Shāṭibī in which under the rubric, "what is obligatory for a seeker of the Hereafter to observe and do", he wrote the following:

"If a certain thing distracts someone from his prayers even for a while, he must free his inner self from this distraction by getting rid of it, even if these distractions number as many as fifty thousand." 73

Shāţibī objected to this statement strongly. He disputed its obligatory claim. He argued that if freeing the inner self were a universal obligation, it would lead to absurdity because it demands that people should get rid of their property and abjure their towns, villages and families since these things constitute distractions. He adds that poverty is the major source of distraction, especially if people are occupied with the worries of supporting large families.⁷⁴

2) Submission to a Shaykh

: :....

With the introduction of tariaas, suffism passed into a new phase. In the previous phase, more significance was attached to books on tasawwuf.⁷⁵ In the new phase, however, as we have pointed out earlier in the story of Abū'l Ma<ālī, the initiation without a <u>shaykh</u> was considered forbidden.⁷⁶ Such an emphasis on submission to a <u>shaykh</u> generated a debate among the scholars.

According to Shātibi submission to a <u>shaykh</u> led to a belief in the superiority of the <u>shaykh</u> to all other religious leaders, even to claim to be equal to Muḥammad.⁷⁷ According to some <u>suffis</u>, including Qushayri, suffism was nothing more than spiritual <u>fiqh</u> (<u>fiqh al-bātin</u>)⁷⁸ It was, therefore, questionable for Shātibi that one should submit oneself totally to a <u>shaykh</u> to be initiated into a discipline; the discipline could be known from books.

Shātibī composed a query in which he summarized the arguments of both parties and sent this to a number of scholars in North Africa. Three of the responses to this query have come down to us. Those of Ibn al-Qabbāb (d. 779/1377) and Ibn 'Abbād of Ronda (d. 792/1389) were preserved by Wansharīsī in his <u>Al-Mi'yār al-Mughrib</u>. They are reproduced by Paul Nwiya in <u>Al-Rasā'il al-Sughrā</u> of Ibn 'Abbād de Ronda. ⁸⁰ The third answer was written by Ibn Khaldūn in <u>Shifā'al-sā'il Ii tahdhīb al-masā'il</u>, available in two editions by Ţavīt Ţanjī⁸¹ and by Khalifé.⁸²

101

Ibn ^cAbbād maintained that, "on the whole, the (submission to a) <u>shaykh</u> is an essential fact in the journey on the path of <u>tasawwuf</u>; no one can deny that".⁸³ He, however, distinguished between two kinds of <u>shaykhs</u>: <u>Shaykh al-Tarbiya</u> (educator) and <u>Shaykh al-Taclim</u> (instructor). The former is not essential for every "traveller", while the latter is necessary for everyone. He also pointed out that reliance on the "educator" <u>shaykh</u> is the approach of the modern (<u>muta'akhkhirin</u>) <u>sūfi</u>s, while the ancients relied on the "instructor" <u>shaykh</u>.⁸⁴

Ibn ^cAbbād stressed that the initiation to the mystic state (<u>hāl</u>) exclusively belonged to special individuals. No one could open its doors except those whom God had chosen for that purpose.⁸⁵

3) Invocation after Prayers

The mention of the ruling Sultān or Khalīfa as a symbol of legitimacy had long been accepted in practice. Al-Muwaḥḥidūn gave the practice much more significance by making some additions. Especially the Muwaḥḥid Caliph ^cAbd al-Wāḥid al-Rashīd (630-640/1232-1242), fearing the dissensions among various groups of the family and in order to check a general decline of al-Muwaḥḥidūn, re-established **bnTūmart**'s institutions which had been discontinued by such caliphs as al-Manṣūr (580-595/1184-1199) and al-Idrīs Ma²mūn (620-630/1229-1232).⁸⁶ One of such institutions was the invocation of the name of the ruling caliph after the prayers in congregational form. This was an innovation, but it gradually became so much established that opposition to it was considered a political as well as a religious offence, punishable by death.⁸⁷

. . . .

Contrary to the claim of the other <u>fuqaha</u>[?] about the consensus on the acceptance of this practice, Shāṭibī argued that it was a <u>bid'a</u> and that scholars had always expressed their dissent against this practice. When this practice was introduced into Spain in the twelfth century, some of the Mālikī <u>fuqahā</u>[?], namely Abū ^CAbd Allāh b. Mujāhid (d. 574/1178) and his disciple Abū ^CImrān al-Mīrtalī, opposed it at the risk of their lives.⁸⁸

The practice continued even after the Muwaḥhidūn, obviously for political reasons. Most probably it was Shāṭibī who publicly opposed this practice by disregarding it whenever he was leading the prayers. This public act of defiance raised havoc for Shāṭibī. The issue became a subiect of heated discussion; Shāṭibī, however, did have some followers. From a letter written by Shāṭibī to one of his followers, it appears that an <u>imām</u> who rejected this practice in favour of Shāṭibī's position was deposed from his <u>imāma</u> and was denied all other privileges and was put to trial.⁸⁹

The first two refutations offered against Shāṭibī were the following: one by the Qāḍī of Andalusia, Abū'l Ḥasan al-Nubāhī's <u>Mas'alat al-đu'ā</u> <u>Ba'd al-</u>salāt, ⁹⁰ the other by the <u>muftī</u> and <u>mushāwir</u> of Granada and Shāṭibī's teacher Abū Sa'īd ibn Lubb. The book is called <u>Mas'ala al-</u>

1.55

'Ad^ciyya ithr al-şalāt.⁹¹

Shāţibī's disciple Abū Yaḥyā ibn ʿĀṣim (d. 813/1410) then wrote, refuting Ibn Lubb and supporting Shāţibī.⁹² Muḥammad al-Fishtālī, the Qādī al-Jamāʿa in Fez wrote a refutation of Ibn ʿĀṣim, supporting Ibn Lubb, entitled <u>Kalām fī 'l-duʿā baʿd al-ṣalāt ʿalā al-hay 'a al-maʿhūda</u>.⁹³ IbnʿArafa (d. 803/1400), the Qādi of Tūnis, also entered into the discussion when he was asked for a <u>fatwā</u> on this issue by someone in Granada.⁹⁴

Shāţibī considered this practice of <u>du'ā</u> a <u>bid'a</u>, while the other <u>fuqahā</u>' accused him of introducing a <u>bid'a</u> by opposing the practice. One result of this discussion was that a rather clear definition of "<u>bid'a</u>" emerged in Shātibī's discussion of this issue.

Allowance for the Disagreeing Opinion (Murāʿāt al-Khilāf)

It has been stated earlier that the aspect of disagreement in Māliki figh was the problem that struck Shāṭibī's mind early in his career and which continued to perplex him even in later times.⁹⁶ He wrote to many scholars and disputed with them on the many facets of this matter. His contemplations of this issue and his discussion about it led him to the conclusion that formed the basis of his doctrine of <u>maqāşid al-sharī'a</u> (the objectives of islamic law). In view of its significance, a detailed discussion of this problem is in order.

Because of various historical reasons which do not concern us here, Mālikī <u>fiqh</u> abounded with disagreement on a number of cases. This was a very perplexing phenomenon for a tradition which upheld the consensus of scholars and the unity of the practice. Consequently scholars were occupied in a perennial discussion on this issue. Very broadly speaking, in the course of time, during the development of the Mālikī tradition in Spain, four positions were taken on this issue. First, some scholars, foremost among them Ibn ^cAbd al-Barr (d.463/1079), denied the existence of "disagreement" in Mālikī <u>fiqh</u>. This position was taken generally by some other ancient scholars also.⁹⁵ It was essentially this position that Shātibī came to adopt after lengthy discussion. Shātibī's views are discussed towards the end of this section.

V.

Secondly, the position was taken under the influence of the <u>taşawwuf</u>. Since Şūfīs feared that an indulgence in cases where disagreement in opinion existed might lead some astray in seeking for lenient opinions, they regarded it as an obligation to avoid the cases of disagreement. They considered these lenient opinions as instances of <u>rukhşa</u> (concession) in contrast to <u>cazīma</u> (regular) cases which were the only path to be followed by a resolute person.⁹⁶

Shāţibī traced this trend to the teachings of Qushayrī on the basis of which in a later period upholders of the position had adopted the following formulation: "al-wara' bi'l khurūj 'an al-khilāf" (piety consists in avoiding (the cases of) disagreement).

Under the impact of <u>tasawwuf</u>, this position had been accepted by a number of <u>fuqahā</u>' as well. Shāṭibī did not question the attitude of <u>sūfīs</u> towards <u>rukhṣa</u> as an attitude appropriate to the <u>khawāṣṣ</u> (special, élite) and <u>Arbāb al-Ahwāl</u> (the people of mystical states), but he did oppose this trend insofar as it meant the

÷.

185

imposition of an impossible obligation for general people. He took this stand because the jurists had gone as far as to consider "<u>wara</u>" as obligatory for every one.⁹⁷

Shātibī wrote to scholars in Spain and North Africa. Among the <u>fatwā</u>s in answer to his query, Ibn ^CArafa's <u>fatwā</u> is available to us as preserved by Wansharīsī.⁹⁸

In his question, Shāţibī states that scholars such as Ghazālī, Ibn Rushd and Qarāfī maintained that piety consisted in avoiding "disagreement". The basis of their argument was that the cases disagreed upon, in the details (<u>furū</u>⁴) of <u>shar</u>⁴ were like <u>mutashābihāt</u> (equivocal statements) which the sayings of the Prophet urged to be avoided. Shāţibī found it logically impossible to maintain such a position, as the seven points which he used to refute it, show.

Ibn ^cArafa's answer, however, can be summarized as follows. He explained that the cases of disagreement were very few and that to avoid them was not only possible but obligatory. The reason was that these cases, being equivocal, had equally forceful arguments in favour and against the issue; such a situation would then be conducive only to an arbitrary decision. Ibn ^cArafa insisted that to opt for the less convenient was the result of the fear of severe punishments from Allāh. This fear was the reason why Ibn Hazm condemned those who sought for convenience in the <u>sharī'a</u>. Ibn ^cAbd al-Salām also condemned the trend to choose the more convenient of any two <u>fatwā</u>s.⁹⁹

The third position regarding differences of opinion was that held by scholars who

considered the existence of "disagreement" as proof of permissibility. Shāṭibī distinguished "disagreement" from <u>murā'āt al-khilāf</u>¹⁰⁰ which is discussed below. He stated this position in the following words:

"Often a fatwa on a certain question recommended abstention (man^c) (from the matter in question). It was said, "Why do you recommend abstention whereas the problem is disagreed upon?" Thus the disagreement becomes the proof of permissibility simply because it is disagreed upon; neither because of certain evidence in favour of the soundness of the argument for its possibility, nor on the basis of some authority more worthy to be followed than the one who demanded abstention." 101

The fourth position was that of <u>murā <āt al-khilā f</u>. This principle not only admitted the existence of disagreement but also stressed the need to give its full consideration, so as to regard both conflicting opinions as valid. Although it was a commonly accepted position, Shāţibī differed and disputed it with a number of scholars. Among them the names of Ibn Qabbāb, Fishtālī, Ibn ^cArafa and Sharīf^{*kl*}_hTilimsānī are known to us. ¹⁰² To help in appreciation of the problem, it is advisable to summarize this discussion.

The main points of the question that Shāṭibī posed to the scholars are the following: In Granada there arose a problem in which different opinions were attributed to Mālik. According to the <u>uṣūl al-fiqh</u> rules about contradiction as explained below, every one of these different opinions had to be rejected. It was further realized that such disagreement existed in the major part of the Mālikī tradition. If the rules of contradiction were applied, most of the Mālikī tradition would have to be rejected. As a measure of necessity (<u>darūra</u>), Mālikī <u>fuqahā</u>' adopted

7

the principle of murāʿāt al-khilāf, but the application of this principle posed a number of problems.¹⁰³

Shāțibi illustrated the use of this principle in a number of cases. In a particular case of marriage when scholars disagreed on the validity of the marriage, it was to be considered void. Yet in reference to its effects an allowance was to be made for the opinion that favoured its validity. Hence matters such as inheritance...etc., were to be applied as if the marriage were valid. The problems that this position raised for Shāțibi were the following:

i) It disregarded the established principle of usul al-figh, that the consideration of time could declare one of those opinions as 'later' and hence more reliable.

Another principle relevant to cases of contradictory opinions was also disregarded. It stated that if two contradictory opinions are attributed to a <u>mujtahid</u> both of them should be suspended until one of the two can be established with certainty.

- ii) The Mālikī scholars were not consistent in applying this "allowance" (<u>murāćāt al-khilāf</u>). In some cases they denied the "allowance", while in other cases they insisted on it. This inconsistency makes the soundness of this principle doubtful. On the other hand, it renders its application arbitrary.
- iii) Thirdly, assuming the soundness of this principle, its basis in the <u>Shari'a</u> as a principle of <u>figh</u> is not known. Apparently this problem refers to the evidence (<u>dalil</u>). The difference between two statements (<u>qawl</u>) must inevitably be so because they are based on two different evidences which are contradictory to each other in the sense that the opposite of one is the requirement of the other.¹⁰⁴

Thus murā^cāt al-khilāf would mean granting each one of such statements what is required by the other, entirely or partially.

Sharif, Tilimsāni answered the question 105 by refuting Shātibi's argument that among two statements of an imam or a mujtahid, the later in time eliminates the earlier. Tilimsānī questioned the consideration of the time factor in such His argument was that this consideration implied the principle of abrogation cases. (naskh) which is applicable only to statements originating from the lawgiver (shāri^c). He distinguished among shāri^c, "mujtahid mutlaq" and "mujtahid Since it was the shari' alone who could institute laws and who fi'l madhhab". could withdraw them, it was, therefore, in his statements alone, that in case of contradiction the later would abrogate the earlier. The Mujtahid, whether "mutlaq" or "fi madhhab," did not make laws but rather sought and decided in favour of one of the evidences. The mujtahid mutlaq sought evidence in the commands of the sharifa; the mujtahid fi madhhab sought evidence in the statements of a mujtahid mutlaq whom he considered the imam for his madhhab. The differences of opinion in the case of mujtahids was, therefore, based on the The evidences, which were derived from the difference in choice of evidence. shari'a, in the instance of each of the opinions could not be invalid. Hence the question of later and earlier, with the effect of one eliminating the other, could not arise in the case of multahid.

Although Tilimsānī did not spell out his view yet it can be concluded from his answer that he did not oppose the principle of <u>murāfāt al-khilāf</u>. If one follows his

189

argument more closely, one may see that he regarded this principle as necessary. Since all the different opinions of <u>mujtahids</u> are supposed to be based on certain evidences from the <u>shar</u> if by neglecting any of them one would be committing the wrong of rejecting <u>shar</u> evidence.

Tilimsäni's elaboration, however, did not answer Shāṭibi's question. It admitted that the basis of difference lay in the choice of legal evidence, but it did not explain how one could claim the existence of two or more contradictory pieces of evidence in the shari's bearing on the same case.

Ibn 'Arafa's answer¹⁰⁶ was longer than others. His answer consisted partly of the arguments already seen in Tilimsānī's answer and partly of whittling down Shāṭibī's use of terms to contradictions.

He explained the principle of $\underline{\text{mura}} \cdot \overline{at} \ al-khilaf$ from a different perspective. He defined $\underline{\text{mura}} \cdot \overline{at}$ as abiding by the implications of $\underline{\text{shari}} \cdot \underline{a}$ evidence in a given case ($\underline{\text{mad}} \cdot \overline{ul}$) in such a way as also to abide by the implications of other evidence in another case. In other words, as a matter of fact, the principle of $\underline{\text{mura}} \cdot \overline{at}$ implied abiding by the implications of both evidences in those aspects in which a $\underline{\text{mujtahid}}$ prefers one piece of evidence to another. In this way, he was neglecting neither of them but was rather abiding by the both at the same time.

Abū 'Abd Allāh, Fishtālī adopted Ibn 'Abd al-Salām's view in his answer. In reference to an action which is considered wrong by one <u>mujtahid</u> and is regarded as correct by the opponent, Fishtālī distinguished between two situations, one before the occurrence of the action and the other after its occurrence. According to him, the prohibition was absolute in the former situation, but once the action had taken place, an allowance must be given to the opponent's opinion for the sake of public convenience.

Ibn al-Qabbāb's answer was very succinct and brief. He regarded <u>murā 'āt</u> <u>al-khilāf</u> as one of the best principles of Mālikī <u>fiqh</u>. He defined it as granting to each one of the two pieces of evidence its value (<u>hukm</u>). He, however, distinguished between two situations; one was a case of disagreement where it was inevitable to prefer one opinion over the other. Of this type are the cases of <u>ta'ārud</u> (conflict) and <u>tarīīh</u> (preponderance). Second was the situation where both evidences led to the same conclusion or in some sense complemented each other. Such an instance was a case of <u>murā'āt al-khilāf</u>. ¹⁰⁸

Shātibi's Views

÷\$

Shāṭibī was not satisfied with these answers. They were either irrelevant or they tried to explain away the evident meaning. For the most part these answers treated the problem of <u>murā¢āt</u> like that of <u>tarjīḥ</u>. The only answer that pleased Shāṭibī was that of Ibn al-Qabbāb who agreed with him that the problem was really very abstruse.

Shāṭibī contemplated this problem for some time and reached his own conclusions. He came to believe that there was no place for "disagreement" in <u>sharīfa</u> such as that which constituted the basis of <u>murāʿāt al-khilāf</u>. Hence the principle of <u>murāʿāt</u> was a false problem. The main conclusion that Shātibī reached was the unity of the origins of <u>sharī</u>. He maintained that, "all rules of <u>sharī</u> originate from one statement, even though there may be a diversity of rules."¹¹⁰

The basis of this conclusion was the following five points:

**

- (i) A large number of Quranic verses stress the original unity of <u>shari</u> and, further, they condemn "disagreement".
- (ii) If disagreement were permissible, there would be no place for the question of abrogation. The need for abrogation means only that two evidences are so contradictory to one another that one has to be replaced by the other.
- (iii) If the existence of disagreement were permitted, it would imply the imposition of an impossible obligation. In other words, to command someone to obey two contradictory orders at the same time is to put him under an impossible obligation.
- (iv) The legal theorists (<u>uşūliyin</u>) recommend a decision in favour of preponderance of one of the contradictory evidences over the other.
 This fact implies the non-permissibility of "disagreement".
- (v) It would be absurd to maintain that both of the contradictory commands are intended by the lawgiver because one would negate the other.

Apart from the linguistic, geographical and historical causes of "disagreement", there were certain factors in <u>shari'a</u> itself that seem to favour disagreement. Among these three factors are worth noting:¹¹²

First, the existence of <u>mutashābihāt</u> (equivocations) in the Qur³ān. These equivocations make allowance for disagreement of opinions, expressed either in interpretation or in suspension of the judgment. Further, it cannot be denied that

<u>mutashābihāt</u> were intended to be equivocal by the Lawgiver. Shāṭibī discussed the problem of <u>mutashābihāt</u> in detail in <u>al-Muwāfaqāt</u>. He maintained that there was no <u>tashābuh</u> in the fundamentals of <u>sharīʿa</u>.¹¹³

Shāṭibī also disputed the assertion that <u>tashābuh</u> was the intention of the Lawgiver. Dealing with the matter in detail, he distinguished between two intentions (<u>irādāt</u>) of the Lawgiver. One was $\frac{al}{k}$ halqiyya ada tiyya (creational predestined intention) in which human will had no place.¹¹⁴ The second was $\frac{al}{k}$ <u>amriyya</u> al -<u>tashrī (iyya</u> (imperative legal intention) in which Divine Will did not impose itself on human will. Shāṭibī argued that <u>mutashābihāt</u> belong to the second category of Divine intention. There, disagreement is not intended by the Lawgiver, because the Qur'ān states that only one of the interpretations is correct. If disagreement were allowed, then every interpretation would have to be regarded as correct.

Second, an analogy is drawn from the <u>shar</u>^ci permission for the exercise of <u>ijtihād</u> (legal reasoning) which, it is maintained, would naturally lead to disagreement. Shāṭibī refuted this argument by referring it back to the problem of <u>tashābuh</u>. He maintained that not every conclusion reached by a <u>mujtahid</u> was correct. Its correctness or error depended on its correspondence with the intention of the Lawgiver 115 which does not favour disagreement.

Third, an analogy was drawn from the existence of the principle of <u>rukhsa</u> in <u>sharica</u>. This principle, which means to opt for a concession from regular rules in special

1.000

cases, allows for the existence of disagreement.

Shātibī refuted this argument by stressing that <u>rukhşa</u> does not mean to opt for one of two equally applicable rules in a case. If it were arbitrary, it would not be allowed in <u>sharī</u>ca. The principle of <u>rukhşa</u> is applicable only in those cases where it becomes hard or impossible to abide by the regular rules. Thus, in fact, <u>rukhşa</u> has to do with two different rules in two different cases, not two different rules in a single case, which is the meaning of disagreement.¹¹⁶

To conclude Shāṭibī's arguments, it may be said that he understood <u>khilāf</u> (disagreement) essentially as taʿāruḍ al-bdilla (contradiction of evidences) while for others it meant essentially tasāwī al-adilla (equal validity of evidences). Hence, for Shāṭibī <u>khilāf</u> involved the problem of tarjīh al-adilla (preponderance) while for others it involved only the problem of jam^c (combining) or <u>murāʿāt</u> (making allowance).

Shātibi's methodological objection concerned the distinction made by Mālikī scholars between <u>muttafaq 'alayhi</u>(agreed upon) and <u>mukhtalaf fih</u>i (disagreed upon). They stressed that in case of the former, consideration could be given only to that evidence on which it was decided to be "agreed". In case of <u>mukhtalaf fihi</u>, however, the evidence on which the opposing decision was based must also be considered. Shātibi viewed the above standpoint as inconsistent. If it were <u>shari</u> evidence which provided the basis of a decision, then why was it to be disregarded if it opposed a <u>muttafaq 'alayhi</u>? Why should a decision be considered as <u>mukhtalaf</u> To agree with the upholders of <u>khilāf</u> would mean, for Shāṭibī, to believe in the existence of contradictions or diversity in the principles of <u>sharī¢a</u>. This belief would be a negation of the unity of the origins of <u>sharī¢a</u>.

-

It was, however, difficult to explain this unity in the presence of an obvious diversity of evidences in the sharica. Shātibi, in his investigation of this problem, came to conclude that the unity of sharica could be explained by the unity of the intentions of the Lawgiver. The result of these investigations was his doctrine of

<u>maqāşid al-sharīća</u> (the objectives of <u>sharīća</u>). This doctrine constitutes the basis of Shāțibī's legal thought. An elaboration of this doctrine and its theoretical and methodological implications are discussed later.

HIS WORKS

The following is a list of Shājibi's works known to us. They belong mainly to two fields; Arabic language and grammar, and jurisprudence.

 Sharh ^calā al-khulāşa fi al-nahw. A commentary on <u>Alfiya</u> by Ibn Mālik, in four parts:

Ż

Mentioned in:

.

 (i) Al-Maqqari, <u>Mafh al-Ţib</u>, Vol.VII, 275; (ii) Kaḥḥāla, <u>Mu^cjam</u> <u>al-Mu^sallifin</u>, 1118, (iii) Sarkis, <u>Mu^cjam Matbū^cāt al-Arabiya</u>, 1090; (iv) <u>Fihris al-Azhariya</u>, IV, 255; (v) <u>Nayl</u>, 48; (vi) Al-Makhlūf, <u>Shajarat al-Nūr al-Zakiya</u>, 231; (vii) Zirkali, <u>al-A^clām</u>, 1,71.

Ms. al-Azhariya / 1487/ 10806. Beginning:

اللسجسم الساخسين ويشكون على ما العمت ، ويشكون على ما العمت Four volumes containing Parts I, II, III and V, written in old naskh. Copyist's name: 'Umar b. 'Abd Allāh al-Manzarāwī. The completion of the third part by the copyist is dated 868 and the fifth 872 A. H. Each page contains 27 lines: 27 cm.¹¹⁷

2. ⁽Unwān al-ittifāq fī ⁽ilm al-ishtiqāq.

Mentioned in:

(i) <u>Nayl</u>, 48; (ii) <u>Al-A^clām</u>, 1,71; (iii) <u>Shajara</u>, 231; (iv) Kaḥḥāla,
 <u>Mu^cjam</u>, 1,118; (v) <u>Idāḥ al-maknūn</u> <u>Al-Baghdādī</u>, <u>Idāḥ al-Maknūn</u>,
 (Cairo, 1945), 127.

3. Kitāb usūlal-nahw.

Mentioned in:

e .

(i) <u>Nayl</u>, 49; (ii) <u>Al-A^clām</u>, 1, 71; (iii) <u>Shajara</u>, 1, 231. Shāṭibī mentions both of the above books (i.e. nos. 2 and 3) in his <u>Sharḥ al-Alfīya</u> but Aḥmad Bābā recalls reading elsewhere that Shāṭibī destroyed both of those works in his life-time.¹¹⁸

. . .

4. Al-Ifādāt wa'l inshādāt / inshā'āt.

Mentioned in:

(i) <u>Nafh</u>, VII, 187-192, 276-301; X, 139-140; (ii) <u>Nayl</u>, 48; (iii) Sarkis, <u>Mu'jam</u>, 1090; (iv) <u>Al-A'lām</u>, 1, 71; (v) Kaḥḥāla, <u>Mu'jam</u>, 1, 119;
(vi) <u>Shajara</u>, 231; (vii) Nwiya, <u>Ibn 'Abbād</u>, 252. As mentioned earlier, the extracts of this work in <u>Nafh</u> and <u>Nayl</u> show that this was Shātibi's collection of class notes and discussions.¹¹⁹ Maqqari and Aḥmad Bābā, both have used it as a source of information about the scholars whom Shātibi mentioned in this work.¹²⁰

- <u>Kitāb al-Majālis</u>. A commentary on the chapter of sale (<u>buyū</u>^c) in the <u>Şahīh</u> of al-Bukhārī. Mentioned in: (i) <u>Nayl</u>, 48; (ii) <u>Shajara</u>, 231; (iii)
 Sarkīs, Mu^cjam, 1090; (iv) Al-A^clām, 1, 71.
- <u>Al-Muwāfaqāt</u>. The original title being <u>Unwān al-ta</u>rīf bi asrār al-taklif.
 An epitome of this work was done by Qādi Abū Bakr b. <u>Ā</u>sim (d.829 A.H.)¹²¹
 Published: (a) First published in 1302/1884 in Tunis by the Tunis government

press, edited by Şāliḥ al-Qā'ijī, 'Alī al-Shanūfī and Aḥmad al-Wartantānī. (b) Reprint of the first part of the above in Kāzān in 1327/1909 with an introduction in Turkish by Mūsā Jār Allāh.

(c) Third (in fact, the second complete) print in 1341/1923 in Matbac Salafiya, Cairo, edited by Muhammad al-Khidr Husayn, the rector of Al-Azhar, and partly by Muhammad Hasanayn al- ^cAdawi, the administrator of the Religious Department, Government of Egypt.

(d) Fourth print in Maţba[<] Muştafā Muḥammad (n.d.), edited with extensive notes by Shaykh [<]Abd Allāh Darāz.

(e) Fifth print in Matba^c Muḥammad ^cAlī, Cairo, in 1969, edited by Muḥammad Muḥiy al-Dīn ^cAbd al-Ḥamīd.

A summary view of its contents is presented in Appendix A.

7. Kitāb al-l^etisām.

; . .-

1

>

(a) Partly published in <u>Al-Manār</u>, XVII, (1333/1913).¹²²

(b) Published in Matba^c Mustafā Muhammad, probably in 1915. This edition was edited by Muhammad Rashid Ridā, the editor of <u>Al-Manār</u>. This edition is based on an incomplete Ms. from the library of Shanqiti.
(c) The book was briefly reviewed by D. S. Margoliouth in J.R.A.S.,

1916, (p.398).

A summary view of its contents is presented in Appendix B.

 A Medical treatise. Ms. University of Leiden: 139r-140r; CCO 1367; Warn/Or. 331-Gb). The University of Leiden holds this Ms.¹²³ The treatise is not mentioned by any major authorities on Shāțibi. The catalogue, however, attributes this treatise to Shāțibi and, significantly enough, it describes it as having been written down by his (Shāțibi's) pupil (?) Ibn al-Khațib.¹²⁴

The probability that Shāţibī was the author of this treatise is heightened by the following points. Among Shāţibī's teachers, there is mention of one al-Shaqūrī.¹²⁷ We have no further information about him. From other sources we know that a family from Shaqūra was known as a family of physicians. Among them Abū Tamām Ghālib al-Shaqūri and Abū ʿAbdullāh al-Shaqūrī are known as the authors of medical treatises.¹²⁶ We also know that Ibn al-Khaţīb was associated with both of these men. He is also the author of certain medical treatises.¹²⁷

From these facts, it might conceivably be argued that Shāṭibī, having been taught by one of these Shaqūrīs, had an education in medicine and hence could be the author of a medical treatise.

NOTES: CHAPTER IV

- This is Ahmad Bābā (d. 1036/1626), the author of Nayl al-Ibtihāj. For details on his life and works see M. Cheneb, "Ahmed Bābā" in El., (1st ed.), Vol. 1, 191-2; Levi Provençal, "Ahmad Bābā", El., (2nd ed.) Vol. 1, 279-280; J. O. Hunwick, "Ahmad Bābā and the Moroccan Invasion of the Sudan (1591), Journal of Historical Society of Nigeria, II (3, 1962), 311-28; same author, "A new source for the biography of Ahmad Bābā al-Tinbukti (1556-1627)", Bulletin of the School of Oriental and African Studies, XXVII (1964), 568-593; Muhammad Makhlūf, Shajarat al-Nūr al-Zakiyya (Cairo, 1349 A.H.), Vol. 1, 298.
- Available to us in two editions; in Maghribi script, (Fas: Matba^c Jadida, 1317 A.H.); second edition, printed on the margin of Ibn Farhun, <u>AI-Dibaj</u> <u>al-Mudhahhab</u> (Cairo, 1351). (Henceforth the reference <u>Nay</u>] will refer to the latter edition).

The question of Ahmad Bābā's sources for <u>Nayl</u> has been dealt with by scholars with varied competence. To my knowledge the best review is still that by Cherbonneau which is mainly a re-enumeration of the sources which Ahmad Bābā himself mentions towards the end of <u>Nayl</u> (p.361). Cf. the following:

1. E. Fagnan, "Les Țabaqāt Malikites" in D. F. Saavedra, <u>Homenaje à</u> D. F. Codera, (Zaragoza, 1904), 110.

 Cherbonneau, "Lettre a M. Defremery sur Ahmed Bābā le Tombouctien, Auteur du <u>Tekmilet ed-Dibādi</u>", <u>Journal Asiatique</u>, 5^e serie, 1 (1853), 93-100.

- 3. Ibn al-Fakhkhär al-Biri, Abū 'Abd Allāh al-Maqqari, Abū 'Abd Allāh al-Tilimsāni and Abū'l Qāsim al-Sabti are some of such common teachers. Cf. Maqqari, Nafh al-Ţib, (Cairo: Matba'Sa'āda, 1949), VII, 187 gives an extract from al-Shātibi's Ifādāt where al-Shātibi mentions Ibn al-Khaţib among others who attended with him al-Maqqari's lectures in 757 A.H.
- 4. See p. 199.

- 5. Ibn Zumruk whom Ibn al-Khajib patronized and who later replaced lbn al-Khajib when the latter defected to Tlemcen, was a close friend of al-Shajibi. See Nafh al-Tib, X, 139 and F. de la Granja, "Ibn Zamrak", in E.I.(2nd ed.) Vol. III, 972-73.
- 6. Ibn Khaldūn's <u>Shifā' al-Sā'il li Tahdhīb al-Masā'il</u>, ed. by Muhammad b. Ţāvīt al-Ţanjī (Instāmbul, 1957) was written in response to a query sent to scholars in the west of whom the names of Ibn Qabbāb and Ibn Abbād are confirmed by Wansharīsī. The attribution of this treatise

to Ibn Khaldūn has been doubted by scholars (see: Talbī, "Ibn Khaldūn", E.1. (2nd edition), Vol. III, p.828. Tanjī, the editor of this work, however, argues in detail in favour of such attribution. He is of the opinion that it was Shātibī whose taqyīd (query) is referred to in this treatise. See his Introduction, p.t.

- Shātibi Al-Istişām, ed. Rashid Rida, (Cairo, 1915), Vol. II, 84, quotes an extract saying "As narrated by one of our contemporary writers" which exactly corresponds with Al-Ihāta, (Cairo, 1319 A.H.), Vol. 1, p.75.
- Various dates have been suggested for the year when <u>Al-lhāta</u> was completed. For instance Mahdi, <u>Ibn Khaldūn's Philosophy of History</u> (Chicago, 1964), p.35, n.5, suggests 763/1361-62. This date is not possible because (1) Ibn Al-Khatib enumerates among the works of Ibn Khaldūn, already completed, a treatise on logic which he wrote for Sultān Muḥammad V of Granada. We know that Ibn Khaldūn's stay in Granada was in 764-65. Hence the date of <u>Al-lhāta</u> must be after 765/1363.
 2) Secondly, <u>Al-lhāta</u> recounts the events in the year 771/1369 (op.cit. Vol. 11, p.58), which places the date of its completion after 771/1369. It is because of the second evidence that we believe that <u>Al-lhāta</u> must have been finally completed in 771/1369.
- 9. Nafh al-Ţib, IV, 195-201.
- For details on the literature of Māliki Tabaqāt see: E. Fagnan, 'Les Tabaqāt Malikites' op.cit. pp.105–113.
- 11. Nayl, p.30
- 12. For instance, al-Qabbāb was a well-known jurist in this period, but Ibn Farhūn derives his information about him from <u>Al-Ihāta</u> (of which probably the fragments were available to him). Al-Qabbāb's commentary on <u>Qawā^cid al-Islām</u> is noticed in the following manner: "someone among my pupils mentions that he (i.e.al-Qabbāb) wrote a commentary on <u>Qawā^cid al-Islam</u>". <u>Al-Dibāt</u> al-Mudhahhab, Op.cit. p.41.
- 13. The author's note at the end of the book sets the date of its completion in 761 A.H. (Ibid., p.362). There are, however, many entries which mention dates beyond 761 (e.g. pp.83, 330); one of them even mentions the date 803. Fagnan (op.cit., p.110) considers such entries as later interpolations and to him the date of its completion is certainly 761.
- 14. Al-Dibāj, p.2.
- 15. See Fagnan, op.cit. p.111.

- Still in Manuscript form. (Bibliothèque Nationale de Paris No. 4627, aussi No. 4614; Zaitūna, No. 3245) vide Fagnan op.cit. p.111.
- 17. See for instance Nayl pp.51, 52, 88.
- 18. Among these notices we may mention the following:

Ignaz, Goldziher, <u>Streitschrift des Gazālī gegen die Batinija - Sekte</u> (Leiden, 1916), pp.32-34, where he discusses Al-Muwāfaqāt. D. S. Margoliouth, "Recent Arabic Literature" in J.R.A.S. (London, 1916), pp.397-98, where he reviews <u>al-Istisām</u>. Among the biographical notices: Brockelmann, <u>Supp. 11</u>, 374-75; Muḥammad Makhlūf, <u>Shajarat al-Nūr al-Zakiyya</u>, (Cairo, 1349), p.231; Ismā^cīl Pāshā Baghdādī, <u>Idāh al-Maknūn</u>, supp. to <u>Kashf al-Zunūn</u> (Cairo: Bahiyya, 1945), Vol. 11, p.127; Maḥmūd Ḥasan al-Toniki, <u>Musjam</u> <u>al-Musannifīn</u>, (Beirut, 1344), Vol. IV, p.448-454; Abd al-Mut'āl al-Sašīdī, <u>Al-Mujaddidūn fī l-Islām</u> (Cairo, Namūdhajiyya, n.d.), pp.307-12; Fādil b. 'Āshūr, <u>Aslām al-Fikr al-Islāmī fī Tārīkh al-Maghrib Al-Arabī</u>, (Tunis: Najāḥ, n.d.), pp.70-77; Yūsuf Ilian Sarkīs, <u>Musjam al-Maţbū'āt</u> a<u>l-Śarābiyyah wa'l Musaraba</u>, Vol. 1, (Cairo, 1928) p.1090; Khayr al-Dīn Zirkilī, <u>Al-Aslām</u>, Vol. 1 (2nd ed., 1954), p.71; Kaḥhāla, <u>Musjam</u> <u>al-Mu'allitīn</u> (Dimashq, 1957), Vol. 1, pp.118-19.

- 19. The extract of the relevant entry of this work is available to us in <u>Al-Muwāfaqāt</u>, (Tunis edition, 1302), Vol. IV, as an appendix, pp.1-4.
- 20. Nayl was completed in 1005 A.H. (Fagnan, op. cit.). For details on this invasion and Ahmad Bābā's life, see the sources mentioned above in n.l.
- 21. Nayl, p.12.
- 22. Compare the sources mentioned towards the end of al-Dibāj by Ibn Farhun and those mentioned by Ahmad Bābā for his Nayl. Ahmad Bābā's sources mostly relate to the Muslim West, while Al-Ihāja is the only source related to the Muslim West, which is mentioned by Ibn Farhun.
- 23. Nayl, p.16.
- 24. Nayl, p.46.

í

25. A well-known and influential scholar in Tunis in Shāţibi's time. He was imām of Zaytūna mosque for 50 years. He was the foremost among Ibn Khaldūn's rivals when the latter was staying in Tunis. He had correspondence and discussions with Shāţibi on the question of murā'āt al-Khilāf. See Ibn Maryam, Al-Bustān fī Dhikr al-Awliyā' wakulamā' bi Tilimsān, Ed. Muḥammad b. Cheneb (Algiers, 1326 A.H.), pp.194-195.

- 26. Nayl, p.277.
- 27. Ibid.
- 28. Ibid., p.217.
- 29. Abū'l ^SAbbās Aḥmad al-Wansharīsī (d. 914/1506), <u>Al- Mi^cyār al-Mughrib</u> wa'l Jāmi^c al-Mu^carrab ^can fatāwā ^culamā' lfrīqiya wa'l Andalus wa'l Maghrib, (Fās, 1314 A.H.)
- 30. See Nay!, pp.69, 283, 346.
- Maqqari supplies lengthy extracts from Al-Ifadat in Nafh Vol. VII, pp.187-192 (regarding Abū Abd Allah al-Maqqari); pp.276-301 (about Ibn al-Fakhkhar al-Biri), and Vol. X, pp.139-40 (about Ibn Zumruk).

V,

- 32. Al-letişam, op.cit. pp.9-12.
- 33. Al-Muwafaqat, Vol. IV, pp. 150f.
- 34. I. Goldziher, Streitschrift des Gazālī gegen die Batinijja-sekte (Leiden, 1916), p.32, said that Shātibī "dem aus Xativa stammenden, später in Granada lebenden". The same mistake was carried out by Brockelmann, G.A.L.S. II, p.374; "aus Xativa, gest in Granada". Asin Palacios was also misled by the nisba, as he stated that Shātibī lived in Shātiba, see Asin Palacios transl. by M.L. de Céligny. "Un Précurseur Hispano-Musulman de Saint Jean de la Croix". Etude Carmélitaines, 1932, p.121-22, vide P.Nwiya Ibn Abbād, op.cit. p.173, n.2.
- 35. Levi-Provençal, "Shātiba", E.I. (1st ed.) Vol. IV, p.337.
- 36. See p. 86ff.
- 37. See above note no. 3 and Nafh al-Tib, op.cit., Vol. VII, p.275; Shajara op.cit., Vol. 1, p.228.
- 38. Nafh al-Tib, Vol. VII, pp.276-278; 297-301.
- 39. Kaḥḥāla, Muʿjam, op.cit. Vol. VIII, p.252.
- 40. Shajara op.cit. Vol. 1, p.233.
- 41. Nayl, p.219.
- 42. Shajara, p.230.

,

- 43. See pp.182, 222ff.
- 44. Nayl, p.47.
- 45. Nafh al-Ţib, Vol. VII, p.134.
- 46. Al-Ihāta, Vol. 11, p.139.
- 47. Nayl, p.250
- 48. Shajara, p.232.
- 49. Nafh al-Tib, Vol. VII, p.206.
- 50. See p. 117f.
- 51. Nafh al-Tib, Vol. VII, p.232-249.
- 52. Nayl, op.cit. p.245, 346; <u>Shajara</u>, 1, p.234. Zawāwī was alive until 770 A.H.
- 53. Nayl, p.346.
- 54. This extract from Shātibi's al-Ifādāt is quoted by Ahmad Bābā in Nayl, p.346 and by P. Nwiya, in Ibn 'Abbād de Ronda, p.XXXIX, No. 2.
- 55. Muhsin Mahdi, op.cit. p.35, n.2; Nayl, p.256.
- 56. Nayl, p.256.
- 57. Nayl, p.258.
- 58. The lack of interest in uşūl al-fiqh is observed by Ibn Sa^cid as quoted by Maqqari in Nafh al-Iib, Vol. 1, p.206.
- 59. Al-Istişām, op. cit. Vol. 1, p.9.
- 60. See p. 184ff.
- 61. Nayl, p.221.
- 62. Ibid.
- 63. Al-Muwafaqat, Vol. I, op.cit., p.22.

- 64. Shātibi, in a letter to his friend, implies that dismissal from the office of imām or Khatib of a mosque was common after one had opposed bid^ca practices. See Wansharisi, Mi^cyār al-Mughrib, Vol. XI, p.109.
- 65. Leon Bercher, (ed. Transl. and Comments on) Ibn 'Âçim al-Mâlikî al-Gharnâţî, <u>Al-'Âçimiyya ou Tuh'fat al-hukkîm fi nukat al-'uqoûd wa'l</u> ah'kam, (Alger, 1958), Introduction, p.III.
- 66. Nayl, op.cit. p.49.
- 67. Ibid.

÷.,

. ...

- 68. Al-Ictisam, op. cit. p.9f.
- 69. Ibid., p.11.
- 70. lbid., p.11 ff.
- 71. See p. 182ff.
- 72. See p. 183ff.
- 73. Al-Muwāfaqāt, Vol. 1, p. 102.
- 74. lbid., 1.103.
- 75. Ibn Khaldūn, Shifā'al-Sā'il, op.cit., describes the controversy between those who attached significance to books and those who favoured the necessity of Shaykh, in the form of an interesting dialogue. See pp.79-85.
- 76. See p. 118f.
- 77. Al-l'tişām, op.cit., p.208.
- 78. Ibn Khaldun, Shifa'al-Sa'il, op.cit., p.11.
- P. Nwiya (Ed.), Af-Rasāil as-Sugrā, (Beiruth: Imprimerie Catholique, 1958), pp.106–115, and Appendice C pp.125–138.
- 80. P. Nwiya, Ibn Abbād de Ronda, (Beiruth: Imprimerie Catholique, 1956), pp.209–13.
- 81. Javit Janji. op.cit. Our references to Shifa' are based on this edition.

82. P. Khalifé. (ed.) Ibn Khaldūn, Šifā as-Sā'il litahzib al-masā'il (Beyrouth, 1958).

Ŋ

- 83. Ibn Abbād, Ar-Rasāil as-Şugra (ed. P.Nwiya) op.cit., p.106.
- 84. lbid., p.107.

a -

1.

í

- 85. Ibid., p.109.
- 86. A. Bel, "Abd al-Wāhid al-Rashid", E.I. (1st ed.) Vol. 1, p.66.
- 87. Al-I'tişām, op.cit., Vol. 11, p.237.
- 88. Ibid., pp.237-8.
- 89. Al-Wansharisi, Al-Mi^cyār, Vol. XI, p.109.
- 90. Lévi-Provençal, introduction to Al-Nubāhī, <u>Al-Marqabat al-^sulyā</u>, p.t.
- 91. Shajara, op.<u>cit</u>. p.231.
- 92. Ibid., p.247
- 93. Nayl, op.cit. p.266.
- 94. Wansharisi, Al-Mi'yar, Vol. VI, pp.258ff.
- 95. Al-Muwāfaqāt, op.cit. Vol. IV, p.151.
- 96. Ibid., pp.144-45. Sūfīs views on rukhşa are discussed in Al-I^ctişām, Vol. 1, p.169.
- 97. Al-Muwafaqat, op.cit. Vol. 1, p.104.
- 98. Al-Mi'yār, op.cit. Vol. VI, p.254-270.
- 99. Ibid., p.267.
- 100. Al-Muwafaqat, Vol. IV, p.141.
- 101. Ibid.
- 102. Al-Mi^cyār, op.cit. pp.254-280.
- 103. Ibid., p.254.

104. Ibid.

na La

- 105. Ibid. and Nayl op cit., p.262.
- 106. <u>Al-Mi'yā</u>r, op.cit., p.261ff.
- 107. lbid., p.274.
- 108. Ibid.
- 109. Ibid., p.271.
- 110. Al-Muwafaqat, Vol. IV, p.118.
- 111. Ibid., pp.118-132.
- 112. Ibid., pp.211-214.
- 113. Ibid., Vol. III, p.96.
- 114. Ibid., Vol. III, p.119ff.
- 115. Ibid., Vol. IV, p.127.
- 116. lbid., p.144ff.
- 117. Fihris Maktaba Al-Azhariya, V (Cairo, 1946), p.255; also see^SAbd al-Hafiz Manşūr, Fihris Makhtūtāt al-Maktaba al-Ahmadiyya bi Tūnis, (Beyrut: Dār al-Fath, 1969), pp.316-319.
- 118. Nayl, op.cit., p.49.
- 119. See above. Recently P. Nwiya (Ibn Abbād de Ronda, (Beyrouth, 1956), pp. XXXIX, 252), has consulted this Ms. in Morocco. The present writer has, however, failed to locate it.
- 120. See above notes 31, 54.
- 121. Shajara, I, p.247.
- 122. These extracts from al-I^ctişām were mistaken for extracts from Al-Muwāfaqāt by I. Goldziher, and the same mistake was carried on in Brockelmann.
- 123. Voorhoeve, Handlist of Arabic Manuscripts, (Library of University Leiden), (Lugduni: Batavorum, 1957), p.438.

124. Ibid.

1

- 125. Nayl, p.47.
- 126. See Renaud, "Un médecin du royaume de Granade: Muḥammad as-ṣaqūrī," Hesperis Vol. XXXIII (1946), pp.31-64.
- 127. Renaud, "Deux ouvrages perdus d'Ibn al-Hatib: Identifiés dans des manuscrits de Fès – Conclusion sur Ibn Al-Hatib médecin", <u>Hesperis</u>, XXXIII (1946), pp.213–225. Also <u>Nayl</u>, pp.264–265.

V.

ł

CHAPTER V

SOCIAL CHANGES AND LEGAL THEORY

In Chapter III we discussed in general the political, social, religious, economic and legal developments in fourteenth-century Granadian society. In the course of that discussion we indicated how the society was undergoing some significant changes. The spread of <u>sūfī</u> <u>tarīqas</u>, the influence of Razism, and the establishment of a <u>madrasa</u> system were particularly important contributions to the decline of the supremacy of the <u>fuqahā</u>. More significant were the economic changes caused by new developments in Mediterranean trade that geared the Andalusian economy to a type of mercantilism. These changes were immediately felt in the domain of Islamic law. The existing legal system was not prepared to accommodate these new circumstances.

In a number of situations, the new practices apparently came into conflict with the teachings of Islamic law. Perplexed, the people asked the jurists to solve the resultant problems. The jurists, in their responsa (fatāwā) made an attempt to reconcile the new practices with Islamic law or to reject them.

This chapter studies a segment of these answers with the following questions in mind:

- a) What subject matters in Islamic law were affected by these social changes and to what extent?
- b) In which subject matters did the jurists adopt the social changes?
- c) To what extent were these social changes related to the social conditions discussed in the preceding chapters?
d) How did the legal theory respond to these social changes? What methods were used to adopt or reject these changes?

For the purpose of this chapter we have limited our study to the $fat\bar{a}w\bar{a}$ of one jurist - Abū Ishāq Shāṭibī. This study is, therefore, based on Shāṭibī's $fat\bar{a}w\bar{a}$ which are available in the following sources:

- 1) Al-Wansharisi, <u>Al-Mi^cyār al-Mughrib....</u>, 12 volumes.¹
- 2) Lopez Ortiz, "Fatawa Granadinas...."² In this study, in addition to the above-mentioned <u>al-Mi<yār</u>, Lopez Ortiz used another collection of fatāwā that still exists in manuscript form.³

3) Certain references to Shāţibī's <u>fatāwā</u> in the following: <u>Al-Muwāfaqāt</u>, Al-I^ctisām, Nayl al-Ibtihāj.

The total number of <u>fatāwā</u> studied in this chapter is 40 and they may be distributed in these categories:

- i) Exegesis: 1
- ii) Theological matters: 2
- iii) Ritual and worship: 12; cleanliness, rituals, prayers
- iv) Family: 5; divorce, inheritance
- v) Property: 5; objects of property, waqf
- vi) Taxes: 3; zakāt, kharāj
- vii) Contract: 11; sale, hire and lease, society
- viii) Procedure: 1; witness
 - i) Exegesis

Responding to a request, Shātibī explains in this <u>fatwā</u> the meaning of an <u>hadīth qudsi</u>⁴, in which God is quoted showing His affection and closeness by becoming the ears, hands and feet of a person who endeavours to approach Him. Shātibī finds that this <u>hadīth</u> implies anthropomorphism, but without denying the authenticity of the <u>hadīth</u>, he explains how the apparent anthropomorphic implications can be removed by the method 5 of ta'wil (interpretation).

Strictly speaking, "exegesis" is not a <u>fiqhi</u> subject matter; the <u>fiqh</u> books generally do not include discussions on this subject. Yet exegesis often finds a place in <u>fatāwā</u>. Such questions, however, arise out of certain problems which are indirectly related to practices which may come into conflict with the teachings of Islamic law. The response in question was most probably prompted by the spread of sūfism in the Andalus.

ii) Theological matters

2.

Again, discussions about theology are not one of the subjects treated in <u>figh</u> books, yet it is a very common subject in <u>fatāwā</u>. It may also be argued that since a larger part of the provisions of Islamic law are applicable only to Muslims, the question of "who is a Muslim", even though a theological question, is quite relevant to <u>figh</u>.

In addition to the above, Shātibī's two <u>fatāwā</u> reveal another aspect of the relevance of dogma to <u>figh</u>. A dogma may sometimes impose restrictions on certain acts.

Shāţibī was asked about a <u>şūfī</u> who interpreted Qurānic terms to his own advantage, claiming that commands about worship were metaphoric. The <u>sūfī</u> also insisted that direct knowledge of God was possible and that books did not provide true knowledge. Shāţibī in very clear terms declared that the <u>sūfī</u> was a <u>kāfir</u>, and that he must be sentenced to death (<u>wājib al-qatl</u>). This "<u>sūfī</u>" rejected and ridiculed the "<u>sharīća</u>" and its transmission and mocked the names of God.⁶

This fatwā appears to disagree with Shāţibī's view on heresy. As Fazlur Rahman has pointed out, Shāţibī categorically states that "it is not possible to locate absolutely the capital errors of these sects so that they may be stigmatized as <u>kuffār</u>."⁷ Shāţibī is quite clear, Rahman observes further, that erroneous beliefs and practices can and must be exposed but that it is impossible to locate absolutely the holders of these practices."⁸

The above-cited view of Shāţibī does not correspond with his absolute belief in the <u>kufr</u> of an individual <u>sūfī</u> or in the unacceptability of the practices mentioned in the <u>fatwā</u>. We do not, however, here face a contradiction. Rahman's observations are derived from a certain context where Shāţibī is discussing a problem of heresiology.⁹ Is it possible to define <u>firqa nājiya</u> (the saved sect), the sect which is on the right path to the exclusion of others? Shāţibī, there, is dealing with the impossibility of such a definition. This stand, however, does not mean that the beliefs and practices implying <u>kufr</u> cannot at all be located; Shāţibī's stress is rather on the impossibility of locating the one sect with the absolute truth. The other <u>fatwā</u> related to this subject matter concerned the wax industry. For their Christian customers the Muslim artisans manufactured wax candles resembling hands in prayer. This resemblance apparently violated the teachings of Islam about strict monotheism that forbade any representation of the human figure in sculpture or paintings, since such an attempt would resemble God's act of creation. Shāţibī dismissed the objection and declared this industry lawful. Quoting earlier Mālikī jurists, Shāţibī argued that what is forbidden is the representation of the complete figure; a figure without its head in particular had been previously permitted in Mālikī <u>fiqh</u>.¹⁰

iii) Rituals and worship

A number of new practices, mostly under the influence of sūfism, had been introduced in this domain. These new practices were considered <u>**(ibādāt.**</u> In his response to the inquiry about these practices, Shāṭibī condemned them on two grounds: first, that they were <u>bid</u>**(**a) (innovations) and second, that they imposed certain practices as religious obligations, whereas the act of imposing such an obligation belongs only to God. The practices condemned by Shāṭibī in this regard included the following:

a) Reciting in congregation the Quranic chapter Yasin on the occasion of bathing the deceased in preparation for burial.

- b) The practice of the group of people called <u>suffyya</u> who assembled in some <u>zāwiya</u>, performing <u>dhikr</u> (chanting the names of God or some such formula), singing and reciting poetry.¹²
- c) Congregational recital of the <u>Hizb</u>¹³ (certain prayer formulas).¹⁴
- d) Recital of certain books in congregation in the mosques.¹⁵
- e) The congregational invocations after the regular prayers (salāt).
- f) The practice of insisting on the completion of the recital of the Quran in the month of Ramadan.
- g) Saying loudly the takbirs (the formulae declaring the Greatness of God) on the eve of 'Id prayers.¹⁸
- h) Shaking hands and embracing each other after the ^cId prayers.¹⁹
- Adding certain sentences in the <u>adhān</u> (call to prayer).²⁰ In
 <u>Al-ictişām</u>, Shātibi refers to the practice of adding the following in the call for morning prayers: "The day dawned, praise be to God".²¹
- <u>Taşbih al-Ghabir</u>: it had become the practice of the people after the burial of the deceased, to gather for seven days and recite the Quran loudly in congregation. Shātibi considered the custom equivalent to <u>matam</u> (mourning) which was forbidden in Māliki figh.²²

Whereas the above ten responses emphatically rejected the common religious practices as <u>bid</u>^ca, there were two customs in regard to which Shāṭibī showed flexibility. In Mālikī <u>fiqh</u> uncleanliness (<u>najāsa</u>) is a legal qualification (<u>sifa ḥukmiya</u>) in opposition to sensory (<u>ḥissiya</u>) or rational (<u>caqliyya</u>) qualification.²³ Cleanliness (<u>tahāra</u>) can be determined only on legal bases. <u>Khamr</u> (wine) and <u>mayta</u> (a corpse) are unclean according to the Qur'ān.²⁴ Accordingly if either of these two happens to fall into something, they make that thing unclean, and that uncleanliness cannot be removed by sensory or rational methods.

T

.

Two such situations arose, and were referred to Shāţibī for an opinion. In one case a piece of earthenware was made unclean by <u>khamr</u>;²⁵ the other case concerned some unclean thing (in another similar <u>fatwā</u> this "unclean thing" was ink made unclean by the dead body of a mouse in it)²⁶ fallen on the Qur'ān. Other <u>muft</u>īs declared these things unclean and their usage not permissible; the earthenware to be disposed of and the Book to be buried.²⁷

Shāṭibī, however, had a different solution. In the case of the earthenware, he held that if it were enameled, it could be cleaned with water in an ordinary manner. Otherwise, it should be washed thoroughly with hot water. If hot water is not available, then it might be washed with cold water but allowed to soak for a while. Its cleanliness would then be decided by ascertaining that water standing in it does not change its colour, flavour or smell.²⁸

In the case of books, Shāṭibī advised that if water would not harm or efface the writing, the books should be cleaned with water; otherwise the uncleanliness should be removed as much as possible by other means and the book allowed to stay as it was.²⁹ iv) Family

Someone repudiated his wife with the regular expression of the formula of divorce, and after some time he also pronounced \underline{zihar} (another form of repudiation by expressing the formula: "You are for me as the back of my mother"). Afterwards, however, he neither expressed repudiation nor revoked it. Shāțibī was asked about this case; whether the divorce had occurred or not. Treating $\underline{talāq}$ and $\underline{zihār}$ as two distinct acts, Shāțibī advised that in the Mālikī school one declaration of repudiation was revokable ($\underline{raj(i)}$ and not definite ($\underline{ba'in}$). Hence, in this case, since the declaration of repudiation was not repeated, the marriage was not yet dissolved. If the man still wanted to resolve the marriage, the dissolution was possible only after paying the <u>kaffāra</u> (penalty) for $\underline{zihār}$.³⁰

١.

The other three cases under this category concerned inheritance. Whereas the above case of divorce does not appear to have emerged from the changing conditions of the society, the following three were quite possibly related to these changes.

A certain Muslim committed apostacy. Soon after, his father died. Since in Mālikī law an apostate is not entitled to inherit from his Muslim father, this person immediately reconverted to Islam. Shātibī denied the son the right of inheritance on the following ground: First the cause of the transfer of the deceased person's property to the other inheritors was the "death of the owner", not the "disposal of the property", hence the right of inheritance belongs to whoever was rightful heir at the time of death. If the other heirs wished, they might give the son some part of the inheritance as a gift; or alternatively he could be granted assistance from <u>Bayt al-Mal</u>. ³¹

An opposite opinion in favour of the son was possible but Shāţibī insisted that the common practice of the Mālikī school be adhered to. It appears that a strict attitude was adopted to discourage apostacy, the growth of which is conceivable under changing circumstances. Ŋ.

In another case of inheritance, the wife of a cloth merchant, on the death of her husband, withheld a certain amount of clothing. The heirs claimed that this clothing was part of the inheritance. The wife claimed that her husband gave the clothing to her as gifts, but she could not produce any witnesses. Shātibī advised that in such a case, where there was a possibility that the clothing was part of the merchandise belonging to the deceased husband, the wife's statement could not be accepted without witness. Nevertheless the heirs should be asked to declare under oath that they did not know whether the deceased had made such gifts.

Shāṭibī, however, explained that there would be no dispute if the clothing belonged among the household articles or had already been in use by the wife.³²

Shātibī took a similar stand in another case of inheritance, where the wife claimed that the house in which she and her husband had lived had been given to her by her husband as a marriage gift (shawār).³³

v) Property

-

a) <u>Objects of property</u>: As referred to earlier³⁴, some of the cultivated land around Granada along the river Mansūra (?) was quite steep. For the purpose of irrigation small dams had to be built and the users had to take turns using the waters. These turns were strictly determined and were often passed on to the heirs as transferable rights. At times, however, some heirs either gave up cultivation or allowed their land to become barren, so that they had no use for the water. They, therefore, began to sell their portion of water to the actual users.

Ň

A dispute arose out of such a situation, and Shāṭibī was asked about it. He emphatically declared that the water was not an object of property, and that its use could not be owned by any person. He, however, distinguished between two kinds of water; such water as is in rivers and in desert ponds was not the object of property, while those waters which were either purchased with or belonged to a land, which itself was private property could become the object of property. Yet no right of ownership could be claimed on the waters of rivers by virtue of the building of dams.³⁵ b) <u>Habs</u>: The question on this subject was most probably asked by Abū ^cAbd Allāh al-Haffār, who was appointed as supervisor of <u>awqāf</u>. ³⁶ Someone willed that one third of his estate be demarcated as <u>waqf</u> (trust), for the purpose of celebrating the birthday of the Prophet.

Shāṭibī, in his response to the inquiry about this will, resolved that such a will was unlawful and hence could not be executed. The reason for its unlawfulness, according to Shāṭibī, was that the celebration of the Prophet's birthday was an innovation and hence unlawful.³⁷

The other two responses relating to <u>waqfs</u> indicate the confusion in the practice of <u>waqf</u> as well as the juridical strictness in abiding by the rules of waqf.

For the maintenance of mosques certain \underline{ahbas} (trust properties) were attached to them. The officer in charge of these \underline{ahbas} decided to rearrange the distribution of the income among various mosques, so that the income of some of the mosques be increased. Shātibī was consulted; he explained that the income of the mosques could be increased either from <u>bayt al-māl</u> or from <u>ahbās</u>. Whereas there were some restrictions in the case of <u>ahbās</u>, there was nothing against such an increase from <u>bayt al-māl</u>. This view is based on the distinction between the opposing motives of <u>bayt al-māl</u> and <u>habs</u>; whereas the essence of the latter is <u>tacyīn</u> (specification), the basis of the former is <u>cadam tacyin</u> (non-specification). Because of <u>tacyin</u>, the increase from <u>ahbās</u> would become problematic. Re-arrangement of the distribution of trust income was not possible if it was definitely known that the trust was specified for a certain mosque or a certain purpose. It would be possible only if it were known that a certain number of <u>ahbās</u> were specified for mosques but that the mosques were not specified individually. Shāțibī, however, explained that, formerly, these <u>ahbās</u> had been specified, but later, due to negligence, or because they were considered analogous to <u>bayt al-māl</u>, these specifications became confused. Then the share of each mosque was decided at the discretion of the officer in charge. In fact, it was not permissible to combine various trusts in order to increase the income of mosques.³⁸

i

•

Shāṭibī took a similar view in another case of aḥbās. Someone bought the trees on a tract of land that was adjacent to a <u>habs</u> property. A doubt passed through his mind that this tract of land might be the <u>anqād</u> (the demolished and unused part of an estate) of that <u>habs</u>. In Māliki fiqh, the act of sale of a <u>habs</u> property is legally void and if this act were knowingly committed, it was punishable by the court.³⁹ Yet it was a common practice in the Andalus to sell the <u>anqād</u> of a <u>habs</u> and, after the <u>habs</u> and <u>milk</u> (ordinary property) were confused or joined deliberately, to share the income of such a sale. The person concerned asked Shāṭibī what to do. Shāţibī replied that the practice of combining <u>habs</u> and <u>milk</u> is like mixing <u>halāl</u> (lawful) and <u>harām</u> (forbidden). As for the sale of <u>anqā</u>d the legal view was not as categorical as on <u>habs</u> itself. The Mālikī scholars had different opinions. Yet Shāţibī explained that this difference of opinion, in fact, emerged from the different bases of analogy. Ibn Mawwāz made the <u>anqād</u> analogous to <u>arādī</u> <u>suljān</u> or <u>arādī bayt al-māl</u> (crown land) and therefore, permitted flexibility in the sale and long-term lease of <u>anqād</u>. Shāţibī differed on this point on the basis of his distinction between <u>bayt al-māl</u> and habs.

He advised the person in question to go to the court for the cancellation of the sale contract; otherwise, he should appeal to the Sultān. This person accordingly appealed to the Sultān after securing <u>fatāwā</u> from ^CAbd Allāh Ibn al-Haffār and Ibn ^CAllāq which were endorsed by Shātibī. The Sultān accepted the opinion of the <u>muft</u>is and referred the case to the <u>qād</u>ī concerned. Despite the Sultān's orders, the upholders of the practice prevailed upon the <u>qād</u>ī. They shouted and condemned the plaintiff for opposing the practice. The <u>qād</u>ī, for fear of disturbances, gave a verdict in favour of continuing the practice.⁴⁰

vi) Taxes

4

In the three <u>fatāwā</u> pertaining to taxes, Shāṭibī departed from the traditional viewpoint. In fact, Lopez Ortiz interpreted this departure as "the skill of an economist from the fiscal point of view".⁴¹ Two of these fatawa concerned kharāj and one was about <u>zakāt</u>.

In view of the deteriorating financial conditions, the Sultān levied a few additional taxes. One of these new sources of revenue was a tax levied on the building of walls in or around Granada. The <u>mufti</u> of Granada, Ibn Lubb, declared such taxes unlawful, because they were not provided for in <u>shari</u>^{< a}.

Y

Shāṭibī disagreed with Ibn Lubb. He viewed taxation from the point of view of <u>maşlaḥa</u> (public weal). His idea was, and he quoted Ghazālī and Ibn al-Farrā' in his support, that the safeguarding of public interests was essentially the responsibility of the community. In situations when they could no longer carry out this responsibility, the community transfer it to the public treasury and contribute from their wealth for this purpose. With this aim in view, the public treasury is in constant need of such contributions. Especially in circumstances similar to those found in Shāṭibī's period, when the treasury had to pay a heavy tribute to the enemy, the levying of new taxes was quite in order.⁴²

Shāțibi applied this criterion even to <u>zakāt</u>. According to <u>al-Mudaw-</u> <u>wana al-Kubrā</u>, <u>zakāt</u> on merchandise for sale could be levied only after the merchandise was sold and after one year had passed; it was to be levied on the price earned from the merchandise.⁴³ Accordingly the artisans did not pay any <u>zakāt</u> on their products, because, first, only a few of these products would be sold immediately and the rest would remain as potential money not yet taxable. Second, the condition of allowing one year to pass would be hard to meet if the investment in these products was an on-going process.

11.

Shāţibī viewed this practice in the light of the changing economic conditions, which gave these artisans ample opportunity for production and yet allowed them to avoid <u>zakāt</u>. Shāţibī, therefore, opined that the products of the artisans should be taxed, as they were potentially sold merchandise.

vii) Contracts and Obligations

One very conspicuous impact of the changing economic conditions can be seen in the area of contracts and obligations. The demand for raw materials in foreign markets generated extensive trade activities within Spain and with neighboring principalities. On the other hand, these trade demands were confronted with the rising number of the population and the scarcity of resources within Andalus. It was quite understandable that such a situation necessitated the freedom of contracts to meet social demands.

In practice, as we shall see below, a number of new and complex forms of contracts emerged, but they did not always satisfy the stipulations of Islamic law. Islamic legal theory did not lay down any general principles of contract and obligation; yet its insistence on avoiding <u>ribā</u> (unjustified enrichment) and <u>gharar</u> (hazard, risk) put restrictions on a number of contracts of sale and association. Despite such restrictions, the scholars of Islamic law have observed that Islamic commercial law showed much flexibility and that custom played an important role.⁴⁵ We should keep this observation in mind as we turn now to the responses which Shātibī made to inquiries about contracts.

a) <u>Contracts of Sale</u>: Shāţibī was asked about a widespread commercial practice of Muslims in the Andalus who traded commodities such as weapons with the Christians; such trade was prohibited by the Mālikī scholars for obvious reasons. But in the particular case of the Andalus, the Muslims were forced to trade such commodities for food and clothing. The question was whether special concessions might not be granted to the Muslims of Andalus because of their peculiar circumstances. The second question was whether that prohibition applied to the sale of candles to Christians, candles were used to invoke prayers against Muslims. The third problem was whether the ^cattārs (pharmacists and general merchants) were obliged to abide by that prohibition.

In his response, Shāțibi, first of all, denied any special concession

to al-Andalus. Cities (or countries) could not be classified on these bases; even the <u>hādin</u> (the inhabitant of a country which was on truce terms with another) or <u>harbi</u> (at war) territories could not claim such concessions. The only distinction that the Mālikī jurists maintained concerned the sale of food commodities. He allowed such sales to an <u>hādin</u> but not to an <u>harbi</u>. Shāțibī did not allow such contracts of sale to Christians even on the basis of dire need for food articles in the Andalus.

X

As to the question about candles, if they were known to be used against Muslims, their manufacture and sale would both be unlawful.

Prohibitions, however, could not be imposed on the <u>attārs</u>, because they are merely salesmen; they do not know for what purpose their merchandise may be used and have among their customers both Muslims and Christians.⁴⁶

It is obvious, in this response, that Shāṭibī did not allow the sale of arms and other such articles which would eventually be used against the Muslims; yet this did not mean that trade with Christians was to be stopped altogether. The circumventing method of permitting the <u>caṭṭārs</u> to make such contracts of sale shows that the jurists did allow consideration for the dire needs of the people, even though as a general principle they would deny it.

It appears that Shātibi considered such demands from the merchants as

LLJ

pedantic. Someone asked him if the common practice among the saffron merchants to mix the yellow stigma of saffron with the white styles of its pistils was not <u>ghashsh</u> (adulteration), analogous to the mixing of saffron with yellow colouring powder. Shātibī agreed that adulteration of saffron with yellow colouring powder was not permitted, but he disagreed with the analogy made to the practice of mixing the stigmas and styles of saffron. Rather, in his opinion, such 'mixing' was analogous to the 'mixing' of fig seeds with figs and raisin stems with raisins. In fact, the matter comes down to the question of cutting the stigma of the saffron to remove it from its styles. In common practice, to do this is considered inconvenient. Since failure to cut the stigma does not make much difference in weight and its removal is not considered necessary, this practice should not be regarded as adulteration.⁴⁷

Shāţibī was consulted in another case of sale contract. Someone handed his merchandise over to a sales agent on the basis of a suggested price. A buyer suggested a different price, the agent informed the owner, and the latter agreed to that price. The agent, however, asked the buyer to raise the price to which he agreed. Thus the agent sold the merchandise for more than the price agreed upon between him and the owner. Shāţibī was asked if such a sale contract was valid.

*LL*0

ł

He responded that, since the stipulation of a contract of sale (the offer, 'ijāb, and acceptance, qubūl) had been fulfilled, the contract was valid and it could not be revoked. As to the question of the agent charging a price higher than the one consented to by the owner, this fact did not invalidate the contract, because the owner's acceptance and bid to sell at a particular price was commonly under-stood as "sell it at this price if there be no higher offer", not as "sell it at this price only and do not accept higher offers".

Ä.

b) <u>Contracts of Lease and Partnership</u>: Beside the cases mentioned above in the category of sales, the rest of the cases pertaining to contracts overlap with the categories of lease and partnership. Two of the cases are even related to the category of 'joint ownership'. We have juxtaposed all these cases here, without imposing our own classification. The purpose of such treatment is to indicate the confusion in the original treatment of the cases.

With the exception of one which concerns the 'joint ownership of food', the rest of the cases in this category are related to agricultural contracts. For a full explanation of the context of the problems in these cases a few remarks about the Mālikī law on agricultural contracts must be made.

In broad terms the agricultural contracts are considered analogous to 'contracts of sale', and in a specific sense they are 'contracts of the sale of usufruct' (ijāra). Having inherited the confusions and un-

certainties about sale contracts in the early development of Maliki theory and practice in Medina, Maliki figh has become very complicated in regard to such questions. First confusion arose between two types of contracts for the lease of land; musaqat the lease of a plantation of fruit trees, and <u>muzārata</u> the lease of a field. Early Malikis maintained distinctions between the two and regarded muzarata as valid only if the field were situated in the middle of the plantation.⁴⁹ Later, however, it seems that this stipulation was no longer observed, and muzarata came to be closer to a contract of partnership and musägät to that of hire of services. The second source of confusion was the prohibitions that concerned ribā al-fadl (inequality in exchange of the same stuff), which implied the prohibition of the lease of one agricultural property for another and gharar (hazard, risk) or juzaf (undertermined quantities) which invalidated most agricultural contracts since the object of the contract, e.g. wages, was often undetermined. The third source of confusion was the failure to distinguish among contracts for hiring of services, contracts of lease of land and contracts of partnership. All three are treated as contracts of lease but the stipulations are often borrowed from other types of contract of sale. Furthermore, the stipulations of ijāra, that the period of time must be determined and the task be defined, were often ignored in muzāra<a and musāqāt.

× .

Santillana marked out four basic types of contracts of <u>muzāra</u> in Mālikī <u>fiqh</u> involving situations where:⁵⁰ ал у 1 ж. на

- a) The land, the labour, the seed, the animals and the tools of cultivation are shared by two parties, the produce to be shared by both.
- b) The land is common, one party provides the seeds, the other party provides the labour and the animals.
- c) One party provides the land and the seeds, the other the labour and the animals.
- d) One party provides the land and part of the seed, the other provides the other part of the seed and the labour as well as the animals.

These types of arrangements indicate that <u>muzāra'a</u> in Māliki figh is a contract of partnership (since it is also called <u>shirka</u> fi al-zar'), rather than a contract of sale; yet in reference to the distribution of the produce there is much similarity to a contract of sale of usufruct or to a contract of hire and lease. We turn now to the specific responses.

The <u>mukhtass</u> lands belonging to <u>bayt al-mail</u> were leased to cultivators with the stipulation that every thing needed for cultivation would be provided by the cultivator himself and, furthermore, that he had to pay 1/5 plus 1/10 (or 1/9 if the land was provided with irrigational or other facilities) of the produce.

When Shāṭibī was asked about this practice, he declared such contracts invalid because the contract confused two distinct obligations; the obligation to pay 1/5 which was the rent on the land and the obligation to pay 1/10 which was the tax on the land.⁵¹

Lopez Ortiz further observed that such contracts were not even valid instances of <u>muzara</u> according to Maliki figh since the landowner did not contribute anything more than just the land.⁵²

Another source of confusion was the practice of hiring the farm labour. The regular types of contract of <u>muzāra</u> a did not allow this. If the hire of labourers was considered to be the hire of services, then it was restricted by two stipulations:

a) the wages could not be paid from the produce of the land, andb) an uncertainty existed in the payment of the wages.

These questions were raised in the case of contracts of labour and partnership regarding the collection of olives and the rearing of silk worms.

In the case of picking olives Ibn Sirāj responded to an <u>istiftā</u> that the contract for the hire of labour to collect olives could be considered <u>musāqāt</u> if the olives were not yet ripe. The contract would then consist of taking care of trees, irrigation...etc. The labour could be contracted for in this case on the promise of payment of 1/4 or so of the produce. But if the olives were ripe and the task was only to collect them, such a promise of wages would make the contract invalid because the task was uncertain and the price of the olives undertermine-able.

Shātibi, however, explained further that if the task were to collect the

olives by picking them from the branches or by shaking the trees, the contract was invalid and the wages unlawful. However, if the task were to collect olives that were already on the ground, a contract for the hire of labour may be allowed because in this case the labourer could guess how much he would eventually receive in wages.⁵³

In the case of rearing silkworms, Haffār, drawing an analogy with the <u>musāqāt</u>-type of contract argued that a contract could be made only when the mulberry trees had grown leaves. The owner of the trees would contribute his share of leaves (1/2, 1/3, 2/3 or whatever had been agreed upon). The partner would also contribute his share of leaves. The owner of the silkworms would pay wages to the other in proportion to his share of leaves. In other words, this would be a case of partnership, and each of the partners would contribute his known shares. The common practice, on the other hand, was to contract before the appearance of the leaves and to pay the wages from the leaves or in silkworms.

Shāţibī responded that, in principle, the wages not be paid from the produce, but if the case were made analogous to <u>muzāra</u> by equal partnership in leaves, silkworms...etc., such payment could be allowed because, in that case, the labour would stand equal to half of the partner's share of leaves.⁵⁴

A more complex case of partnership was the practice of pooling milk

to make cheese. Shātibi explained that, in principle, the mixing of milk in unequal quantities to make cheese could not be allowed because it resembled <u>muzābana</u> (a contract for barter of dried dates for fresh dates). The practice may also be associated with <u>gharar</u> and <u>ribā</u> which are prohibited. Yet the mixing of milk could be allowed by consideration of <u>tashil</u> (convenience) and <u>rafé baraj</u> (removal of hardship). The consideration of <u>baraj</u> becomes relevant in this case because it would be inconvenient to produce cheese individually by keeping every partner's share of milk separate.⁵⁵

A peculiar case of partnership had to do with the produce of a tree owned by more than two persons. The question was asked whether it were permissible to distribute the produce equally.

Shāţibī did not allow such distribution because the matter of the tree was actually a case of partnership and not of joint ownership. Hence the distribution of the produce must be according to known shares. Shāţibī suggested that to make such a distribution convenient the branches of that tree should be marked for every partner and then the produce be divided accordingly.⁵⁶

c) <u>Contract for hire of services</u>: Shājibī was asked whether it was allowed for an <u>imām</u> (leader of prayers) to live on income from the <u>habs</u> of a mosque, without any other vocation. Shājibī responded that the office of <u>imām</u> was a vocation, and if the person in question performed his duties, it was lawful for him to live on such income.⁵⁷

▲ >> 2 3 ≤ × An interesting case was the emergence of the appointment of <u>mu</u>^cin <u>al-dhabh</u>. In the meat market a person was hired by the butchers to supervise the killing of animals and to keep accounts of animals, meat and skins. He was paid partly by the butchers and partly by the sale of the meat. Shātibi was asked whether such an appointment was lawful.

N

4

Shāṭibī replied that if the consideration governing the appointment of <u>muʿīn al-dhabḥ</u> was to safeguard <u>maşlaḥa</u> (public interest) in the observance of <u>sharīća</u> rules about the killing of animals, then in view of <u>fasād al-zamān</u> (corruption of contemporary conditions) or the ignorance of religious teachings, such an appointment could be allowed. If such considerations did not exist and the person was not qualified to carry out the rules, his appointment would fall into the category of reprehensible things. Furthermore, such a practice would impose upon the people hardships in those matters in which God allowed convenience. The Prophet Muḥammad used to eat meat brought by <u>badawis</u> after simply saying the name of God.⁵⁸

Abū Abd Allāh Haffār asked Shāţibī if an increase in his salary received from the <u>bayt al-māl</u> was lawful. Shāţibī replied that it was lawful only on two conditions:

- i) that the amount of work had increased.
- ii) that the increased wages were commensurate with the increased work.

When Haffār received that answer he wrote again saying that he had been receiving that increase already for thirty years and that the practice of the community had allowed him to do so. He asked Shāṭibī what he should do. Shāţibī answered that he was not obliged to return the overpayment; what deserved consideration in this case was the morality of receiving such an increase, not whether the practice of the community allowed it or not.⁵⁹

viii) Procedure

Shāţibī was asked about the legal nature of <u>lawth</u> (incomplete evidence leading to presumption of guilt in case of homicide), in a case where there was one eye witness of the murder and two witnesses of the culprit's confession. According to Mālikī law the requirements of proof in such a case are the following:

- i) the confession of the culprit, and
- ii) two eye witnesses of the murder,
- iii) one witness on the basis of <u>qasāma</u> (declaration on oath by several persons), and
- iv) strong circumstantial presumptive evidence. 60

The evidence in the above case did not fulfil the requirements completely, but it was evidence of presumption. Shāţibī was asked whether it was <u>lawth</u>. He responded that <u>lawth</u> could be described as accepting weaker evidence that could prove harm. Accordingly, the evidence in this case would be regarded as <u>lawth</u>. Shāţibī, however, explained that the difference of opinion among Mālikī jurists on the legal effectiveness of <u>lawth</u> is based, in fact, on the fact that the efficacy of lawth depends largely upon the discretion of the judge (<u>nāẓir al-qaḍiyya</u>).⁶¹

Conclusion:

.

To conclude this chapter we recapitulate the above with reference to the questions that we raised in the introduction. A fuller statistical analysis of the <u>fatāwā</u> is given in a tabulated form. Brief answers to the questions are attempted below.

The first question concerned the subject matter of Islamic law that was confronted by the impact of social change. Out of the 40 cases 34 implied social change. Among them the following categories are included:

- 1) Theological matters (2/2 implied change)
- 2) Ritual and worship (11/12)
- 3) Family (3/5)
- 4) Property (4/5)
- 5) Taxes (3/3)
- 6) Contracts and obligations (11/12)

Subject matter		Total number of cases	Those which imply social change		Related to the general social conditions			Shāțibi's attitude		
	marter		yes		yes r	10	n/a	accepted	rejected	<u>n/a</u>
1.	Exegesis	1	-	1	1	-	-	-	-	١
2.	Theological matters	2	2	-	2	-	-	۱	1	-
3.	Ritual and worship	12	11	1	10	2	-	1	11	-
4.	Family	5	3	2	3	-	2	3	2	-
5.	Property	5	4	١	4	-	1	-	4	1
6.	Taxes	3	3	-	3	-	-	3	-	-
7.	Contract and obligation		11	-	11	-	-	6	5	-
8.	-	1	-	١	-	1	-	-	-	1
тс	DTAL	40	34	6	34	3	3	14	23	3
	RCENTAGES	100	85	15	85	7.	5 7.5	35	57.5	7.5

The following table shows, in detail, the attitude towards the adaptation to social changes in Shātibi's fatāwā.

The second question concerned the adaptation to social change by the jurists. Out of the 40 cases Shātibi adapted law to social change in 14 and rejected The details are as follows: adaptation in 26.

÷.,

Ŋ.

	Adaptation to social change	Rejected adaptation to social change		
1. Exegesis		-		
2. Theological matters	1	1		
3. Ritual and Worship	1/12	11/12		
	3/5	2/5		
•	· .	4/5		
5. Trust (Property)	-	·		
6. Taxes	3/3	-		
7. Contract and Obligation	6/11	5/11		
8. Procedure	-	-		

The subject matters in which Shātibī rejected adaptation to social change most often were ritual and worship, family and trust. He showed flexibility with respect to theological matters and taxes. In contracts and obligations, he accepted and rejected cases almost equally, although he accepted adaptability more often than he rejected it.

As to the third question, whether these changes were related to the general social conditions or not, it may be seen that out of 40 cases, 34 were related to such conditions; two cases of ritual and one case of procedure were not the result of such conditions, whereas the rest of the cases were posed by social changes.

The fourth question concerns the method of adaptation to or rejection of adaptation to social change by the jurists. Broadly speaking, in rejecting adaptation of law to several social changes two different principles are invoked by Shāțibī. In matters relating to theological matters, ritual and worship, and

r.

trust he rejects adaptability of law to the social changes by declaring these changes <u>bid</u>^ca (innovations). It is obvious that these matters particularly concern religion or relate primarily to a matter between man and God. It is very important to note that Shātibī does not invoke this principle in other matters. Apparently a social change affecting the above mentioned subject matters implied, for Shātibī, the imposition of a new obligation or the rejection of an earlier obligation in the name of religion. This must have led him to an investigation of the philosophical question of religious authority – to whom did it belong? An analysis of Shātibī's concept of <u>bid</u>^ca can provide us with an answer to this question in the particular context of our dissertation.

The second principle employed in rejecting social change, especially in cases of contracts and obligations, was that of 'unjust enrichments' and 'risk'. This principle can be understood as the negative side of the other principles such as <u>tashil</u>(convenience), <u>Cadam haraj</u> (removal of hardship) and <u>maslaha</u> (public good) on the basis of which he accepts social changes especially in matters of taxes and contracts.

The use of such general principles was prompted because of the failure or confusion of the regular methods of interpretation usually employed by the jurists. Among such regular methods the following are used in these <u>fatāwā</u> but found insufficient.

First is the method of analogy. This method is used in three ways:

~ ° •

200

a) Proximate analogy with a precedent that is very close to the case in question.

1

- b) New analogy to refer to a precedent which was not usually employed to make analogy. e.g. - the production of making cheese analogous to olive oil.
- c) To adjust the case in a way to suit the requirements of analogy. As in the case of rearing silkworms, analogy was sought with <u>muzāra^ca</u> by restructuring the form of the contract. This method of analogy forced the jurists to be casuistic, and even then the results were not very satisfactory. Consequently, they had to refer to general principles.

Second was the method of abandoning strict adherence (taqlid) to Māliki figh in order to borrow from other schools. This method, though employed by other jurists, was rejected by Shātibi as it led to a diversity of legal practices and also because it did not help to solve the problems. The method still depended on analogy.

The third method which was used in contracts was to divide the contract into different moments and parts of the contract so as to find analogies applicable to each.

In short the usual <u>fiqhi</u> methods generally proved to be insufficient to meet the new changes, and, hence, the jurists turned to general principles. An obvious result of such a trend was that more attention was paid to <u>usul al-fiqh</u> in order to investigate the foundations, objectives and purposes of Islamic law. It would not be accurate to presume that Shāţibi alone was facing these problems. It is true that Shāţibi invoked the general principle of <u>maşlaha</u> more often than other jurists, still, it is important to note that other contemporaries of Shāţibi, Ibn Allaq (d.707 A.H.), Ibn Āşim (d.811 A.H.), Ibn Sirāj (d.818 A.H.) and others whose <u>fatāwā</u> were studied by Lopez Ortiz also frequently refer to such principles as tashil, darūra...etc.

Not only did lengthy discussion go on between various jurists about such matters as <u>qiyās</u>, <u>ikhtilāf</u>, and the role of custom, but the question of <u>mashhūr</u> <u>madhhab</u> and such subject matters were also discussed, indicating the interest of the jurists in legal theory.

Another important factor stimulating the interest in legal theory was the fact that the nature and form of contracts, which have a fundamental significance in every legal system were changing in that period. The factor of labour, and especially of seasonal labour, had brought a new dimension to the problem of wages. The new forms of contracts did not fit into the old framework of agricultural contracts for lease of land. The <u>fuqahā</u>' who still considered hired labour a <u>sharika fī-l-zar</u> found in application contracts considered in this way too complicated and too unjust to be convenient for any of the parties.

The above analysis has revealed how the impact of social change was felt in the <u>fatāwā</u> in this period. It has further shown that the older legal concepts failed to answer the problems raised by the social changes. We have also seen that because of this failure, Shātibī and other jurists resorted to general philo-

4-TV

sophical principles such as <u>maşlaha</u>. The failure of older legal concepts and resort to general principles caused the jurists to reflect on basic matters of legal theory.

Finally, the above analysis has shown that a change in method and substance of <u>fiqh</u> had taken place. Such a change logically called for a theoretical justification of the adaptation of law to social changes. Shājibī sought this justification in the principle of <u>maşlaha</u> as we shall see in the following chapters. There were, however, certain theoretical and methodological objections raised by the jurists against using <u>maşlaha</u> as a method of legal reasoning. Shājibī's analysis of <u>maşlaha</u> cannot be fully understood without a general understanding of such objections. The following chapter, therefore, outlines the development of the concept of <u>maşlaha</u> in <u>uşūl al-fiqh</u> prior to Shājibī.

N,

471

NOTES: CHAPTER V

1. These <u>fatāwā</u> are recorded by Al-Wansharīsi, <u>Al-Mi'yār al-Mughrib'an</u> <u>fatāwā 'ulamā' Ifrīqiya wa'l Andalus wa'l Maghrib</u> (Fās, 1314–1315 A.H.) at the following places:

Vol. 1, 22,24,229,267,267-68; Vol. 11,230,401-403; Vol.111,163; Vol.V,17,18,19,50-51,186-189,192; Vol.V1,254-279; Vol.V11,68-69,70-74; Vol.V11,238; Vol.IX,163-165,181,478; Vol.X1,31-37,82-83,87,88-91,96-98,107-111; Vol.X11,6,7,8,11,16,19,28,201-211.

- 2. Lopez Ortiz, "Fatwas Granadinas de los Siglos XIV y XV", <u>Al-Andalus</u>, VI(1941), pp.85,89,93,97,98,102,109,114,120,123.
- 3. Caisiri, <u>Bibliotheca Arabica Hispana Escurialenisis</u>, 1, 460/no:1,096. <u>vide</u> Lopez Ortiz, op.cit.
- 4. Hadith qudsi, in opposition to the hadith nabawi, represents those sayings of the Prophet in which the actual "word of God" is believed to be found, whereas hadith nabawi represents the sayings attributed to Muhammad. See Shorter Encyclopedia of Islam, article "Hadith", 117.
- 5. Al-Mi'yar, Vol. XI, 96f.
- 6. Ibid., Vol. II, 401-403.
- 7. F. Rahman, Islamic Methodology in History (Karachi, 1965), 166.
- 8. Ibid.
- Al-Ktişām, Vol.II,208ff. Rahman's reference to Shātibi (Al-Ktişām, III, 178) needs to be corrected. According to our text of <u>Al-Ktişām</u> the quotation is located at Ibid., Vol. II,p.251.
- 10. Al-Mi'yār, op.cit., Vol.XI,87.

- 11. Ibid., Vol. 1, 267.
- 12. Ibid., Vol. XI, 31-33.
- 13. Most probably the practice of hizb was introduced into the Andalus with the Shādhiliya. The most well-known hizb, hizb al-bahr, which was supposed to be chanted while crossing the sea, is attributed to Abū'l Hasan Shādhili. See D. B. MacDonald, "Hizb", E.I. (2nd ed.), Vol. 111, 513-14.
- 14. Al-Mi^cyar, Vol. XI, 88.
- 15. Ibid.

5.

- 16. Ibid., see also Ch. IV. p. 182f.
- 17. Ibid., 89.
- 18. Ibid.
- 19. Ibid.
- 20. Ibid., Vol. 1, 229.
- 21. Al-Ktisām, Vol. 1, 207.
- 22. Al-Mi^cyār, Vol. 1, 267.
- Ibn Arafa (d. 803 A.H.), Hudud Fiqhiyya, vide Abu Abu Allah Muhammad al-Raşã^c, Sharh Hudud Fiqhiyya (Tunis, 1350 A.H.), 12–13.
- 24. The fight conclusion on the prohibition of khamr is based on the following verses: 2:219; 5:90-91. "O you who believe, intoxicants (khamr) and games of chance are only an abomination of Satan's handiwork, so be ye away from it so that ye may be successful". The prohibition of mayta is based on the following verses: 2:173; 5:3; 16:115. "Verily He hath forbidden unto you what dieth of itself..."
- 25. Al-Misyar, Vol. 1, 22.
- 26. Ibid., Vol. 1, 25.
- 27. Ibid.
- 28. Ibid., 22.
- 29. Ibid., 25.

- 30. Ibid., Vol.IV, 146.
- 31. Ibid., Vol.IX,163.
- 32. <u>lbid</u>., Vol.IX, 478.
- 33. Ibid., Vol.III, 163.
- 34. See p. 124.
- 35. Al-Mi^cyār, Vol.VIII,238.
- 36. Ibid., Vol.VII,74.
- 37. <u>lbid</u>.

-

S. 2

- 38. Ibid., Vol.VII,68.
- 39. Ibn 'Āşim, <u>Tuḥfat al-Hukkām</u>...Ed. by L. Bercher. (Alger, 1958), 174. See also Bercher's note: 873, p. 387. المراجع علم أسا See also Bercher's note: 873, p. 387.

Y.

- 40. Al-Micyar, Vol.VII, 70-74.
- 41. Lopez Ortiz, op.cit.,97.
- 42. Ibid., 85; Nayl, 49; Mi^cyār, Vol. XI, 101–107.
- 43. <u>Al-Mudawwanat al-Kubrā</u>, Vol.1, (Baghdād, 1970), 268. Sahnūn asked Ibn Qāsim about a case where someone inherited certain merchandise, which he decided to sell; would there be <u>zakāt</u> on this merchandise. Ibn Qāsim said no, because: المتقبل بالمحن هولاسن يوم بامعا، لائه يوم باعها صاربت للتجاري.

On p.294, Mālik states about <u>zakāt</u> on fruit as follows:

فليس فيعا زكاة ولافى انمانها حتى يعول على انمانها الحول سنايرم تعبض انمانها

- 44. Lopez Ortiz, op.cit.,97.
- 45. Udovitch, Partnership and Profit in Medieval Islam (Princeton, 1970), pp.7f.,9,10.
- 46. Al-Mi^cyār, Vol.V, 186-187.
- 47. Ibid., 19.
- 48. Ibid., Vol.V, 192.

- 49. D. Santillana, <u>Istituzioni di dritto Musulmano Malichita</u>, Vol. II, (Rome, 1938), <u>306</u>. See also <u>Mudawwana</u>, Vol. VI,21.
- 50. Santillana, loc.cit.

•••

1.

51. Lopez Ortiz, <u>op.cit.</u>,97. Shāţibī's <u>fatwā</u> is understood better in the context of the following two stipulations of <u>musāqā</u>t, mentioned by Ibn ^cĀşim

Ŋ

(i) ولاتصح [المسامّاة] مع كماء
 ولاتصح [المسامّاة] مع كماء
 والدفع للزكاة المام ليشتوط بينيها بنسبة العبزء فقط (ii)
 المام المعنوط المعنوط المعنوط المعنوم المعن

- 52. Lopez Ortiz, op.cit.
- 53. Ibid., 109.
- 54. Al-Mi'yār, op.cit., Vol.V,50.
- 55. Ibid., 187.
- 56. Lopez Ortiz, op.cit., 102.
- 57. Al-Mi^syār, op.cit., Vol.VII, 68.
- 58. Ibid., Vol.IX, 98.
- 59. Ibid., Vol.VII, 74.
- 60. Ibn Asim, op.cit., 226.
- 61. Al-Mi^cyār, Vol.11, 230.
CHAPTER VI

1

THE CONCEPT OF MASLAHA BEFORE SHATIBI

Shāțibi's doctrine of <u>maqāşid al-Sharī</u>^ca, which is analysed in the following chapter, is, in fact, a continuation of the discussion of the concept of <u>maşlaha</u> that had appeared in major works of <u>uşūl al-fiqh</u> prior to Shāțibi. It is, therefore, necessary to review briefly the major problems in the treatment of this concept in traditional Muslim jurisprudence.

Etymologically the word <u>maşlaha</u> is an infinitive noun of the root <u>s-1-h</u>. The verb <u>saluha</u> is used to indicate when a thing or man becomes good, uncorrupted, right, just, virtuous, honest, or alternatively to indicate the state of possessing these virtues. When used with the preposition <u>li</u> it gives the meaning of suitability. It is also said of a thing, an affair or a piece of business which is conducive to good or that is for good. Its plural form is <u>maşalih</u>. <u>Salāh</u> is its synonym, and <u>mafsada</u> is its exact antonym. <u>Mafsada</u> is the synonym of <u>fasād</u>. In Arab usage, it is said: <u>nazara fī maşālih al-nās</u>, which means: "He considered the things that were for the good of the people." The sentence <u>fī'l- amr maşlaha</u> is used to say: "In the affair is that which is good \angle or cause of good 7.¹

In the Qur'ān various derivatives of the root $\underline{s-1-h}$ are used, but the word <u>maslaha</u> does not appear there. The Qur'ān uses \underline{zalama} ('He did wrong') $\sqrt{V:397}$ and <u>fasada</u> ('He/it corrupted') $\sqrt{XXVI: 125; XXVII: 48; VII:142, II: 2207}$ as

opposite terms to <u>saluha</u>. <u>Salih</u>, the active participle of <u>s-l-h</u>, occurs very frequently in the Qur'ān. On one occasion the meaning of this term is elaborated textually as follows:

> "They believe in God and in the last day and enjoin goodness and forbid evil and hasten to do good deeds, and these are the righteous ones (Sāliḥīn)".²

It is quite often claimed that <u>maşlaha</u> as a principle of legal reasoning – broadly speaking, to argue that "good" is "lawful" and that "lawful" must be good – came to be used at a very early period in the development of <u>figh</u>. The use of this principle is attributed, for instance, to the early jurists of the 'Ancient schools of law' or even to the companions of the Prophet. Among the founders of the schools of law, it is associated with Mālik b. Anas. There seems, however, to be a confusion in these statements in equating the use of <u>maşlaha</u> as a general term with its use as a technical term. The early use of <u>maşlaha</u> may have been in its general sense similar to other terms such as <u>ra'y</u>. Rudi Paret has observed that the word <u>maşlaha</u> as a technical term is not used by Mālik or Shāfi^cī; hence this concept must have developed in the post-Shāfi^cī period.³

Paret's observation, however, does not refute the possibility that considerations similar to maşlaha were employed by pre-Shāfi^cī jurists. Such considerations do not, however, seem to have been formulated in technical legal terms. The proponents of the use of maslaha in the early period have, apparently, confused the early similar considerations with maslaha. It is, therefore, not incorrect to say that the post-Shāfi^cī development of the concept of maslaha was a continuation of such early methods of reasoning as were not yet formally defined. Later, when Shāfi^ciⁱ's definition of the method of reasoning in terms of sources and his insistence that all methods must be linked with the revealed texts through <u>qiyās</u>, prevailed over other methods of reasoning, the concept and method of <u>maslaha</u> was also seen, especially by Shāfi^ci jurists in terms of 'sources'.

From Imām al-Haramayn Juwaynī's (438/1047) <u>Al-Burhān</u>, it appears that by his time the validity of reasoning on the basis of maslaha had become a problem controversial enough to bring forth three schools of thought in this respect to it. Some Shāfi^cīs and a number of <u>mutakallimūn</u> are claimed to have maintained that the acceptable <u>maşlaha</u> is only that which has a specific textual basis (<u>aşl</u>). The <u>mursala</u> (a <u>maşlaha</u> not based on such an <u>aşl</u>) and the like are contradictory to the textual evidence (<u>dalīl</u>), hence not valid. The second school of thought is attributed to some followers of Shāfi^cī and to Hanafīs in general. They believe that <u>maşlaha</u>, even if it is not supported by a specific basis, can still be used, provided that it is similar to those <u>maşālih</u> which are unanimously accepted or which are textually established. The third school is attributed to Mālik who held that a <u>maşlaha</u> is abided by without any consideration of the condition of similarity or whether it corresponds with the texts or not.⁴

This comment by Juwayni does not help us in determining the dates of the use of <u>maslaha</u> but it is very significant to note what divides these schools on <u>maslaha</u>. First the comment shows that the method of reasoning on the basis of <u>maslaha</u> was different from an other method of reasoning which sought its basis in the revealed texts. Secondly if we also accept the attribution of <u>maslaha</u> to the

240

ĺ

names of the jurists given in this comment, the comment also shows that the method of <u>maşlaha</u> in its early formulation by Mālik and his followers was independent of the consideration of 'sources' or 'bases' and further that <u>maşlaha</u> was accepted by others if it conformed to 'sources' - to the text in the case of the first group and to <u>ijmā</u> in the case of the second group. They rejected only <u>maşlaha</u> <u>mursala</u> because it did not conform with the sources. This explains why the concept of <u>maşlaha</u> which originally was not necessarily conceived and confined within the framework of 'sources', came to be seen, particularly by later Shāfi'īs, in reference to 'sources'. This confused the discussion of the concept of <u>maşlaha</u> as we shall see at a later moment. One indication of this confusion that may be noticed in the following analysis, is the tendency to discuss <u>maşlaha</u> at two levels, i.e., first in terms of need and effectiveness, and second in reference to sources. When talked about in terms of validity these two levels were confused.

Juwayni analysed maşlaha, or munāsib which is often used synonymously, as an <u>cilla</u> and divided it as such into five categories. First is the category where its ma<u>n</u>a (significance) is rationally understandable and where it is related to certain essential necessities (<u>darūra</u>) which are inevitable. The second category concerns what is a general need (<u>al-hājat al-cāmma</u>), but below the level of <u>darūri</u>. Third is the category which belongs to neither of the above, but rather concerns something which is noble (<u>mukarrama</u>). The fourth category is similar to the third, yet, in terms of priorities, the fourth comes later. The fifth category concerns those ma<u>s</u>ālih, whose <u>ma</u><u>n</u>ā (significance) is not obvious, and is not demanded by darūra, nor by hāja nor is it required by <u>mukarrama</u>. Examples of this category are the purely physical <u>ibādāt</u>.⁵

Í,

Maslaha as a technical term is not used in the Zāhirī jurist Ibn Hazm's (456/ 1065) <u>Al-Ihkām fi Uşūl al-Ahkām</u>, or in Hanafī jurist, Pazdawī's (d. 482/ 1089) Uşūl.

The terms maslaha and masalih are used by the Mustazili Abü'l Husayn Basri (d. 478/1085) both in a general sense and as technical terms. To him masalih are good things, and maslaha means goodness. Basri discusses maslaha in reference to <u>lstidlāl</u> (reasoning) and <u>silla</u> (reason), and in arguments against his opponents who maintain that masalih cannot be known through reasoning at all. At one point he defines masalih <u>Sharsiyya</u> as those acts which we are obliged to do by the <u>Sharisa</u> such as <u>sibādāt</u>.⁶ Related to these acts are the means to achieving the <u>Sharsi</u> commands; these means are also connected with masalih. These means are <u>dalīl</u>, <u>amāra</u>, <u>sabab</u>, <u>silla</u>, <u>shart</u>. The illustrations of these terms are given respectively as follows: the validity of consensus, analogy, the sunset for <u>salāt</u>, measurability for <u>ribā</u>, the conditions in contracts of sale. All of these means are connected with <u>maslaha</u>.⁷ For instance, the connection of amāra and silla is evident in what follows:

> "When a correct sign (amāra) indicates (dallat) a quality (wasf) being (illa) reason, we decide that it is the basis of maslaha...lt indicates that the basis of maslaha is to be found wherever an cilla is found".

For Basri, then, maslaha is an end for which silla and other related terms are

means. Başrī, however, does not elaborate what these maṣāliḥ are and what is the connection between maṣāliḥ Shar'iyya which he mentions, and other maṣāliḥ which he does not mention.

In the following centuries, however, the concept of <u>maslaha</u> advanced quite significantly. There are two main stages in the development of this concept, before Shātibī. One is represented by Ghazālī in the early **Tweitth**. Century, the other by Rāzī in the early Thirteenth Century.

In Ghazāli's <u>al-Mustasfā</u>, the problem of <u>maslaha</u> is discussed more clearly and fully than by Bașri.

Ghazali defines maslaha as follows:

"In its essential (aşlaŋ) meaning it [maşlaha] is an expression for seeking something useful (manfa^ca) or removing something harmfu! (madarra). But this is not what we mean, because seeking utility and removing harm are the purposes (maqāşid) at which the Creation (khalq) aims and the goodness (şalāh) of Creation consists in realizing their goals (maqāşid). What we mean by maşlaha is the preservation of the maqsūd (objective) of the law (Shar^c) which consists of five things: preservation of religion, of life, or reason, of descendents and property. What assures the preservation of these five principles (uşūl), is maşlaha and whatever fails to preserve them is mafsada and its removal is maslaha." 9

<u>Maşlaha</u> as understood in the above definition is then divided into the following three categories. First, the type <u>maşlaha</u> which has a textual evidence in favour of its consideration. Second is the type which is denied by a textual evidence. The third is the type where there is neither a textual evidence in favour, nor in contradiction. The first category is valid and can be the basis for <u>aiyās</u>.¹⁰ The second is obviously forbidden. It is the third category which needs further consideration. Accordingly, the element of <u>maslaha</u> contained in the third category is further examined from the viewpoint of its strength (<u>quwwa</u>). From this angle there are three grades of <u>maslaha</u>: <u>darūrāt</u>, <u>hājāt</u>, ta<u>hsīnāt</u> or <u>tazyīnāt</u>. The preservation of the above-mentioned five principles is covered in the grade of <u>darūrāt</u>. This is the strongest kind of <u>maslaha</u>. The second grade consists of those <u>masālih</u> and <u>munāsabāt</u> which are not essential in themselves but are necessary to realize <u>masālih</u> in general. The third grade is neither of the above but exists only for the refinement of affairs.¹¹

Keeping this classification in mind, only that <u>maşlaha mursala</u> - i.e. that which is not supported by textual evidence, will be accepted which has three qualities: darūra, <u>qat^ciyya, kulliyya</u>. Ghazālī illustrates the point with an example:

> "If unbelievers shield themselves with a group of Muslim captives, to attack this shield means killing innocent Muslims - a case which is not supported by textual evidence. If Muslim attack is withheld, the unbelievers advance and conquer the territory of Islam. In this case it is permissible to argue that even if Muslims do not attack, the lives of the Muslim captives are not safe. The unbelievers, once they conquer the territory, will rout out all Muslims. If such is the case, then it is necessary to save the whole of the Muslim Community rather than to save a part of it. This would be the reasoning which is acceptable, as it refers to the above three qualifications. It is daruri because it consists of preserving one of the five principles, i.e. protection of life. It is qat'i because it is definitely known that this way the lives of the Muslim community will be safe. It is kulli, because it takes into consideration the whole of the community, not a part of it.¹²

The other two grades of masalih, however, are not admissible if they are not supported by a specific textual evidence. If these are supported by the text, the reasoning is then called <u>qiya</u>s, otherwise, it is called <u>istislah</u> which is similar to <u>istihsan</u>, ¹³ and, hence invalid.

Ghazālī counts <u>istislā</u>h along with <u>istih</u>sān among the methods of reasoning which do not have the same validity that <u>qiyās</u> has. He calls such methods "<u>usūl</u> <u>mawhūma</u>" - those principles in which the <u>mujtahid</u> relies on imagination or on his discretion rather than on the tradition.¹⁴

The above definition and classification of maslaha have a particular place in Ghazālī's structure of the discussion of uşūl al-fiqh. A brief analysis of this structure will reveal the place that Ghazālī gave to the concept of maslaha. Ghazālī divides the discussion of uşūl in al-Mustaṣtā into six parts. Apart from the first two parts which deal with introductory matters such as definition of uşūl and an introduction of methods of logic, the remainder of the four parts discusses the following subject matters of uşūl: Hukm (command); Adilla arbata, the four evidences, i.e. Qur'ān, sunna, ijmāt and saql; method of reasoning (istithmār), i.e. interpretation and analogy; and taqlīd and ijtihād. The above treatment of maslaha appears as an annex to the discussion of the four evidences, where he argues that maslaha is not one of the four reliable evidences.¹⁵ Also it is significant that it is not discussed in the part dealing with methods of interpretation and analogy, although its connection is implied.

References to maslaha, however, appear in other parts also. In the part on hukm, where Ghazālī discusses the essential meaning (haqīqa) and its four components, maslaha is mentioned occasionally. The four components of hukm, according to

.

253

Ghazālī are the following: (1) Hākim (the one who gives judgment; the legislator, sovereign); (2) Hukm (the judgment); (3) Maḥkūm ^cAlayh (subject of command, <u>mukallaf</u>); (4) Maḥkūm fihi (the object of command, the act <u>of Mukallaf</u>).

Discussing the meaning of hukm, he deals with the question of whether the goodness or badness of acts (both human and div.ne) is known objectively or through <u>shar</u>. His description of hasan is similar to his above definition of <u>maslaha</u> in its essential meaning.¹⁶ At one point he even uses the term <u>masalih</u> in place of hasan.¹⁷ He frequently refers to <u>mafsada</u> in the course of his analysis of <u>mahkūm fihi</u>, in dealing with the question whether only voluntary acts are objects of command or not. He regards it a <u>mafsada</u> if involuntary acts are also considered as objects of command.¹⁸

Y.

Reference to <u>maslaha</u> is made again in the part on methods of reasoning. Dealing with the method of <u>qiyās</u> (analogy), he explains that qiyas has four components: (1) <u>asl</u>, the root to which analogy is made; (2) <u>far</u>^c the branch for which analogy is sought; (3) <u>cilla</u>, the reason on the basis of which analogy is made; (4) <u>hukm</u>, the judgment to which the analogy leads. Ghazālī clarifies that <u>qiyās</u>, here, must be distinguished from <u>qiyās</u> in philosophy. This distinction lies, apart from the difference in the form of reasoning, in the conception of <u>cilla</u> itself. The <u>cilla</u> in <u>fiqh</u> is not 'cause' but merely a 'sign'.¹⁹ Naturally then the methods of finding the <u>cilla</u> are also different. The evidence in which the <u>cilla</u> is sought is <u>naqliyya</u> (traditional), meaning the Qur'ān, S<u>unna</u> and <u>ijmā</u>^c. The <u>cilla</u> is either explicit (<u>sarīh</u>), or it is implicitly indicated (<u>īmā</u>'), or it is known from the sequence and order of the command (<u>sabab</u> and <u>tartīb</u>). The fourth manner of finding the <u>cilla</u> is istinbāţ (inference). The only valid methods of istinbāţ are two: 1) <u>Al-sabr</u> wa'l_taqsīm (observation and classification; method of exclusion), and 2) <u>munāsaba</u> (affinity).²⁰ It is in reference to <u>munāsaba</u> that <u>maslaha</u> as a main element of affinity with Shar^c is frequently discussed.

Ghazali defines <u>munasib</u> as "that which, like <u>masalih</u>, becomes regulated (is achieved rationally <u>intazama</u>) as soon as it is connected with the command (<u>hukm</u>)".²¹ For a discussion of the meaning, classification and grades of <u>munasib</u>, Ghazali refers to the annex which is significantly enough the discussion of <u>maslaha</u> and its grades.

<u>Munāsaba</u> and <u>maşlaha</u> are, however, not identical. Although Ghazālī analyzes <u>munāsib</u> also in terms of effectiveness and validity in the same way as he does with <u>maşlaha</u>, yet the details vary. Among the various classifications of <u>munāsib</u>, one is of particular significance for us, as it explains the relationship of <u>munāsib</u> to <u>maşlaha</u> as well as the difference between <u>istihsān</u> and <u>istişlāh</u> in the eyes of Ghazālī. <u>Munāsib</u> is divided into four categories: first, the <u>munāsib</u> which is suitable to and is supported by a specific textual evidence. Second, that <u>munāsib</u> which is neither suitable to nor is supported by the textual evidence. Third, that <u>munāsib</u> which is not suitable to but is supported by textual evidence. Fourth, that <u>munāsib</u> which is classification the first category is acceptable to all jurists. The second category is called <u>istiķsān</u> which clearly means to make law according to personal discretion. The fourth is called <u>istişlāḥ</u> or <u>istidlāl mursal</u>. It is clear from this classification that <u>maşlaḥa</u> is the basic consideration for deciding the suitability or <u>munāsaba</u> of something which <u>istiķsān</u> lacks. But again the <u>munāsaba</u> of <u>maslaķa</u> further depends on its suitability or conformity to the text in general; otherwise it will fall into the category of istiķsān.

From Ghazālī's treatment of <u>maşlaha</u>, it can be concluded in general, that his predilection for theologization of <u>figh</u>²³ and for <u>qiyās</u> as a method of reasoning, led him to examine the concept of <u>maşlaha</u> with reservations. From the point of view of theology, he rejected the conception of <u>maşlaha</u> in terms of human utility; furthermore, he subjected it to scrutiny on the basis of revealed texts. Secondly, he made the method of reasoning by <u>maşlaha</u> subordinate to <u>qiyās</u>. He did not reject <u>maşlaha</u> altogether, as he did with <u>istihsān</u>, but the qualification he provided for the acceptance of <u>maşlaha</u>, did not allow it to remain an independent principle of reasoning.

Furthermore, with the above limitations on the concept of <u>maşlaḥa</u>, he could not bring into focus in the discussion the other elements which are quite relevant to <u>maş-</u> <u>laḥa</u>, such as <u>taklif</u>, <u>haqiqa al-ḥukm</u>, fahm al-khiṯāb, niyya, ta<abbud, etc. The discussions of these elements are scattered through various chapters in his <u>al-Mustaṣfa</u>. Also, he did not see the necessary relationship among different categories of <u>maşlaḥa</u>.

Some of the above points were taken into consideration by some jurists after Ghazālī, but more systematic consideration was given them by Shāṭibī, as we shall see later.

Ghazāli's classification and definition was followed by a number of jurists. At least according to the channel of the <u>uşūl</u> works that is mostly known to us, Ghazāli's influence, particularly in reference to <u>maşlaha</u>, is very strong. As Ibn Khaldūn

....

noticed, Başrī's book <u>al-Mu^ctamad</u> and Ghazālī's <u>al-Mustaşfā</u> remained a major source of influence for later writers on <u>uşūl</u>, until the appearance of Rāzī's monumental work <u>al-Maḥşūl</u>. ²⁴ <u>Al-Maḥşūl</u> combined the above two works and reformulated a number of concepts. Rāzī's <u>Maḥşūl</u> then in turn became a source of considerable influence for later <u>uşūl</u> works. This influence is evident from the number of commentaries and abridgements on <u>al-Maḥşūl</u> that were written in later periods. This work influenced even Mālikī and Ḥanafī <u>uşūl</u> which had so far taken exception to <u>Shāfi^cī</u> influence. We need not go into details, but it must be mentioned that Qarāfī (684/1285), Ibn Ḥājib (646/1249) and Ibn ^cAbd al-Salām, whom Shāțibī knew and in general opposed, were largely under the influence of Fakhr al-Dīn *al*-Rāzī's (606/1209) Maḥşūl.

Razi's <u>Maḥşūl</u>²⁵ is structured more on the pattern of Baṣṣī's <u>al-Mustamad</u> than on Ghazāli's <u>al-Mustaṣfā</u>. Rāzi deals with definitions of the basic terms in the introduction. Significantly enough, the discussion about the meaning and classification of <u>hukm</u> and the question of the goodness of acts constitute more than half of this chapter. The scheme of the rest of the chapter is exactly the same as that of Baṣrī. The references to <u>maşlaḥa</u> are made, therefore, in the introduction, where the question of the goodness of acts is discussed, in the chapter concerning <u>qiyās</u>, where the question of <u>munāsaba</u> as a manner of finding <u>silla</u> is dealt with, and in the last chapter where <u>al-Maṣāliḥ al-mursala</u> are discussed as one of the ways of knowing the commands of <u>Sharī'a</u> in addition to <u>qiyās</u>.

Razi does not define maslaha but it seems that in his thinking munasib and maslaha

are quite closely associated with each other. He gives two definitions of <u>munasib</u>. First, <u>munasib</u> is defined as "what leads man to what is agreeable (<u>yuwafiq</u>) to him both in "acquisition" (<u>taḥṣīl</u>) and "preservation" (<u>ibqā'</u>)." ²⁶ He explains that <u>taḥṣīl</u> means to seek "utility" (<u>manfa'a</u>), and <u>manfa'a</u> is pleasure (<u>ladhdha</u>) or its means. <u>Ladhdha</u> is to achieve what is suited (<u>mulā'im</u>). <u>Ibqā'</u> is explained similarly as removing harm, <u>madarra</u>, which is'alam (pain) or its means. Both <u>ladhdha</u> and <u>'alam</u> are evident and cannot be defined. Thus <u>munasib</u> in its final analysis is related to <u>ladhdha</u> in the positive sense and to <u>'alam</u> in the negative sense.

N

The second definition of <u>munasib</u> is as that which is usually suited (fi'l 'adat) to the actions of the wise.²⁷

Rāzī then clarifies that the first definition is accepted by those who attribute <u>hikam</u> and <u>maṣāliḥ</u> as causes or motives to God's commands. The second definition is employed by those who do not accept the above causality. ²⁸ This explanation takes us back to Rāzī's own view on the problem of causality and God's commands. This question is first dealt with in the course of discussion whether the goodness or badness of things is rational or established by <u>Shar</u>. He argues that inasmuch as the definition and understanding of good as something "suited to nature (of man)", or as "a quality of perfection" is concerned, undoubtedly good and bad are rational. The point in question is, however, whether good and bad can be defined in reference to praise or blame as the <u>Mu<tazila</u> have done. ²⁹ Rāzī, after detailed analysis concludes that, if defined in the latter sense, good and bad can be established only by Shar^s. ³⁰ The question then is whether what is praised in God's commands corresponds with the rational good or not. If it corresponds, can this correspondence be understood as cause or motive?

į

Razi answers this question in detail in his discussion of munasaba as a manner of 'illa. He argues that to prove that munāsaba can be 'illa, there are three premises to be established: first, that God issued the commands for the masalih of the people; second, that the case in question consists of a maslaha, and third, that it can be shown that the probable reason for God's issuing this particular command is this particular maslaha. ³¹ Giving six proofs, he establishes the first premise that the commands are issued because of masalih. He explains, however, that in contradiction to the Mustazila the fuqahā' do not regard maşlaha as gharad (personal motive); they rather view it in terms of ma<nā (significance) or hikma (rationale). In fact, there is not much difference between the two positions. The difference is as follows: whereas the Mu<tazila believe that God is obliged to consider maşlaha, the fuqahā'stress that He is not obliged to do so. God has done so because of His grace. ³² The second premise needs no explanation. The third premise, that this particular command attributes a specific motive to God's acts and Commands, is a position which Razi does not accept. 33 Razi resolves this problem by explaining it in the following terms:

Muslims believe that the revolving of the heavens, the rising and the setting of the stars, the continuity of their forms and the lights are not obligatory, yet it has been God's custom to continue them in one state. Inevitably it provides the probability that this will continue tomorrow and after tomorrow with the same qualities... To sum up, if a certain thing occurs repeatedly many times, it gives the probability that when it happens it will happen the same way... Now, when we observe Sharā'i^c, we find that the commands and <u>masālih</u> occur together, without being separated from each other, this is known inductively.....34 To sum up, Rāzī stresses that no motives or causes can be attributed to God's acts or commands; yet he admits that God's commands are for the <u>maşlaha</u> of the people, and this <u>maşlaha</u> or <u>munāsaba</u> can be considered <u><illa</u> for that command. The paradox in this position is resolved by two explanations: first, that these <u>maşālih</u> have occurred together with God's acts, only accidentally, not in terms of cause and effect and, secondly, that it has happened this way not as a necessary correlation between <u>maşlaha</u> and command, because God is not obliged. Rather, God has acted as He has as a Grace, so that a sign may be established to make known His command.

3

ł

Razi has offered these explanations in view of the possible objections against his admission of <u>taclil afcal-Allah</u>, (to attribute causes to God's acts). It is significant to note that Razi recounts the possible criticism of his position in lengthy detail while his own defense is very short and quite unsatisfactory. The criticism consists of more than ten objections. 35

Razī's answer to this criticism is very short. Two main points in his answer are as follows:

"We have explained that God's commands are issued (mashrū^ca) because of the masālih. As to the rational arguments that you have enumerated, they are not applicable here (ghayr masmū^ca). Because if they are established they would infringe upon the legal obligation (taklīf), whereas the controversy over analogy (qiyās), whether in favour or in opposition, is based on the acceptance of the obligation. This well-considered answer suffices all what you have mentioned. 36 Secondly, your criticism applies to those "who maintain that to attribute <u>masalih</u> as <u>cilla</u> to God's commands is rationally necessary. It is not applicable to the one who holds that it is not obligatory for God but He has done so because of His Grace." 37

1

ŗ

Thus Razi could maintain that <u>munasaba</u> or <u>maslaha</u> were evidences for <u>silla</u>, and could still insist that God's commands had no motives. It is with this reservation that Razi apparently accepted the first definition of <u>munasib</u>. This is also the reason that he divides <u>munasib</u> into two categories: <u>Haqiqi</u> (true) and <u>iqna</u>; (apparent). <u>Haqiqi</u> is that <u>munasib</u> which consists of either a <u>maslaha</u> in this world or one in the hereafter. <u>Iqna</u> i only appears to be a <u>munasib</u>; in fact, it is not.³⁸

Like Ghazālī, Rāzī also divides <u>maşla</u>ha into <u>darūri</u>, <u>hāj</u>ī and <u>tahsīnī</u>. He divides <u>munāsib</u> according to <u>tarthī</u>r and <u>shahādat al-shar</u> (textual evidence), and <u>mulārama</u>.³⁹ With the exception of certain differences of detail, he is generally in agreement with Ghazālī.

In general, the attempt at theologizing the concept of <u>maşlaha</u> in Ghazālī was completed by Rāzī with much more emphasis. Ghazālī objected that a conception of <u>maşlaha</u> in reference to human utility alone and independent of God's determination, is not theologically possible. Rāzī gave this general objection a specific theological content. He made it clear that even to attribute the consideration of <u>maşlaha</u> in terms of human utility to God's commands, is to attribute causality to His acts and hence theologically impossible. Both of these positions led to a kind of <u>ijbār</u> (determinism). ⁴⁰ Both implied that God's commands demand obedience in their own right, not because of maşlaha. If there existed the content of <u>maşlaha</u> in Sharīća, it was to be explained by the grace of God or by accident, as Rāzī held. These positions rendered the question of moral and legal responsibility meaningless. Rāzī admitted such implications of his position for the question of <u>taklif</u> as well as for the problem of reasoning by analogy, but he did not elaborate it further.

į

i.

Briefly, the concept of maslaha which was originally a general method of decision for jurists and as such a free principle, came to be limited by the opponents of this concept through two considerations. First, there was theological determinism which tended to define maslaha as whatever God commands. Second, there was a methodological determinism which, aiming to avoid the apparent arbitrariness of the method, tried to subject maslaha to givas so as to link it with some more definite basis. Both considerations were inadequate. First, in order to decide that something is maslaha, even to say that God's commands are based on maslaha, some criterion outside these commands has inevitably to be accepted. This was precisely what theological determinism denied. Second, to proceed by <u>qiyas</u>, one must seek the <u>cilla</u>, which was either denied because of theological reasons or was interpreted so as to mean "sign". The implications of this position are obvious. On the one hand, it insisted that further extension of rules must be in units; every new deduction must have a specific link in Shari'a. It denied the extension of law as a whole. On the other hand, it refused to take social needs into consideration, because it insisted upon deducing laws from specific rulings of Sharisa, not even from the general intent of the law.

If we may take general note of major works on usul during the period between Rāzī and Shāțibī, we can see in these works four trends. The first trend refers to those whose conception of <u>maslaha</u> is either dominantly similar to that of Rāzī or who have simply juxtaposed Ghazālī's and Rāzī's definitions of <u>munāsib</u> and <u>maslaha</u>. Among Mālikī jurists Shihāb al-Dīn \int_{1}^{d} Qarāfi (684/1285)⁴¹, and among Hanafi's Şadr al-Sharī'a Maḥbūbi (747/1346)⁴², stay closer to Rāzī. Accepting Rāzī's criticism of <u>maslaha</u>, Qarāfi even went further. He raised serious doubts whether 43 maslaha could ever be defined and justified in clear terms.

1

Jamāl al-Dīn Isnawī (771/1370)⁴⁴ and Tāj al-Dīn Subkî (771/1369)⁴⁵ combine Ghazālī and Rāzī. Sa<d al-Dīn Jaftāzānī (792/1290)⁴⁶ interprets the Hanafī position, mainly that of Pazdawī (482/1089), in reference to Rāzī.

The second trend refers to those jurists who reject <u>maşlaha</u> <u>mursala</u> as a valid basis of reasoning. In this category fall the Shāfi^ci jurist Sayf al-Din Amidi (631/1234)⁴⁷ and the Māliki, Ibn Hājib (646/1249).⁴⁸ In their arguments against <u>maşlaha</u> <u>mursala</u> both follow Ghazāli rather than Rāzi. To them <u>maşlaha</u> is acceptable only if it is textually supported.

The third trend is illustrated by the Shāfi⁴ i jurist, ⁴⁹ Izz al-Din Ibn ⁴⁹ Abd al-Salām (660/1263). He was inclined towards t<u>asawwuf</u>. ⁴⁹ There is a noticeable inclination towards sufficient interpretation of law in his treatment of the concept of <u>maslaha</u>. This needs a detailed observation.

To Ibn ^CAbd al-Salām <u>maşlaha</u> means <u>ladhdha</u> (pleasure) and <u>farah</u> (happiness) and the means leading to them. ⁵⁰ The <u>maşālih</u> are then divided into two kinds, <u>maşālih</u> of this world and the maşālih of the hereafter. The former can be known by reason, while the latter can only be known by <u>maql</u> (tradition, revelation). ⁵¹ In view of the people's knowledge, however, <u>maşālih</u> differ according to the level of the approach of the people. The lowest level of <u>maşālih</u> is that which is common to all men. Higher than this is the level on which the adhkiyā' (the wise people) conceive

<u>maşāli</u>h. The highest level is peculiar to the <u>awliyā' Allāh</u> (friends of God, sufis) alone. The <u>awliyā'</u> and <u>aşfiyā</u> prefer <u>maşālih</u> of the hereafter to those of this world. "The reason is that the <u>awliyā'</u> are anxious to know His commands and laws \angle in their reality \angle , hence their investigation and reasoning (<u>ijtihād</u>) is the most complete one". ⁵²

Elsewhere, Ibn ^CAbd al-Salām divides <u>maşālih</u> as "rights" into two major divisions. First are the Rights of God, and second, the Rights of men. The Rights of God fall into three categories: rights which belong purely to God such as <u>ma^cārif</u> (gnosticism) and <u>aḥwāl</u> (mystic states); second, rights which combine rights of God and those of men such as <u>Zakāt</u>; and third, those which combine rights of God, and of His Prophet, and of the people in general. The rights of men are also of three categories: rights of <u>nafs</u> (self), rights of men toward each other, and rights of animals toward men.⁵³

The above references, which are recurrent themes in his Qawā^cid al-Anām, indicate that lbn ^CAbd al-Salām's legal thinking was deeply influenced from a mystic viewpoint. For instance, he did not reject <u>huqūq al-nafs</u>, but a <u>maşlaha</u> aiming at the realization of such rights was lower in rank than one which aimed at <u>ma<rifa</u> and aḥwal. In fact, Ibn 'Abd al-Salām represents the stage where the Şūfī conception of <u>maşalih</u> came to permeate <u>uşūl al-fiqh</u>. It is not possible at this point to go into details of the Şūfī conception of human <u>maşālih</u> and its history. It must, however, be pointed out that at a very early stage in sufism, rejection of <u>huzūz al-nafs</u> (pleasures of the animal soul) became significant as a means of controlling the <u>nafs</u>. In Sarrāj's (378 A.H.) <u>al-Luma^c</u>, <u>huzūz al-nafs</u> are frequently opposed to <u>huqūq al-nafs</u>. ⁵⁴ <u>Zuhd</u> is defined as abandoning the <u>huzūz</u>. ⁵⁵ The <u>huqūq</u> are defined as <u>ahwāl</u>, <u>maqāmāt</u>, <u>ma'ārif</u>, etc. ⁵⁶

ł,

-

<u>Huzuz</u> had its apparent connection with <u>maşāli</u>h, and more particularly, with the question of <u>rukhşa</u> (legal allowance) in case of hardship. The <u>Suff</u> stress on <u>zuhd</u>, <u>wara</u> and <u>ikhlās</u> required abandoning of <u>huzuz</u>. An obvious example of this encroachment of <u>taşawwuf</u> on fiqh and <u>usul al-fiqh</u> may be seen in Qushayrī's <u>waşiyya</u> (will) to his disciples where he advised them against opting for such allowances because "when a <u>faqīr</u> falls down from the level of <u>haqīqa</u> to that of <u>rukhşa</u> of <u>Sharī</u>'s, he dissolves his covenant with God and violates the mutual bond between him and God." ⁵⁷

Closer to the period of Shāṭibī, the opposition to <u>huẓūẓ</u> appears still stronger. Abu'l-Ḥasan al-Shādhilī (656 A.H.) with whom Ibn ⁶Abd al-Salām's connections are claimed, ⁵⁸ used to define <u>tawhīd</u> (unification) in terms of abandoning the <u>huẓūẓ al-nafs</u>. ⁵⁹ He also explained it as a curse from God when someone is found indulging in the <u>huẓūẓ</u> so as to be barred from <u>^cubūdiyya</sub> (servitude). ⁶⁰</u> Ibn 'Abbād al-Rundī (792/1390), the famous Shādhili Ṣūfi, with whom Shāṭibī was in correspondence on matters relating to <u>taṣawwuf</u> and <u>fiqh</u>, also stressed the rejection of <u>huṣūṣ</u>. Commenting on the <u>Hikam</u> of Ibn 'Aṭā Allāh, Ibn 'Abbād said that "the <u>nafs</u> always seeks <u>huṣūṣ</u> and runs away from <u>huqūq</u>; hence if you are confused in two matters, always choose what is harder for the <u>nafs</u>".

Elsewhere, commenting on the <u>hikma</u>: "The coming of <u>fāqāt</u> (trial by wants and needs) is a happy occasion for the disciples", Ibn ^cAbbād explained that the <u>Sūfī</u>, contrary to a common Muslim, finds pleasure by losing his <u>huzūz</u>. Situations of neediness provide a disciple with purity of heart, which is not achieved by <u>sawm</u> (fasting) or <u>salāt</u> (praying), because in <u>sawm</u> and <u>salāt</u> there is a possibility of <u>hawā</u> (desire) and <u>shahwa</u> (lust).

The Şūfī view of obligation to God, thus, had serious implications for <u>maşlaha</u> in terms of human utility. It not only denied human interest as a basis of consideration, but also insisted on abandoning human interests to purify the obligations as "complete obedience to God". These implications were not generally recognized by the jurists. Ibn ^cAbd al-Salām accepted the Sūfī view, but in his attempt at synthesis between the two he was led either to deny <u>maşālih</u> of this world altogether, or to accept the two on separate grounds.

The fourth trend is represented by Ibn Taymiyya (728/1328) and Ibn Qayyim al-Jawziyya (751/1350). Ibn Taymiyya tried to find a middle way between the two extremes of total rejection and total acceptance of maşlaha. He considered maşlaha <u>mursala</u> similar to the methods of <u>Ra'y</u>, <u>Istihsān</u>, <u>kashf</u> (mystic revelation) and <u>dhawa</u> (mystic taste) of whose validity he was suspicious, ⁶⁴ and hence rejected them. On the other hand, he refuted the moral implications of the denial of <u>maşlaha</u> to the commands of God.

Ibn Taymiyya also counts <u>maşlaha</u> <u>mursala</u> as one of the seven ways of knowing the commands of God, along with the traditional sources of law. He defines <u>maşlaha</u> mursala as follows:

" (It is a decision) when a <u>mujtahid</u> considers that a particular act seeks a utility which is preferable, and there is nothing in <u>shar</u> that opposes this (consideration). 65

Ibn Taymiyya, however, concludes that to argue on the basis of <u>maşlaha mursala</u> is to legislate in matters of religion, and God has not permitted this. To do so is similar to <u>istihsān</u> and <u>tahsīn <aql</u>ī. ⁶⁶ He admits that <u>Sharī</u> a is not opposed to <u>maşlaha</u>, but when human reason finds <u>maşlaha</u> in a certain case where there is no supporting citation in the text to be found, only two things are meant. Either there definitely is a Text which the observer does not know or one is not dealing with a <u>maşlaha</u> at all. ⁶⁷ The obvious assumption in Ibn Taymiyya's arguments is that all the possible <u>maşālih</u> are already given in the Text. The other assumption is, of course, that all of God's commands are based on <u>maşlaha</u>. The latter assumption is of particular significance to Ibn Taymiyya, as it has to do with the moral responsibility of man, a matter which he stressed very much. He condemned both the Mu<tazila and the Jabriyya in reference to the question of <u>maşlaha</u>. The Mu<tazila argued that God is

i

ĺ

obliged to command only what is good for man. They conceived God's actions as analogous to man's actions. They assumed that whatever is morally obligatory for man must be obligatory for God. Ibn Taymiyya refuted this. But he also refuted the Jabriyya position that God's commands are not based on <u>maşlaha</u>. He questioned their assumption that the intention of <u>maşlaha</u> is a limitation upon God's acts. The Jabriyya argued that a command does not necessitate will (<u>irāda</u>). Ibn Taymiyya saw in this argument a theological advantage, but morally such a doctrine was harmful. Ibn Taymiyya, therefore, set out to analyse this generally-accepted doctrine. He clarified that in reference to God there are two kinds of wills (<u>irāda</u>), $\frac{\alpha l - \alpha}{\sqrt{1 - \alpha a}} \frac{\alpha l - \alpha \alpha}{\sqrt{1 - \alpha \alpha}} \frac{\alpha l - \alpha$

The consideration of <u>maslaha</u>, or as Ibn Qayyim, following Ibn Taymiyya, often calls it, <u>Siyāsa</u>, plays an important part in explaining legal obligations, legal reasoning and legal change in Ibn Qayyim's <u>I<Iām al-Muwaqatīn</u>. He expounds the principles of Hanbalī <u>fiqh</u>, and enumerates the following five as sources and principles: (1) <u>Nusūs</u>, (2) the <u>Fatāwā</u> of the companions of the Prophet, (3) selection from the opinion of the companions, (4) <u>Hadīth Mursa</u>l (a report of a saying of the Prophet which lacks a link in the chain going back to the Prophet.) (5) <u>Qiyās l'il-darūra</u>.⁶⁹ Thus it is in reference to the three sources that the consideration of <u>maslaha</u> is expounded. Ibn Qayyim explains that it is valid to attribute <u>filla</u> to the commands of God, because the Qur'ān and the <u>Sunna</u> of the Prophet themselves are replete with examples where reasons are given to explain the command.⁷⁰ The larger part of the <u>I<</u>Tām set. on certain reasons which he calls hikma or maslaha.

The following passage contains a clear statement of his views on maslaha. In a chapter where he explains how "fatāwā may change according to the change in

times and places, etc...", he says:

 (\mathbf{r})

"This chapter is of great utility. Out of ignorance, grave errors have been committed regarding the <u>Sharī'a</u>, which have caused hardship, difficulty and obligations that are not required by <u>Sharī'a</u>; as is known, the magnificent <u>Sharī'a</u>, which keeps the highest level of public interest, does not bring forth these things. The foundations of <u>Sharī'a</u> are laid on the wisdom and on the interest of the people in this world and in the hereafter. <u>Sharī'a</u> is all justice, kindness, interests [of the people] and wisdom. Hence any case which departs from justice to injustice.... from <u>maslaha</u> to <u>mafsada</u>, is not part of <u>Sharī'a</u> even though it has been imposed by literal interpretation [of the texts of <u>Sharī'a</u>]." 71

The fifth trend is illustrated by Najm al-Din Λ^{el} awfi (716/1316). He justified the use of <u>maşlaha</u> even to the extent of setting aside the text. He stressed <u>maşlaha</u> as the basic and overriding principle of <u>Shari'a</u>. <u>Maşlaha</u>, therefore, prevails over all other methods such as <u>ijmā</u>.⁷² Țawfi regards maslaha as a fundamental principle.

Tawfi's preference of maşlaha over against texts and <u>ijmā</u> was also prompted by his belief that textual sources as well as the opinions on which <u>ijmā</u> is claimed were diverse, inconsistent and often self-contradictory. The principle of <u>maşlaha</u> provided a consistent method of decision.⁷³ Tawfi, however, did not elaborate on a concrete criterion of <u>maşālih</u>, how they are to be decided, especially in a case where there is a question of choosing among more than one <u>maşlaha</u>. He goes on to the extreme of suggesting a decision by drawing of lots.⁷⁴ To sum up, the concept of maslaha with its simple beginnings unfolded its various aspects as it came into contact with theology, tasawwuf, logical analysis and, most significantly, with social and legal changes. Theological determinism introduced by Ash'ari jurists appears largely in the discussion of taklif. To Ash'aris, obligation is created by divine command. The Mu^ctazila refuted this sense of They differentiated between two senses of obligation: theological determinism. taklif and wujub, the latter was rational and ethical, while the former was theological.⁷⁵ In other words, mere command does not oblige man to act; it What obliges man is the knowledge of good and bad, or of only informs him. useful and harmful. Commenting on this position, G.F. Hourani concludes that this interpretation should have been acceptable to the legal concept of obligation. Yet there were certain complexities. First, if legal obligation is based on one's knowlege of utility, it may lead to arbitrariness, and furthermore this criterion in its absolute sense is not universally applicable. All the things which are apparently useful also have certain elements which are harmful either to the person concerned or to others. Second, all the rules of Shari a do not conform to the rule of utility; there are obvious hardships and disadvantages in obeying them. Third, to preserve an order and a system the decision of utility cannot be left to the individual; who should then decide?

Ņ

Still another aspect of the relationship of <u>maşlaha</u> and <u>taklif</u> was brought forth by Sūfīs. The consideration of seeking utility and avoiding harm leads one to view obligation in a formal sense. Whenever there is a choice between hard and soft, a

270

maşlaha-oriented person chooses the latter. Not only that, to avoid harm to himself, one seeks devices which are legal; and since he is a utility seeker, he feels satisfied by escaping the full implications of legal obligation. To Şūfis, this attitude, even in its lawful aspects, was quite opposite to the meaning of obligation towards God. They opposed this attitude as <u>huzūz</u> of <u>nafs</u> (lower soul) who is one of the enemies of the traveller on the path to God.

Shāțibi tried to find an answer to above questions. He concentrated on the concept of <u>maşlaha</u> itself, in contrast to other jurists who focussed on <u>maşlaha</u> <u>mursala</u>. At a point where Shāțibi rejects the connection of the method of reasoning by <u>maşlaha</u> with <u>bid</u>^{ca}, we find an elaborate discussion of why and how he did not agree with the general understanding of the term <u>maşlaha</u> <u>mursala</u> by other jurists.⁷⁶

Refuting the association of <u>maşlaha mursala</u> and <u>bidéa</u> as maintained by some jurists, ⁷⁷ Shājibī asserts that the two are completely opposed to each other. ⁷⁸ He argues that first of all the jurists are not agreed upon the definition <u>of maşlaha mursala</u>. Even Ghazālī expressed two different views on this point. ⁷⁹ Secondly, Shājibī explains, that <u>munāsib mursal</u> which is neither specifically support ed by the legal texts nor is it rejected, is not a <u>bidéa</u>. One finds in it two categories. First, where <u>munāsib</u> <u>mursal</u> agrees with the general function (<u>taşarrufāt</u>) of Sharīća, <u>Maslaha mursala</u> belongs to the second category. The validity of the first category is limited. On the contrary, <u>maşlaḥa mursala</u> is supported by the existence of the genus which is common between <u>maşlaḥa mursala</u> and Shari a, and this genus is considered valid by <u>Sharīća</u>. This validity is not based on a specific evidence but on its consideration as a whole.⁸⁰ Shāţibī illustrates maṣāliḥ mursala with ten examples. Among them are the following: the collection of the Qur'ān; determining the penalty for using intoxicants; allegiance to a less qualified person for an office in the presence of a better qualified one.⁸¹ He finds three elements common in all the ten examples. First is the element of suitability with the objectives (maqāşid) of Sharīća.⁸² <u>Maşāliḥ mursala</u> do not conflict with the fundamentals or with the evidences of Sharīća. Second, they are rationally intelligible. <u>Maşāliḥ mursala</u> do not belong to <u>ta<abbudāt</u> because the latter are not rationally intelligible in detail. Shāţibī gives more than ten examples to prove this point.⁸³ Thirdly, <u>maşāliḥ</u> <u>mursala</u> refer to the following principles: protection of (human) necessities; removal of impediments which are harmful to religion; and protection of an indispensible means to the end of law.⁸⁴

Shāţibī, thus, shows that the acceptable <u>maşālih</u> cannot be equated with <u>bid</u> and that they are not limited to the category of <u>darūr</u>ī, as some jurists have maintained; they cover other categories as well. In fact, the above explanation of <u>maşlaha mursala</u> conforms to Shāţibī's concept of maslaha which is of fundamental significance to his doctrine of maqāşid al-sharī^ca.

Shātibi's doctrine of maqāsid al-Shari'a is an attempt to establish maslaha as an essential element of the ends of law. He treats the problem of the relativity of maslaha, the relationship of taklif and maslaha, huzūz and maslaha in sufficient detail. He tries to refute the implications of theological determinism and the dilemma of the relativity of maslaha first by suggesting study this problem

ì

on two levels. On the first level he discusses the maqāsid of the lawgiver and on the second level he deals with the maqasid of the <u>mukallaf</u> (subject of law). By proposing that <u>maşlaha</u> is the objective of the lawgiver on the first level, he suggests that it is the legislator who decides what is <u>maşlaha</u>. Still, Shāţibī stresses that this decision is not final for all times to come. But the objective of the <u>mukallaf</u> (the subject of law) which also includes the legislator insofar as he is <u>mukallaf</u>, is obedience.

N.

The scheme of Shātibi's discussion of magasid is as follows:

- 1. Qasd of the Shāri^c (lawgiver and legislator)
 - i) First aspect: The primary intention of the lawgiver in instituting law as such.
 - Second aspect: His intention in instituting it so as to be intelligible (ifhām).
 - iii) Third aspect: His intention in instituting it to demand obligation (taklif).
 - iv) Fourth aspect: His intention in including the <u>mukallaf</u> under its command.
- II. Qasd of the mukallaf.

į

1

The discussion in the first aspect deals with <u>maşlaha</u>, its meanings, grades, characteristics and its relativity or absoluteness. The second aspect discusses the linguistic dimension of the problem of <u>taklif</u> which was overlooked by other jurists. A command constituting <u>taklif</u> (obligation) must be understandable by all of its subjects, not only in words and sentences but also in its linguistic and cultural meaning. Shāţibī discusses this problem by explaining two terms: <u>dalāla aşliyya</u> (essential meaning) and <u>ummiyya</u> (intelligible to commonality). The third aspect analyses the notion of <u>taklīf</u> in reference to <u>qudra</u> (power), <u>mashaqqa</u> (hardship) etc. The fourth exposes the aspect of <u>huzūz</u> in relation to <u>hawā</u> and <u>ta<abbud</u>.

On the second level, i.e. that of <u>mukallaf</u>, Shāṭibī is largely discussing the question of intention and acts.

For details we turn to the next chapter.

į

í

NOTES: CHAPTER VI

1.	See:	Lane,	An Arabic-Ei	nglish Lexicon,	Book IV	(London:	Williams N	Norgate,
	1863-93), pp.1714-1715.							

- The Qur'ān: III, 114. Rāghib Işfahānī, <u>Al-Mufradāt fi Gharib al-Qur'ān</u>, (Karachi: Tijārat Kutub, 1961), p.286, also confirms that in the Qur'àn Salāh is often opposed to fasād and sayyi'a.
- R. Paret, "Istiķsān and Istişlāķ" Shorter Encyclopedia of Islam (Leiden: Brill, 1961), p.185.
- 4. As quoted by Mustafa Shalabi, Ta<li al-Ahkam...(Cairo: Azhar, 1949), p.292 ff.
- 5. Vide Ibid., pp.285 ff.
- 6. Abul Husayn Başrī, <u>Al-Mu^ctamad fī Usül al-Fiqh</u>, Vol. II (Dimashq: al-Ma^chad al^cIlmī al-Firansi, 1964), p.888.
- 7. Ibid.

Į.

- 8. Ibid., p.805.
- 9. Ghazālī, Al-Mustasfā min ⁽Ilm al-Usūl, Vol. 1 (Baghdād: Muthannā, 1970), p.286-87.
- 10. Ibid., p.284.
- 11. Ibid., p.290.
- 12. <u>Ibid.</u>, p.294-295.
- 13. Ibid., Vol.II, p.306.
- <u>Ibid</u>., Vol.1, pp.274, 284, following Shāfi^ci, Ghazāli says:
 من استعسرت نقده شرع محلها ان من استعسرت نقده شرع p.315.
- 15. <u>lbid</u>., Vol.1, pp.284-315.
- 16. Ibid., pp.56-57.
- 17. Ibid., p.60.
- 18. lbid., p.87.

ł

- 19. Ibid., Vol. II, p.230.
- 20. Ibid., Vol.11, p.295ff.
- 21. Ibid., p.297.
- 22. Ibid., Vol.11,p.306.
- 23. Ibid., Vol.1, pp.5–7. Ghazālī complains that Transoxian jurists, such as Abū Zayd have tried to bring too much figh into usul al-figh.(p.10).
- 24. Ibn Khaldūn, <u>Muqaddima</u> (Cairo: Bulāq, 1320 A.H.), p.431.
- [•] 25. Fakhr al-Din Rāzi, <u>al-Maḥṣūl fi Uṣūl al-Fiqh</u>, Ms. Yale University, Nemoy, A -1039 (L-643).
 - 26. Ibid., part 11, f.87a.
 - 27. lbid.
 - 28. Ibid.
 - 29. Ibid., part 1, f.9a.
 - 30. Ibid., f.13a.
 - 31. Ibid., part 11, f.90b.
 - 32. Ibid., f.91b.
 - 33. Ibid., f.92a.
 - 34. Ibid., f.92b.
 - 35. Ibid., ff.92-97.
 - 36. Ibid., f.97b.
 - 37. Ibid.

í

- 38. Ibid., f.88b.
- 39. Ibid., ff.87-90.
- 40. Most probably this is the ibar to which Shātibi refers. See Ch. VIII, p. 368.

- 41. Qarāfī, <u>Tanqī́h al-Fuşūl fī́ Ilm al-Uşūl</u>, in <u>Al-Dhakhīra</u> (Cairo: Matbas kulliya al-Sharīʿa, 1961), pp.144-46.
- 42. Şadr al-Shari'a Mahbūbi, Al-Tawdih wa'l-Tanqih, Vol.II (Cairo: Matba' Bosanawi, 1304 A.H.), pp.536-540.
- 43. See p. 299.

an 15

- 44. Jamāl al-Din Isnawi, Nihāyat al-Su²al, commentary on Baydāwi's Minhāj al-Wuşūl, on the margin of Ibn Amir al-Hājj, <u>Al-Taqrir wa l-Tahbi</u>r (Cairo: Bulāq, 1317 A.H.), Vol.III, pp.134-140.
- 45. Taj al-Din Šubki, Jam^c al-Jawāmi^c, in ^cAbd al-Raḥmān,Bannāni, Ḥāshiya... <u>ʿAlā matn Jam^c al-Jawāmi^c, Vol.</u>11 (Cairo: Muṣṭafā Bābi, 1937), pp.270-285.
- 46. Sad al-Dîn Taftazanî, <u>Sharh al-Tawdih wa'l-Tanqih</u> (Cairo: Bosanawi, 1304 A.H.), Vol.II, pp.548-683.
- 47. Sayfal-Din, Ámidi, <u>Al-Ihkām fi Uşulal-Ahkām</u>, Vol. IV (Cairo: Matbas Masārif, 1914), p.215-217.
- 48. Ibn Häjib, Mukhtasar Muntahā al-Usūlī, Vol.II (Cairo: Būlāq, 1317 A.H.), p.289.
- 49. Ibn ^CAbd al-Salām was initiated into the Suhrawardiya Tarīqa. He is also claimed to have joined the Shādhiliya. His relationship with Ibn ^CArabī has, however, been a subject of dispute. For details see Ridwān ^CAlī Nadwi, <u>Al-^CIzz b. ^CAbd</u> al-Salām (Damascus: Dār al-fikr, 1960), pp.103-110.
- 50. Ibn ^cAbd al-Salām, Qawā^cid al-Aņkām fī Maşāliņ al-Anām, Vol. I (Cairo: Istigāma, n.d.), p.10.
- 51. Ibid., p.6.
- 52. lbid., p.24
- 53. lbid., p. 129
- 54. R. A. Nicholson (Ed. Comm.), in Abu Naşr al-Sarrāj, <u>Kitāb al-Luma^e fīⁱl</u> <u>Taşawwuf</u> (London: Luzac, 1914), p. 134. For this opposition of terms cf. Abū Bakr/Kalābādhī, <u>al-Ta^carruf li Madhhab Ahl al-Taşawwuf</u> (Cairo: Îsā al-Bābī, 1960), p.23. Najm al-Dīn Kubrā, <u>Fawā'ih al-Jamāl</u>, ed. Fritz Meier, (Wiesbaden: Steiner, 1957), p.71 (Introduction), pp. 5-6 (Text).

- 55. Sarrāj, op.cit. p.47. (quoting Şūfi Ruwaym b. Aḥmad.)
- 56. Ibid., p. 336.
- 57. Abū'l Qāsim al-Qushayrī, <u>al-Risāla fī 11m al-Taşawwuf</u> (Cairo: Muḥammad ʿAlī, 1948), p.181.
- 58. See above, n . 48.
- 59. ^cAbd al-Halim Mahmūd, <u>Al-Madrasa al-Shādhiliyya wa Imāmuhā Abū'l Hasan</u> al-Shādhilī, (Cairo: Dar al-Kutub al-Haditha, 1387 A.H.)
- 60. Ibid., p.137.
- 61. Ibn ^cAbbād al-Rundī, <u>Sharh al-Hikam l'il-Imām Abī'l-Fadl Ahmad b.</u> Ajā'Allāh al-Iskandarī (Cairo: Mustafā Afandī, 1320 A.H.), p.99.
- 62. Ibid., p.106.
- 63. See p. 300.
- 64. Ibn Taymiyya, Qā'ida fi'l Mucjizāt wa'l karāmāt wa Anwā' khawāriq al- 'Ādāt in Majmū'a al-Rasā'il wa'l Masā'il, Vol.V (Cairo: Matba Manar, 1349 A.H.), p.22.
- 65. Ibid.
- 66. Ibid., p.23.
- 67. Ibid.
- 68. Ibid., p.30
- 69. Ibn Qayyim, I'lām al-Muwaqqi'in, Vol.1 (Cairo: Sa'āda, 1955), pp.29-32.
- 70. Ibid., pp. 197 ff.
- 71. Ibid., Vol.III, p.14.
- 72. Najm al-Din al-Țawfi, Sharh al-Araba in, included as appendix to Mustafă Zayd, al-Mașlaha fi'l Tashri'al-Islâmi wa Najm al-Din al-Țawfi (Cairo: Dâr al-fikr al-ʿArabi, 1954), p. 18.
- 73. lbid., pp.35-37.
- 74. Ibid., p.47.

í

- 75. G. F. Hourani, Islamic Rationalism, The Ethics of [<]Abd al-Jabbar (Oxford: Clarendon, 1971), p.57, 115, 118.
- 76. Shāțibī, <u>Al-Istişām</u> (Cairo, 1332 A.H.), Vol.II, pp.95-140.
- 77. Shātibī, here, refers to Mālik whose reliance on masālih mursala is strongly criticised by other jurists. Shātibī defends Mālik in the following manner

Mālik, adhering to the principle of not applying rational explanations in matters of 'ibādāt "revolves entirely around his /approach to/ stop at the limits prescribed by <u>Shari'a</u>, /and thus/ disregarding what <u>munāsib</u> requires.../This is/ in contradiction to 'ādāt which are governed according to suitable reason (al-ma'nā al-munāsib) which is evident to human reason ('uqūl). He employed laxity (istirsāl) with self-confidence and with deep insight in reasoning by maslaha.../He employed this laxity so frequently/ that the scholars often condemn him because of this laxity. They imagined that Mālik threw off the yoke / of <u>Sharī'a</u>/ and opened the gate of law-making. How far is it /from truth /!" Al-l'tiṣām, p.113.

78. lbid., p.115.

Ĺ

1

- 79. lbid., p.96.
- 80. lbid., p.98.
- 81. Ibid., pp.99-110.
- 82. Ibid., p.111.
- 83. <u>Ibid.</u>, For instance, Shātibi explains, <u>Shāri</u>s prescriptions about cleaniless from human excretions are not rationally uniform. In case of urine and stools one is obliged only to wash certain parts of one's body, (i.e. ablution), but in case of nocturnal discharge, washing of the whole body is obligatory. <u>Ibid.</u>, p.111.
- 84. <u>Ibid</u>., pp.113-115.

CHAPTER VII

SHAŢIBĪ' S DOCTRINE OF MAQASID AL-SHARIS

K. 🖗

- be

As we have said elsewhere, in order to appreciate Shāţibī's concept of <u>maşlaḥa</u>, one must study it within the structure and formulation he himself devised. This chapter, therefore, aims at presenting Shāţibī's concept of <u>maşlaḥa</u> as it emerges from his philosophy. The concept will be developed according to his own formulation. For this reason, the present chapter is almost entirely based on Volume Two of Shāţibī's <u>Al-Muwāfaqāt</u>, which is wholly devoted to an exposition of <u>Maqāşid</u> al-Sharī<a.

It must be made clear that the following is neither a translation nor a summary of the said volume. We have summarized only those discussions from <u>Al-Muwāfaqāt</u> which, in our opinion, are relevant to our problem. To keep Shāţibī's structure of analysis intact, his method of dividing and subdividing the concept into its various components, has been faithfully followed.

The scheme of Shātibi's discussion of <u>Maqāşid</u> has been given in the previous chapter. Accordingly the present chapter is divided into five sections which analyse the following concepts and terms: <u>maşlaha</u>, <u>dalāla</u>, <u>taklīf</u>, <u>ta<abbud</u> and niyya.

As a preamble to the exposition of the <u>maqāşid</u>, Shāṭibī states that the whole of the discussion is based on a generally accepted premise which is theological in

 \overline{z}

Ĩ

origin. The premise is that God instituted <u>sharā'i</u> (laws) for <u>maşālih</u> (benefits, good) of the people, both immediate and future. (6)¹. There exists, however, a difference of opinion among scholars concerning the details of this premise. For details of the discussion Shāţibī refers his reader to <u>lim al-Kalām</u>. For the purpose of this dissertation, however, the point needs to be explained briefly.

The <u>mutakallimūn</u> (theologians) accept the general and apparent meaning of the premise of <u>maşālih</u>, yet they differ from one another if <u>maşālih</u> are understood in terms of <u>(Ilal</u> (pl. of <u>(Ila</u>)) meaning "causes" or "motives". The Ash arī theologians reject explicit as well as implicit causality in reference to God. For them, the premise implies that God is obliged by the consideration of <u>maşālih</u> to act in a certain way. Since such an obligation proposed limitation on God's omnipotence, the Ash arīs reject the idea that <u>maşālih</u> are the <u>(ilal</u> of <u>sharā'i(</u>). They, however, accept the premise by interpreting <u>maşālih</u> to be the 'grace' of God, rather than the 'cause' of his acts. On the other hand, the Mu tazila, even though they too maintained God's omnipotence, yet believed that God is obliged to do good. Consequently they accepted the above premise, regarding maşālih as <u>(illa of sharā'a</u>).

The theological disagreement initially concerned God's acts, but it was extended to God's commands in the Qur'ān as they constitute His acts of speech. Thus the theological disagreement manifested itself in <u>Uşūl al-fiqh</u> as well. Theological arguments penetrated into <u>uşūl al-fiqh</u> also because a number of writers on uşūl were theologians.
<u>Uşūl al-fiqh</u>, however, required a manner of thinking and a method of reasoning different from that of <u>kalām</u>. Legal thinking necessitated that the volition for voluntary human acts must be attributed to man himself if man is to be held legally responsible for his acts. Since obedience to Divine Commands thus depends on human volition, the Command must be shown to be motivated by the consideration of human interests. Consequently, the premise of <u>maṣālih</u> must be accepted in uṣūl in terms of "cause", "motive" and "purpose".

The premise of <u>maşālih</u> came to be generally accepted in <u>usul</u>. Some usuliyyin, such as Ghazāli and others, in order to be consistent with their theological views, redefined the term <u>cilla</u> so as to rid it of the connotation of "causality" and "motivation" in which sense it was used and disputed in <u>Kalām</u>. Passing from <u>Kalām</u> to <u>usul</u>, the term <u>cilla</u> thus underwent a semantic change. For the explanation of the meaning it acquired in <u>usul</u>, we now turn to Shātibi.

Shātibi argues that the premise of <u>maṣālih</u> is established in <u>Sharira</u> by the method of induction, both as a general theme and by the evidence of the description of

the <u>'ilal</u> of various commands in detail. For instance, the Qur'ān explains the reasons for ablution, fasting and <u>jihād</u> as being cleanliness, piety and eradication of oppression, respectively. (7)

ان **مىد** را س راب

After explaining this premise, Shātibī proceeds to discuss the details of the <u>maqāsid</u>, which are analyzed in five aspects; four in relation to the lawgiver, and one in relation to the <u>mukallaf</u> (subject of law).

SECTION ONE

MAŞLAHA, THE FIRST MAQŞID OF SHARICA

The primary objective of the lawgiver is the <u>maşlaha</u> of the people. The obligations in <u>Shari'a</u> concern the protection of the <u>maqaşid</u> of <u>Shari'a</u> which in its turn aims to protect the <u>maşalih</u> of the people. Thus <u>maqaşid</u> and <u>maşlaha</u> become interchangeable terms in reference to obligations.

Shāṭibī defines <u>maşlaḥa</u> as follows: "I mean by <u>maşlaḥa</u> that which concerns the subsistence of human life, the completion of man's livelihood, and the acquisition of what his emotional and intellectual qualities require of him in an absolute sense". (25)

This is the definition of <u>maşlaha</u> in its absolute sense. Shāṭibī, however, takes into account various other senses in which <u>maşlaha</u> can be studied. The <u>maşālih</u> belong either to this world or to the world hereafter. Further, these <u>maşālih</u> can be seen as a system; belonging to different grades and with a definable relationship with each other.

The second element in the meaning of <u>maşlaha</u> is the sense of protection. Shāţibī explains that <u>be Sharīka</u> deals with the protection of <u>maşālih</u> either in a positive manner as when to preserve the existence of <u>maşālih</u>, <u>Sharīka</u> adopts measures to support their bases. Of in a negative manner, to prevent the extinction of <u>maşālih</u> it adopts measures to remove any elements which are actually or potentially disruptive of maşālih. (8) Shāţibī divides <u>maqāşid</u> into <u>darūrī</u> (necessary), <u>bāj</u>ī (needed) and <u>taḥsīnī</u> (commendable). The <u>darūrī maqāşid</u> are called necessary because they are indispensable in sustaining the <u>maşāli</u>h of <u>Dīn</u> (religion and the hereafter) and <u>Dunyā</u>, in the sense that if they are disrupted the stability of the <u>maşāli</u>h of the world is disrupted. Their disruption results in the termination of life in the world, and in the hereafter. It results in losing salvation and blessings. (8)

Ŋ,

The <u>darūrī</u> category consists of the following five: <u>Dīn</u> (religion), <u>Nafs</u> (self), <u>Nasl</u> (family), <u>Māl</u> (property) and <u>'Aq</u>l (intellect). (10)

Scholars, says Shāṭibī, have observed that these five principles are universally accepted. Analyzing the aims of the <u>Sharīʿa</u> obligations, we find that <u>Sharīʿa</u> also considers them as necessary. The <u>Sharīʿa</u> obligations can be divided from the viewpoint of positive and negative manners of protection into two groups. Falling into the positive-group manner are <u><ibādāt</u> (rituals, worship), <u><ādāt</u> (habits, customs) and <u>mu<āmalāt</u> (transactions), and falling into the negative group are jināyāt (penalties).

<u>Clbādāt</u> aim at the protection of <u>Din</u> (religion). Examples of <u>Cibādāt</u> are belief and the declaration of faith (the Unity of God, the Prophethood of Muḥammad), <u>salāt</u>, <u>zakāt</u>, <u>siyām</u> and <u>haij</u>. <u>Cādāt</u> aim at the protection of <u>nafs</u> (self) and <u>Caql</u> (intellect). Seeking food, drink, clothing and shelter are examples of <u>Cādāt</u>. <u>MuCāmalāt</u> also protect <u>nafs</u> and <u>Caql</u> but through <u>Cādāt</u>. Shāṭibī defines <u>jināyāt</u> as that which concerns the above five <u>maṣāliḥ</u> in a preventive manner; it prescribes the removal of what prevents the realization of these interests.

. .

To illustrate <u>jināyāt</u>, he gives examples of <u>qişāş</u> (retaliation) and <u>diyāt</u> (blood money) for <u>nafs</u>, and <u>hadd</u> (punishment for drinking intoxicants) for the protection of <u>(aql.</u> (8-10)

₹ ₹ 25

> The <u>hājiyāt</u> are so called because they are needed in order to expand (<u>tawassu</u>^c) the purpose of the <u>maqāşid</u> and to remove the strictness of literal sense, the application of which leads to impediments and hardships and eventually to the disruption on the <u>maqāşid</u>(objectives). Thus if the <u>hājiyāt</u> are not taken into consideration along with the <u>darūrī maqāşid</u>, the people on the whole will face hardship. The disruption of <u>hājiyāt</u> is, however, not disruptive of the whole of <u>maşāliḥ</u>, as is the case with the <u>darūriyāt</u>. Examples of <u>hājiyāt</u> are as follows: in <u>cibādāt</u>, concessions because of sickness and because of travel which otherwise may cause hardship in prayers, fasting, etc.; in <u>cādāt</u>, the lawfulness of hunting; in <u>mucāmalāt</u>, permission for <u>qirād</u> (money lending), <u>musāqāt</u> (agrarian association) and in <u>jināyāt</u>, allowances for weak and insufficient evidence in de cisions affecting public interest. (10-11)

<u>Taḥsīnāt</u> means to adopt what conforms to the best of practices (<u>cādāt</u>) and to avoid those manners which are disliked by wiser people. This type of <u>maşlaḥa</u> covers noble habits (ethics, morality). Examples of this type are as follows: in <u>cibādāt</u>, cleanliness (<u>tahāra</u>) or decency in covering the parts of the body (<u>satr</u>) in prayer; in <u>cadāt</u>, etiquette, table manners, etc.; in <u>mucāmalāt</u>, prohibition of the sale of unclean (<u>najis</u>) articles or the sale of surplus food and water, and depriving a slave of the position of witness and leadership, etc.; for <u>jināyāt</u>, prohibition of killing a free man in place of a slave, etc. (11-12)

286

Shātibi regards the above division of <u>maşālih</u> as a structure consisting of three grades, connected to one another. His detailed analysis reveals two aspects of their relationships with one another. First, every grade separately requires annexion of certain elements which supplement and complement this grade. Second, every grade is related to the others. (12)

Every one of the three grades requires certain elements to achieve the fuller realization of its objectives. For instance, $\underline{qis}\overline{as}$ (retaliation) cannot be realized without a condition of $\underline{tam}\overline{a}\underline{thul}$ (parallel evaluation). This position, however, calls for two clarifications: first, a lack of these complementary elements does not amount to a negation of the essential objectives; second, the consideration and realization of the complementaries must not bring about a negation of the original objectives – that is to say, if the consideration of a complementary results in the annulment of the original objective, its consideration will not be valid. The reasons for this stipulation are, first, because the complementary elementary element is like a quality (\underline{sifa}). If the consideration of a quality results in the negation of the the consideration of the consideration is negated as well. Second, even if it is supposed that the consideration of the original objective, its is the complementary results in the realization of its interests at the cost of the original objective, it is stressed that the realization of the original be preferred. (14)

The above situation is illustrated by the following example. The eating of carrion is allowed in <u>Sharica</u> to save life. The reason is that the preservation of life is of the utmost importance, and preservation of <u>murura</u> (manliness, honour) is only complementary (<u>takmili</u>) to the protection of life. Impure things are prohibited

201

in order to preserve honour and to encourage morality. But if the preservation of the complementary, i.e. to preserve honour by avoiding eating impure things, leads to the negation of the original interest, i.e. the preservation of life, the consideration of the complementary is forsaken.

. .

Another example may be seen in the act of sale which is a <u>darūrī maşlaha</u> while the prohibition of risk and ignorance in sale transactions is complementary. If the complete negation of risk is stipulated, the result will be complete negation of the act of sale.

The relationship of the above three grades of <u>maşālih</u> with one another is the same as that of the complementary <u>maşālih</u> to the original objective of the law. The <u>tahsīniya</u>t are thus complementary to the <u>hājiyāt</u> which are complementary to the <u>darūriyāt</u>. The <u>darūriyāt</u> are the fundamentals of <u>maşālih</u>. In view of the above explanation, Shāțibī deduces the following five rules in this relationship:

- 1. The darūrī is the basis of all masalih.
- 2. The <u>ikhtilal</u> (disruption) of <u>darūri</u> necessitates the <u>ikhtilal</u> of other masalih absolutely.
- 3. The <u>ikhtila</u> of other <u>masalih</u>, however, does not necessitate an ikhtilal of and within, the <u>darūri</u> itself.
- 4. In a certain sense, however, the <u>ikhtilāl</u> of <u>taḥsīnī</u> or <u>ḥājī</u> absolutely necessitates the ikhtilāl of <u>darūrī</u>.
- The preservation (<u>muḥāfaẓa</u>) of <u>ḥājī</u> and <u>taḥsīnī</u> is necessary for the sake of darūrī. (16-17)

These rules may be illustrated by the rule of <u>qişāş</u> (lex talionis). <u>Qişās</u> is <u>darūri</u>, and <u>tamāthul</u> (consideration of equality) in <u>qişāş</u> is <u>taḥsini</u> and <u>takmīlī</u>.

. .

To illustrate the first rule, <u>tamāthul</u> (<u>taḥsīnī</u>) is complementary and exists only because of <u>qiṣāṣ</u> (<u>darūrī</u>). Thus a <u>darūrī maṣlaḥa</u> (<u>qiṣāṣ</u>) is the basis of a <u>taḥsīnī maṣlaḥa</u> (tamāthul).

To illustrate the second rule, if there is no <u>qişāş</u>, there is no consideration of <u>tamāthul</u>. In other words, the <u>ikhtilāl</u> of the <u>darūrī</u> means the same for the other grades of <u>maşālih</u> necessarily.

To illustrate the third rule, the <u>ikhtilāl</u> of <u>tamāthul</u> does not require <u>ikhtilāl</u> of qi<u>s</u>ās.

The fourth and fifth rules can be appreciated if one grasps the sense in which <u>darūri</u> is affected by the <u>ikhtilāl</u> of other <u>maşālih</u>. Shāṭibī explains the effect of other <u>maşālih</u> on <u>darūri maşālih</u> with the following four similes:

- 1. The relationship of other <u>maşālih</u> to <u>darūrī maşālih</u> is like that of protective zones (<u>himā</u>). The interruption (<u>ikhlāl</u>) of one protective zone amounts to the interruption of the next zone and eventually to the disruption of the <u>darūrī maşālih</u> which are at the centre of these zones.
- 2. This relation may also be understood as that of the part and the whole; other masalih together with the darūri masalih make one whole. The disruption of the parts obviously means the same as the disruption of the whole.

÷ .

. . .

- The hājiyāt and tahsiniyāt can be understood as individuals in relation to the universal, i.e. darūriyāt.
- 4. The hājiyāt and tahsiniyāt serve the darūri masālih as a prerequisite (muqqaddima), or as associates (muqārin). (16-24)

As mentioned above, the <u>masālih</u> are also divided into those belonging to this world and those which concern the Hereafter.

First are the <u>maşālih</u> of this world. There are two angles from which the <u>maşālih</u> of this world can be observed. The first angle is to observe them as they actually exist, and the second is to observe them on the basis of <u>shars</u> proclamation.

١,

Examining <u>maşālih</u> as they exist in this world, we see that they are not found as pure <u>maşālih</u>. Rather, they are mixed with discomfort and hardship, however big or small, and which may precede, accompany or follow the <u>maşālih</u>. Similar are the <u>mafāşid</u> (opposite of <u>maşālih</u>) which also are not pure but are found to be mixed with a certain amount of comfort and enjoyment. The whole phenomena in this world point to the fact that this world is created from a combination of opposites and that it is impossible to abstract (<u>istikhlāş</u>) only one side. The proof of the matter is the completely universal experience of this fact. It is for this reason that the <u>maşālih</u> and <u>mafāsid</u> in this world are known only on the basis of the pre-dominant side; if the side of <u>maşlaḥa</u> dominates, the matter at issue is considered, customarily a <u>maşlaḥa;</u> otherwise a <u>mafsada</u>. In these matters, thus, the determining factor is the prevalent practice (<u>cāda</u>). (26) It must be noticed here that this principle is applicable only to acts relating to <u><āda</u>, and only to the determination of <u>maşlaha</u> or <u>mafsada</u> in this world through knowing them as they exist. Acts which are not <u><ādāt</u> are not affected by this principle. (26)

The second approach to considering the <u>maşālih</u> of this world is to observe them in reference to their connection with Shar^c i proclamation (Khiţāb). The basic rule in this approach is that the <u>maşālih</u> or <u>mafāsid</u> as taken into consideration by the <u>Shāri</u>^c are pure. If they are supposed to be mixed (<u>mashūba</u>), they are not so in the reality of <u>shar</u>^c. (27) As explained above, <u>maşlaḥa</u> or <u>mafsada</u>, in this world, is determined by the predominant side (<u>al-jiha al-ghāliba</u>) of a matter. It is the predominant part which is the object of Shar^c i proclamation. The dominated (<u>al-maghlūba</u>) part, whether <u>maşlaḥa</u> or <u>mafsada</u> is not the objective of the Shāri^c. Why is it then that the dominated elements, even though they may be <u>maşlaḥa</u>, are not the objectives of <u>sharī</u>^ca, still be <u>maşlaḥa</u>? Shāṭibī solves this apparent contradiction with the following explanation.

He argues that <u>maşlaha maghlūba</u> is that which is considered as such according to the acquired habitude (<u>al-l<tiyād al-kasbī</u>) alone, i.e. without adding the <u>Shar<ī</u> requirements of <u>maşlaha</u>. Customarily such a <u>maşlaha</u> is not considered worth seeking. This is the part of <u>maşlaha</u> which is also not the objective of the lawgiver insofar as the <u>shar<iyya</u> (legality) of rules (ahkām) as a whole is concerned.

• • •

Further, if the dominated part were also taken into account by the <u>Shāri</u>, no act could have been the subject of command alone or of prohibition alone. Obviously such is not the case. If it is supposed that the dominated part in a mixed <u>maşlaha</u> is the object of prohibition and the dominating part that of command, then one and the same act becomes an object of command and prohibition at one and the same time, which would have been a <u>taklīf mā lā yutāq</u> (impossible obligation) as well as an absurd situation. (28)

The above explanation, however, does not clarify the existence or occurrence of <u>mafsada</u> despite the <u>Shāri</u>'s intention to the contrary. Shātibī elaborates the matter further by saying that the above position may appear to be that of the philosophers' and the Mu^ctazila on the existence and occurrence of evil. According to the philosophers, God created a world in which the good is mixed with evil. It is the good, however, which is the purpose of creation. He did not create the world for evil, even though evil may occur along with the good.

The Mu<tazila believed that evils are not intended to occur; their occurence is against God's will (irāda).

Shāṭibī first refutes the apparent similarity between his and the above positions. He argues on the basis of a distinction between two intentions (<u>qaşd</u>) of God. First there is the intention of creation (<u>al-qaşd al-khalqī al-takwīnī</u>) and second the intention of legislation (<u>al-qaşd al-tashrīčī</u>). The positions of the philosophers and the Mu<tazila concern the former and Shāṭibī's the latter. As he argues, the occurrence of <u>mafsada</u>, despite God's will and intention for <u>maslaha</u>, is justifiable in the case of <u>dasd al-tashric</u>, because a man is held free (<u>mukhtār</u>) so as to be legally responsible for his acts. This position is not justified in the case of <u>ac-irāda takwīni</u>, as this would imply imperfection in God's powers. (30).

18 C

The above discussion of <u>maşlaha</u> has been concerned with the cases where the actual practice may be used as the basis of determining a <u>maşlaha</u>. There are cases where the judgment of habitude is not so definitive. For instance, eating carrion in case of dire need and killing a murderer for the prevention of crimes, are considered <u>maşlaha</u> despite the fact that the acts themselves are not so. In other words, unlike the cases in the above discussion where the acts, despite their consisting of certain aspects of <u>mafsada</u>, are regarded as <u>maşlaha</u> in themselves, become <u>maşlaha</u> because of certain external considerations. The supposition in this case is that the external consideration can dominate the internal consideration. How this domination is decided needs elaboration.

In view of the above situation, logically, there are two positions; either both considerations are equal in such a manner that one cannot be preferred to the other, or one of them can be preferred. The former position probably does not exist in <u>Sharī'a</u>, because it necessitates that <u>Sharī'a</u> should intend prohibition and permission simultaneously.

Furthermore, if one side is preferrable, it is still possible that the <u>Shāri</u>^c might have intended the other side. Both sides will always remain to be weighed by a

<u>mujtahid</u>. We are obliged only to do what, after weighing both sides, appears to us (<u>yanqadihu</u>) the intention of the <u>Shāri</u>, not what is intended by the Shāri in reality (in His mind). (31) In this way, after the decision of a <u>mujtahid</u>, the possibility of the other side being intended has to be disregarded insofar as fulfilling an obligation is concerned. The possibility is, however, not finally disregarded insofar as <u>nazar</u> (examination, investigation) is concerned.

A group of Scholars who believed the above possibility to be applicable in the case of obligations as well, maintained the principle of <u>murā^cāt al-khilāf</u>. As mentioned elsewhere, this principle, to Shāṭibī, meant an impossible and hence void obligation.²

Shāţibī sums up the above discussion by saying that <u>al-jiha al-marjūha</u> (the dominated aspect) when it is found mixed with <u>al-jiha al-rājiha</u> (the dominating aspect) is not the objective of a legal obligation. This principle governs all problems which are subject to <u>ijtihād</u> (legal reasoning) irrespective of whether one believes a <u>mujtahid</u> to be always correct or not. Hence reasoning by analogy must go on (<u>al-qiyās mustamirrun</u>) and the demonstrative proof must remain free and unqualified (<u>al-burhān muţlaqun</u>). (32)

So far the discussion has been concerned with the masalih of this world. <u>Masalih</u> of the hereafter are also pure, such as the blessings of paradise, as well as mixed (<u>mumtazija</u>), such as the punishment of hell meted out to believers in the Unity of God. The basic rule in such <u>masalih</u> and <u>mafasid</u> is that they are all determined according to <u>Shari</u>, because the reason has no place in matters relating to the hereafter.

3 1

Sometimes a confusion may arise because of considering the pure <u>masalih</u> or <u>mafasid</u> as mixed. For instance the blessings bestowed upon the prophets in paradise differ from those given to others. Those in lower ranks may be regarded as being punished by the absence of the blessings given to those in higher ranks. According to Shātibî, this confusion arises because a distinction is not maintained between a species and its individual exemplifications. The individuals may differ in special characteristics, etc., but they do not differ in relation to their species; they are all equal as members of the species. This membership is the thing that determines their <u>wasf</u> (quality). (36).

١,

From the above discussions, Shāṭibī deduces the following rules as characteristics of maşlaḥa:

- The purpose of legislation (tashri^c) is to establish (iqāma) masālih in this world and in the hereafter, but in a way that they do not disrupt (yakhtall) the system of Shar^c.
- 2. The Shari' intends masalih to be absolute.
- 3. The reason for the above two considerations is that <u>Shariča</u> has been instituted to be <u>abadi</u> (eternal, continuous), <u>kulli</u> (universal) and <u>Cāmm</u> (general) in relation to all kinds of obligations (<u>takālif</u>), mukallafin (subjects of law) and <u>aḥwāl</u> (conditions, states). (37)

The above three characteristics thus require maslaha to be both mutlaq (absolute)

and <u>kulli</u> (universal). The absoluteness means that <u>maşālih</u> should not be relative and subjective. Relativity is usually based on equating a <u>maşlaha</u> with one of the following: <u>'ahwā' al-nufūs</u> (personal likings), <u>manāfi</u> (personal advantages), <u>nayl al-shahawāt</u> (fulfilment of passionate desires) and <u>aghrād</u> (individual interests). According to Shāțibi all of the above considerations render the concept of <u>maşlaha</u> relative and subjective, which is not the consideration of <u>Shāri</u> in <u>maşlaha</u>, though it may be so in 'āda.

Ŋ,

He argues on the following grounds: First, the objective of <u>Shari'a</u> is to bring the <u>mukallafin</u> out of the dictates of their desires so as to make them servants of God. This objective negates the consideration of personal liking as an element in the consideration of <u>Shāri</u>^c.

Second, <u>masālih</u> cannot be considered as mere <u>manāfi</u> because in <u>cāda</u> as well as in <u>Shar</u> they are mixed with disadvantages. The point of emphasis here is that <u>naf</u> is not essential in the consideration of maslaha – neither in <u>cāda</u> nor in <u>shar</u>.

In <u>cada</u> some higher goal like the subsistence of life forms the basic consideration in determining <u>maslaha</u>. In <u>sharc</u> the consideration must still be higher, and that is the consideration of the hereafter.

Third, the consideration of the fulfilment of personal desires also renders the concept of <u>maslaha</u> highly relative. The consideration of personal desire varies from state to state, person to person, and time to time. It is so relative that it cannot be an essential consideration for determining maslaha.

. .

Fourth, consideration of individual interests leads not only to a divergence but, more significantly, also to a conflict with others and to the deprivation of others' interests.

1 - 2 - 2 - 4 Consequently, relativity and subjectivity are excluded from the <u>shar</u> consideration of <u>maşlaha</u>; it must, therefore, be absolute. In <u>shar</u> this absoluteness is provided by the stipulation that <u>maşlaha</u> must aim at the subsistence of life in this world commensurately with life in the next world.

The second characteristic of maslaha is its universality (kulli). This universality is not affected by the takhalluf (falling short) of its particulars. For instance, the penalties are imposed on the basis of the universal rule that they generally restrain people from committing crimes. Yet, there are people who, despite being punished, do not abstain from committing a crime. Nevertheless, such exceptions do not affect the validity of the general rule about the penalty. (52) In shari a it is al-ghālib al-aktharī (the major dominant) which is the general-definitive element (al-" \overline{a} mm al-qat" \overline{i}) in the consideration of a maslaha. This is the characteristic (sha'n) of inductive universals (al-Kulliyāt al-istiqrā'iyya). An illustration of this universality may be found in the universal rules of a language. The universals of a language are closer to those of sharisa, because both are wadsi (instituted, conventional) not 'aqli (speculative). The inductive universals (in Arabic grammar, for instance) remain valid even if some of their particulars do not conform to the majority of particulars. (52-53).

In reference to the characteristics of maşlaha, Shātibi takes into consideration

the criticisms of this concept by other jurists. Among them he specifically refers to Fakhr al-Din Rāzi, Shihāb al-Din Qarāfi and Ibn^CAbd al-Salām. He has answered their criticism. As these criticisms and answers are quite relevant to the discussion of <u>maşlaha</u>, a brief summary of this debate is given below.

Analysing the position of those who favour <u>maşlaha</u>, Rāzī refers to their argument that the basic rule in <u>manāfi</u> (useful things) is <u>'idhn</u> (permission, lawfulness) and in <u>maqārr</u> (harmful things) is <u>man</u> (abstention). (40). Shāţibī rejects this analysis as an unfaithful representation of the <u>maşlaha</u>-view. It is possible to speak about <u>manāfi</u> and <u>maqārr</u> only in absolute terms as they do not exist as absolute in reality; actually they are largely relative. Second, since <u>maṣāliḥ</u> refer to <u>shar^cī</u> proclamations which take into consideration the differences among persons, times and states, it is inadequate to talk in absolute terms. Third, since no <u>manāfi</u> are to be found that are not mixed with <u>maqārr</u>, if we accept Rāzī's principle, we will have also to accept that <u>'idhn</u> and <u>nahy</u> (prohibition) can apply to one and the same thing – which is absurd.

Shihāb al-Din^AQarāfi, the commentator on Rāzī's <u>al-Maḥşūl</u>, had some doubts about the principle that <u>maşlaḥa</u> constituted the basis of legal obligations. He argued that <u>maşlaḥa</u> cannot be the basis of <u>ibāḥa</u>. This is true, first, because <u>maşlaḥa</u> cannot be realized and hence defined in simple and absolute terms, because no <u>maşlaḥa</u> can be gained without <u>'alam</u> (pain) and <u>mafāsid</u> (evils). Thus to maintain that every mubāḥ must be based on maşlaḥa amounts to a com-

£70

plete negation of <u>mubāh</u>. Second, in order to argue that <u>maşlaha</u> is the basis of obligation, <u>maşlaha</u> must be defined in absolute terms and not in reference to certain specific factors, because this process of the preference of one specific consideration to another is never ending and because it does not provide a universally accepted basis of definition. Furthermore, this position cannot be argued on the grounds that <u>maşlaha</u> is that whose violater is punished by God. This definition is not acceptable because it is based either on the assumption that God punishes only evil and this manner of argument is <u>dawr</u> (arguing in circle) or on the assumption that every obligation from God is a <u>maşlaha</u>, simply because it is an obligation.

Qarāfī adds that the <u>maşlaha</u> view is difficult to maintain for our people (<u>aşhābunā (ash'arīs?</u>)), as well. They cannot say that God takes <u>maşlaha</u> into consideration over against <u>mafsada</u>, because there are many <u>mubāhāt</u> in which this consideration is lacking. The only proof they have is an argument on the basis of the induction of the obligations, and this also is based on a claim to know the <u>asrār</u> (secrets, rational explanation) of <u>fiqh</u>. They are necessarily thus led to the position thatGod's actions, commands and considerations are entirely dependent on His will and nothing else. The Mu^ctazila are also led to the same conclusion. (42)

To answer Qarāfī, Shāṭibī refers to his own discussion of the relativity of <u>maşlaha</u>. Second, he answers that a survey of the rules of <u>Sharīća</u> by the method of induction is claimed to have proved that sharīća has taken into considerati on what is

_ , ,

١,

regarded as <u>maşlaha</u> in customary practice as well. He argues that such a survey on the basis of the method of induction provides the <u>dawābit</u> (determining factors) of <u>maşlaha</u>. The examination of the events by way of induction where the <u>takālīf al-s harīka</u> (legal obligations) have been realized in practice shows that these <u>takālif</u> and <u>mubāhāt</u> did not harm human interests (or <u>masālih</u>) but have conformed to them and established them.

Ibn ^cAbd al-Salām distir. Juished between <u>maşāliķ</u> <u>al-dār al-ākhira</u> and <u>al-maşāliķ</u> <u>al-dunyawiyya</u> on the basis that the former can be known only by <u>shar</u> while the latter are known by needs, experience, custom and by considerations of probability. He even says that when one wants to know a <u>maşlaķa</u>, he may simply consider it rationally, supposing that the <u>shāri</u> has given no indication. Judgment is reached rationally in this manner except in the case of <u>ta</u> <u>abbudāt</u> where maşāliķ or mafāsid are not given.

Shāṭibī, quoting Ibn Abd al-Salām here, probably to indicate his disagreement, refers to him not by name but by terms such as <u>bard al-nās</u> (some person) and

<u>hādhā al-qā'il</u> (this speaker). To Shāţibī <u>maşālih</u> in the hereafter are not independent of <u>maşālih</u> of this world. Hence not $only_{A}^{el}$ <u>maşālih</u> <u>ukhrawiyya</u> but also the <u>dunyawiyya</u>, as long as they are obligations, are known by <u>shar</u> alone. If the distinction between the two <u>maşālih</u> were absolute, the <u>shar</u> would have been concerned only with <u>al-maşālih</u> <u>ukhrawiyya</u>. In fact, to realize the <u>ukhrawiyya</u>, the establishment of the <u>dunyawiyya</u> is inevitable. Shāţibī refutes the implication in Ibn Abd al-Salām's statement that the <u>dunyawiyya</u> are rational and hence the consideration of shar only supplementary. (48)

300

Į.

SECTION TWO

. . .

AN ANALYSIS OF THE TERMS DALALA AND MACNA

The preceding section discussed the first aspect of <u>maqāşid</u> which focussed on <u>maşlaha</u> as being the primary objective of law. In the present section Shāţibī goes further to argue that the second <u>maqsid</u> of Sharī^ca is its intelligibility; <u>Sharī^ca</u> was revealed in such a manner that it was to be intelligible for every <u>mukallaf</u>. Although Shāţibī does not say so explicity, his analysis of <u>dalāla</u> develops an argument against the Zāhirīs and the Hadīth-group who discouraged any interpretation of <u>Sharī^ca</sub> on the basis of <u>maşlaha</u>. Zāhirīs attach more significance to the letter of the law (<u>lafz</u>: words) than to the spirit of the law (<u>ma^cnā</u>: meaning). Shāţibī, on the contrary contends that it is the meaning which is important, and not the word. Thus, he indirectly leads to the conclusion that interpretation of Sharī^ca by maşlaha serves to fulfil the objectives of <u>Sharī^ca</sub>.</u></u>

The idea of <u>Sharīća</u> being universally intelligible has been accepted generally. There have been, however, some points which had posed some difficulty for the scholars. One such point was the question of foreign words in the Qur'ān. Generally, the jurists found it necessary to reject the foreign origin of these words in order to maintain that the Qur'ān was revealed in pure Arabic. Before proceeding to discuss his theory of <u>dalāla</u> (indication of words to meaning), Shāṭibī first discusses the problem of foreign words in the Qur'ān.

Shātibi opens his discussion by analysing this very fact of revelation in Arabic.

He explains that in the claim 'that : sharifa is all Arabic and there is nothing a'jami (foreign) in it', the point of emphasis is not whether there are foreign words in the Qur'an or not. Unfortunately, many a jurist has understood the problem in this sense. In fact, the point to be stressed is that Qur'an was revealed in the language of the Arabs as a whole, and it is in this general sense that Shari'a aims to be understood. It was revealed in such a manner that the particular words and styles of expressing the meanings are the same as used and understood by the Arabs. For instance the Arabic language uses 'āmm (general) sometimes to mean zāhir (apparent), sometimes to mean cāmm in one sense and khāss in another sense, and sometimes to mean khāss only. The Qur'an follows the same styles of expression. In other words every language has particular styles of expression, and styles of one language cannot help in understanding another language. The language of the Arabs cannot be understood on the basis of the language of non-Arabs. Similarly, the language of Arabs cannot help in understanding non-Arab languages. Shafi^ci noticed this aspect of shari^ca and stressed its significance for usul al-figh, but the later jurists have generally disregarded this aspect. (66) Shātibi retakes from Shāfi'i and develops the theme of the universality of the understanding of shari'a by an analysis of the meaning-indication process in the Arab language.

í

Shāțibi's discussion of the universality of the intelligibility of <u>Shari'a</u> does not seem to solve directly the contradiction which emerges in case of those who know no Arabic. We may, however, infer from the general trend of his argument two levels of the universality of intelligibility which may serve as an indirect answer to the question. On the first level the universality of intelligibility is confined to Arabs. Shāţibī maintains that <u>Sharīća</u> is cast in a language which is understood by all Arabs and it is in this sense that it is Arabic. On the second level <u>Sharīća is universally intelligible</u>, even by the non-Arabs. Here, intelligibility refers to a more special sense of 'meaning'; it does not refer to the indication by words, syntax or grammar. This is the special sense of 'meaning' in which the meaning is separated from words, syntax, grammar, etc., and thus, actually disconnected from any language. In this state of abstraction they are ready to be understood by speakers of all languages. These meanings are ready to be translated into other languages. This 'meaning' nevertheless, initially comes from the first level of intelligibility which is achieved from the context of a speech in a particular language.

Shāțibi calls the process, which indicates this special 'meaning', $\frac{d}{d}alāla a silvya}{\frac{d}{d}alāla}$ which may explain how Shāțibi proposes that Sharīća can be understood even by those who do not know Arabic. $\frac{d}{d}alāla a silvya}{\frac{d}{d}alāla}$ is explained in detail as follows:

The Arabic language, insofar as it consists of words to express meanings, has two aspects:

First, the absolute aspect of its words and expressions which denote absolute meanings. This denotation is $\frac{d}{dal\bar{a}la} \frac{d}{a\bar{s}liyya}$ (essential denotation). Second, the limited aspect in which the words and expressions denote subsidiary meanings. This denotation is $\frac{d}{dal\bar{a}la} \frac{d}{dal\bar{a}la}$ (subordinate denotation).

The first aspect is common to all languages and is the ultimate aim of a speaker. For instance, if A performs a certain action, let us say standing, all languages can state this fact. Although with different words; yet all languages will state the same fact. It is in this aspect that statements in one language can be translated into another. This is the sense in which one speaks of universal understanding of a language. (66)

ĺ,

í

The second aspect concerns particular languages, in this case Arabic. The statement in the above example, "Qāma Zaydun", will vary depending on the emphasis on subject, predicate, condition, context and on the variations of styles. As examples may be given the following: Zaydun qāma; <u>lnna Zaydan qāma</u>; <u>Wallāhi inna Zaydan qāma</u>; <u>Qad qāma Zaydun</u>; <u>Zaydun qad qāma</u>; <u>lnnamā qāma</u> Zaydun, etc. (67).

1

These kinds of variations, though they change the meanings and emphasis in a statement, are, nevertheless, not the original objective (<u>al-maqşūd al-aşlī</u>) of the speaker, but rather they are supplementary and ameliorative to the essential meaning. This, however, does not mean that they are to be disregarded. Rather they are to be taken together with the first aspect of indication as attributes (<u>awşāf</u>) of the essential meaning. These attributes depend on the essential meaning and will be disregarded if the essential meanings exist no more or are disrupted. (68)

To satisfy the requirements of universality and absoluteness in the comprehensibility of <u>Shari'a</u>, it is necessary not only to confine the comprehensibility to the essential meaning as evident from the context, but also to the fact that the meanings so found must accord to Arab usage. For this, the following two aspects may be considered as determinative factors: first the Arab usage in word-meaning relationship and second, the Arab intellectual background. The consideration of Arab usage is so essential that "if the Arabs have an incessant custom in their language, it cannot be validly disregarded in the comprehensibility of <u>Sharīća</u>, and if there is no such custom even then it is not valid to adopt for its comprehension something which is not well known to them (Arabs)". (82)

i,

The Arab usage in this regard is that the words are not followed slavishly in their indication of meaning. The Arabs do not confine themselves to one and the same word, and the replacement of words does not seem to affect their statements. The above fact can be illustrated by the following examples.

The Arabs often disregard the general rules of language. For instance, they frequently employ the styles of poetry in prose, even though such a style is not required and despite the fact that it is contrary to prose styles. What is significant to note, however, is that customarily such a deviation does not seem to affect the speech. (83)

Second, one of the characteristics of Arab usage is that they frequently replace original words with their synonyms, and this practice is not considered to imply contradiction or confusion in speech as long as the intended idea ($al-ma^{r}n\bar{a}$ $al-maqs\bar{u}d$) subsists. The seven readings of the Qur'ān are examples to this effect.

Further, a number of evidences are found in the transmission of verses. For instance, Ibn al-A^crābī (d.848), the famous linguist, once recited:

Wa <u>mawdi'in zīrin</u> lā <u>'urīdu mabītahū</u> ka'annī bihī min shidda (t) <u>al-raw'i ānisū</u>

(I do not want to spend night in a place of <u>zīr</u> (like a conical jar), as if, because of intensive fright, I am familiar with it).

One of his listeners corrected, reminding him that on another occasion he had recited 'wa mawdi'in diqin' (a narrow place) instead of 'wa mawdi'in $z\bar{i}rin'$. Ibn al-A'rābi replied regretting that the enquirer had been with him for such a long time and yet did not know that ' $z\bar{i}r'$ and ' $d\bar{i}q'$ are one and the same. (84)

Arabic Poetry has been transmitted according to varying reports and with a diversity of words. On the whole, one learns that the Arabs do not strictly adhere to particular word specifically so as to regard synonymous words as weaker and defective. The few exceptions from this usage belong to peculiar cases where only one meaning is possible. (84).

The Arabs often disregard part of the grammatical rules of a word, although never as a whole. An example of such disregard is the subtle rules (<u>al-abkām</u> <u>al-latifa</u>) which the words demand according to theoretical analogy (<u>al-qiyās</u> <u>al-nazarī</u>) but which are, nevertheless disregarded. To illustrate, Shāţibī says that the words "<u>`amūd</u>" and "<u>ya`ūd</u>", and "<u>sa`īd</u>", strictly speaking, do not rhyme, yet they are often used to rhyme in Arabic poetry. The reason is that the Arabs' aim for the refinement of their language does not lead to a pedantic concern (taʿammuq) for these rules. (84). The best appreciated piece of literature, according to the Arabs, is that which avoids unnecessary artificiality. When a poet is found indulging in refinement of his diction the is no more regarded as worthy to be followed. (84)

To sum up, Arab usage pays more attention to meaning than to words, because "the word is only a means to reach the desired meaning, whereas the meaning is the goal". (87)

Y.

It must, however, be noted that not all the meanings of a word are intended at one time. Shāṭibī makes a distinction between <u>al-ma^cnā</u> <u>al-ifrādī</u> (single meaning), and <u>al-ma^cnā</u> <u>al-tarkībī</u> (contextual meaning). The <u>ifrādī</u> is disregarded whenever it does not agree with the latter. (87)

The purport of the above discussion of meaning is Shāţibī's contention that neither the words, nor even their abstract meanings are the goals of language in a speech. It is rather the meaning obtained within a context, written or oral, which is the goal. It is this sense of meaning, i.e. $\frac{\lambda t}{\sqrt{dalāla}}$, which according to Shāţibi assures the universal intelligibility of speech within the circle of the speakers of a certain language.

The second consideration for universal intelligibility is the consideration of the intellectual level of the addressees of a speech. Obligation depends on comprehension in the sense that one cannot be held responsible for more than he can understand. Comprehension, however, does not depend simply on the familiarity of words and meaning, but also on many other things.

The degree of comprehensibility may differ from person to person in specific matters because men are not equal in their individual mental make-ups. They, however, come to agree with one another in general matters, and this is the condition according to which masalih function in this world. (85)

Since <u>Shari'a</u> concerned the <u>masalih</u> of the Arabs who were <u>ummiyyin</u> (unlettered), the <u>Shari'a</u> had also to be <u>ummiyya</u>. Shāṭibi explains that <u>ummiyya</u> means that the Arabs did not possess the sciences of the Ancients (Greeks). Literally, <u>'ummi</u> comes from <u>'umm</u> (mother) to connote one who remains as he was originally at the time of his birth, that is to say, in the state of not yet having learned anything. (69)

To call the Arabs '<u>ummiyyin</u>, however, does not mean that they were completely ignorant and uncultured. On the contrary they did possess certain branches of knowledge such as astronomy, knowledge of weather, history and medicine etc. They also possessed their own code of ethics. (71–79)

This consideration implies that in understanding <u>Sharīča</u>, (particularly as, in the case of exegesis of the Qur'ān, many scholars introduced matters which were not intelligible for the common people), one should not demand more than what <u>'ummīs</u> can generally understand. This consideration would also require that the obligation whether pertaining to beliefs (<u>i<tiqādiyāt</u>) or to actions (<u>camaliyāt</u>) must be within the intellectual capacity of an <u>'ummī</u>. Otherwise, obligations would concern only the élite and not people in general. If an obligation surpassing the intellectual capacity of all were made to apply to

4

× .

i. I se people in general, it would constitute an impossible obligation. Both of these consequences are absurd. This conclusion is strongly supported by the attitude of the companions of the Prophet who did not indulge in speculative discussions. Also in practical matters <u>Sharifa</u> uses commonly observable facts rather than complicated speculations as criteria, as for instance, the rising or setting of the sun rather than an astrologically (or astronomically?) defined schedule of times of prayers. (90)

÷. .

It must, however, be made clear that by insisting on the comprehensibility of <u>Sharī'a</u> to'ummiyyīn, Shāţibī neither claims that everything in the Qur'ān or <u>Sharī'a</u> is and must be understood by an'ummī, nor does he discourage any thinking or action beyond the comprehensibility of an 'ummī. Rather what he stresses is the minimal essential requirement in matters of obligation without which the sense of obligation is not complete. Additional considerations may supplement or ameliorate an obligation but the absence of such considerations does not make it any the less obligatory so long as the minimal essential requirement is present. The question of comprehension is restricted furthermore, to those matters which are relevent to the fundam@ptals of <u>Sharī'a</u> (<u>qawā'id</u> <u>al-Sharī'a</u>) and has no meaning for theological matters (<u>umūr'ilāhiyya</u>). The latter are additional matters which are not primarily obligatory. (91)

SECTION THREE

AN ANALYSIS OF THE TERM TAKLIF: LEGAL OBLIGATION AND PHYSICAL CAPABILITY OF THE MUKALLAF

Ŋ,

4. 11

15

In this section, Shāṭibī discusses the concept of <u>taklīf</u> which is the term used for 'obligation' in <u>Uşū</u>l. Etymologically the term has the connotation of 'toil', 'pain' and 'hardship'. On the other hand the principle of <u>taklīf mā lā yuṭāq</u> (no obligation which is impossible to fulfil), which is theological in origin, does not encourage the literal meaning of <u>taklīf</u> to be extended to its extreme. The discussion of the term <u>taklīf</u>, thus, naturally takes into account both of the above extreme aspects of obligation.

For a definition of <u>taklif</u>, Shāṭibī, therefore, indulges in an analysis of the terms <u>qudra</u> and <u>mashaqqa</u>. According to Shāṭibī <u>qudra</u> is an essential element in the concept of legal obligation. He says that the premise of his discussion of <u>taklif</u> which is again theological in origin, is that the <u>sharṭ</u> (condition) or <u>sabab</u> (cause) of <u>taklif</u> is the <u>qudra</u> of doing that for which one is obliged. Hence, any obligation which is not within the <u>qudra</u> of the <u>mukallaf</u>, is not valid according to <u>shar</u>, though it may be so <u>caglan</u> (rationally). (107)

To define <u>qudra</u>, Shāṭibī chooses to analyse what is considered <u>ghayr maqdūr</u> (that which is not within the power of a man to do) in <u>Uşū</u>. Shāṭibī's term ghayr maqdūr is synonymous to mā lā yuṭāq.

Shātibi observes that ghayr maqdur may be used in four senses. First, it may

refer to those obligations which are impossible to fulfil (<u>mā lā yuţāq</u>), either because they are beyond human capability, as for instance the demand to give up eating or drinking or to command someone not to die, etc., or because the obligations demand something which a man has or does not have because of his individual nature, as for instance the demand for bravery in a man who is a b orn coward. (108–109) Shāţibī also uses the term <u>mā lam yakun dākhilan</u> <u>taḥta kasbihī</u> (that which is absolutely not acquireable by man) to refer to this sense of ghayr maqdūr.

s....

чĩ

The second sense of <u>ghayr</u> <u>maqdur</u> refers to obligations which cannot be fulfilled because of the following grounds:

- (a) Where the obligation concerns acts which depend on other acts in such a manner that the latter acts are means to realize the former. In such cases obligation itself becomes <u>ghayr</u> <u>maqdur</u> without the performance of the latter acts. (109)
- (b) Where an act occurs as an inevitable consequence of a certain other act. This case may seem similar to (a), but, in fact, it is different, because in (a) one has to do a certain act before being able to fulfil the act which is obligatory, while in (b) one does not perform the obligatory act itself and only by performing the precedent act does the obligatory act come to occur inevitably. Shāţibī illustrates his meaning by the example of the obligation to know. Apart from <u>a priori</u> knowledge, other kinds of knowing occur inevitably following <u>nazar</u> (observation, reasoning, syllogism). (111)

The other two senses with which the term ghayr maqdur is associated are mashaqqa (hardship) and haraj (impediment). Shātibī maintains that, strictly speaking, mashaqqa and <u>haraj</u> are not <u>ghayr maqdūr</u>. He explains it by arguing that legal obligations in <u>Shari^ca</u> are related with <u>mashaqqa</u> and <u>haraj</u>, but not with the above-mentioned first and second senses of <u>ghayr maqdūr</u>, and since <u>Shari^ca</sub> is not mā lā yuṭāq, the mashaqqa and haraj are not ghayr maqdūr</u>.

Shāṭibī does not deny the fact that in <u>Sharī</u>ca there are occasions where a command is apparently directed to a certain <u>ghayr maqdūr</u> act, yet he maintains that the close examination reveals that the obligation is not actually related to the <u>ghayr</u> maqdūr act. He elaborates it in the following arguments.

Shāṭibī observes that, as a principle, the realm of <u>ghayr maqdūr</u> is not object of <u>taklīf</u> -- whether in respect to demand or prohibition. If the apparent sense of a <u>shar'ī</u> command is to make <u>ghayr maqdūr</u> obligatory, the command must be understood to refer to a <u>maqdūr</u> act which (or the mention of which) either precedes (<u>sābiq</u>) this <u>ghayr maqdūr</u> as a means or cause, or occurs simultaneously (<u>qarīn</u>) with it or succeeds (<u>lāḥiq</u>) it. To illustrate, the Qur'ānic command: "Do not die but as Muslims (lit. Do not die except if you are Muslims)" (2:122), literally demands not to die, which is <u>ghayr maqdūr</u> to fulfil. Naturally the obligation must be connected with the phrase that follows the actual command, i.e. to be Muslims. (108) This example shows that command may be related with ghayr maqdūr but that ghayr maqdūr is not obligatory.

There are further instances in <u>Shari</u>² where a command is directly related with a <u>ghayr maqdu</u> and even aims at it, yet it does not constitute the actual obligation. In such instances ghayr maqdur is capable of being the object of either

а 24 г.

4

the desire (<u>hubb</u>) or the detestation (<u>bughd</u>) of the <u>Shāri</u>^{ϵ}. Even though the acts which are <u>ghayr maqdūr</u> are neither within the capability of the <u>mukallaf</u> nor within his intention, yet they may be desired by the law-giver. To illustrate, Shātibī refers to the above-mentioned example of the obligation to know. If the object of knowing is something <u>darūrī (a priori)</u>, then there is no action involved to fulfil the obligation. In other cases, the knowing is a result of some other act, and even then it necessarily and immediately follows the act of arranging the premises. In short, the act of knowing itself is <u>ghayr</u> maqdūr and yet desired by the <u>Shāri</u>^{ϵ}. (111)

In the latter category of ghayr maqdur, Shātibi refers, in fact, to acts which are involuntary, being fitri and idtirāri and musabbab. (110, 112)

Shāţibī's argument is that such <u>ghayr maqdūr</u> acts as mentioned above, are not object of obligation, though they are desired by the law-giver. The fact that they are desired is proven either in literal expression by the law-giver to such effect or by his making it subject to <u>Jazā'</u> (reward and punishment). (112) On this point Shāţibī's position rather appears puzzling. How an act despite being the object of <u>Shāri</u>'s desire and subject to <u>Jazā'</u>, be not the object of obligation?

Shātibi explains his position in the following manner.

The jurists have taken three positions in answer to the above question. One group has held that the reward and punishment do not concern <u>ghayr maqdūr</u>. Another group believes that reward and punishment both attach to <u>ghayr</u> <u>maqdūr</u> at the same time. In contrast to these groups, Shāțibī maintains that either reward or punishment attaches to <u>ghayr maqdūr</u> to the exclusion of the The first group argues that since <u>ghayr maqdur</u> acts are subject to obligation, they are not subject to reward or punishment. If there is no obligation, there can be no reward or punishment. (115)

Shāṭibī refutes this argument by rejecting the assumption of the necessary relationship of reward and punishment to obligation. He illustrates his view with examples showing that there are obligations which entail no reward or punishment. (117-118).

Another argument advanced in favour of the first position proceeds by showing contradiction in the second position. This argument is as follows. Reward and punishment, if their connection with <u>ghayr maqdur</u> acts be accepted, will either concern the acts in question in their essence or in terms of related acts. If reward and punishment concern their essences, then no distinction is possible between one act and another and between reward and punishment. Consequently, both reward and punishment may concern one and the same act at the same time, which is impossible. If reward and punishment concern related acts, instead of essences, then the point is proven that in neither case does reward and punishment concern <u>ghayr maqdur</u> themselves. (115)

Shātibī refutes this argument by showing that by not maintaining a distinction between reward and punishment in respect of one and the same act, the above argument implies that one and the same act can be the object of both desire and detestation of <u>Shāri</u>^c at the same time, which is absurd.

He argues further that reward and punishment cannot be supposed to be concerned

314

× .

4 -

with related acts, in this case to the exclusion of the act in question. If a connection between <u>ghayr maqdur</u> act and related act is necessary for reward and punishment, the meaning is that <u>ghayr maqdur</u> act is certainly effective in determining reward and punishment. (118)

Shāţibī, therefore, concluded that an act even though not object of obligation may still be subject to reward. Also, that being a subject to reward does not make an act to be the object of obligation Thus a ghayr maqdūr may be desired or rewarded, yet it does not mean that it is obligatory. To be obligatory, an act must be maqdūr.

From here, Shāţibī proceeds to an analysis of <u>mashaqqa</u> and <u>haraj</u> which, he maintains, are not to be equated with <u>ghayr maqdūr</u> in the senses which have been discussed so far. <u>Mashaqqa</u> and <u>haraj</u> make an act hard and difficult, but they are capable of being object of obligation. Shāţibī, however, lays stress that acts consisting of <u>mashaqqa</u> and <u>haraj</u> may be object of obligation, yet <u>mashaqqa</u> and <u>haraj</u> are not objectives of obligation for their own sake. Shāţibī develops his views in a detailed analysis of the term <u>mashaqqa</u>.

Mashaqqa

<u>Mashaqqa</u> is often confused with <u>ghayr maqdūr</u>. The discussion below contends that a distinction among <u>taklif mā lā yuṭāq</u>, (<u>ghayr maqdū</u>) and <u>mashaqqa</u> must be observed. <u>Sharīća</u> aims at none of them <u>per se</u>, but it does impose the latter though not the former. (119) This discussion calls for an investigation into the meaning of mashaqqa.

4

a 14

Literally, $\underline{sh}-\underline{q}-\underline{q}$ as in <u>shaqqa</u> <u>calayya</u> <u>al-shay</u>?(the matter became difficult for me), denotes something "tiresome" and "hard". The Qur'ān says, "You could not reach it save with great trouble to yourselves (<u>bi</u> <u>shiqq</u> <u>al-anfus</u>) (17:7). This meaning when taken in the absolute sense -- without reference to its conventional (<u>wad</u><u>ci</u>) meaning in Arab usage -- acquires five particular technical (<u>istilāḥiyya</u>) senses. These five senses, in fact, stem from three considerations: (1) from the general literal sense of the word <u>mashaqqa</u>, (2) from the viewpoint of <u>cāda</u> i.e. whether a certain act is considered <u>mashaqqa</u> by <u>cāda</u> or not, and (3) from the concept of <u>taklīf</u> itself i.e. a <u>mashaqqa</u> is so neither in its literal sense nor in its customary sense but is rather derived from the concept of obligation itself. These three viewpoints provide the following five senses of mashaqqa.

- 1. First, in a very general sense, <u>mashaqqa</u>, applies to all meanings of "toil" or "trouble" disregarding their being <u>maqdūr</u> or not, or being real or metaphorical. It is in this sense that <u>taklīf mā lā yuţāq</u> is also called <u>mashaqqa</u>, because in order to fulfil a command which is supposedly <u>mā lā yuţāq</u> man puts himself into vain trouble. For instance if a man tries to fly in the air his attempt will be in vain. But here a distinction must, however, be recognized; "flying in the air" is called <u>mā lā yuţāq</u> not <u>mashaqqa</u>; <u>mashaqqa</u> is rather the effort made to achieve the end (i.e. flying...). Thus it becomes obvious that even linguistic usage associates mashaqqa with maqdūr only.
- In the second sense <u>mashaqqa</u> is applied to acts which are extraneous to the <u>mu^ctad</u> (customary). That is to say to perform these acts

.

2 ...

means to incur hardship upon oneself. For instance, to observe fasting during sickness or a journey is not according to <u><u><u></u></u><u>ada</u>, and thus it incurs <u>mashaqqa</u>. It is here that the <u>Shari</u><u>a</u> makes certain allowances which are called rukhsa by the <u>fuqahā</u>.</u>

A ..

3. The third sense of <u>mashaqqa</u> is an extension of the second one. While the second concerns particular acts, the third concerns the totality of actions.

It is persistence in uninterrupted performance of acts, although initially easy to fulfil, that creates <u>mashaqqa</u> and makes them difficult to carry out. In such cases the <u>Shari'a</u> recognizes the principle of <u>rifq</u> (leniency, moderation) by commending the choice of acts which are not tire some.

- 4. In the fourth sense of <u>mashaqqa</u>, the hardship of an act does not result from its being against <u>Gada</u> but rather because it is additional to <u>Gada</u>. In other words customarily it is not <u>mashaqqa</u> but it becomes so because one is obliged to do it. It becomes <u>mashaqqa</u> also because it creates responsibility in addition to the acts required by this worldly life.
- 5. The fifth sense of <u>mashaqqa</u> also flows from obligation, but in a manner different from the fourth. Whereas in the fourth an act is <u>mashaqqa</u> merely because it is an obligation, there being no additional hardship other than this fact alone; in the fifth, there is an additional hardship. The additional element comes about because <u>taklif</u> requires one to reduce (<u>mukhālafa</u>) his own desires which incurs toil and hardship, since hardship is quite evidently seen in prevailing customary practices (<u>fādāt jāriya</u>). (119-121)
These five senses of <u>mashaqqa</u> constitute the framework for investigating whether <u>mashaqqa</u> is included under the requirement of obligation or not. Shāțibi conducts this investigation by analysing the intention of the law-giver, the understanding of the term in <u>cada</u>, and the intention of the <u>mukallaf</u>.

Y.

The first question is whether the <u>Shāri</u> intends <u>mashaqqa</u> or not. There are two kinds of answers to this question. One is given through the <u>Shāri</u> declaration of his own intentions, known through the Qur'ān or tradition. The second may be known through an analysis of the notion of <u>mashaqqa</u> in <u>Shar</u> as distinguished from that in <u>cāda</u>. Both kinds of answers agree on the point that the <u>Shāri</u> does not intend <u>mashaqqa per se</u>. The first kind of answer is manifested in the following:

- a) various statements in the Qur'ān and Ḥadīth categorically deny any intention by the <u>Shāri</u> to impose hardship. (121–122)
- b) the existence of well-known allowances (<u>rukhaş</u>) in <u>Shar</u> prove the existence of concessions to remove hardship. (122)
- c) the consensus on the absence of any intention by the <u>Shāri</u> to make <u>shāqq</u> acts obligatory. If it were supposed that e <u>Sharī</u> did such a thing, it would be guilty of self-contradiction and hence self-negation; e <u>sharī</u> cannot and does not aim at both comfort and hardship. (122-123)

The second kind of answer is sought by investigating the notion of <u>mashaqqa</u> in relation to <āda.

Not every bit of toil and hardship is called mashaqqa in cada. For instance,

seeking one's livelihood through following a craft and trading, although it involves toil (<u>kulfa</u>), is not called <u>mashaqqa</u>. Rather a person is reproached if he avoids such efforts. All states of the human being in this world are toilsome (<u>kulfa</u>), yet they are not called mashaqqa. (123)

Having established this distinction, Shāṭibī points out that <u>mashaqqa khārija an</u> <u>al-mu^ctād</u> is obviously not <u>maqsūd</u> by <u>Shar</u>. Even <u>mashaqqa mu</u>ctāda is not <u>maqsūd</u> by itself in an obligation. It is required rather because the obligation serves the maşlaha of the mukallaf. (124)

There are three possible objections to this position which are discussed in the following lines.

First is the fact that the very term, taklif, which is used as an appellation for

319

∢

these acts entails the meanings of <u>kulfa</u> and <u>mashaqqa</u>. An act is demanded only insofar as it entails <u>mashaqqa</u>, and this is why it is called <u>taklif</u>. Hence <u>mashaqqa</u> is the <u>maqsūd</u> of the <u>Shāri</u>. (124)

2.

. .

Shāṭibī answers this objection by explaining that taklīf can be directed to the mukallaf in two aspects:

(1) First because taklif is mashaqqa and (2) Second because there is an immediate

or forthcoming <u>maşlaha</u> and good to be achieved for the <u>mukallaf</u>. Shājibī obviously favours the second aspect as the only <u>maqsūd</u> of the <u>Shāri</u>[<]. The first cannot be <u>maqsūd</u> because both of these two aspects cannot exist together. The fact of <u>maşlaha</u> being the <u>maqsūd</u> has been established in the first section. Hence <u>mashaqqa per se</u> cannot be <u>maqsūd</u>. Why, then, is an obligation called <u>taklīf</u>? Shājibī answers that, in the usage of Arabs, a thing derives its name from its inseparable attribute, although, in usage, this inseparable attribute is not intended. It is on the basis of this rule of <u>cilm al-ishtiqāq</u> (etymology) that an act is called <u>taklīf</u> because it entails <u>kulfa</u> and <u>mashaqqa</u>, not because <u>taklīf</u> in the sense of <u>Kulfa</u> is the aim or purpose of this act. The consideration of <u>Kulfa</u> is possible only when the term <u>taklīf</u> is applied in a <u>majāzī</u> (metaphorical) sense to a certain act rather than using the term in its <u>haqīqat al-wag</u>^c <u>al-lughawī</u> (the essential posited meaning of a word in a language). (125-126)

(2) The second objection is that the <u>Shāri</u> knows what a <u>taklif</u> is and what it incurs, and since it is known that every <u>taklif</u> incurs mashaqqa it follows that the shāri knows that a taklif incurs mashaqqa. It is, therefore, evident that by imposing a taklif, the Shāri purposes also to impose mashaqqa. (124-125)

~~ .

-

÷.

÷.,

Shāṭibī answers this objection by refuting the equation of the knowledge of <u>sabab</u> and <u>musabbab</u> with <u>qaşd</u> (intention). He argues that even if in this particular case knowledge of <u>sabab</u> and <u>musabbab</u> is considered as intention (<u>qaşd</u>) it would be considered only as leading to the whole; the intention for <u>mashaqqa</u> is only secondary. But even within this supposition the position comes to a contradiction because, even though secondary, the intention for <u>mafsada</u> (<u>mashaqqa</u>) is posited together with intention for <u>manfa^ca</sub> (<u>maşlaba</u>). Hence the Shāri^c does not intend mafsada i.e. mashaqqa. (126–127)</u>

Secondly, it is evident from the Qur'ān and etc. that the <u>Shāri</u> intends to remove hardship. How can it be then maintained that the <u>Shāri</u> intends to impose and remove <u>mashaqqa</u> at one and the same time?

To sum up the discussion, Shāṭibī maintains that: "The obligation of <u>muctadāt</u> and the like does not entail <u>mashaqqa</u> as explained. Hence what necessarily follows from <u>taklīf</u> is not called <u>mashaqqa</u>; irrespective of whether the knowledge of its occurrence necessarily requires it or necessitates the intention for it." (127)

There is, however, another dimension of the problem. Granted that the <u>Shāri</u>⁴ does not intend <u>mashaqqa</u> in his imposition of taklif, should a mukallaf intend

mashaqqa while fulfilling his obligations or not?

ĺ

Shātibi's general answer is in the negative. The <u>mukallaf</u> should not intend <u>mashaqqa</u> because the <u>shāri</u> does not do so and because the <u>mukallaf's</u> intention must correspond to that of the <u>Shāri</u>. Consequently the <u>mukallaf's</u> intention should be concentrated on act rather than on <u>mashaqqa</u>. (128)

In details, however, the problem is more complicated when acts and <u>mashaqqa</u> are looked upon from different points of view.

First, the acts themselves, in this case, can be considered in two categories, those In the latter case, the which are permissible and those which are not so. (133) The problematic matter intention to perform such acts is obviously forbidden. is those acts which are nothing but mashaqqa in themselves but which the Shāri Shātibī maintains imposed as such, as for instance punishment (<uqübāt). that even here the intention of the Shāri^c is not to impose <u>mashaqqa</u> as such, but Accordingly, the to acquire maslaha or to remove mafsada by this mashaqqa. mukallaf's intention must also be maslaha and not mashaqqa as such. This is the reason why if a mashaqqa (such as a half (oath) to give all his property for charitable purposes) contravenes some darūri or hāji principle in din (i.e. the limitation of such a voluntary distribution to only one third of one's property), it will be deemed as void. (149)

Next is the category of acts which are permissible. These are to be considered in relation to mashaqqa whether this mashaqqa is ikhtiyārī (by man's own choice) or <u>idtirārī</u> (imposed on man not by his choice). Another point to be considered regarding <u>mashaqqa</u> is whether it is so called in <u>cāda</u> or not or whether it is extraneous to all such considerations. (133) To simplify, we can divide Shātibī's discussion of <u>mashaqqa</u> into the following 3 categories:

- 1) Ikhtiyārī, where the mukallaf intends mashaqqa by his own choice.
- 2) <u>Idtirārī</u>, where <u>mashaqqa</u> is an inevitable consequence of a certain action.
- <u>Khāriji</u>, where <u>mashaqqa</u> is neither of the above but rather falls upon the <u>mukallaf</u> without having any connection with them.

We will deal with these three categories one by one.

Mashaqqa lkhtiyāriyya

ð_ <

As already mentioned, Shāţibī maintains that since <u>Shāri</u> does not intend <u>mashaqqa per se</u>, one must not seek for <u>mashaqqa</u>. <u>Mashaqqa ikhtiyārīyya</u> therefore, is condemnable according to him. There is, however, one point where one may argue that a <u>mukallaf</u> may intend <u>mashaqqa</u> to augment his reward on the assumption that reward is enhanced in commensuration with the hardship suffered. (125)

Shātibī rejects this kind of reasoning. First, because, to him, the whole concern of taklīf is with action (<amal) and this is also that at which the <u>Shāri</u>^c aims. It is, therefore, action and not <u>mashaqqa</u> which increases reward. (127)

Secondly acts depend on intentions. The intention must, therefore, correspond to the intention of the Shāricso as to produce acts which are intended by the Shāric. To seek <u>mashaqqa</u>, in this case, would be to violate the intentions of <u>Shāri</u>. This violation cannot earn reward. (129)

In opposition to Shāṭibī's view a considerable number of traditions are quoted to the effect that a reward is connected with the hardship of the act, and the more the hardship the greater the reward. (129–130) Second evidence to oppose Shāṭibī is the situation of <u>arbāb al-aḥwāl</u> (sūfīs) who try their utmost to increase <u>fazīma</u> and hardship in rejection of rukhṣa. (130)

Shāțibi refutes these evidences on the following grounds:

- All such reports are <u>akhbār abād</u> and relate only one matter. They do not constitute <u>istiqrā' qaf^cī</u>. Our concern is <u>Qaf'iyya</u> not <u>zanniyya</u>. Hence these <u>zanniyyāt</u> cannot invalidate our position. (130)
- In the final analysis these traditions do not favour the intention of mashaqqa; rather they stress the acts themselves. The intention to bring about mashaqqa is a secondary (tābića) not the primary (matbūća), concern. (130)
- Rather there are traditions in which the Prophet reproached those who opted for hardship. His proscription (<u>nahy</u>) of hardship (<u>tashdid</u>) is so well known in Sharica that it has become a definite principle (<u>aşl qat</u>ci). (132-133)
- 4. As for <u>arbab</u> <u>al-aḥwāl</u>, even in their case it is not correct to say that they intend to bring about <u>mashaqqa</u> only. Their purpose is to disregard their own <u>huzūz</u> (self-considerations) so as to fulfil their duties toward God. Shāṭibī explains this point more fully in the case of <u>haraj</u>. <u>Haraj</u> is an act which causes an impediment in fulfilling the <u>huzūz</u>. The <u>arbāb</u> <u>al-aḥwāl</u> prefer to forego their <u>huzūz</u> in favour of their duty towards God, because of fear or love of God. (132, 147-148)

sŧ

Mashaqqa Idtirāriyya

ie.

 In general terms, hardship can be seen in three ways. First, there is the hardship which has become part of daily life and is no more called <u>mashaqqa</u> but is rather expressed by terms such as <u>kulfa</u>, <u>tacb</u> etc. This is called by Shāţibī <u>mashaqqa muctād</u>. Second, there is the type of hardship which is not habitual. It may not be impossible to bear, but it might be so painful as to be too difficult to endure. This is called by Shāţibī <u>mashaqqa ghayr muctād</u>. The third category lies on the fringes of the second one. In itself it may neither be impossible nor painful to bear, but it becomes an impediment to the performance of other acts. This is called haraj. (133)

According to Shāţibi the first type of <u>mashaqqa</u> is not in question at all because it is, in fact, not considered <u>mashaqqa</u>. The discussion here does concern the second type when the mukallaf chooses it for its own sake. This type has been dealt under the category <u>ikhtiyārī</u>. If it becomes so difficult as to be impossible to carry out, this type is discussed under the category <u>ghayr maqdūr</u>.

What concerns the category of <u>idtirāri</u> is, in fact, the third type of <u>mashaqqa</u>. This kind of <u>mashaqqa</u> is usually either an inevitable result of a certain act, in that case called <u>baraj</u>, or it comes about from without; neither from the <u>mukallaf's</u> own choice nor as a result of his action. This kind is discussed further below under the heading <u>khārijī</u>. The category <u>idtirārī</u> thus deals with <u>baraj</u> actions. On baraj actions, Shātibī's basic position is that they are revoked where they become impediments in fulfilling essential obligations.

According to Shātibi haraj is revoked in the following two cases:

- First where one fears being cut off from the Path (<u>al-khawf min</u> <u>al-inqitā</u> <u>an al-tariq</u>, That is, when inconvenience in performing a certain act amounts to abhorrence of it or creates a dislike for one's obligation, that inconvenience is called <u>haraj</u> and is revokable. The revoked acts include all that may cause any harm to occur to one's body, intellect, property or condition. (136)
- 2. Second, where the fear of falling short of fulfilling all of one's duties occurs, or, at least, where one's indulgence in one act comes into conflict with his other duties or results in neglecting other duties. In some cases this indulgence prevents one from fulfilling his duty to others. Thus he stands condemned because he is required to carry out all his duties without neglecting any one of them. (136)

Shātibi's argument in favour of the above observations are based on evidences from the Qur'ān and <u>Hadith</u> to the effect that "God made this blessed upright <u>Shari'a</u> generous and convenient and by making it so He won people's hearts and evoked in them love for <u>Shari'a</u>. If they had to act in a way against convenience, they could not honestly fulfil their obligations." (136)

There are, however, instances from the Prophet's own actions (and from others) when people opted for the harder acts. Nevertheless the Prophet is quoted frequently prohibiting or promoting the deliberate creation and seeking of hardship. This poses an apparent contradiction to Shāţibī's position.

Shāṭibī resolves this problem, still maintaining his original position, by concluding on the basis of an analysis of verses of the Qur'ān and of certain aḥādīth that, "The maqṣūd of the Shāri'is that the prohibition be based on some intelligible <u>'illa</u>. (138)

4 : + Shāţibī maintains that the <u><illa</u> of the prohibition in this case is the fatigue or impediment which results from an action and which makes it difficult or tiring to carry the action on further. In the case of the second situation, the <u><illa</u> lies in the fact that the action impedes carrying out other duties or others'. duties. The contrary is also true; ^{if} an action does not constitute an impediment in the above sense, it will not be prohibited even though it may be hard.

Shātibi thus concludes:

<u>}</u> ж.е

> "In fine, prohibition based on some intelligible <u>cilla</u> is the <u>maqsud</u> of the <u>Shari</u>. Since this is true, the prohibition depends on there being an <u>cilla</u> both for its affirmation and its negation." (138)

There is, however, one situation of hardship worth considering. That is a situation where an obligation involves a risk of losing one's life and yet a person opts for it. Is his option valid? Shātibī examines this situation by asking the following question: Did the <u>Shāri</u> remove <u>mashaqqa</u> because it is His right (<u>haqq</u>) or because it is the right of the <u>cabd</u>? (142) In his answer, Shātibī takes into consideration his previous arguments about God's not intending <u>mashaqqa</u> and observes that "when someone chooses to see the act as a right of (rather duty towards) God, the act is absolutely forbidden, (because God has removed hardship from religion). But if one regards it as a right of the <u>cabd</u>, it is not absolutely forbidden, but rather be left to one's choice." (143)

In this context Shātibi reconsiders the case of arbāb al-aḥwāl and their like, the people who choose extraordinary hardship in preference to Sharfi allowances or who indulge in certain duties in order to disregard others. Shāțibi considers the attitude of arbāb al-aḥwāl towards <u>Shar^ci</u> obligations as extraordinary.

Shāṭibī explains his view by making a distinction between two kinds of people:

 <u>Arbab al-huzūz</u>: those for whom carrying out a particular act causes extraordinary hardship, or for whom not availing of <u>Shar</u> allowance means inviting harm. Such people must not carry out an act of this kind and should avail themselves of <u>Shar</u> irukhsa.

Shātibi, however, warns against the other extreme of following one's <u>huzūz</u> absolutely so that one departs from the bondage of ^cubūdiyya. (146–147)

Ŋ,

"The true position according to <u>Sharira</u> is a combination of both aspects with a view of balance(<u>radl</u>); to pursue one's <u>huzūz</u> as long as the pursuit does not interfere with an obligatory duty, and to abstain from huzuz as long as the abstinence does not lead to prohibition." (146)

2. <u>Ahl isqāt al-huzūz</u>: those for whom such acts do not bring about fatigue and hardship because of their acts being governed by fear, hope or love. The fear makes the hardship feel lesser; the hope relaxes the hardness of the act, while the love renders the act rather enjoyable. This group is so engrossed in fulfilling their duty to God on the basis of fear, hope and love that they even forget their own huzūz. They give up personal considerations. (147-148)

Mashaqqa Khārija

der.

А 2 л 2 л There is a third category of <u>mashaqqa</u> which falls upon <u>mukallaf</u> from without; it is neither intended by the <u>mukallaf</u> nor is it a result of any of his actions. In the above discussed categories, <u>mashaqqa</u> was a necessary part, or a consequence, of <u>mukallaf</u>'s intention or action. In the present category, <u>mashaqqa</u> is khārijī (external) to his intention as well as to his action.

Shāțibi maintains that the Shāri^c does not intend the continuation of a mashaqqa

as he did not first originally intend to impose it. The only explanation for the imposition of <u>mashaqqa khārija</u> when it is known to be intended by God, is that He intends it in order to test and examine the faith of the <u>mukallafīn</u>. It is, neveretheless, understood from the totality of <u>Sharī^ca</sub> that it is per-</u> missible to remove <u>mashaqqa</u> absolutely to eliminate the related <u>mashaqqa</u> and to protect the permissible huzūz from being affected by mashaqqa.

30

<u>Sharīša</u> even allows preventing <u>mashaqqa</u> before it occurs. (150) This permission is known a priori (darūratan) in din. (151)

Shāțibī illustrates <u>mashaqqa</u> <u>khārija</u> with the following: hunger, thirst, cold, heat, sickness, bodily harm, etc. Removing all of these mashaqqas is allowed. (150–151)

Shāṭibī, however, observes an important detail. The obligatory nature of the demand to do away with the <u>mashaqqa</u> differs in two kinds of <u>mashaqqa</u> <u>khārija</u>. The first is that where the obligatory nature of the removal of <u>mashaqqa</u> is proven, such as in case of an attack upon Muslims to destroy Islam. In such cases, the <u>mashaqqa</u> consists of an attack or a possible domination of non-Muslims. The obligation to do away with this mashaqqa is undoubtedly proven.

In the second kind of <u>mashaqqa</u> <u>khārija</u>, for example, an incurable sickness, its elimination is not irrefutably demanded. In such a case the imposition of hardship and the endurance of trial must be borne. One must submit to such a mashaqqa as a qaḍā' (decree of destiny).

Shātibi sums up the discussion on taklif in reference to mashaqqa with the

following three conclusions:

1.

- 1. Whether <u>mashaqqa</u> falls upon the <u>mukallaf</u> particularly and singularly (in such a case, called <u>mashaqqa</u> <u>khāssa</u>), or falls upon others together with him or falls upon others because of him, (called <u>mashaqqa</u> <u>famma</u>), in every case, a <u>mashaqqa</u> is not required by <u>Shāri</u> neither in its essence nor in the act that leads to it. If there be a conflict between two obligations to eliminate two <u>mashaqqas</u>, the elimination of a <u>mashaqqa</u> which is <u>famma</u> (general) will prevail over the elimination of a<u>mashaqqa</u>
- 2. <u>Mashaqqa</u> may be <u>mu<tād</u> or <u>khārij</u> <u>an mu<tād</u>. In case of its being <u>mu<tād</u>, its removal is not intended by the <u>Shāri</u> just as its imposition was also not intended. The removal of this kind of <u>mashaqqa</u> means the discontinuation of taklif.

In case of a <u>mashaqqa</u> which is <u>khārij</u> <u>an <u>mu</u><u>stād</u>, since it is conducive</u> to disruption in either <u>dīn</u> or <u>duny</u><u>a</u>, its total removal is the <u>maqs</u><u>u</u><u>d</u> of the Shāri</u>.

There is, however, one consideration. The hardship involved in acts is not the same in all cases; it varies from time to time, place to place and state to state. This is the reason why the same <u>mashaqqa</u> may appear to be <u>khārij</u> <u>can</u> <u>mu</u><u>t</u><u>r</u><u>ad</u> in certain cases while, in fact, it is <u>mu</u><u>t</u><u>ad</u>. Shā<u>t</u><u>i</u><u>b</u><u>i</u> explains this difficulty by saying that a <u>mashaqqa</u> following from a single act has two ends and a middle. The higher end of <u>the mashaqqa</u> is such that when something is added to it <u>the mashaqqa</u> ceases to be <u>mu</u><u>t</u><u>ad</u>. This does not, however, exclude <u>mashaqqa</u> from being essentially <u>mu</u><u>t</u><u>ad</u>. The lower end is such that were something subtracted, there would remain no more <u>mashaqqa</u> attributable to that act.

3. <u>Sharira</u>, according to its requirements, follows precisely the middle way in its obligations, taking both sides equally. Obedience to law comes within the capacity of man without necessitating any <u>mashaqqa</u>

or any leniency.

18 90

Now if <u>Sharī'a</u> legislates in view of the <u>mukalla</u>f's deviation from the middle point to one of the above-mentioned ends, the legislation will aim at returning the <u>mukallaf</u> to the just middle. But in this process it will lean on the other side so as to restore a balance.

001

Following this line of argument, it is to be concluded that $every_kulliyya al - \frac{shar^{i}yya}{shar^{i}yya}$ (universal legal principle) essentially takes the middle position. But if it leans toward one of the extremes, it will do so because of actual or possible inclination towards the other end. The tendency to <u>tashdid</u> (severity) is brought forward to balance the laxity in a <u>mukallaf's</u> regard for <u>Din</u>. The tendency to <u>takhfif</u> (laxity) is brought forward to balance hardship and severity.

The departures from the middle position, as reported in traditions, must be understood in the light of the above explanation. This departure is meant to balance the severity or laxity, whichever the case may be, inherent in the act, the object of obligation. Similarly the stress on piety (wara^c) and asceticism and the like, when they appear to be departures from the middle position, should also be taken as an attempt to balance the laxity in obligation.

SECTION FOUR

~~-

AN ANALYSIS OF THE TERM <u>TA^cABBUD</u>: DISTINCTION BETWEEN LEGAL AND MORAL AND RELIGIOUS OBLIGATIONS

This section deals with the purpose of the lawgiver in making the <u>mukallaf</u> subject to the rules of <u>Shari</u>. In other words it seeks an answer to the question of the nature of legal obligation.

The preceding section discussed the aspect of legal obligations which concerns the legal command and the <u>mukallaf's</u> physical capability to perform it. This section deals with that aspect of obligation which has to do with the <u>mukallaf</u> himself - his intention and motive. The argument here is, again, that legal obligation is essentially motivated by the <u>maşlaha</u> of the <u>mukallaf</u>. To explain this, Shāțibī clarifies and analyses the notions of <u>maşlaha</u> and <u>ta'abbud</u> which are often considered to be opposed to each other, in reference to obligation. As elaborated earlier, the notions of <u>zuhd</u> and <u>ikhlāş</u>, as expounded by the <u>Şāfīs</u>, laid special stress on <u>tark</u> <u>huzūz</u> al-nafs as a necessary qualification of <u>cubūdiyya</u> or the Şūfī understanding of obligation.³ Shāțibī maintains that although legal obligation also aims at <u>ta'abbud</u>, yet <u>huzūz</u> are not denied by <u>ta'abbud</u>. It is in fact the conformity of action with the objectives of the lawgiver which is the real meaning of <u>ta'abbud</u>. The sense of hardship contained in the meaning of <u>taklīf</u> (obligation) is not the denial of the necessities of life; it is rather perseverence in fulfilling the obligation and its universality that makes it hard. The Şūfī sense of <u>ta'abbud</u> is further refuted by the limitation of the scope of <u>ta'abbud</u> in the sense of mere obedience. For Shāṭibī, this sense applies only to the <u>'ibādāt</u>, while <u>'ādāt</u> are governed by <u>maşlaḥa</u>. Since according to him, in the final analysis, the <u>ta'abbud</u> in <u>'ibādāt</u> is only one aspect of <u>maşlaḥa</u>, and <u>maşlaḥa</u> in <u>'ādāt</u> is not opposed to <u>ta'abbud</u>, Shāṭibī concludes that legal obligation is motivated by the <u>maşlaḥa</u> of the mukallaf.

The discussion in this section is arranged in twenty problems. The three main topics discussed are as follows: 1) <u>ta</u> abbud and the problem of <u>huzuz</u>; 2) <u>awa'id</u>; 3) the division of obligations into <u>ibadat</u> and <u>adat</u> in accordance with the considerations of ta abbud and maslaha.

Tarabbud and the Huzūz

1

Shāṭibi opens the discussion by saying that "the legal objective in instituting the law is to relieve the <u>mukallaf</u> from the stimulus of his passions (<u>hawā</u>) so that he be a servant of God voluntarily (<u>ikhtiyāran</u>) as he is so naturally (<u>idțirāran</u>, by compulsion)". (168)

To prove this point he argues from the Qur'ān and sayings of the Prophet where following one's passions (hawā) is condemned. (169) He further contends that human experience in society (al-tajārib wa'l-'ādāt) also tellus that <u>maṣāliḥ</u>, be they those concerning religious matters or be they mundane, cannot be achieved by following passions and selfish motives. (170) The above position may appear to agree with the anti-maslaha viewpoint in denying the interests and desires of the people, and may imply a demand for absolute obligation. It is at this point, however, where Shātibī makes a significant distinction. He denies the identification of masalih with shahawat (desires), hawā (passion) and aghrād (personal interests). He stresses that Sharī^sa aims at the masalih, not at realizing hawa. He does not accept the idea that 'akhdh huzūz can be equated with hawā. (172) In order to distinguish between hawā and <u>buzūz</u>, Shāțibi argues in detail that following the passions is condemned even in cases where the act concerned is in itself praiseworthy, but this is not so insofar as $huz\bar{u}z$ are concerned. (174) The reason is that an action performed in obedience to the stimulus of passion, obviously, pays no attention to the Command or Prohibition of the law, whereas seeking fulfilment of Contract and aghrad is not opposed to the objectives of Shari'a in the above sense. (174, 172). One can seek huzūz by making them subservient to masālih which are the pupose of law. Referring to the Sufis' states and experiences, Shātibi argues that by denying huzūz al-nafs these people aim at something praiseworthy; but by suspending the observance of the legal obligations or by aiming at things which may bring happiness to them, they are merely obeying the demands of passions. (175) On the contrary, Shāțibi argues that one cannot avoid huzuz in fulfilling legal obligation. He says furthermore that ikhlāş or more specifically takhlīs al-hazz (purification of hazz) does not mean denial of $huz \overline{u} z$. The main points of of Shāṭibī's arguments are as follows:

≰્ર સ્ટ્રેટ

> **.** . .

magasid may be divided into two types: From the standpoint of hazz, maqāsid as liyya (essential objectives) in which the mukallaf has no hazz and maqāşid tābisa in which the hazz is provided. $\int_{\lambda}^{\mu_{c}}$ Maqāşid at asliyya means universal necessary obligations consisting of the Five Maṣāliḥ. (176) Examples of tābi'a are obligations in which the natural desires (shahawat) and pleasures are also aimed to be satiated. Shātibi argues that in_{l}^{al} maqāsid tābića, the shahawāt are, in (178)fact, a means to achieve the maqasid asliyya and, thus, no longer In fact, God knows that din and dunya are remain ittibās al-hawā. maintained and well preserved by these stimulii in man which excite him to acquire what he and his fellow beings need. The desires to eat and drink are created so that when he is hungry and thirsty, they motivate him to seek means to fulfil this need. But there are certain desires which one individual cannot fulfil alone; hence he needs the co-operation of others. Thus, although each one fulfils his own desires, in fact, at the same time, he is also, working for the benefit of others. Hence his seeking of huzūz is, in a sense not entirely a hawa. On the basis of huzūz is made permissible, not prothis consideration seeking of hibited. (178-179)

1.

ан 19 ал

- 2. Through a detailed analysis of $\frac{d^2}{maqasid}$ as invariant $\frac{d^2}{ds}$ and $\frac{d}{ds}$ an
- 3. <u>Takhliş</u> (purification) or <u>tajrid</u> (abstraction) from <u>hazz</u> is thus achieved in those cases where <u>hazz</u> is permitted or demanded even when one is actually seeking <u>huzūz</u>. This occurs for the reason that if the seeking of <u>hazz</u> is qualified by legal provisions and other such conditions, there is,

in fact, no more a hazz for mukallaf insofar as hazz is a requirement. (186)

4. The legal penalties in which there apparently figures no <u>hazz</u> for the <u>mukallaf</u>, are, in fact, a means to protect or realize <u>hazz</u> of the <u>mukallaf</u>. The penalties are meant to prevent persons from harming others' <u>maşlaha</u> so that <u>maşalih</u> in general are maintained in a better way. (190–191)

In cases where the act accords with $maq\bar{a}_{sid}$ $t\bar{a}_{bi'a}$, the case is somewhat different. Here the criterion cannot be $t\bar{a}_{bi'a}$, hence it must be seen whether the act is connected with $t\bar{a}_{b}$ $maq\bar{a}_{sid}$ a_{sliyya}^{d} . If it is so connected, even though it seeks <u>hazz al-nafs</u>, the act is undoubtedly one of obedience. (207) This connection is either actual such as a declaration of intention by the <u>mukallaf</u>, or potential such as acts which are means to the permitted act. If this connection with <u>asliyya</u> is absent, then the act is simply one of <u>hazz</u> and <u>hawa</u>. (207)

í

Shāţibī explains the matter further by saying that if the seeking of <u>huzūz</u> were the absolute opposite to obedience, it would not have been permissible for anyone to perform any act of <u>cada</u> unless there were no intention and effort to achieve the <u>hazz</u> <u>al-nafs</u>. In fact there is no such command in <u>Sharī'a</u>, nor is the goal of <u>huzūz</u> in <u>al-camal al-cadiyya</u> prohibited, even though the lawgiver always lays stress on ikhlāş. (208)

If the intention to achieve <u>hazz</u> is denied in <u>al-a'māl al-'ādiyya</u>, any hope for paradise or fear of hell in reference to acts of <u>ibādāt</u> would render them invalid (<u>'amal bighayr al-haqq</u>). Such a conclusion is obviously absurd in view of the numerous verses in the Qur'ān and of the sayings of the Prophet which promise reward and punishment for such acts. To act in hope of reward or with a fear of punishment is certainly an act of seeking huzūz. (210)

To defend his conclusion, Shāţibī, in addition to rational and traditional criticism, particularly mentions Ghazālī's views on <u>huzūz</u> and clarifies his own position by criticizing Ghazālī (214-215), Shāţibī explains that obligations are divided into two categories. First, there are the <u>lbādā</u>t, by which one seeks closeness to God. They consist of Belief (<u>lmān</u>) and its subsidiaries as funda-mentals of Islam and all <u>sibādā</u>t. The second category is <u>cāda</u>t. Satisfaction of <u>cādā</u>t obligations means spreading <u>masālih</u> absolutely, and opposition to meeting these obligations means spreading <u>masālih</u> of the people. The first has to do with the rights of God in this world. It does not aim to yield <u>masālih</u> in

this world but rather in the hereafter. (215)

.a. 11

-s. ->

Now in the first category $d \in hazz$ in the hereafter is established and lawful. The seeking of hazz in this sense cannot be called <u>shirk</u> (polytheism), nor is it a denial of <u>ikhlāş</u>. Furthermore, even according to Ghazālī, the highest aim of <u>ubūdiyya</u> is <u>nazar</u> <u>ilā al-mahbūb</u> (the vision of the beloved) in <u>ākhira</u>, which is also a <u>hazz</u>. (216) In fact Ghazālī calls it <u>hazz</u> <u>cazīm</u> (great joy). Also to demand complete negation of <u>huzūz</u> is an impossible obligation. (216) Seeking $d \Rightarrow hazz$ in this world in <u>cibādāt</u> such as to perform <u>cibādāt</u> in order to earn the praise of the people, or for some strictly personal considerations like fasting in order to save money, etc., are matters which affect the <u>ikhlāş</u> of ^cibādāt. (218-219)

As to the second category of obligations, i.e. <u>sadāt</u> such as <u>nikāķ</u> (marriage), <u>bay</u> (sale), etc., it is well known that the lawgiver intends through these things the maintenance of the immediate <u>masāliķ</u> of the people. Since such is the case, seeking <u>hazz</u> in performing this category of obligations cannot be contradictory to the intention of the lawgiver. Further, if it were wrong to seek these <u>huzūz</u>, the Qur'ān and <u>Sunna</u> would not have mentioned them as being part of God's Grace and favour. (222)

The distinction in <u>cadat</u> and <u>cibadat</u> may be observed from the point of view of <u>niyaba</u> (proxy) as well. <u>Niyaba</u> is not allowed in <u>cibadat</u>, while it is lawful in <u>cadat</u> with the few exceptions where the obligation is specific and individual.

000

The criterion in this regard is the consideration whether the hazz which one aims at can be realized by someone else for him or not. If this obligation can be realized by another, then <u>niyāba</u> is valid; otherwise, not. For instance, in matters of sale etc. <u>niyāba</u> is valid, but it is not in matters such as eating, drinking, marrying, etc. (227)

Ĩ

Since <u>hazz</u> is distinguished from <u>hawā</u>, Shāṭibī enumerated three characteristics of the obligation which provide assurance that the effort to achieve <u>hazz</u> in obedience to the lawgiver will not reduce one's act to <u>hawā</u>. These characteristics are <u>dawām</u> (perseverance) (242) universality (<u>kulliya</u>) and the generality (<u>umūm</u>) of the obligation.

It is a test of one's obedience when one has to meet an obligation constantly. (243) The characteristic of <u>kulli</u> (universal) requires that all obligations, and each obligation in its entirety, must be met without there being any possibility of getting exemption from some or part of an obligation. All particulars and parts of an obligation are obligatory without preference of one above others. (244) Being <u>camm</u>, the obligations are obligatory upon each <u>mukallaf</u> without distinction. The only exception to this <u>cumum</u> is the Prophet, in respect to his regular obligations as well as to his special distinctive privileges (<u>mazāyā</u>) This case is unique, partly because khawāriq al-<u>c</u>ādāt (deviation from regular habits) are

often equal to <u>cada</u>t in the case of the prophets. Since as a general rule the acts of the Prophet are obligatory, as models to be followed, and the cases of khawāriq al-cādāt are impossible to be followed, the latter must be considered

as special to the Prophets. They are not obligatory to be followed unless the Sharica explicitly demands so and then only if they do not disagree with The main argument that underlies this discussion is that Sharī'a. (249-266) the extraordinary acts of the Prophet where he appears to be abandoning huzuz are in fact khawariq al- 'ādāt in the case of common men. (269) Since Sharica is universal, it cannot oblige all men with things which are khawāriq. (275) Invalidity of the khawārig, however, does not mean that law does not or cannot be changed. What Shātibi is stressing is the fact that the khawarig do not convey the sense of legal change; they are rather exceptions to laws of nature. In addition to Prophetic revelation, Shātibī includes Kashf (mystic revelation) and of the awliyā'in khawāriq. (266–269) In order that it ru'ya (dreams) may be understood fully, this discussions requires a rather detailed analysis of the notions of 'adat and khawariq, and their relationship to the rules of Shari'a. The analysis of 'āda is presented in the following chapter, as it is more suited to Briefly, Shātibi uses <u>cādāt</u> both in the sense of habits, 5 the discussion there. customs and human behaviour and as an opposite term to^c ibadat. Essentially, مقطع belongs to the physical world، محكمة are constant; and when some event happens contrary to 'āda it is called kharq al-fāda. Not all of the 'ādāt are constant, however; it is, in fact, only the universals of being which are constant; Shātibi calls them cawa'id mustamirra. Some of these cawa'id are either introduced or sanctioned by <u>Sharira</u>, hence called $\frac{d^2}{d^2}$ and $\frac{d^2}{d^2}$. Others are current in the practice of the people, hence called $\frac{d^2}{d^2}$ and $\frac{d^2}{d^2}$. Sharira does not oppose $\frac{c_{awa}}{c_{awa}}$ in fact, it shows a constant regard for them. There are,

however, variations in the practice of these <u>cadat</u>. Also they change with time and place.

A detailed analysis of $\frac{l}{kawa'id}$ shar'iyya by Shāţibi reveals that <u>maşlaha</u> is the basic consideration both in the change and the continuity of these 'awa'id. In the light of this view it may be seen that <u>ta'abbud</u> toward $\frac{l}{kawa'id}$ shar'iyya is not devoid of <u>huzuz</u> and <u>maşlaha</u>.

Ta^cabbud and Maslaha

From the above analysis Shāţibī concludes that the essential consideration in <u>(ibādāt</u>, insofar as the <u>mukallaf</u> is concerned is <u>ta</u><u>(abbud</u> without regard for <u>ma</u><u>(ani</u>) (inner meanings). In <u>(ādāt</u>, on the other hand, the essential consideration is that of <u>ma</u><u>(āni</u>). (300) This conclusion is further demonstrated by the following points of argument. First, from a survey of <u>Sharī</u><u>(a</u> it may be inductively known that provisions such as <u>tahāra</u> (ritual cleanliness) and <u>tayammum</u> (ablution with dust) in the realm of <u>(ibādāt</u> are difficult to explain, except in terms of <u>ta</u><u>(abbud</u>). (301) In the realm of <u>(ādāt</u>, it is obvious that such provisions are based on <u>maslaha</u> of the people. It is thus inductively discoverable that the lawgiver relies on a regard for <u>maslaha</u> in <u>(ādāt</u>). (305)

Secondly, in <u>ibadat</u> the extension of the scope of <u>ta</u> abbud is not intended. (301) In other words, the obligation is limited to the specific commands comprised in

★ibādāt. This is why no explicit reason is given for promulgating such commands. In the case of <u>fādāt</u>, on the contrary, the extension of the rules is the purpose. Hence the lawgiver generously explains the rules of law relating to fādāt in respect to their <u>filal</u> (reasons) and <u>hikam</u> (wisdom). (306)

ч-т і

<u>Ta<abbud and ma<nā/maslaha</u>, however, are not opposite terms for Shāţibī. He characterizes <u>ta<abbud</u> by various statements: "<u>al-rujū</u><<u>ilā mujarrad mā haddahu</u> <u>al Shāri</u><" (recourse only to what the lawgiver has determined); (304) <u>Al-inqiyād</u> <u>li 'awāmir Allāh</u>" (being bound by the commands of God). (301) "<u>Mā huwa</u> <u>haqqun lillāh khāssatan</u>" (that which is the exclusive right of God). (315) "Rāji < un ilā 'adami ma<quliyat al-ma < nā (that which refers to the nonintelligibility of its meaning). (318) Shāţibī defines <u>ma < nā</u> in this context as follows: that is "dabţu wujūh al-masāliḥ" (to define the aspects of masāliḥ). (308)

- ----

3

ł

The distinction between <u>ta</u> abbud and <u>ma</u> nā or <u>maslaha</u> occurs initially in reference to the question whether the reason for a command is intelligible or not. If the reason is intelligible, the command is based on <u>ma</u> nā; otherwise, it is <u>ta</u> bbud. (314) This explanation is as yet insufficient, however, because the "intelligibility" needs further to be qualified. Shā tibi explains that "intelligibility" applies where the <u>ma</u> nā or <u>maslaha</u> can be extended as an <u>cilla</u> to other similar cases. If the <u>ma</u> nā is extendable, it will still be taken as <u>ta</u> abbud. (309) To illustate,

"The requirement of dowry in marriage is one of those matters in which the human reason cannot understand (determine) the specific masalih in these commands, so that they could be made analogous to other cases. We know that the required conditions in marriage such as that of the guardian and the dowry, etc., are laid down to distinguish marriage from fornication (<u>sifāb</u>)...But (if they are considered as being the <u>cilla</u> of marriage) they are but general principles just as humility and submission to the Sublime are the reasons for the obligation of <u>cibādāt</u>. This amount (of <u>cilla</u> is not sufficient to establish an analogy, to extend the above rule to further cases; so that one might say that were a distinction between marriage and fornication to be established by some other factors, the above conditions would no more be required." (308)

This explanation implies at least two things: one, that tasabbudāt according to Shāțibī, are absolute obligations in the sense that they must be fulfilled without asking for the reason, and second, that tacabbudat cannot be made the Shātibi seems to be stressing the second implication, rather basis of analogy. than the first. In other words, he is implicitly arguing that the absoluteness of obligation in matters of ta abbudāt is maintained only in the sense that they are not to be extended. There is no denial of <u><u>`illa;</u> in fact it is only after the</u> search for an <u>filla</u> in the command that one can decide whether the <u>filla</u> given extension by taclil or implied is general or specific. What is denied is the amounts to placing a limitation on the scope The denial of tasii and qiyās. of application of these commands. It is in the sense of specifically limited command that tacabbud is spoken about in this context. As Shāțibi himself says, "In all those matters where a consideration of tacabbud is established, there can be no tafri^c (deduction, extension by analogy) from them." (310)

Shāțibi, however, also accepts other senses of <u>ta</u>abbud in addition to the one mentioned above. He explains that even matters, where the consideration of meaning without <u>ta</u>abbud (în the sense mentioned above) is established, are not free from <u>ta</u>abbud (in the general sense of the term). (315) This general sense of <u>ta</u>abbud is demonstrated by the following considerations. First a <u>mukallaf</u> is bound to obey a command because of the sense of demand (<u>iqtiqa</u>) and option (takhyīr) imposed by the command, not because he finds in it a certain maslaha. (311) Second, even if a decision about an <u>cilla</u> is taken, this process does not assure us that the <u>cilla</u> decided upon is the only illa of that command or that it is the only <u>maslaha</u> to be realized. This state of indecisiveness (<u>wāqifīn</u>) is removed by recourse to <u>tacabbud</u>. (312) Shāţibī further explains that <u>qiyās</u> means a search for an <u>cilla</u> only insofar as it is ordinarily possible. <u>Qiyās</u> does not exhaust all the <u>cilal</u>; it is rather based on the most probable (<u>ghalbat al-zann</u>) <u>cilla</u>. On this basis '<u>qadā'bi'l</u> <u>tacaddī'</u> (judicial decision by extension of the original ruling) is not contradictory to <u>tacabbud</u> which, here, means ' not based on reason'. (312)

١.

Third, the obligations are known to us in two ways: either through well-known methods such as <u>ijmā</u>, <u>naşs</u>, <u>ishāra</u>, <u>munāsaba</u> etc., or through instances where none of these methods can be applied. The obligations of the latter kind are known only by <u>wahy</u> (revelation). In this category of obligation the absence of <u>cilla</u> and <u>ta</u><u>caddī</u> within command demands <u>ta</u><u>cabbud</u> only. This <u>ta</u><u>cabbud</u> means to stop at the point where the <u>shāri</u> has defined the limit; if the <u>cilla</u> is not given, <u>ta</u><u>cabbud</u> demands that the command must not be extended by <u>qiyā</u>s. (313) "A <u>maşlaha</u> is so from God in such a manner that it is verifiable (ya<u>s</u><u>diq</u>u) by human reason (<u>caql</u>) and reassuring (<u>ta<u>t</u><u>ma</u><u>cinn</u>) to the soul (<u>nafs</u>)". (315)</u>

The takālīf can also be viewed as rights of God. In this sense they become ta'abbudi. Shāţibi, however, regards ta'abbud as a general sense of the rights of God. He divides these rights into three categories. First are those rights which belong exclusively to God, such as the <u>cibādāt</u>. Second, are those rights

of God which involve the rights of men as well, but the consideration of the former dominates. The third category consists of those rights of God in which consideration for the rights of men dominates. It is to the last category that <u>maşlaha</u> or <u>ma<nā</u> belong directly, and hence this category is not essentially ta<abbudi. (318-320).

1

Shāţibī clarifies the distinction between <u>ta'abbud</u> and <u>maşlaha</u>, and <u>'ibādāt</u> and <u>'ādāt</u> from the point of view of <u>huqūq</u> (rights). He says that the right of God means a situation "where it is understood from <u>Shar'</u> (law) that the <u>mukallaf</u> has no option (<u>khiyara</u>), whether the <u>ma'nā</u> is intelligible or not." (318) The right of man is defined as "what refers to his (man's) <u>maşālih</u> in this world". (318) The <u>maşālih</u> in the hereafter are generally rights of God. Thus <u>ta'abbud</u> means something, "the meaning of which cannot be specifically understood". (318) In view of these definitions Shāţibī concludes that <u><ibādāt</u> essentially refer to the rights of God and 'ādāt to the rights of men. (318)

SECTION FIVE

THE MUKALLAF'S MAQSID IN LEGAL OBLIGATION:

Analysis of the term <u>niyya</u>

So far the discussion has been concerned with the objectives of the lawgiver. This present part discusses the objectives of the <u>mukallaf</u>. On the whole these objectives have to do with the intention of the <u>mukallaf</u> and its effect on the validity or utility of the act. The discussion is arranged in twelve problems. At the end is an epilogue on the problem of knowing the objectives of the lawgiver.

The main points for discussion are the following terms: <u>niyya</u> (intention) and <u>maqāşid</u>, <u>takālif</u> and <u>jalb al-maşālih</u> (to seek <u>maşlaḥa</u>); <u>maşlaḥa</u> and <u>taḥayyul</u> (seeking legal devices to escape the severity of the law).

Niyya

Shāṭibī opens the discussion by saying that "acts are (judged) by <u>niyyāt</u> (intentions)." (323) Thus an interrelation between 'act' and 'intention' is established. But this raises a question about the details of this relationship. Does it mean that intention without act and act without intention will not be considered? Further, what is intention? By intention of the <u>mukallaf</u> does one mean the correspondence with the intention of the lawgiver in that particular act or something else? It may be noted here that Shāṭibī uses the terms <u>niyya</u>, <u>qaşd</u>, <u>maqşid</u>, ibtigha interchangeably, all of which have the sense of English "intention".

The relationship of niyya and act: Shāṭibī says that the maqasid make a distinction between fada and fibada. The same act, such as the act of prostration, is cibada according to one intention, but it is not cibada according to others. (324) Thus acts are judged by the intention of their authors. Shājibi, however, maintains a distinction at this point between al-aḥkām al-waḍ<iyya and al-aḥkām Al-Taklifiyya are those rules of law which come into effect by al-taklīfiyya. the declaration of the lawgiver. They are declared to be 'amr (command), or nahy (prohibition), etc. The five well-known values of obligatory, recommended, Since al-aḥkām al-taklīfiyya produce etc., belong to this category of rules. direct obligations, a necessary condition for their being fulfilled is the intention of Wad 'iyya are those rules which are not the effect of a the mukallaf to do so. direct command but which become effective because they are auxiliary to direct commands.

Ŋ,

With the above distinction in mind, Shātibī says that if an act is connected with a <u>qaşd</u>, <u>al aḥkām al-taklīfiyya</u> become effective in connection with this act. If the act is performed without a definite intention, <u>al aḥkām taklīfiyya</u> will not be effective.

One possible objection to this position may be drawn from the cases of acts done under <u>ikrāh</u> (duress) and <u>haz</u>l (joke) where the intention of the <u>mukallaf</u> is not connected with the acts in question, yet, juridically the acts are considered to be valid. (325) Shātibī's answer to this objection entails very significant

ľ.

points of philosophical interest. In brief, he seems to be maintaining a distinction between two standpoints of deciding the validity of an act; from the standpoint of religion and morality the act is subject to <u>ankām</u>, <u>taklīfiyya</u> and here the intention must correspond explicitly with the act, otherwise, the act is not valid. From the juridical standpoint, in cases other than <u>cibādāt</u>, expression of intention and its correspondence with the said act is not a necessary requirements; an act is valid and subject to juridical consequences even in the absence of a corresponding niyya.

i \ . . .

> The source of confusion has been the question of consideration of niyya in the above cases of duress and joke. The niyya, here, is not lacking in an absolute Shātibi, therefore, begins his answer by explaining various senses of the sense. considerations of niyya. In its general sense, niyya (in the sense of volition) is a necessity ("darūra") for the validity of an action. This is so because the doer of an action insofar as he is mukhtar (one who has a choice, freedom of will), has intention implicity necessarily in his action, whether his intention is to be obedient to the command of the lawgiver or not. From this standpoint intention is absent only in such cases as, for instance, when a certain action is performed by a na'im (a person in sleep) or by a majnun (an insane person). Having no ikhtiyār, individuals in these states, are not mukallafin. Those acts which are done with ikhtiyar, however, cannot be considered as lacking niyya. Hence acts performed under duress or as jokes will be judged, juridically, by such intentions. This sense of the consideration of niyya is from the standpoint of ahkam wadeiyya. As has been explained earlier, from the standpoint of

wad^ciyya, an act becomes valid and its juridical consequences are effective, if the necessary conditions of the said acts are fulfilled, even though a corresponding <u>niyya</u> be absent in that act. For instance, if a person returns the deposit to its owner, even though unwillingly, juridically his act of returning the deposit is valid. (327)

Unlike the above-mentioned general sense of the consideration of <u>niyya</u>, the consideration in the special sense demands the intention to obey law. In this specific sense the consideration of <u>niyya</u> becomes a necessary condition for the validity of an act in cases of <u>(ibādāt</u>). It is also necessary when one wants to transform all his acts, <u>(ibādāt</u>) or <u>(ādāt</u>, into <u>tafabbudāt</u>. Free actions (<u>al-afmāl</u> <u>al-dākhila</u> <u>taḥt al-ikhtiyār</u>) can be changed into <u>tafabbudā</u>, if the intention of obedience accompanies them. This sense of consideration of <u>niyya</u> is from the standpoint of <u>(abkām taklīfiyya</u>). As discussed earlier, from the standpoint of <u>taklīfiyya</u> an act becomes valid and the <u>jazā</u> becomes effective only if the act is accompanied by the intention to obey the <u>Shāri</u>.

The <u>niyya</u> of obedience is understood as meaning that the intention of the <u>mukallaf</u> in performing an act will be in conformity with the intention of the lawgiver in instituting the law, i.e. with the <u>maşlaha</u> of the people. (331) From this standpoint any act by which one intends what is unlawful, becomes void (<u>bāțil</u>). The reason for this judgment is that things are allowed in order to achieve <u>maşlaha</u> and remove <u>mafsada</u>. A contrary intention with respect to these lawful things would be equivalent to seeking <u>mafsada</u> and preventing <u>maşlaha</u> which is contrary to human interest as well as to <u>Sharī'a</u>. (333)

ţ

In the light of above discussions, acts may be of the following four types. First there are those acts in which the act and the intention both conform with the objectives of the lawgiver. Second, there are those acts in which both do not conform. Third there are those in which the act conforms, but the intention does not. Fourth there are those in which the intention conforms, but the act does not. (337)

8 n

The legal value of the act in the first and second type is obvious. In the third type the doer will be considered disobedient only for his intention but not for his act. In other words, he has violated the right of God, not the right of men. (338) If a man knows, however, that his act conforms to the objectives of the lawgiver, although his intentions do not, then he is to be the more blamed because he is taking advantage of his act for some other objectives. (339)

In the fourth type, if the doer of the act knows that his act is contrary to the objectives of the lawgiver then his conduct is similar to <u>ibtidā</u>^c (<u>bid</u>^ca, innovation in religious matter). <u>Bid</u>^ca as such is <u>madhmūm</u> according to Shāţibī. He does not accept the judgment of <u>bid</u>^ca made by some scholars. What is called <u>bid</u>^ca <u>muḥarrama</u> or <u>bid</u>^ca <u>madhmūma</u> is understood by Shāţibī in reference to the second type of acts above where intention and act both are contrary to the objectives of the lawgiver. (340) In <u>bid</u>^ca per se the intention conforms but the act does not. Shāţibī, however, excludes those cases where the doer does not know that his act does not conform. In such a case he will not be regarded as disobedient, but his act will still not be considered as compliance (imtithāl). (342)

Jalb al-Maslaha

ĺ.

1

It has been mentioned above that <u>jalb al-maslaha</u> within the limits of <u>sharifa</u> becomes a necessary requirement of <u>niyya</u>. The act of seeking <u>maslaha</u> occurs, however, not always in isolation; often it is connected with other acts as well. Hence the questions that need be considered in regard to <u>jalb al-maslaha</u> have to do with the following situations: striving for <u>maslaha</u> when the result will be harmful to others, and secondly, striving for <u>maslaha</u> for someone else. (348)

Shāțibī devides the situations where one's own maslaha may be harmful to others into eight types of cases, according to the types of harm done. Harm may be general to the whole community, or may be specific to someone; it may be inevitable; it may be avoidable, etc. (349-362) The main principle upheld in these discussions is that if there is an alternative to harm, the bad result Disregarding an alternative would mean that harm becomes must be avoided. the only purpose of one's action. (349) Furthermore, striving for maslaha even though it may be harmful to others will be allowed if there is maslaha for more people than are harmed. The right of striving for maşlaha will be given preference to the consideration of avoiding harm if it is well known that a prohibition to strive for maslaha will cause harm to the seeker. In cases where the seeker himself does not meet any harm but engages in efforts to achieve maşlaha that customarily lead to harm, it must be seen whether this potential harm is qatci (definite) <u>nadir</u> (rare) or <u>zanni</u> (probable). A man will be prevented from striving for maslaha only if the harm done to others is <u>atri</u>. (348)

The second question that needs to be considered in regard to jalb al-maşlaha is that of seeking the maslaha of others. As a general rule Shatibi states that if some one is obliged to seek his masalih, it is not obligatory for others also to This rule is similar to the rule of nivaba discussed seek his masalih. (364) earlier. The main points that Shātibi brings forth in this discussion serve to show that no man is under obligation to fulfil the specific obligations of others. We are not concerned here with the obligation of < ibādāt, as was made clear earlier in reference to nivaba in that ibadat cannot be fulfilled by proxy. The obligations under discussion are those that concern this world. Such obligations, however, become binding upon others when the original mukallaf is unable to fulfil them, although they are necessities for him. For instance, the following obligamasālih of others, can be justified in terms of tions which aim at striving for the above explanation: Zakāt, lending money, burying the dead body, looking after the affairs of minors and the insane, etc. Among these obligations are some which are general (or public) ($\frac{kifa^2iya}{A}$) and some which are specific (<u>cala</u> al-ta^cyin) and individual obligations. The specific obligation cannot be fulfilled In such cases an individual is required to seek masalih for others, but by proxy. only if his own masalih are not affected. A situation meeting this condition is possible if either the individual is capable of fulfilling his own as well as others'obligations, or if other people are looking after his masalih. If he cannot fulfil both his and others' obligations at the same time, his obligation to others will give way in instances of particular obligations to a particular person. His own maslaha is to be preferred to others. If the matter at issue is a general

obligation to others, then others must look after the individual's obligations while he fufills his duty. (364-368)

Shāţibī's conclusions regarding the above two questions of striving for <u>maşlaha</u> are very significant to his legal philosophy. He seems to admit that by doing good, or trying to do good, i.e. to strive for one's <u>maşlaha</u>, one may also actually do evil, i.e. to harm others. This would make $2 \Rightarrow sharīta$ in some instances result in evil deeds. To rectify such a consequence, Shāţibī stresses that obligations be undertaken after considering their ends and consequences, and not on their appearance of good or badness. Furthermore, the goodness of obligations, or the ultimate criterion of <u>maşlaha</u>, is good of the larger number of people and harm to less of them. If the good of the few is harmful to many, it no longer remains good.

The above conclusion shows that in Shāţibī's legal thinking there are certain elements which imply law's consideration for society rather than being an individual commitment towards the lawgiver. In fact, Shāţibī even implies that by disregarding the social implications of the legal obligation, one's individual commitment to do good may result in evil.

Shāțibi's view of legal obligation as also a social obligation is further explicated in his conclusions regarding the situation where one strives for the maşlaha of others. If a person has devoted himself to look after <u>maşālih</u> of society it becomes a kind of societal obligation for others to look after <u>maşālih</u> of that individual. Shāțibi states that this is why the obligation to pay <u>zakāt</u> is

353

١
prescribed; mutual lending of money is allowed; and looking after the maintenance of wife and children is required. In all above cases the individuals in question, e.g. the poor in case of $zak\bar{a}t$, and wife and children, are unable to look after their own <u>masalih</u>, either because they are occupied with serving h =masalih of others, as in case of wife, or they are simply incapable of doing so.

Tahayyul

•

Shāţibī defines <u>hīla</u> and <u>taḥayyul</u> as follows: "When a <u>mukalla</u>f uses certain means in order to escape an obligation or to make some forbidden thing permissible for him, this use of means which causes an obligatory thing to become apparently non-obligatory and a forbidden thing apparently to become permissible, is called <u>hīla</u> or <u>taḥayyul</u>". (379) These means are either apparently permitted in <u>Shar</u>, or are not permitted. They work either by rendering a rule inapplicable or by transferring the consideration of the matter at issue. (378)

<u>Tahayyul</u>, according to Shāțibi, works on two premises: 1) it strives to transfer the value of one legal act to another legal act externally, i.e. merely on the basis of apparent similarity between the two acts. 2) It disregards the inner meaning (i.e. <u>maşlaha</u>), of the acts on the basis of which the acts were originally intended by the <u>Shāri</u>^c, and by doing so reduces the value of these acts to be means to certain other acts, whereas they were meant to be the end. Shāțibi illustrates it with the following example: Someone wishes to sell ten dirhams in cash for twenty on credit. Because of the prohibition of usury, such a transaction is not allowed. This person evades this prohibition by the following <u>hīla</u>. He buys a piece of cloth for ten dirhams and sells it for twenty on credit. To refer to the above premises, he transferred the value of the act of selling the cloth (which is permitted) to the selling of dirhams (which is prohibited) only on the basis of their external similarity (both are acts of selling). But he disregarded the <u>maşlaha</u> in both acts; he rather used the <u>maşlaha</u> of one act, i.e. to earn profit in selling cloth, as a means to achieve his own end.

In the light of the above expalanations Shātibi regards tahayyul as unlawful on the whole. (380) He supports his argument from the Qur'ān and Hadith. He contends that rules of law are made for the masalih of the people; hence acts will be judged according to their relation to masalih because that is the objective of the lawgiver. If an act is based externally and internally on lawful grounds, it is evidently a valid act. On the other hand, if it is externally lawful but the maslaha of that act is against a legal base, it is not lawful. (385)

Shāțibi stresses that legal acts are not intended (maqşūda) for their own sake, but for their marānī which are in maşālih. The maşālih gained in <u>Gibādāt</u> are closeness to God, devotion to him alone with submission and humility and conformation of the heart and other parts of body in obedience to him. If these masalih are not sought, even in <u>Gibādāt</u> become unlawful. There is a clear instance in the case of <u>şalāt</u> done for the sake of <u>rirā al-nās</u> (to put up a show for the people). Such şalāt is not lawful. (385)

In view of the above explanation <u>hiyal</u> (p. <u>hila</u>) are of three types. First come the <u>hiyal</u> of the hypocrites which are unanimously regarded as void and illegal. (387) Second are those <u>hiyal</u> which are unanimously held to be lawful such as uttering phrases of unbelief under duress. Shātibi, here, however, excepts from this rule

the case of one who uses the confession of Islam to save his life. In their motives both cases are similar, for both aim at a <u>maslaha</u> <u>dunyawiyya</u>. In the latter case, however, since the real intention is different from that of a <u>hila</u>, i.e. one confesses but does not believe in Islam, he is seeking a <u>mafsada</u> in the hereafter, and hence the use of the confession is not lawful. (387)

The third type of <u>hiyal</u> are those the legal validity or invalidity of which cannot be decided as clearly as in the above types. Neither is it clear that such <u>hiyal</u> agree with the intentions of the lawgiver nor can it be said that they oppose it. Hence it has been controversial. Shāțibī illustrates this type with two cases; <u>nikāḥ al-muḥallil</u> (marriage of a divorcee with a person other than her husband in order to make remarriage with the husband lawful) and <u>buyū al-ājāl</u> (sales on credit). Shāțibī finds it impossible to decide in favour of or against the practice of those two <u>hiyal</u>. He is of the opinion that those who regard this type as forbidden, believe that it is against the <u>maşlaḥa</u>, or in other words, is an internctional violation of Sharīfa. He disagrees with this conclusion. (388) Shāțibī only provides the arguments of those who are in favour of these two <u>hiyal</u>, but does not give his opinion in favour or against them. (391)

CONCLUDING REMARKS

Having investigated Shātibi's doctrine of <u>maqāsid</u> according to his own formulation and the structure of his own presentation, we are now in a better position to infer the basic components of Shātibi's concept of <u>maslaha</u> and its significance in his legal philosophy. What follows is not a conclusion in the proper sense of

JJU

the word, but rather a reconstruction of Shāṭibī's concept of maslaḥa from his own statements presented above and their implications.

ţ

ć

As must be evident from Shātibi's definition of maşlaha and its various aspects, the essential element in his concept of maslaha is the consideration for and protection of the necessities of human life. The five aspects of the maqasid serve further to establish this point. The first aspect reveals the necessary relation between human needs and maslaha and sets out further details of these human needs in different areas. The second aspect discusses intelligibility as a qualification of legal commands, which implies that a major role is allowed to human reason, in the interpretation, justification and extension of the rulings of The third aspect discusses the doctrine that harmful things which se sharīća. impede the satisfaction of human needs are revokable. Contrary to the views of the suffis and some jurists the maslaha of man or the goal of law does not result in the negation of these needs. The fourth aspect reveals the meaning of obedience. In its narrow sense, obedience means to comply without asking for the reason lying behind the command. This meaning of obedience applies essentially to the The other areas of life, for which Shāṭibī uses the term 'ādāt, are based 'ibadat. on maslaha. There is a second meaning of obedience, therefore in which obedience signifies to conform to the objectives of the lawgiver, or to obey the intent of the law. This sense applies both to <u>"ibādā</u>t and <u>"ādāt</u>, but implies that obedience in matters of ibadat means ta abbud and in matters of adat to follow maslaha, because these are the objectives of the lawgiver. This point is elaborated in detail in the fifth aspect of the magasid. The basic components

of Shātibī's concept of <u>maşlaha</u> are, therefore, the following: 1) the consideration for the needs of man, 2) the rationality of law and the responsibility of man, 3) protection from harm, and 4) conformity with the objectives of the lawgiver.

The jurists preceding Shāţibi had divided <u>maşlaha</u> into <u>darūri</u>, <u>hāji</u>, and <u>taḥsini</u> types only to reject the latter two as less satisfactory bases of legal reasoning. Shāţibi, in contrast, sees the latter two categories of maslaha as layers or zones that are meant to protect the <u>darūri</u> type; they complete and supplement <u>darūri</u> <u>maşlaha</u>. The rejection of the <u>hāji</u> and <u>taḥsīni</u> categories may not immediately affect <u>darūri maslaha</u> but, eventually, such a rejection may disrupt the <u>darūri</u> type as well. This structural approach to maslaha makes Shāţibi's conception more integral than that of others.

Shātibi, however, distinguishes between two conceptions of <u>maşlaha</u>. <u>Maşlaha</u> as conceived in <u>cada</u> is essentially <u>maşlaha</u> <u>dunyawiyya</u>, which does not look beyond this world.

<u>Maşlaha</u> conceived in connection with <u>shari</u> a takes into consideration <u>admaşlaha</u> <u>alukhrawiyya</u> in addition to <u>maşlaha</u> <u>dunyawiyya</u>. Another factor that distinguishes the conception of <u>maşlaha</u> in <u>shari</u> is its simple and abstract nature. <u>Maşlaha</u> in <u>cada</u>, although conceived as not-mixed, yet is found always to be mixed with <u>mafsada</u> and non-maşlaha. In <u>cada</u>, the <u>maşlaha</u> in an act is determined by weighing the elements of <u>maşlaha</u> and <u>mafsada</u>; whichever dominates gives its name to that act. <u>Al-</u> <u>Maşlaha</u> <u>shar</u> iyya does not reject this process and the conclusion drawn from it, yet as <u>maşlaha</u> <u>shar</u> iyya constitutes a legal obligation, it accepts only

А. 14 ро

í

the dominant aspect as a requirement of obligation and rejects the other part for this purpose.

The relativity of <u>maşlaha</u> in <u>cada</u> and definition of $\int_{\Lambda}^{M} \frac{d}{maslaha} \frac{d}{shar}$ in reference to dominating $\int_{\Lambda}^{M} \frac{d}{adiyya}$ is fundamentally important in Shatibi's legal thinking. Such a conception of <u>maşlaha</u> gives him the means to free Islamic legal theory from the rigidity with which traditional view had invested it on both the conceptual and the methodological level.

On the conceptual level there were two main deterministic factors that discouraged any trend towards adaptability in Islamic legal theory. One of these factors was theological determinism springing from the concept of God as Omnipotent and Absolute Authority. The negation of causality in relation to God's actions and the denial of man's free will provided this determinism with further rigour. Shātibi's conception of legal obligation which takes <āda into consideration along with Sharika, making maslaha the common element of the two, provides justification for man's responsibility for his legal acts, a responsibility that theological determinism would deny. The distinction between Cada and shari'a as two different aspects of Divine Will, is a further attempt to solve the dilemma which theological determinism creates for Islamic law. The theological understanding of God's Omnipotence, which demands, by necessity, no disjunction between God's willing something and the actual occurrence of that thing, forced most of the theologians to hold that legal commands are not necessarily backed by the Divine Will; otherwise, they would be actualized immediately.

Shāţibī rejected this mode of thinking. He emphasized that there is Divine Will behind legal commands, but this Will is <u>tashrī^cī</u> and, thus, distinguished from the type of Divine Will which is <u>takwīnī</u>. Man is not involved as an agent in the actualization of God's <u>takwīnī</u> Will, but he is involved in God's <u>tashrī^cī</u> Will. Since man is a <u>mukhtā</u>r, the actualization of legal commands depends upon his choice. This position upholds the responsibility of man in legal acts; yet it does not reject the connection of Divine Will with legal commands.

The second deterministic factor was mora! and ethical and was introduced to Islamic thought by sūfīs. The sūfīs viewed the whole concept of obligation as devotion to God even to the extent of denying the necessities of human life. This attitude resulted in virtual neglect of the major part of <u>Gadāt</u> as being <u>huzūz</u> pursued for the sake of <u>zuhd</u>. In relation to <u>Gibādāt</u> their view of obligation demanded much more than formal fulfilment of the requirements in law for the sake of <u>ikhlāş</u>. <u>Zuhd</u> and <u>ikhlāş</u> thus constituted the basic elements of the <u>sūfī</u> concept of obligation which they termed <u>wara</u>?

In his analysis of <u>ta</u> abbud and <u>huz</u> z Shātibi shows the irrelevance of ethical determinism for legal obligation. <u>Ta</u> bbud means conformity with the objectives of law. Legal obligation does not demand more than what law has specified, and any additional requirement above and beyond the specifications of the law cannot constitute legal obligation.

Shāțibi's concept of mașlaha freed Islamic legal theory from its traditional rigidity on the methodological level as well. On the methodological level the question

7

of how to apply and extend law to new situations was hampered by theological, linguistic and logical factors. On the theological plane, major opposition to <u>maşlaha</u> came from the denial of cause (<u>cilla</u>) in legal reasoning. Shāţibī tried to solve this problem by distinguishing between the <u>afcāl</u> and the <u>awāmir</u> of God. He argued that <u>cilla</u> can be attributed to God's <u>ahkām</u> and His <u>awāmir</u>, if not to His <u>afcāl</u>. Secondly he demonstrated that the Qurfān even mentions <u>cilla</u> for specific commands. Thirdly, after making an analysis of Divine legal commands, Shāţibī concluded that these commands not only have a purpose and motive but also that this purpose is <u>maşlaha</u>.

On the plane of language legal formalism and literalism had been acceptable to jurists in general. Even the method of analogy and interpretation by implication, in the final analysis, inclined towards literalism. Shāţibī rejected this method in two ways. First, by his theory of $\frac{L}{dalāla} \frac{d}{a_sliyya}$, he laid stress on the significance of meaning, more precisely on contextual meaning, rather than the letter of the law. Second, he emphasized that even in interpretations by implication the maqāşid of the shārifa should be the basis of reasoning. Such an interpretation required induction rather than deduction in the process of legal reasoning.

On the plane of logic, the fear of arbitrariness had become a major source of rejection of <u>maşlaha</u>. By giving substance to the concept of <u>maşlaha</u> through conceiving it as a structure and confining it to five specific areas of human needs, Shātibī defended the concept against its becoming merely personal and relative. Moreover, by suggesting that <u>maşlaha</u> is based on istigrā' rather than the method of analogy from particular to particular, Shātibī argued that maşlaha is based

on surer grounds. The proponents of analogy argued that a decision reached by analogy having been deduced from a specific ruling of a legal text logically constituted <u>yaqin</u>. Reasoning in terms of <u>maslaha</u> provided only <u>zann</u>. Using the same terms, Shāţibī argued that the method of analogy led, at the most, to <u>ghalbat al-zann</u>, and not to <u>yaqīn</u>. A decision in favour of one <u>cilla</u> does not remove the doubt that there may be another <u>cilla</u> which is more valid. Secondly there is no way to ascertain that the <u>cilla</u> for which one has decided is also the one in the mind of God. These decisions are based on one's best judgment which amounts to probability, not to certitude. If this be the case, a ruling based on induction is more valid than one based on deduction from one particular ruling.

ł

In the light of the above analysis we can discern a trend towards a view of Islamic theory by Shāţibī that permits adaptability. His understanding of <u>maslaha</u> as a principle of adaptability to human needs is based on certain distinctions that evolved out of his analysis of the concept. The most significant among these distinctions were those between <u>cāda</u> and <u>sharī</u> and <u>between</u> <u>cādāt</u> and <u>cibadāt</u>. For a better understanding of Shāţibī's view of legal theory these distinctions need to be further analysed.

NOTES: CHAPTER VII

The numbers in the parenthesis in the text of this chapter refer to the following:
Shāţibī, Al-Muwāfaqāt, Vol. II (Kitāb al-maqāşid), ed. and

comments. Darāz, (Cairo: Mustafā Muhammad, n.d.)

Ÿ,

- 2. See above Chap. IV, p. 191ff.
- 3. See Chap. VI, p.265ff.
- 4. Kwame Gyekye, "The Terms 'Prima Intentio' and 'Se cunda Intentio' in Arabic Logic", <u>Speculum</u>, XLVI,1(1971),32–38, also suggests that the term <u>ala al-qasd al-awwal</u> should be translated as 'primarily', 'initially', or 'directly' instead of 'in the first intention'.
- 5. See p. 373ff.

Ì

6. See p. 349.

CHAPTER VIII CONTINUITY AND CHANGE

In Chapter V we noticed that in his fatāwā, Shāţibī accepted 14 cases of social change and rejected 23 others. Among the rejected cases 12 belonged to changes in <u>*ibādāt*</u> and 11 to changes in laws relating to family, property, and to contracts and obligations. He rejected changes in <u>*ibādāt*</u> because he considered them to be <u>bid^cāt</u>. He rejected changes in cases of laws relating to family and property where they amounted to either confusion or violation of the individual right of ownership and partnership. He rejected changes in cases of contracts and obligations where they hampered the freedom of trade and commerce.

The fact that Shāţibī did not accept or reject social change <u>in toto</u> and further, that he distinguished among various cases of change, indicates that Shāţibī had a clear notion of change and of the interaction between social and legal change. In fact, as we shall see, in Shāţibī's legal thinking social change and legal change are so much interrelated that one cannot be understood without the other. Although this relationship makes Shāţibī's views on change importantly relevant to the problem of our dissertation, yet this complexity renders the analysis of his concepts much more difficult. This chapter, nevertheless, attempts to outline Shāţibī's concept of social and legal change in Al-Muwāfaqāt and Al-l^etişām.

In reference to Shāṭibī's terminology, this chapter will deal with the following concepts: Sharī'a, 'Āda, bid'a, and ijtihād. An analysis of the term sharī'a

reveals Shātibi's concept of law in reference to change. The concept of $\leq \overline{a}da$ explicates the notion of social change and its relation to law. <u>Bid'a</u> presents a concept of legal change which is generally linked with social change. The concept of ijtihād explains the interaction of social and legal change.

Sharīsa

-**T** -

2. >

 Shāţibī has defined most of the essential terms which he uses, but a definition of <u>Sharī<a</u> does not seem to be attempted. An understanding of a term can, however, be obtained from other words used as its opposites or used in connection with it.¹ Accordingly, we find that Shāţibī's concept of <u>Sharī<a</u> is associated essentially with the notion of "revelation".

On the epistemological level the terms <u><aql</u> (human reason) and <u>hawā</u> (desire)² are used as terms opposed to <u>Sharī<a</u>. Ontologically <u>Sharī<a</u> is contrasted with <u>kawn</u> (being)³. This semantic opposition has significant implications for the concept of <u>Sharī<a</u>. Firstly, it indicates that law is not arbitrary and merely based on personal likings. Secondly, the values on which <u>Sharī<a</u> is based are not determined by human reason. Thirdly, it implies that being opposed to <u>kawn</u> which is changing, <u>Sharī<a</u> is eternal and abstract. Shāţibī distinguishes between <u>kawn</u> and <u>Sharī<a</u> also as two different aspects of Divine Will. <u>Kawn</u> is the expression of the creative aspect of Divine Will, and <u>Sharī<a</u> is the expression of the legislative aspect. This distinction implies that in the first aspect there is a necessary connection between will and the occurrence of an event. This connection is not implied, however, in the Legislative Will. The details on this point have been discussed earlier.

€.

2.4

The term <u>Sharī</u> is also used as synonymous with <u>wahy</u> (revelation)⁴. Since <u>wahy</u> is a process and <u>sharī</u> is not, the synonymous use of the two makes sense only if we understand <u>Sharī</u> as the substance of the process of <u>wahy</u>. As for explanations of the term through synonyms or substitutes for it, the Qur'ān is equated with the <u>Sharī</u> a.⁵ The application of the term is also extended to <u>Hadī</u>th of the Prophet, the <u>Sunna</u> of the Prophet and that of his companions; ⁶ but whereas the rulings in the Qur'ān are certain (<u>qat</u> i), in general and in details, the <u>Sunna</u> is certain only in general and is but probable (zannī) in detail. ⁷ The absolute and original <u>Shāri</u> is Allāh only. The Prophet, muftis and <u>mujtahids</u> are also considered to be <u>shāri</u> by Shāțibī, but they function on God's behalf, ⁸ and not in their own right.

The characteristics of <u>sharīća</u> that Shāţibi has enumerated are the following: blessed (<u>mubāraka</u>)⁹, Arabic ¹⁰, general (<u>ummiya</u>)¹¹, universal (<u>cāmma; kulliya</u>)¹², liberal (<u>samḥa</u>)¹³, convenient (<u>sahla</u>)¹⁴, protected (<u>ma<şūma</u>)¹⁵.

The other terms that are associated with the term "Islamic law" are figh and uşūl al-figh.

The term <u>figh</u> is used by Shāṭibī more in its literal and essential meaning than in the technical sense. The phrase <u>figh al-sharī</u>^(c) as used by Shāṭibī may mean "under-

standing of the <u>sharifa</u>", "investigation of the <u>sharifa</u>" or "establishing the meaning of <u>sharifa</u>".

Shāṭibī, however, uses the term "uṣūl al-fiqh" more often and in a certain technical sense. In al-I<tiṣām, he defines it as follows: "Uṣūl al-fiqh" means f to infer, by method of f induction, universal f principles f from the evidences f of Sharī<a f until the mujtahid finds them conspicuous (naṣb 'ayn), and the searcher 17

The equations he uses to explain the term show that his concept of $\underline{usul al-fiqh}$ is very closely connected with that of <u>Sharīća</u>. He argues that $\underline{usul al-fiqh}$ have the same relationship to <u>sharīća</u> that the <u>usul al-dīn</u> (the principles of religion; 18 <u>kalām</u>) have. He explains that to Qādī Ibn al-īayyib <u>usul al-fiqh</u> meant the principles of the science of <u>sharīća</u> (in the epistemological sense), whereas to 19 Imām al-Juwaynī they were the proofs (<u>adilla</u>) of <u>sharīća</u>. Shāţibī did not consider them either as proofs or directives for <u>sharīća</u>, but as the principles derived 20 inductively from the underlying universal laws in <u>sharīća</u>.

A summary of Shāṭibī's view on the origins of sharī<a also reflects his concept of sharī<a as a "revelation". According to Shāṭibī, sharī<a is the light of knowledge. In their pre-sharī<a state, mankind sought their ends at random. Because of its inclination to passions and desires (hawā), the human reason was unable to discover the maṣāliḥ (good) of all mankind. Its efforts led only to confusion. It drew conclusions with defective analogies; it sought health from a sick body. Mankind was walking in reverse; yet it believed that it was on the right path. This state of self-

1

assurance led to sheer determinism (<u>ijbār</u>) in the very concepts of freedom, power, and choice (<u>aqdār</u>). Necessitarianist values (<u>hukm al-idtirār</u>) were attached to acquired acts (<u>al-af<āl al-muktasaba</u>). Men were in this plight, when God showed His Grace and sent prophets to every people with <u>sharā'i</u>[<] (pl. of <u>sharī'a</u>). The Prophet explained to the peoples in their own languages what was the true, right path. ²¹

This account of the origin of <u>shari</u> is difficult to be interpreted in terms of time because Shāţibi on other occasions argues that there never was a time without <u>shari</u> a. This account then can be understood either in a mythological sense or in the sense that Shāţibi was referring to what he calls the <u>fatra</u>, the period of interval in between periods of revelations of <u>sharā'</u> is

The last in these series of revelations was sent to Muhammad b. Abdullah. God revealed to him His Book, the Qur'an. This book established the criterion of distinguishing certitude from doubts.

The Qur'ān is the totality (<u>kulliya</u>) of <u>sharī'a</u>, the fountain of widsom. It is the source of <u>sharī'a</u>. ²³ The Qur'ān was revealed first in Mecca and was continued in Medina. The universal principles were revealed in Mecca; they included among other things, belief in God, the Prophet and the Hereafter. These were followed by general rules such as those about prayer, alms, etc. Along with this were revealed general ethical rules about justice, virtue, patience, etc. These rules generally concerned religion and social practices in the pre-Islamic period. Very few specific

4 2

4.9-

rulings were revealed in Mecca. When the Prophet came to Medina the territory of Islam had expanded. From then on, the general principles revealed in Mecca were complemented with additional particular rulings pertaining to contracts, prohibition of intoxicants, proscription of penal punishments, etc.

್ಷ

The need for detailed rules might have been felt because of various reasons. Often there arose disputes among the people which required detailed judgments. There were controversies also because many people had accepted Islam while retaining their pre-Islamic mental attitudes and social habits. God revealed to them all that they needed, ²⁵ sometimes in the Qur'ān and sometimes by the <u>Sunna</u>. Thus the whole of <u>shari<a</u> came to be completed in Medina, and God declared, "Today I completed your religion..."

The<u>fu</u>qahā' attended to the task of applying these rules and prescriptions in further details. They searched the basis of these rules in order to apply them to particular cases. This process was the method of <u>ij</u>tihād (legal reasoning).²⁶

The above account indicates that <u>shari'a</u> insofar as it is a revelation of laws by God, was completed in the days of the Prophet. As to the question of change in the days of the Prophet, Shātibī maintains that the fundamental principles revealed in Mecca were permanent; they were never changed or repealed, because they were the necessary and essential matters. Abrogation (<u>naskh</u>) occurred only in particular details, not in universals.²⁷

In other words, the finality and immutability of Islamic law in the days of the

Prophet meant the non-changeability of fundamentals of <u>shari</u> a only. Legal change is, however, possible in individual cases. The question then is to ask which legal institution does Shātibi regard as responsible for the function of legal change?

As mentioned elsewhere, two legal institutions are involved in this matter, futyā and gada: Shāțibi considers gada? and futyā both as wilāyāt (administrative offices). ²⁸ Like the establishing of a government, they are also $\frac{kifa^2iya}{k}$ (societal obligations). ²⁹ In Shātibi's structure of maqāsid, kifā'iya in contradistinction to sayniya which are specifically individual obligations of each person, are an obligation for the society as a whole, somehow to be fulfilled though each individual may not be involved. Kifa 'iva, however, are still essential and necessary as they are among the maqasid of shariga. They are indeed complementary to <u>sayning</u> because they make the fulfillment of the latter possible. $\int \circ Kifa'i\chi^{\gamma}_{a}$ aim at achieving the common good (masāliņ ʿāmma) for all the people, because one individual by himself cannot take care of his interests or his family. How can he attend to the good of the whole society? One necessarily needs co-operation with others. Consequently one works for his own benefit but also toward the interests of others; thus is the benefit of all achieved by all. Such is the manner in which general (public) institutions such as khilāfa, wizāra, nigāba, gadā' and futyā came into being. They were recognized by shari'a in the public interest because were they to be abandoned, the social order would be destroyed.

1

This clarification was necessary to show that \underline{futya} and \underline{qada} being societal obligation are necessarily linked with society and hence with social change. Their being classified among \dots Kifa $\overset{}_{iya}$ also implies that the <u>mufti</u> and the \underline{qadi} both perform their functions on behalf of the whole of society. Consequently the interests of the society as a whole are required to be considered.

Shāṭibī does not spell out the distinction between the functions of qādī and <u>muftī</u>, but from his discussion of <u>fatwā</u> and <u>iqtidā</u>, which follows, it can be assumed that, properly speaking, the institution of <u>futyā</u> was regarded by Shāṭibī as responsible for the interpretation of law and the adoption of legal changes.

Shāṭibī believed the <u>muftī</u> to be the deputy and successor to the Prophet. Like the Prophet, a <u>muftī</u> relays the commands from God, interprets <u>sharī</u> a for the people and executes them. More important, Shāṭibī regards the <u>muftī</u> to be a lawgiver in a certain sense. He explains this opinion in the following manner.

A <u>mufti</u>'s knowledge of <u>shari</u>'a is gained either through transmission of tradition or through deduction. In the former case he performs as a <u>muballigh</u> (communicator), in the latter he is a law maker (<u>inshā</u>' <u>al-aḥkām</u>) which is the function of a <u>shāri</u>'. This function qualifies the mufti as a true successor (<u>Khalīfa</u>) to the Prophet. ³¹

In regard to the question of authority (\underline{iqtida}') , ³² Shatibi divides the wielders of authority into three categories. First are those in whose actions freedom from error ($\underline{\epsilon_{isma}}$) can be demonstrated. In this category are included the Prophet, and the consensus of those people of whom it is customarily believed either that they can-

4.

not unanimously agree on error, or that such a consensus is sanctioned by sharifa.

4.5

Second are those who by certain specific acts claim the obedience of others. This category includes <u>hukkām</u> (rulers, officers) who pronounce this claim in the form of commands, and prohibitions or by signature. The third category of authority is one in which none of the above features exist. The first category is admitted in law without any doubt. The other two, however, need further consideration. The reason is that the objectives of authority in the case of these two types cannot be unanimously determined. Thus Shāţibī does not admit their authority to command obedience in law.³³ He, nevertheless, accepts the authority of a judge (<u>hākim</u>) in the application and execution of law.³⁴

Ŋ,

Shāțibi's concept of authority seems to be based on two notions: <u>cişma</u> and <u>qaț</u>ci<u>v</u><u>à</u>. Though <u>cişma</u> implies freedom from error, yet it cannot be understood in the sense of infallibility in Shāțibi's terminology. To him <u>cişma</u> is equivalent to <u>hifz</u> (safety, protection, assurance) from change or transformation; but not in a static sense. He explains that the <u>cişma</u> of the Qur'ān has been attained through its wider study, preservation and the development of sciences relating to the Qur'ān. The <u>cişma</u> of the <u>shari'a</u> in the hands of the generation succeeding Muḥammad came to be as they inferred the rules of <u>shari'a</u> by seeking its objectives from the Qur'ān and Sunna, sometimes literally, sometimes from its implications and sometimes by deducing the 'cause' (<u>cilla</u>) of the command. They applied these rules to cases that were unprecedented. In this way they made matters convenient for their successors. "This is the exact meaning of <u>hifz</u>..."³⁵ Shāțibī's notion of <u>aqt</u>^ci<u>y</u><u>a</u> will be discussed later. What is important here is to note that the consideration of certain conditions that would assure the continuity and permanence of the rules of law, is essential in Shāțibi's concept of law.

The institution of <u>futyā</u>, however, does not function in a vacuum. Law can be applied, interpreted or changed in reference to society. The problem, therefore, can be formulated in the following question. Does Shāţibī recognize the interaction of legal and social change? For the answer to this question we turn now to Shāţibī's view on <u>cāda</u>.

Ŋ,

۲Āda

●.(1) 注 透力/

> ***** 2 2 *****

Shāţibī's discussion of sāda turns around three problems; the constancy of sādāt, the possibility of their change and their relationship with <u>sharīsa</u>. Even his definition of <u>sāda</u> shows these predilections. According to Shāţibī <u>sāda</u> means nothing but that a given act, if it is supposed to happen without any impediment, happens only in a certain manner which is known by other similar acts. <u>36</u> Shāţibī's definition partially resembles the theory of determinism. This deterministic element is the constancy which Shāţibī calls <u>istiqrār</u> (persistence) and <u>istimrār</u> (continuity). <u>37</u>

The continuity of <u><u>s</u>d<u>a</u>t is a necessary condition without which the fulfilment of a legal obligation cannot be conceived. The other element is the certainty and predictability of the <u><u>s</u>d<u>a</u>, as Sh<u>a</u>tib<u>i</u> says "the occurrence of <u><u>s</u>d<u>a</u> in the world of existence is a known (<u>ma</u><u>c</u><u>l</u><u>u</u><u>m</u>) matter not a conjecture (<u>ma</u><u>z</u><u>n</u><u>u</u><u>n</u>)."³⁸ Both elements are such that their absence makes any law impossible. "If a divergence (<u>ikhtil</u><u>a</u><u>f</u>)</u></u></u>

is presumed in <u>'awā'id</u>,³⁹ it would necessitate a divergence in law-making (<u>tashrī'</u>); in classification of law (<u>tartīb</u>) and in promulgation (<u>khiţāb</u>), and the sharī'a would not be the same as it is now." 40

₹

1

Shāţibī uses the term <u><u><u></u>ada</u> in various meanings; sometimes he means simply habits and human behaviour, ⁴¹ on other occasions it is equivalent to custom. ⁴² It is also contrasted with <u><u></u><u></u>ibādāt so as to mean what the other jurists generally call <u>mu</u><u><u><u></u>amalāt</u>. ⁴³ In fact Shāţibī's use of the term is inclusive of all these senses. This interpretation is admissible if we recall that Shāţibī contrasts <u>sharī</u><u>a</u> or <u>amr</u> with <u>kawn</u>. ⁴⁴ The <u><u>ada</u>, then, would be related to <u>kawn</u> or the physical world, as the counterpart of <u>ahkām al-sharī</u><u>a</u>.</u></u></u></u>

Shāṭibī's concept of the continuity of <u>ada</u>, is questioned on two points, first on the ground that the continuity of a certain thing in this world is equivalent to the beginning of its existence, because for its continuity as for its existence it also requires an agent who, however, may possibly become non-existent. During the first period the continuity of the non-existence of that thing was possible, but when it was brought into existence, one of the two possibilities was achieved, i.e. its existence; the other possibility i.e. its non-existence still remains. When it is possible to conceive the possibility of its discontinuity, how can one talk with certainty about its continuity?

The second objection is that very often events occur contrary to <u><u><u></u></u><u>a</u>da (<u>Khawariq</u> <u>al-<u></u><u>a</u>da). This fact of actual occurrence supplements the above argument about the potential possibility of non-continuity of <u><u></u><u>a</u>dat. How then can it be main-</u></u></u>

374

tained that the occurrence of sadat is known with certainty?

Shāţibī replies that it is by tradition (sam^c) that we know the possibility of continuity. The possibility of discontinuity maintained in the objection does not contradict the position of tradition, because the notion of possibility is logical (<u>al-jawāz al-^caqlī</u>), while tradition is not concerned with possibility but with occurrence (<u>wuqū</u>^c). Many a thing happens although logically it is possible for it not to happen. In fact the term "possibility" (<u>jawāz</u>) refers to the "possible" itself, while "necessary" (<u>wujūb</u>) and impossible (<u>imtinā</u>^c) refer to some external factor. Thus the latter cannot be contradictory to the former.

Ŋ;

Second, the certainty and predictability in <u>sādāt</u> do not concern each and every ada. Essentially they concern the universals of being (<u>kulliyyāt al-wujūd</u>), not the indivduals. Hence if an individual deviates, this does not destroy the universal. The argument from <u>khawāriq sāda</u> refers to individuals. Furthermore, it is the occurence of <u>khawāriq sāda</u> that assures our knowledge about the universal <u>sādāt</u> and <u>vice versa</u>.

From the above it can be seen that in a fashion similar to his views on <u>shari'a</u>, Shāţibī believes in the continuity of only those <u>cawā'id</u> which are universal. The acceptance of their continuity is not only an actual fact, but it is also necessitated by the requirement of a stable base for law. The <u>cawā'id</u> which accept change are more in number than those which are immutable. <u>Kharq cāda</u> is not a proper example of changing <u>cādāt</u>; in fact <u>kharq cāda</u> is a breach of a universal <u>cāda</u>, and hence it happens seldom. Shātibī, therefore dismisses <u>kharq cāda</u> as a

-4

÷

serious objection to the continuity of cadat, as well as an example of change.

Shāţibī classifies $\int_{a}^{b} \frac{d^2}{dt} \frac{d^2}{dt} \frac{d^2}{dt}$, in reference to shāṯi^c into two kinds: shaīy², which are introduced or sanctioned by shaī² and <u>awā'id jāriya bayn al-khalq</u>, 46 those which are current among people. These two categories are not exclusive of each other; the first category also belongs to the habits and customs of the people. Shar^c necessarily gives consideration to <u>awā'id jāriya</u>, because in fact Divine law (<u>sunnat</u> <u>Allāh</u>) corresponds with the <u>awā'id</u> in general; hence the <u>sharī^c</u> was instituted commensurate with the institution of <u>awā'id</u>. 47

Shāṭibī believes in the relationship of <u>sharīča</u> to <u><u>c</u><u>ada</u> more than in the relation between <u>sharīća</u> and <u><u>caql</u>. As mentioned earlier, it was in the periods of <u>fatra</u> that the philosophers (<u><u>cuqalā</u>) claimed to know good and evil by reason alone.⁴⁸ According to Shāțibī, this was possible, in fact, only because the values of good and evil already existed as instituted in <u><u>c</u><u>adāt</u>, although they were confused. This is why Shāțibī finds that the sharā'ic have not rejected <u><u>c</u><u>adāt</u> entirely. In the case of the <u>sharīča</u> of Muḥammad, indeed, the <u>sharīča</u> confirmed most of the <u><u>c</u><u>adāt</u> practiced by the people in the pre-Islamic period. Examples of such laws are the following <u><u>c</u><u>adāt</u> which were regarded as good in the pre-Islamic period and were adopted by Islam: <u>diya</u> (blood money), <u>qasāma</u> (compurgation), gathering on <u><u>carūba</u> (the ancient Arabic name of Friday) for sermons, <u>qirād</u> (loan), etc.⁴⁹ Shāțibī illustrates the relation of <u>sharīča</u> and <u><u>c</u><u>ada</u> by the case of <u>khamr</u> (intoxicant). "It was habitually used in pre-Islamic days. Islam came, and left it intact in the</u></u></u></u></u></u></u></u></u>

period before Migration and a few years after. The share did not

376

pronounce any law regarding kham r until the verse 'they ask you about kham r and maysir...'Then he explains that "the fundamental rule of sharis a is that when an evil (mafsada) in a thing transgresses the good (maslaha), it will be evaluated as evil. The evils are prohibited, hence the reason for its prohibition is clear. In cases where the sharis has not pronounced the prohibition, even though this aspect of its evaluation is apparent, the people will act upon the supposition that the original law established by the continuity of practice (sāda) remains in tact."

Shāțibi's discussion of the relationship of <u>shari'a</u> and <u>sāda</u> implies the aspect of change as well. The <u>shari'a</u> can change <u>sāda</u> in certain cases, and <u>vice versa</u>, but more important is the fact that when a change takes place within an <u>sāda</u>, it also effects a change in the <u>shari'a</u> rule. A thing which was relatively good becomes evil or vice versa; the <u>shari'a</u> has to adjust itself accordingly. This takes us to Shāțibi's view of legal change. First we will discuss the aspect of change in <u>sāda</u>; then the problem of legal change will be dealt with.

It should be noted here that in the usage of the term "change" Shāţibī includes both 'horizontal' and 'vertical' senses of change. The former is the change which manifests itself in the differences in <u>cadat</u> among various societies, cities, countries, etc. The latter is the replacement of old <u>cadat</u> by new ones, or the development of these <u>cadat</u> by additions or modifications. For the 'horizontal' his term is <u>ikhtilāf</u> and for the 'vertical' he uses the terms "taghyīr" and "tabdīl".

Beside <u>awā'id shar'iyya</u> which do not change, Shatibi divides <u>awā'id jāriya</u> into two: first, <u>al-'awā'id al-'āmma</u>, which do not change with time, place or

· · ·

* *

.

state and second, those which change. In the first category, Shājibī mentions the <u>awā'id</u> of eating, drinking, joy, sorrow, acquiring nice things, etc. The evaluation in these categories has been established on the basis of the awa id of past generations; they have never changed. In fact they are based on the Divine law in Creation [law of nature].

In the second category are the <u>cawa</u>^{*}id such as the forms of dress, styles of dwellings, etc., which change with time, place, and states. In this category, therefore, it is not correct to evaluate <u>cawa</u>^{*}id absolutely on the basis of past experience. Even if there is found some external evidence which proves the continuity of such an evaluation, it must be kept in mind that this decision of evaluation was made because of some external factor, not because of <u>cada</u> itself. Similarly the decision of evaluation in the present cannot be carried on into the future, or to the past. The reason for this temporal limitation is the probability of change.⁵¹

Shāţibī discusses five senses of this change.⁵² The first is exemplified where the change is from good (<u>husn</u>) to evil (<u>qubh</u>) and <u>vice versa</u>. For example, keeping the head uncovered is regarded as evil in eastern countries but not so in western countries. Second, there is the change that results from the different interpretations of objectives. This change usually takes place among various peoples (<u>'uman</u>), but it also occurs within one people, like the differences of the technical vocabularies among men of various trades and professions. Third, there is the difference of acts in <u>muʿāmalāt</u> (dealing with each other), like the <u>cāda</u> (custom) of receiving a dowry (<u>sadāq</u>) before the consummation of a marriage. Fourth, there is the change resulting from the difference of considerations which are external to the acts in question, for example, the difference in the criterion of maturity (<u>bulūgh</u>)

among various people, whether on the basis of puberty or on the basis of age. Fifth, there is the case of irregular <u>awa</u>'id which have become regular <u>adat</u> for some people, for instance a person who is injured in such a place that he can no more urinate in the regular manner. The irregular manner of his urinating is an <u>ada</u> for him.

••

Ŋ

The illustrations of changing 'awā'id show that Shāţibī admitted change in 'āda in both 'horizontal' and 'vertical' senses. This would then imply that sharī'a insofar as it is related to 'āda must also admit change.

Shāţibī's discussion of the problem of change in <u>sharī'a</u> can be analysed in at least six aspects. The first aspect is that of the universal principles on which the Sharī'a is based and which underlie the Meccan part of the Qur'ān. These principles are also called <u>maqāşid al-sharī'a</u>, and they never change. What are these universal principles? Examples drawn from the Meccan revelations have been mentioned earlier.

The second aspect is that of $\int_{1}^{d} awaid \int_{1}^{d} shar'i ya}^{d}$. As mentioned earlier, they are the <u>'awai'id</u> introduced or sanctioned by <u>shari'a</u>. In contrast to the universal principles, they are more specific and concrete rules of law. According to Shātibī they also do not change. "Because," Shātibī explains, "they are among the matters included in the rules of <u>shari'a</u>. Hence they do not change. Even if the opinions of the <u>mukallafin</u> (subjects of law) differ about them, it is not correct to change good into evil... For instance it cannot be argued that since the acceptance of the witness of slave is not disdained by the <u>'adat</u>, hence it is allowed...If

Į.

this were permitted it would constitute abrogation of the rules which are constant and continuous, whereas abrogation after the death of the Prophet is not valid (<u>bāțil</u>)."⁵³

The third aspect concerns those <u>`awā'id</u> which are either a means or a mediate cause (<u>sabab</u>) to the fulfilment of certain rules of sharīća, like the physical capabilities to perform an act, the <u>'awā'id</u> about maturity, etc. "Since they are mediate causes (<u>sabab</u>) for the 'caused act' (<u>musabbab</u>), they are also commended by the lawgiver. Hence there is no difficulty in giving them due consideration and accepting them as the basis of rules."⁵⁴ The problem then is to ask whether the rules of <u>sharīća</u> would change if this basis changed. Shāțibī replied in the affirmative, saying "The rule of <u>sharīća</u> will always be in consonance with these <u>'awā'id</u>."⁵⁵

The fourth aspect is that of $\frac{d}{\sqrt{\frac{\alpha}{2}} + \frac{\alpha}{2}} + \frac{\alpha}{\sqrt{\frac{\alpha}{2}} + \frac{\alpha}{2}} + \frac{\alpha}{2} + \frac{\alpha}{2}$

The fifth aspect concerns the legal changes which imply that certain matters are not covered by <u>Shari'a</u> so that additional rules are required. Shātibi regards this aspect as requiring further investigation. A certain <u>hadith</u> lays down the rule that 'the matters on which the lawgiver is silent, are forgiven (<u>'afw</u>)'. This <u>hadith</u> admits that <u>Shari'a</u> does not cover everything. The <u>hadith</u>, however, renders the position of those jursists questionalbe who maintain that there is nothing <u>maskūt</u> <u>'anhu</u> (where lawgiver is silent), because they claim that every case is either covered by the text (manşūş) or is coverable by analogy with the text. To avoid

ĺ

the conflict with the above $had\bar{i}th$, they explain that the Shari^{c's} silence can be removed in a number of ways; for instance, by way of istishab, or by referring to the Shari^{c's} explicit proclamations in laws revealed before Muhammad, or by confining the interpretation of the text to the generality of a specific ruling by disregarding its modifications if they are not mentioned immediately after the ruling. ⁵⁶

Shāţibī goes into a detailed analysis of the nature of this silence. He divides this silence into two types. First is the type of silence where there was no immediate cause for issuing a command and hence the lawgiver did not say anything. Second is the type where such a cause existed but the lawgiver still kept silent. Quite naturally, the second type is a form of prohibition. To interpret this silence as absence of ruling would thus lead to introducing a bid^ca.

The first type is what can be properly called "silence". Because of its obvious significance, Shātibi's explanation on this point needs to be quoted in his own words.

One of them / the aspects of silence/ is, that he keeps silent because there is no motive $(d\bar{a}ciya)$ that necessitates it / the ruling/...For instance the events that occurred after the (death of the) Prophet. They certainly did not occur in his lifetime so that one can say that the lawgiver said nothing about them, even though they occurred. They took place later and hence the people of <u>Sharica</u> were obliged to examine those events and to execute them according to what had been established as universal principles.

The new things that the righteous ancestors introduced in Islamic law belong to this type. The examples of this type are the <u>masalih</u> <u>mursala</u> such as the collection of the Qur'an etc....These are some of those things that were not discussed in Prophet's days, nor were they enquired about. Nor did they find place in social practice so that a cause for such an enquiry might arise. ⁵⁷

4

As a second instance, we may cite the category of <u>safw</u> which according to Shāțibi falls between the <u>halāl</u> (lawful) and the <u>harām</u> (forbidden). This category also proves that he not only admitted the possibility of matters not covered by <u>sharis</u>, but also that they fall under the category of <u>safw</u> (silence or indifference of the lawgiver). ⁵⁸

The sixth aspect of change is what Shātibī calls <u>ihdāth fī al-sharīča</u> (innovation). Shātibī does not believe in the legitimacy of <u>ihdāth</u>. He argues that <u>ihdāth</u> occurs in <u>sharīča</u> in three ways. It happens first because of ignorance of the objectives of law. This is either ignorance of the tools that lead to an understanding of the objectives, such as ignorance of the Arabic language and its grammar, or it is ignorance of the objectives themselves. N

A second reason for <u>ihdath</u> is <u>tahsin al-zann bi'l 'aql</u>, to decide the value of a thing on the basis of rational speculation. A third cause is following one's own desires in seeking the truth. In such a case the desire dominates and even conceals the true evidences and leads to false ones.⁵⁹

<u>Bid</u>^ca is one aspect of this ihdath. <u>lhdāth</u> can occur in all subject matters of <u>sharī</u>^ca, while <u>bid</u>^ca, according to Shātibī, is limited to certain aspects. This distinction requires a rather detailed analysis of Shātibī's concept of <u>bid</u>^ca.

Bid'a

Shātibi's book <u>al-litisām</u> is specifically designed to discuss the problem of <u>bid'a</u>. We need not go into the details of his arguments; what concerns us here is to discuss

A

2.5

4

 $\Delta_{\mathbf{x}}$

bid^ca as a legal change and the problem of its legitimacy.

17

ۍ.

1.0

Shāţibī vehemently condemns <u>bid</u>^ca on at least nine grounds.⁶⁰ His reasons for condemnation can be summed up by saying that since <u>sharī</u>^ca is complete and final, anyone who innovates, commits among other sins two grave errors. One is the implication of equality or rather superiority to God, the original lawgiver, because the promulgation of <u>bid</u>^ca implies that the innovator knows more than God about <u>sharī</u>^ca. Second, he relies more on human reason and desires than on the intentions of the lawgiver.⁶¹

Ň,

Shāţibī's condemnation of <u>bid</u>^ca must not, however, be taken as condemnation of any and all legal changes. Not only would such a conclusion not conform with his views discussed above, but it would also give a wrong idea about Shāṯibī's understanding of the concepts of <u>bid</u>^ca and <u>ijtihād</u>.

Shātibī explains that etymologically <u>bid</u> comes from <u>bada</u> which means to invent something new, the like of which has not existed. In a technical sense, however, this "new-ness" and "invention" is meant in reference to <u>sharī</u>.⁶² In reference to <u>sharī</u>.⁶² human acts can be of three kinds: required, prohibited, or voluntary. The category of prohibited actions is governed by two considerations. First, simply that it is prohibited by law, second, that it literally opposes the rules of <u>sharī</u>.⁶³

Shāțibī gives two definitions of bid^ca. The first is a definition that does not include cādāt; the second includes both <u>Sibādāt</u> and <u>Cādāt</u>. The first definition is as follows: "A way (<u>tarīqa</u>) of innovation in religion (<u>din</u>) that resembles the way of <u>sharīća</u> (<u>tudāhī al-tarīqat al-sharćiyya</u>) and which is intended to be followed in order to strive in the utmost toward obedience (<u>taćabbud</u>) to Allāh."⁶⁴ The second definition replaces the phrase 'in order to strive...' with the following "with the same intentions that Shārić aims for."⁶⁵

× .

4

1

Shāţibī even goes further to clarify the qualifications included in the definition. It is relevant to note some of these qualifications. The qualification of 'religion' (<u>dīn</u>) is significant because according to Shāţibī "if this way of innovation belongs to <u>dunyā</u> (mundane matters) exclusively, it would not be a <u>bid</u>. Examples cited are innovations in crafts, in plans of cities, etc." ⁶⁶ The qualification of 'innovation' excludes those matters which have their bases in shar .

The qualification of 'intention of similarity with <u>Shari'a'</u> is also very important. Shāţibī is admitting that the intentions of the innovators are not bad in themselves, but he implies that they misunderstood the purpose of <u>Shari'a</u>. Shāţibī does not equate <u>bid'a</u> with heresy only because it is a new thing. The key terms in this respect are 'intention' and the 'right understanding of the purpose of Shari'a .' How is this right understanding to be judged?

In one respect right understanding means correspondence of both intention and acts with the purpose of <u>Shari'a</u>. Shātibi elaborates the relationship of intention and act to the purpose of <u>Shari'a</u>, by describing four situations. First, if the intention of an act and the act itself conform with the purpose of <u>Shari'a</u>, the act certainly is valid. Second, the act is not valid if the act and the intention do not conform with <u>Shari</u>ⁱa. Third and fourth are the cases where one of them (the intention or the act) conforms and the other does not, Shātibī makes a distinction; if the intention conforms and the act does not, it is to be called <u>bid</u>ⁱa. If the act conforms but the intention does not, the act belongs to the category of <u>riva</u> and hyprocisy.⁶⁷

AMasalih Mursala illustrate the type of new things where the intention and the act both conform to the purpose of <u>Sharīća</u>.⁶⁸ An example of this type is the levying of new taxes in addition to those prescribed in the texts. The conformity of the act with the purpose of <u>Sharīća</u> and the intention in this case show right understanding of <u>Sharīća</u>, and, further, the intention does not conflict with the objectives of Sharīća.

In the case of <u>'Ibādāt</u> this intention leads to an exaggeration in <u>ta abbud</u>. For instance, in Shātibī's period the practice of chanting the names of God (<u>al-dhikr</u>) in congregation was considered obligatory.⁶⁹ This intention is absent in <u>adāt</u>. Nevertheless wherever this intention (of similarity) is absent in a new thing, even though there be similarity in actuality, such a new thing will not be regarded as <u>bid'a</u>.⁷⁰ Shātibī gives the following as the examples of the last type of new things: taxes levied on property in a specific proportion and amount that resemble <u>zakāt</u>; use of sieves; washing the hands with <u>'ushna</u> (potash); erecting lofty buildings, etc.⁷¹

Lack of such distinctions as above, in various types of 'new things' in Islamic law, had made the concept of 'bid'a' both confusing and controversial. The jurists who would accept nothing new in Islamic law rejected bid'a in any sense. Some jurists maintained a broad distinction between good and bad <u>bid</u>^ca. Scholars such as lbn ^SAbd al-Salām and Qarāfī have even divided <u>bid</u>^ca into five categories corresponding to the five categories of legal valuation: obligatory, recommended, indifferent, reprehensible and forbidden.⁷² Shātibī regards such a division as meaningless and irrelevant. With the exception of those <u>bid</u>^ca mentioned by these scholars in the categories of 'reprehensible' and 'forbidden' the others are not bid^ca at all.

Shāţibī refined the meaning of the concept of <u>bid</u>^ca and made it more precise by clarifying his terminology and fitting it into a proper framework of legal philosophy. He showed that the <u>bid</u>^ca are of two kinds only, <u>haaīqiyya</u> (absolute), '<u>idāfiyya</u> (relative). $\frac{AL}{A}$ <u>bid</u>^ca <u>bid</u>^ca <u>ida</u><u>fiqiyya</u> is that which is not proven by any <u>shar</u>^cī evidence like the Qur'ān, Sunna, <u>ijmā</u>^c or a reliable basis of reasoning, neither in general nor in particular. $\frac{AL}{A}$ <u>bid</u>^ca <u>id</u><u>fi</u><u>ya</u> is that which mingles both aspects. In one aspect it is connected with <u>shar</u>^cī evidence; in the other it is not. It is only in the latter aspect that it is <u>bid</u>^ca.⁷³

The common point in the two definitions of bidsa given by Shāṭibī, is the intention of the innovator to equal the lawgiver, and this is possible in saāt as well. Obviously this common point can be taken as the essence in Shāṭibī's concept of bidsa. Real bidsāt according to Shāṭibī, however, are only those which belong to sibādāt. Shāṭibī argues this point in two ways. First he refutes the thesis of his opponents who maintain that were innovation possible in sibādāt, it would also be possible in sādāt. Furthermore, there are a large number of ahādīth which predict the occurrence

3 8 - of new things in later periods.

Shāţibī dispels this objection by saying that the controversy is not about the possibility but about the actual occurrence of bid'a in 'ādāt; hence the argument of 'possibility' is not valid. As for predictions of changes in <u>ahādith</u>, the argument is misleading. These particular <u>ahādīth</u> do not call all of these changes <u>bid'a</u>, and moreover, these matters are not condemned there because they are innovations. Shāţibī continues by saying that were every new thing in <u>'ādāt</u> regarded as <u>bid'a</u>, then every change in matters such as eating, clothing, speaking, etc., would stand condemned.⁷⁴ He sums up his argument in the following fashion:

> There are <u>awa'id</u> which change with time, place and name. If every change is condemned then everyone who differs in this respect with those Arabs who were in contact with the companions of the Prophet... will be considered as not following them and hence deviating from the right path. This is quite difficult to accept.⁷⁵

The implications of the above statement are fundamentally important for the question of legal change. Shāṭibī, here, is saying that there are large areas of life --- in fact everything except <u>`ibādāt</u> --- where the concept of <u>bid</u>^ca does not apply. The implication is that <u>Sharī</u>^ca does not control these areas of life or at least does not control in the same sense that it controls the relations of man and God.

Shāţibī's second manner of argument against including <u>cādāt</u> among <u>bidcāt</u> is the consideration of <u>tacabbud</u>. As mentioned earlier, from the viewpoint of <u>sharc</u>, acts of the <u>mukallafīn</u> are of two kinds; <u>cādāt</u> and <u>cibādāt</u>. It is generally agreed that <u>cibādāt</u> are <u>tacabbudī</u>, but there is disagreement whether <u>cādāt</u> are also <u>tacabbudī</u>.

Shāţibī defines tasabudī as "that the meaning of which cannot be rationally understood from the act itself". Matters such as ritual cleanliness, prayers, fasting and pilgrimage are all tasabbudī. Matters such as whose meanings can be rationally understood and whose goodness or badness can be known are <u>sādī</u>. Examples may be seen in the acts relating to sale, marriage, lease and punishment for crimes. It is in this sense that <u>sādāt</u> are not <u>tasabbudī</u>, and hence the term <u>bids</u> is not applicable.⁷⁶

In the light of what has been said so far, it is possible to reconstruct Shāṭibī's theory of social and legal change as follows.

One finds a significantly elaborate conception of social as well as legal system in Shāţibī's thought. The conceptions of these systems emerge from Shāţibī's analyses of <u>cawā'id</u> and <u>Sharīća</u>. It must, however, be noted that Shāţibī sees both of these systems originating from one source, God, yet as they represent two different levels of Divine Will, they do not function in the same way. <u><u>c</u>Ada represents the level of <u>dirāda takwīnī</u>, where man has no choice but to obey the rules. In <u>Sharīća</u>, obedience depends on man's choice. Human acts insofar as they belong to <u>irāda</u> <u>takwīnī</u>, obey the laws of <u>takwīn</u> necessarily, but those acts which belong to <u>irāda</u> <u>at tashrīčī</u> <u>hecessarily need man's intention and volition for obedience. <u>Awā'id</u> which cover habits in reference to individual and social practices in reference to individual and social practices in reference to the community and laws of nature (<u>kulliyāt</u> <u>al-wujūd</u>) in reference to kawn, provide the determinism that stabilizes the function of a social as well as a legal system. There are, of course, some deviations (khawārig) from this continuity of <u>cawā'id</u>, but these deviations establish (rather</u></u>

a f

 \mathbb{C}^{\ast}

than destroy) the factor of stability. Such <u><u>awa</u>'id</u> which continue are called <u>al</u> <u><u>cawa</u>'id <u>mustamirra</u> and the rules of <u>sharī'a</u> have their basis in this type of <u><u>'awa</u>'id.</u></u>

The connection of <u>Shari</u>^ca with the recurring <u>cawā</u>ⁱid makes possible for Shari^ca to be eternal and continuing. The eternity of <u>Shari</u>^ca does not originate from the continuity of <u>cawā</u>ⁱid, in the sense that the concepts and rules of <u>Shari</u>^ca become eternal because of these <u>cawā</u>ⁱid. In fact <u>Shari</u>^ca forms the ultimate basis which are abstract, universal and general and, thus, is believed to be unchanging. The continuity of <u>cawā</u>ⁱid makes the actualization of these ultimate bases possible.

Shāţibī clarifies that human reason alone could not discover these ultimate bases, hence this knowledge was revealed in two ways: on the one hand it was instituted in <u>awā'id</u>, and on the other, it was revealed through <u>Sharā'i</u>. Human reason was led either to total laxity or to sheer determinism in its attempt to discover these ultimate bases from <u>awā'id</u>. Consequently revelation of <u>Sharā'i</u> was necessary to save man from both extremes of legal attitudes. Leaving aside the discussion of how the revelation of <u>Sharī'a</u> differed from the attempt of human reason in this respect or how far it is a denial of any role to human reason, what is notable here is Shāṯibī's attempt to explain that <u>Sharī'a</u> aims at the good of mankind. This good is judged in relation to and on the basis of <u>awā'id</u>.

With the exception of universal principles, the <u>cawā'id</u> are, however, subject to change. <u>Sharī'a</u> is based on the unchanging principles of <u>cawā'id</u>, which are thus called <u>cawā'id</u> <u>shar'iyya</u>. Nevertheless the <u>cawā'id</u> which belong to human beings al-<u>(cawā'id jāriya bayn al-khalq)</u> may change. Since <u>Sharī'a</u> governs these <u>cawā'id</u>
as well, it must respond to these changes. The mechanism of this response gives birth to a legal system.

Shāţibī illustrates some such changes. The legal system faces one type of change when an individual, coming from a different social system, becomes the subject of another legal system, or the legal system is introduced where a different social system is in function. Obviously this change does not affect the fundamentals as it is supposed that the <u>cawā'id</u> on which <u>Sharī'a</u> is based are universal. Nevertheless this change requires to be accommodated in order to maintain the stability of the legal system. The second type of change occurs when the old practice no longer satisfies human needs, or when some new elements either from without or from within are introduced. Yet another type of change is introduced when social practices or institutions come into conflict with each other or with the purpose of law; this conflict may arise from a clash of personal interests or because of certain new developments in society. Whatever the cause, the change in a social system takes place in such a manner that it requires a legal system to respond to these changes.

The need to respond to social changes is essentially the result of the aim of the legal system at its own as well as at the stability of the social system. Since the possibility of change is unending and the applicability of the rules of law to these changes is limited, it is out of this necessity that the legal system is organized on rational basis both in its principles and methods, so that it is manageable by human reason. According to Shātibī since human reason alone cannot achieve such organization, Sharī^ca has provided men with general guidelines. Among these guide-

~ / ~

lines some can be tested in social practice and some not. Those which cannot be tested are <u><u>ibādāt</u></u> and they are to be obeyed as such. Of those which can be tested and which are rationally intelligible – they are <u><u><u>s</u>dāt</u>. The latter constitute the major area of human acts. Since it is possible to rationally organize the <u><u>s</u>dāt</u>, Sharī<u></u> has left the details to be worked out by legal reasoning.</u>

Y.

The Islamic legal system, insofar as the principles are concerned, is revealed in its entirety in the Qur'ān. Shāṭibī divides the injunctions of the Qur'ān into three categories: First the injunctions declaring lawfulness of things, second the declaration of prohibition and the third category is <u>cafw</u> which refers to those situations that are not covered by <u>Sharī</u>. Such situations will be decided by leagal reasoning, the guidelines for which are provided in the other two categories.

The decision about the situations not covered by <u>Shari'a</u> may mean application of established rules or it may mean extension of these rules. Shātibi does not accept extension in the case of <u>ibādāt</u>, but only in <u>adāt</u>. The reason is that in<u>ibādāt</u> it is only God who can decide what is good for men. Consequently, the Qur'ān being the last and complete revelation, contains all that man needs. Hence there is no need of extension of <u>ibādāt</u> beyond what the Qur'ān prescribes. Shātibī regards such an extension as <u>bid'a</u> which is to be condemned.

While <u>ibadat</u> are not rationally intelligible, the <u>adat</u> are. Moreover, often in the Qurran, an <u>illa</u> is mentioned in case of <u>adat</u> which means that <u>Shari</u> not only considers them intelligible, but also extendible.

Since the human reason is considered incapable of discovering the masalih, yet as

071

there will be no more revelation of Sharā'i after the Qur'ān, the situation demands that some system must be evolved to respond to the changes and to extend and apply the rules of law. According to Shātibi this is achieved through the A mufti is a successor of the Prophet both in institution of futyā and qaḍā. communicating the previous rules of law and in making new laws. A gadi applies and executes these laws. Whenever a situation arises where a member of the community feels that existing laws do not cover or satisfy this situation, he takes this question to the mufti, who investigates the problem and provides an answer on behalf of the legal system. Most often these enquiries arise out of ignorance of the rules of law. Nevertheless a layman as well as a qadi may often feel the rules of law to be insufficient in a particular case. In such an event, they are supposed to refer to <u>muffis</u>. Implicitly there is a rejection of social changes relating to law, without formally accommodating them in the legal system. A more significant implication is that law is to follow social changes, not to initiate 'To follow' here means 'to adjust itself', not 'to obey'. or plan them.

The process of legal reasoning through which a <u>muffi</u> responds to a social change in the framework of the legal system is called <u>ijtihad</u>. <u>ljtihad</u> is not simply a process of adaptation of legal theory to social changes, but it also aims at a rational attempt to accommodate the change and still maintain the continuity of a legal system.

UTL

6

Ľ,

For a better understanding of Shāțibi's discussion of <u>ijtihād</u> we need to consider a few technical details first.

A new case may either be provided for in the body of the rules of law or not. Further, this provision may either be implicit or explicit. An implicit provision may either be in form of general rules or in the form of permission derived from the absence of any prohibition. The need and method of legal reasoning both depend on the nature of these provisions. In some cases <u>ijtihād</u> may be continuously needed, while in other cases it may not be necessary.⁷⁷

The dependence of the method of legal reasoning on the legal provision means that to justify the validity or invalidity of the new case it needs to be axamined in reference to these provisions. This justification is exercized by demonstrating the correspondence of the essential elements in the new case with the basis of the legal provision. These bases which are called <u>manāt</u>, may be explicity known, or can be known by further ijtihād.

Shāṭibī divides <u>iļtihād</u> in reference to these <u>manāṭ</u>, into four types: 1) <u>Taḥqīq</u> <u>al-manāṭ al-ʿāmm</u>: General verification of the basis of the rules of <u>Sharīʿa</u>. In this case, the rule (<u>hukm</u>) in its <u>sharʿī</u> precept (<u>mudrak</u>), as its basis, is already established. The function of the <u>mujtahid</u> is to <u>verify</u> the application of these general bases in the subjects of law, but still in a general and universal sense.⁷⁸

In other words the basis of the legal provisions are examined so as to be applicable

The generality, here, is further explained by Shatibi to all the mukallafin. to mean that this type concerns anwa^c (species, types) of <u>mukallafin</u>, and not the ashkhāş (persons, individuals). It is called 'general' to distinguish it from the second type of ijtihad, which is specific. Shātibi illustrates this point with the Sharfi ruling that requires a witness to be sadl (just). The general and broad meaning of ^cadl is known, but to determine the characteristics and qualifications on the basis of which a witness can be universally described as ^cadl is the function of a mujtahid. In order to verify this qualification in case of a particular witness <u>ijtihā</u>d is required. Taqlid cannot solve this problem, because this process of verification can never end. Every new case is unique in Furthermore, Shari'a does not pronounce its rulings to itself in this respect.⁸¹ cover all particular cases individually. The rulings of Shari'a are general and abstract so that they can cover any new cases which are infinite.⁸²

This is because of the above reasons that Shātibi regards this type of <u>ijtihād</u> as evercontinuing. If one admits the discontinuity of this <u>ijtihād</u>, one makes the application and extension of the rules of <u>sharifa</u> impossible.⁸³ Human acts never happen in the abstract, they always happen concretely and as individual cases. If this type of <u>ijtihād</u> discontinues, the obligations of <u>Sharifa</u> will exist only in man's minds, and not in practice.⁸⁴

<u>Tahqiq al-manāt al-khāşş</u>: This type is different from the first one, as it concerns <u>ashkhāş</u> (individuals). This is more detailed and specific. For this a <u>mujtahid</u> relies more on <u>taqwā</u> (piety, prudence) and <u>hikma</u> (wisdom, inner reason).⁸⁵
 Tanqih al-manāt (the refinement of the basis of the rule). The proper quali-

fication (wasf) is mentioned in the text of the ruling but in conjunction with another matter; the task of separating and refining this qualification is done by ijtihad.⁸⁶ Shāţibī further explains that this type does not concern with the method of <u>qiyās</u>, but is rather a type of <u>ta'wīl al-zawāhir</u> (interpretation of the literal sense).⁸⁷ In a certain sense it also belongs to what Shāţibī calls <u>al-ijtihād</u> <u>bi-al-istinbā</u>ţ (reasoning by inference).⁸⁸

4) <u>Takhrīj al-manāt</u> (deduction of the basis of the rules). This type refers to a text of a ruling where <u>manāt</u> are not mentioned. The <u>manāt</u> are found through the process of deduction. The method is also called <u>al-ijtihād al-qiyāsī</u> or reasoning by analogy.⁸⁹

Shāţibī maintains that among these four types, the first is ever continuing,⁹⁰ but the continuity of the other three depends on their need. The reasons for the continuity of the first type have already been noted. Shāţibī explains the need of continuing the other three as follows:

> The new events which were not known in the past, in proportion to those which have occurred in the past, are very few because of the expansion of the body of rules due to the investigation and ijtihad of the preceding jurists. It is therefore possible to accept their decisions (taglid) in the major part of Shari'a." 91

The need for ijtihad was often justified by the jurists by arguing on the basis of <u>khilāf</u>. In other words if the opinion of scholars differed on a certain point, the case was considered open for ijtihād. For Shātibi this implied <u>khilāf</u> in <u>Sharīća</u>, which he vehemently rejected. He maintained that in its basis <u>Sharīća</u> is a unity; <u>khilāf</u> is neither intended to exist nor to be perpetuated.⁹² Hence <u>khilāf</u> in this technical sense is not sufficient to justify continuity of <u>ijtihād</u>.⁹³ What justifies <u>ijtihād</u> is the absence of rules to cover new cases.

In reference to legal material required for i<u>itihād</u>, Shāţibī finds in i<u>itihād</u> three processes. One that depends on inference and deduction and hence is connected with written legal material. For this type a knowledge of Arabic language is inevitable. Shāţibī clarifies that he does not mean the knowledge of grammar, syntax, etc., but rather a knowledge of Arab usage.⁹⁴ The second process of ijtihād is that where it is not directly concerned with the text, but with the law itself. For this process of reasoning, one requires a grasp on <u>silm maqāşid al-Shar</u> (the knowledge of the purpose of law).⁹⁵ In reference to the above-mentioned four types of <u>ijtihād</u>, the present process is particularly relevant to <u>taḥqīq al-manāt</u> and takhrīj al-manāt.

The third relates to deductions which require neither of the above types of knowledge.⁹⁶ This process is, in fact, the application of the verified <u>manat</u> to specific cases. Consequently in this type of reasoning, two premises are involved; first <u>tabaqquq</u> <u>al-manāt</u> (certitude of the basis of ruling) and second <u>tabakkum</u> (decision).⁹⁷ Shāţibī explains further that the method of deduction of conclusion in <u>iitihād</u> is quite different from what is followed by logicians. The premises here do not mean the formulation of propositions in accordance with the figures (<u>ashkāl</u>) of syllogism known in logic. Nor does <u>ijtihād</u> depend upon considerations of syllogism, such as <u>tanāqud</u> (contradiction) and <u>caks</u> (conversion). If there is found any similarity, it must not be confused with the technical terms of logicians.

The closest logical figures of syllogism to the method of <u>ijtihād</u> are $\int_{\Lambda}^{\alpha l} \frac{d^2}{dy ds} \int_{\Lambda}^{\alpha l} \frac{d^2}{dy$

च्च १२ -नद्द्रद्वर who rejected logician's claims that there cannot be a conclusion without two premises, and, referring to figh, argued that it is possible to conclude from one premise.⁹⁹

It is in the light of this explanation that Shātibī rejects the requirement of a knowledge of the rules of logic for <u>shar</u>⁴ i purpose, ¹⁰⁰ whereas knowledge of the Arabic language and that of objectives of law is considered <u>sine qua non</u>. As for other sciences such as the science of the readings of the Qur'ān, or that of <u>hadith</u>, or <u>kalām</u>, they are not considered absolutely necessary. In fact, a <u>mujtahid</u> can justifiably accept the conclusions reached by these sciences as <u>muqaddimāt</u> (premises, foundations) in <u>ijtihād</u>.¹⁰¹

Ņ

The above analysis of <u>ijtihād</u> shows that Shātibī saw it as a process of adapting the legal system to social changes. What distinguishes his treatment of <u>ijtihād</u> is his outlook as a jurist. He looks upon <u>ijtihād</u> as a necessary process but neither open to everyone nor at all times. It is exercised only when it is needed. <u>Taqlīd</u> for him is not a theological concept, but a practical necessity in a legal system.

NOTES: CHAPTER VIII

- 1. For the validity of this method, we refer to the following work, which has used this method in a highly successful manner: T. Izutsu, <u>God and Man in the</u> Koran (Tokyo, 1964).
- Al-Muwafaqat, comm. Abd Allah Daraz (Cairo, n.d.), Vol.11, 169. Shatibi, here, argues that the Qurran uses way (revelation), which is opposte to hawa. Ibid., Vol.1, 35, he contrasts the terms aqliyya and shariyya.
- 3. Ibid., Vol.III, 121.
- 4. See above n.1.
- 5. Ibid., Vol.III, 369.
- 6. Ibid., Vol.1, 46-47. To illustrate what is <u>Shari'a</u> on a certain point, Shātibi quotes from the Qur'ān, <u>hadith</u> and sayings of the Companions. This pattern is frequently repeated throughout the book. On p. 56, for instance, he says, "And this is how the <u>Shari'a</u> explains itself", and then he quotes sayings of the Prophet.
- 7. Ibid., Vol.IV, 7.
- 8. Ibid., 245.
- 9. Ibid., Vol.11, 64.
- 10. Ibid.
- 11. Ibid., 69.
- 12. Ibid., 274-275.
- 13. Ibid., 136.
- 14. Ibid.
- 15. Ibid., 58.
- 16. lbid., 59.
- 17. Al-14tişām, Vol.1, 19.

- 18. Al-Muwafaqat, 1,31.
- 19. Ibid.

et :

R,

- 20. Ibid.
- 21. Ibid., Vol.1, 19-20.
- 22. Ibid.
- 23. Ibid., Vol.III, 346.
- 24. Ibid., 103-104.
- 25. Shāţibī's statement that "God revealed all that they needed", may be understood as that nothing outside the Qur'ān belongs to <u>Sharī'a</u> and secondly that there were things that God did not reveal because they were not needed. Apparently these statements reject any need of legal change. To be meaningful, these statements must be understood together with Shāţibī's distinction between 'ādāt and 'ibādāt. Thus totality and completion in reference to 'ibādāt have been revealed and that nothing else by way of 'ibādāt is further needed. The totality, in reference to adat, means that the totality of basic principles or universals have been revealed, the particulars of which will always require ijtihād.
- 26. Ibid., Vol.IV, 233-39.
- 27. <u>Ibid</u>.,236.
- 28. ibid., Vol.11,247.
- 29. Ibid., 180.
- 30. Ibid., 177-79.
- 31. Ibid., Vol.IV, 244-46.
- 32. Ibid.,272ff.
- 33. Ibid.,274,276.
- 34. lbid.,281.
- 35. Ibid., Vol.11,58-61.
- 36. Ibid., Vol.11,281.
- 37. Ibid.,279ff.
- 38. Ibid.

د.

39. Shātibi uses both 'adat and 'awaid as plurals of 'Ada. Etymologically, some

linguists claim that 'awā'id is the plural of 'ā'id and 'ā'ida (something that re-occurs), and 'adāt is plural of 'āda. This distinction is, however, generally disregarded. In Shātibi's use of 'awā'id there is some indication that he uses 'ādāt for those legal acts which are opposite to 'ibādāt, and 'awā'id for habits, customs, etc. But this distinction is not consistently mainted by him.

Ą

40. Ibid., Vol.11, 280.

€.

1

- 41. Ibid., Vol.II, 297. The examples of eating, drinking, etc., are given to illustrate 'āda.
- 42. Ibid., Shāṭibī gives the examples of variety of 'āda in different forms of dwellings; on p. 307, he gives the examples of customs in pre-Islamic period to illustrate ʿāda.
- 43. Ibid., pp. 307-308.
- 44. Ibid., Vol.111, 121f.
- 45. Ibid., Vol.11, 281-82.
- 46. Ibid., Vol.11, 283ff.
- 47. Ibid., Vol.111, 265.
- 48. Ibid., Vol.11, 307.
- 49. Ibid.
- 50. Ibid., Vol.1, 174.
- 51. Ibid., Vol.11, 297-98.
- 52. Ibid., 284-85.
- 53. Ibid., 283-84.
- 54. Ibid., 284.
- 55. Ibid.

-0

с. "ж

- 56. Ibid., Vol.1, 173.
- 57. Ibid., 11, 409-410.
- 58. Ibid., Vol.1, 161.

- 59. Al-Istişām, Vol.11, 293ff.
- 60. lbid., Vol.1, 26-111.
- 61. Ibid., 82.
- 62. Ibid., 18f.
- 63. Ibid., 18.
- 64. Ibid., 19.
- 65. Ibid.
- 66. Ibid.
- 67. Al-Muwafaqat, II, 337ff.
- 68. Ibid., pp.341-42.
- 69. Al-Istişām, Vol.11, 24.
- 70. Ibid., Vol.I, 22. "<u>Fa kullu mā 'ukhturi'a fi'l-dini mimmā yu</u>dāhi al-mashrū' wa lam yuqsad bihi al-ta'abbud, fagad kharaja 'an hādhihi al-tasmiyya".
- 71. lbid., pp.22-23.
- 72. lbid., 147ff.
- 73. Ibid., 232ff.
- 74. lbid., Vol.11,63-67.
- 75. Ibid., 67.
- 76. Ibid., 68.
- 77. Al-Muwafaqat, Vol.IV, p.89.
- 78. Ibid., p.90.
- 79. Ibid., p.97.
- 80. Ibid., p.90.
- 81. Ibid., p.91.

A

< >

- 82. lbid.,p.92.
- 83. lbid., p.105.
- 84. lbid., p.93.
- 85. lbid., pp.97-98.
- 86. lbid., p,95.
- 87. lbid., p,96.
- 88. lbid., p.162f.
- 89. lbid., p.96.
- 90. lbid., p.89.
- 91. Ibid., p.105.
- 92. lbid., pp.118ff.
- 93. Ibid., p. 128.
- 94. Ibid., pp. 162-165.
- 95. lbid., pp. 105-6, 162f.
- 96. Ibid., p. 165f.
- 97. Ibid., p.334.
- 98. Ibid., p.337.
- 99. <u>Ibid</u>., p. 339.
- 100. Ibid.
- 101. lbid., pp.110-111.

€; ,∱:

CHAPTER IX

SUMMARY AND CONCLUSION

Granadian society in the fourteenth century underwent certain very significant changes. These changes were both multidimensional and fundamental for the Granadian legal system; they affected the political, religious, economic and legal structure of the society.

Sultān Muḥammad V Ghanī Billāh's reign (1354-59/1362-91) was replete with depositions, intrigues, and assassinations. He eventually brought political stability to the kingdom by making himself an absolutely independent ruler. The Sultān secured his independence by weakening the political power of the offices of Shaykh al-Ghuzāt, Wazīr and Qādī al-Jamāʿa.

The weakening of the office of the Qādī al-Jamā'a affected the political power of the <u>fuqahā</u>'in general. The <u>fuqahā</u>'as a political and social group were very powerful. They held most of the administrative offices, and, further, they were the principal authority in religious matters and they controlled the institutions of learning. In addition, they were responsible for the administration of a considerably large amount of trust property.

The decline of the political power of the <u>fuqahā</u>'began with the Sultān's skillful manoeuvres to become independent of the <u>fuqahā</u>'. There were a number of factors which facilitated the Sultān's success. One of these was the introduction of the

A

state-controlled <u>madrasa</u> system of learning in the days of the Sultān's father. Despite the opposition of the <u>fuqahā</u>' <u>the madrasa</u> system had succeeded and had been gradually making the <u>fuqahā</u>' dependent on the Sultān.

The second factor was the penetration of taşawwuf and of Sufi tariqas into Granadian society. The Sultan had bestowed his favours on the sufi shakks because the Berber mercenaries who constituted the armies of the Sultan were followers of the suff shakks. To weaken the power of the shakk al-ghuzat and of the fuqaha? and to raise his prestige among these mercenaries the Sultan would eagerly patronize taşawwuf. Furthermore, the suff life, being simple and pious, appealed to the people at large, who compared suff life with that of the fuqaha? who lived in an aristocratic style. The rise of the tariqas which undermined the religious and legal authority of the shari a was a real threat to the fuqaha?.

The above political and religious changes were further solidified by other factors which brought certain fundamental changes in the economy of Granadian society. Due to continuing loss of territories to Christians, agricultural land in Granada had become scarce. Furthermore, the Muslim emigrants from Christian Spain, and the Berber fortune-seekers from Africa were adding to the already over-grown population. Consequently every possible piece of land was being used for agricultural purposes. Thus, new forms of agrarian property, new types of agrarian partnership and the practice of hired seasonal labour had become popular.

To add to the complexity of the economy, the Granadian treasury owed to Christians and to the Berbers huge sums of money which were to be paid in cash. Hence state revenues had to be collected in cash. In addition, a number of new taxes were introduced. Since this economic situation affected the gold and silver reserves in the treasury, a copper <u>dinar</u> was introduced, probably as a devalued currency.

Local crafts and industries supplemented agricultural production, but by this time they had naturally become of prime importance. In the Kingdom of Granada, silk was the most profitable export industry. The rise of the Italian silk industry had, however, reduced the demand for finished Granadian products in the Mediterranean market. Now, raw silk was more in demand. Hence the Granadian economy was geared to such demands.

The Mediterranean trade had also developed rapidly. To meet the demands of Italian manufacturers, raw materials were imported from Africa and Spain. Granada, being connected with Malaga and Almeria, was situated on one of the very significant arteries of trade that linked North Africa with the European countries. The significance of trade was recognized by the rulers in these countries. Strong trade pacts among these principalities assured the safe transit of merchandise.

The affects of the above-mentioned developments were very far reaching for the legal system in Granada. New commodities and ideas were being exchanged. New forms of transactions had emerged. The legal theory had to accommodate all these changes into the system. The existing legal system was not adequate for the new circumstances. The incompetence of the legal system was recognized by lbn al-Khatīb in his criticism of notaries and their outdated legal practice in regard to legal contracts. The internal contradictions of the system were exposed under the impact of these changes. An indication of these contradictions is seen in the

controversy over the demarcation of the functions of the mufti and the gadi.

Such was the milieu in which Shāṭibī (d. 1388) grew up in Granada. His training in figh brought him into touch with these problems quite early in his career. Later, he actively participated in discussions and disputations with other scholars on the problems arising out of the social conditions mentioned above. Quite early on he realized the inadequacy of the legal system in Granada. The centre of his interests were the problems relating to Islamic legal theory and particularly the devices that the Māliki fugahā' had used to adapt Māliki legal theory to accommodate social One such device was that of mura at al-khilaf. By accepting a diverchanges. sity of laws as fact, the Mālikī fuqahā'came into possession of a legal device to accommodate new social practices. For Shātibi, accepting a diversity of laws meant negating the very basis of law. On various aspects of this and other problems, he wrote to Māliki scholars in Andalus and in Africa. After a long search and investigation, he expounded his doctrine of maqasid al-sharita. He examined the traditional legal theory in the light of this doctrine. The result was his book al-Muwāfaqāt in four volumes.

Ń

As Shāțibi had expected, <u>al-Muwāfaqāt</u> was not welcomed. He was called a heretic. Alluding to a number of Shāțibi's actions in his public life, his opponents condemned him as an innovator. He defended himself against these charges by writing his other book <u>al-l<tişām</u> in which he defined the concept of <u>bid</u>^ca.

In preparing his fatāwā, Shāțibi had further actual experience of the inadequacy of the then legal theory to meet the challenge of social changes. We have seen above that out of 40 queries that we have examined, 34 were related to social changes. Shātibī found that the methods of analogy and of borrowing from other schools of law in the name of <u>murāʿāt al-khilā̃f</u> was not sufficient.

The insufficiency of the provisions of Islamic law and the methods of Islamic legal theory to cope with rising needs were more conspicuous in the area of contracts and obligations. Growing economic activities, especially in trade and commerce, demanded freedom of contract. The Mālikī <u>fuqahā</u>' found it difficult to respond to such demands. The new forms of contract had become highly complicated. The older framework of contract in Mālikī legal theory, which still operated on the legal fiction of <u>shirka fi'l-zar</u>^c derived from the early Medinese practice of agrarian partnership, did not provide sufficient analogies to new kinds of contract which were different both in form and in nature. The Mālikī <u>fuqahā</u>' tried to solve these problems by adhering to the method of analogy through various devices, but the search for particular precedents to particular æses proved unsuccessful. A number of <u>fuqahā</u>' were forced to fall back on the original Mālikī general legal principle of maşlaḥa.

Shāțibi also had the same experience in preparing his <u>fatāwā</u>. He too had to refer to principles such as <u>tashil</u>, <u>maslaha</u> and <u>'adam haraj</u>. He, however, realized that he could not apply these principles indiscriminately to all areas of social and legal change. Under the influence of <u>tasawwuf</u>, a number of new rituals had come into social practice. He regarded these rituals as <u>bid'a</u> and rejected them. The need for such distinctions impressed upon him the significance of investigating

-107

the aim and purpose of law, the nature of legal obligation, and the method of legal reasoning.

Shāţibī found the principle of <u>maşlaha</u> to be the essential point at which all the enquiries about the nature and purpose of legal obligation, social and legal change, and the method of legal reasoning converge. At the same time this principle also provides the basis of the unity that underlies the diversity of rules in Islamic law.

The principle of <u>maşlaha</u>, as a legal concept, however, has not been a simple concept in <u>uşūl al-fiqh</u>. Various theological, moral, methodological and more recently <u>sūfī</u> conceptions of <u>maşlaha</u> had posed serious difficulties for the use of <u>maşlaha</u> as a principle of adaptability. The Ash'arī denial of causality in God's actions made it impossible to analyze <u>shar</u>ī commands on the basis of an <u>cilla</u>.

The suffis denied anything that implied any pleasure for the lower soul. Their emphasis on wara^c, zuhd and ikhlās rendered maslaha simply into an indulgence in personal desires.

Methodologically, according to traditional jurists, <u>maşlaha</u> provided only a probable basis of reasoning if it was not supported by a specific legal Text. Traditionally, <u>maşlaha</u> was classified from two perspectives. From one viewpoint it was divided into <u>darūrī</u>, <u>hajī</u> and <u>taḥsīnī</u> with the last two being rejected. From the other angle <u>maşlaha</u> was divided into <u>mu<tabara</u>, <u>mulghā</u> and <u>mursala</u>; as the first two were in fact covered by the legal Text, it was only <u>maslaha</u> <u>mursala</u> which remained to be discussed. Consequently the discussion of <u>maşlaha</u> was reduced to

400

į

ĺ

a consideration of <u>maşlaha mursala</u>. It is evident that Shāţibī's analysis of <u>maşlaha</u> keeps the traditional criticism of <u>maşlaha</u> in view. The first thing that emerges from his analysis of this concept is his stress on human needs rather than on its being simply a Divine prerogative in the absolute sense. From Shāţibī's definition of <u>maşlaha</u> and its characteristics and from his discussion of its five aspects, it becomes clear that the essential element in the concept of <u>maşlaha</u> is consideration for and protection of the necessities of human life in this world and in the hereafter.

Shāțibi accepts the traditional division of <u>maşlaha</u> but rejects the limitations on their validity. He finds <u>hāji</u> and ta<u>hsini</u> types of <u>maşlaha</u> to be complementary and to act as protective zones for the <u>darūri</u> type. The two are indispensible in this sense. He does not seem to accept the other division, however. The term <u>maşlaha mursala</u> is seldom used in his discussion of <u>maşlaha</u>, and when it is used, it does not differ in meaning from <u>maşlaha</u>.

In his analysis of the concept of <u>maşlaha</u>, Shātibī established certain distinctions to clarify the confusions that had gathered around this concept. He analyzed the implications of <u>tasabud</u>, <u>huzūz</u>, and <u>mashaqqa</u> in order to elucidate the concept of legal obligation. He refuted the <u>sūfī</u> conclusion that abandoning of the <u>huzūz</u> was an essential meaning of <u>tasabud</u>. He explained that <u>tasabud</u> has two senses; one to obey without searching for the reasons underlying obligations and the other to conform to the intent of the law-giver. Shātibī concluded that the first sense of tasabbud is applicable only to the <u>sūfī</u> which he distinguished from <u>sādāt</u>. The other sense was applicable to the entire body of legal obligations. Obeying 'the intent of the law-giver meant to regard the <u>maşlaha</u> or <u>ma'ānī in 'ādāt</u> and <u>ta'abbud</u> in the second sense or to obey the explicit meaning, in 'ibādāt. He further explained that <u>ta'abbud</u> in the technical legal framework means that the area of <u>'ibādāt</u> cannot be extended further than what has been revealed by the lawgiver.

Shatibi answers the theological objections to maslaha by pointing to the confusion that had resulted from not distinguishing between two levels of the Divine Will. Divine Will at the legislative level does not operate in the same way as it does at the level of the Creation. The legislative will allows man's freedom to act and holds him responsible for his acts. Human freedom and responsibility logically require that the Divine Commands must be within man's capability to comply with them and, further, that they must be intelligible. Intelligibility refers to both the linguistic and the rational aspects of the commands. Thus the factors of responsibility, intelligibility and rationality taken together, necessitate that Divine Commands should be based on an explicit or implicit <u>cilla</u>, so that they can be understood, generalized and extended to like situations. Ash ari jurists, in order to defend God's Omnipotence, were forced not only to deny i silla in Divine Commands, but were also compelled to say that a Divine Command does not necessitate the Divine Will. Shāțibi differentiated between two Wills; the Creative Will which is to desire someone to produce a certain act. Thus, contrary to Ash aris, Shāțibī was able to make it clear that a Divine Command with a legislative will does not necessitate its actualization, yet it stresses the support of the Command by the

ξ.,

The basic components of Shātibī's concept of <u>maşlaha</u> are, thus, the following: 1) consideration for the needs of man, 2) the rationality of legal commands and the responsibility of man, 3) protection from harm, and 4) conformity with the objectives of the law-giver.

Shātibī, however, distinguishes $\frac{d}{d}$ as a shar iyya from the ordinary concept of <u>maşlaha</u>; the former is abstract and simple. Ordinary <u>maşlaha</u> does not exist in pure and simple form; it always contains certain elements of <u>mafsada</u>. Ordinary <u>maşlaha</u> is known by weighing the aspects of good and evil in an action; whichever dominates characterizes the thing in question. $\frac{d}{d}$ Maşlaha shariyya as a legal obligation takes into account only the dominating aspect which is pure and simple, unmixed with mafsada.

In Shāțibi's understanding, cāda and sharīca are very closely connected. Although both are willed by God, yet the former belongs to the Creative Will and the latter to the Legislative. Temporally cāda is unlimited but Sharīca is limited. Except for certain fundamental laws cāda may undergo changes, whereas sharīca insofar as it reflects the Divine Will cannot change. To find rules for new situations occurring because of changes in cāda one needs to know the exact rule or the intent of the law. This intent can be known through studying cāda in combination with the principles inductively derived from sharīca.

The above-described concept of maslaha was admirably suited to Shātibi's under-

41 I

standing of social change and to his views on legal change. According to Shātibī the <u>awā'id</u> or the habits of individuals and social practice alike are stabilized by certain universal laws which do not change. The changes that occur in society happen because of the movements from one place to another of individuals, or because of the movements of social customs along with the migration of people. More fundamentally, changes are generally produced by human needs. It is when these social changes go beyond the provisions of the rules of law or when they become too complicated for the existing rules, that a <u>muftī</u> or <u>muj tahid</u> is summoned, through the agency of a <u>fatwā</u>, to examine the law and legal theory as they relate to the changes in question.

Ņ

The process of legal change may be called <u>ijtihād</u>. Shāṭibi divides <u>ijtihād</u> into four types. Although the 'gate' of <u>ijtihād</u> is closed in none of the types, yet Shāṭibī was of the opinion that because of cumulative growth of <u>fatāwā</u> and judicial decisions, <u>ijtihād</u> may not be needed in many areas. For Shāṭibī <u>ijtihād</u> and <u>taqlīd</u> are legal necessities and not theological obligations. Thus Shāṭibī comes to a different conclusion about the principle of <u>ijtihād</u>. As has been pointed out, this rather legalistic and positive understanding of <u>ijtihād</u> is quite significant for Shāṭibī's legal philosophy.

Having summarized our findings, we may now draw conclusions in reference to the problem of the adaptability of Islamic legal theory to meet social changes.

We have seen that Shātibi admits that changes take place in society and that the legal changes in the area of <u>cadat</u> accord with social needs. We have also found

Shāțibī to believe that although its general and universal principles remain unchanged, yet Islamic law does accommodate itself to changes and that it favours the consideration of social needs in making its accommodations. According to Shāțibī, <u>ijtihād</u> provides a method and process for legal change; <u>maşlaha</u> gives a basis and direction to change; and the concepts of <u>bid</u> and <u>ta</u> abbud provide limits on social and legal changes.

< ∞

Through his analysis of <u>maşlaha</u> as the purpose of Islamic law, Shātibī has tried to free the operation of Islamic legal theory from a number of factors of determinism and rigidity arising out of theological and methodological considerations. In fact his concept of <u>maslaha</u> provides a correction for many traditional as well as modern misunderstandings of this concept. We need not repear all the points relevant to these corrections; it will suffice to say that contrary to the general understanding, <u>maslaha</u> is neither a totally relative and arbitrary principle nor is it strictly tied to <u>qiyās</u> or to specific legal texts of <u>shaīiʿa</u>. It is connected with social needs at one end, and on the other it is inductively supported by <u>sharīʿa</u>. It is, thus, not a special form of analogy, nor is it an extra legal method of expediency to provide an area of flexibility in legal reasoning along with more strict elements of the law. To Shātibī, <u>maşlaha</u> is an integral principle that unifies <u>sharīʿa</u>, provides stability and gives direction to legal changes.

It can be seen that Shātibi had considerably improved upon the traditional philosophy of Islamic law by refining and clarifying certain basic legal-philosophic concepts, particularly the concept of <u>maslaha</u>. His views were quite fitting for

the needs of Islamic legal theory in fourteenth century Granada. As we have seen, quite similar developments in the philosophy of law took place in Christian Spain that came to bear fruit in the sixteenth century in Suarez's philosophy of law. The difference was, however, that in Christian Spain those activites which continued through the sixteenth century later helped in the development of modern philosophies of law. In Muslim Spain, despite the fact that Shāţibī's philosophy of law was in some respects similar to that of Suarez, it did not gain acceptance, and the traditional view persisted. Why did Shāţibī's philosophy fail?

To explain the failure of Shāṭibī's legal philosophy on the basis of material and historical reasons will not be sufficient. His legal philosophy was revived in the Salafiyya and Liberal movements in the nineteenth century, and as various studies, such as those by A. Hourani and M. H. Kerr, have shown, once again failed although the historical setting and circumstances were this time more favourable.

The reasons for this failure must also be sought within Shātibi's philosophy and in the understanding of it by his recent followers. Since the matter lies beyond the scope of this dissertation we will only suggest in respect of it that Shātibi's recent followers do not seem to have accepted his philosophy as a whole. For instance, they refer to <u>maşlaha</u> as a principle of expediency to be used in cases where the provisions of legal texts and the method of analogy do not suffice. This is not Shātibi's understanding of <u>maşlaha</u> but is rather a repetition of concepts long held in the community. Thus, in fact, these recent followers have not departed from the traditional concept of maşlaha.

One would have expected that in view of modern developments in theories and systems of law, Shāṭibī's philosophy would have been further refined by his modern disciples. Instead, they have remained within the traditional framework of legal methodology and have even interpreted Shāṭibī's philosophy in the same framework. Consequently, it was possible for scholars such as Rashīd Riḍā to blunt the thrust of Shāṭibī's philosophy by giving Shāṭibī the image of a conservative, a crusader against innovation and of a reviver of tradition.

From the above observations it is possible to suggest that there is a significantly visible trend in Shāṭibī's legal thought towards a positive Islamic law. His emphasis on <u>maşlaḥa</u> and his attempt to free legal theory from theological determinism indicate such inclinations. To illustrate, we may refer to his demarcation of two areas of legal change.

He stresses that no innovation can be accepted in <u>ibādāt</u>, whereas in <u>adāt</u> changes are possible. The element of positivism lies in his theoretical justification of the above conclusion. He explains that <u>ibādāt</u> belong to that area of <u>masālih</u> which is known only to God. Generally<u>ibādāt</u> cannot be rationally explained. Since they cannot be observed and tested by human reason, they cannot be extended by analogy to similar situations.

The area of <u>cadat</u> is different, however. Not only are <u>cadat</u> based on <u>masalih</u>, but the commands in <u>shari</u> a relating to <u>cadat</u> usually provide the reason indicating that these <u>masalih</u> can be grasped by human reason. Further, <u>cadat</u> are observable and they can be tested. This is the reason they are extendible by analogy, and why they can be the subject of ijtihad. Such arguments should have led Shāṭibī to positivism in his legal philosophy; yet there are no explicit statements by Shāṭibī showing such a tendency.

1

The implicit positivism in Shāţibī's legal philosophy may be further noted in his attempt to separate law (<u>fiqh</u>) from theology and from <u>sūfī</u> morality as set out in his definitions of legal obligation. Although he believed the origin of Islamic law to lie in religion and morality, yet he was able to maintain that theological and moral elements and the conception of obligation based on them could not be admitted into the definition of legal obligation. He was, however, reluctant to reject entirely the theological and moral implications of legal obligation.

This reluctance, in fact, sometimes resulted in his allowing an element of confusion to creep into his definitions. For instance, we may cite his definition of <u>ta<abbud</u>. His illustrations of <u><ibādāt</u> refer to the well-known Islamic rituals and other such acts which, according to him, should be accepted without rational explanation in contrast to <u><ādāt</u> which have rational bases. There are a number of occasions, however, when he implies that even those legal commands in the Qur'ān, which do not concern <u><ibādāt</u> such as those governing family relations, should also be accepted without rational explanation. Does he mean that he extends the definition of ta<abbud in the sense of <u><ibādāt</u> to all the commands in the Qur'ān?

In the light of Shātibi's philosophy as a whole, it is difficult to explain such departures. Most probably these departures result from Shātibi's reluctance to accept the logical conclusions of his attempt to separate the two levels of conceiving the legal obligation i.e. the level of the origin of legal obligation and the level of its definition and application. The first level may relate to religion and morality, but such a relationship is not necessary on the second level. One can appreciate Shāţibī's reluctance if it is recalled that the legal system in his day, despite certain attempts, did not succeed in separating the jurisdiction of the <u>muftī</u> from that of the <u>qāqī</u>. Furthermore, the <u>muftī</u> was also regarded as a deputy of the Prophet, and as such his jurisdiction included both religious and legal matters and the bases of his authority were somewhat metaphysical; the <u>muftī</u> derived his authority from the metaphysical principle of continuity of Divine guidance through prophets, and after Muhammad through muftīs.

The <u>qādi</u> did not enjoy independence in the legal system; he had to rely on the <u>mufti</u>, who was attached to the court as a consultant, for the validity and legality of his decisions. Such limitations on the institution of the qadi inevitably influenced the concept of legal obligation.

Inspite of his attempts to define legal obligation, Shāţibī did not uphold the independence of the <u>qād</u>ī from the <u>muftī</u>. Hence his legal philosophy, despite certain elements of positivism, did not go far enough and, consequently, could not grow into a positive legal philosophy. This is probably the reason why this philosophy has also failed more recently when modernists have attempted to use it without supplying the necessary correctives.

It may, in fine, be concluded that, in the history of <u>uşūl al-fiqh</u>, Shāṭibī's philosophy of law marks a tendency towards "legal positivism". A proper understanding of its limitations, which had resulted from the particular historical nature of

A A **T17**

Islamic law in this period, and of the ambiguities resulting from these limitations, may help us to reconstruct Shāţibī's arguments to adapt Islamic law to social change. Such a reconstruction might hold a key to a fruitful adaptation of Islamic law to modern circumstances.

.,

....

APPENDIX A

A Summary-view of Al-Muwafaqat

N

Al-Muwāfaqāt is divided into five books:

FIRST BOOK: "muqaddimāt" (preliminaries). In this book thirteen preliminary rules are discussed. The main points made in these discussions are three:

- (i) The <u>uşūl al-fiqh</u> (principles of Islamic legal theory) are <u>qat</u>^ci (definite), not <u>zanni</u> (probable) as had been held by most of the <u>mutakallimūn</u> jurists, especially by Ghazāli and Rāzi.
- (ii) These principles must relate to <u>furū</u>^c(the details of applied law) and to <u>a</u>^cmāl (actions). This position was again taken in order to refute the <u>mutakallimūn</u> jurists who had introduced problems of <u>kalām</u> into <u>usūl al-fiqh</u>.
- (iii) The method of knowing the precepts of law must fulfil the following three

requirements:

- (a) the methods must be close to the level of common capability of understanding.
- (b) it must aim at being a means to tarabbud (bondage to God).
- (c) it must lay stress on a necessary relationship between knowledge and action.

At this point Shāṭibī goes into a detailed discussion about <u>al-</u>^cilm (the knowledge). This entails the following problems:

- 1. Definition of the proper (mu(tabar) shar(i knowledge.
- Division of knowledge: (i) <u>sulb</u> (solid), (ii) <u>mulah</u> (salty, to add flavour), (iii) neither of these. <u>Sulb</u> is the goal, and <u>usul al-figh</u> belong to this category.

€} ≹ja

- 3. The role of reason: human reason (aql) follows, does not take precedence over the transmitted knowledge (naql).
- 4. <u>Al-Adilla al-shar iyya</u> (legal evidences) are the only basis for a proper <u>shar'i</u> knowledge.
- 5. The method of learning: of the two methods of learning, i.e. <u>al-Mushāfaha</u> (direct from the teacher) and <u>mutāla^ca</sub> kutub</u> <u>al-muşannifin</u> (indirect, by studying from books of authors), the former is better, yet the latter must supplement the former.

٩.

6. The signs (<u>amārāt</u>) of a true scholar with whom one should study are three: (i) correspondence of action with knowledge, (ii) direct relationship with his teachers in his education, (iii) imitation of his teachers.

SECOND BOOK: Ahkām (legal values), deals with forty-two problems. The discussion is divided according to the two major categories of legal values, khitāb al-taklīf, concerning the values that result directly from a Divine command, and khitāb al-wad^c concerns the values that are the indirect result of that command.

Khitab al-taklif creates five legal values: <u>Nadb</u> (recommendation), <u>wujūb</u> (obligation), <u>Ibāḥa</u> (freedom), <u>karāha</u> (reprehensibility) and <u>man</u> (prohibition). Shāṭibī considered <u>ibāḥa</u> as a middle value in this structure, hence a major part of his discussion on this category of values is devoted to <u>ibāḥa</u>. The main points of discussion are the following:

- 1. The definition and essence of *ibāha*, the mode of expression of the value and its various aspects.
- 2. Relationship of ibāha to other values.

\$2 2

Ŧ

- 3. Relationship of ibāḥa to the maqāṣid (objectives of law).
- 4. Relationship of ibaha to the problem of Rights and Obligations (huqua).
- Ibāḥa and śafw (foregiveness), a new category suggested by Shāibi as a middle value between the Qurānic values of ḥalāl (permissible) and ḥarām (forbidden).

<u>Khitāb al-wad</u>^c also creates five values which indirectly lead to the above-mentioned five. The above five are created as grades of obligation by direct command, but <u>khitāb al-wad</u>^c creates values by instituting the requirement through one of the following five values:

- (1) Sabab (mediate cause).
- (2) Shart (qualification).
- (3) Mani^c (preventive cause).
- (4) Sihha/butlan (soundness/unsoundness).
- (5) Azīma/rukhṣa (regularity or allowance in the requirement).

In these discussions Shātibī defines these shars values and establishes distinctions among them. His main concern is to know whether they are intended by the lawgiver as such or not. He also attempts to establish the two levels of this intention, the intention of instituting the requirement through these values, and the intention of requiring the performance of the action. For instance, a <u>sabab</u> act is required in performance but not the <u>musabbab</u> act although at instituting (<u>wad</u>s) them the lawgiver's intention embraced both. Shāţibī's interest in the question of the intention of the lawgiver finds more space in the third book. After he has shown the relevance of this question to the discussion of legal values, he proceeds to deal with the question itself.

THIRD BOOK: "<u>al-maqāşid al-shariyya fi al-shari'a</u>" (the legal objectives of shari a), Shāțibi discusses these objectives at two levels. On one level he treats the objectives of the lawgiver through the following four aspects:

- (a) the primary objective of the lawgiver in the institution of shari a. Shātibi maintains that the primary objective is the masālih of the people. He discusses the definition, types and structure of masālih in order to show that the notion of maslaha constitutes the central theme in sharifa.
- (b) the objectives of the lawgiver regarding the intelligibility of shari'a.
 Here Shātibi maintains that shari'a aims to reach all people; hence it must be intelligible to all of them. In this respect he discusses two notions:
 (i) 'umuīniya, generality in the use of terms and (ii) 'ummīya, consideration for the majority of the people in their capability of understanding and their command of knowledge of the sciences.
- (c) the objectives of the lawgiver in instituting the obligation. The main theme here is that <u>sharica</u> does not impose obligation which is impossible; yet this does not mean that the obligation does not entail any exertion. The criterion in this regard is <u>cada</u>, the common practice; if something is considered impossible in common practice, it cannot be regarded as a requirement by <u>sharica</u>.

the objectives of the lawgiver in admitting the <u>mukallaf</u> (one under obligation) as a subject of obligation. Shātibī here goes into detail in defining the objective of <u>taklīf</u> on two levels. On the first level he deals with two types of objectives of the lawgiver; primary and secondary. On the second level he discusses the participation (hazz) of the <u>mukallaf</u> and concludes that his participation is exclusively conditioned by the objectives of the lawgiver.

Ņ

(d)

t L r

1

Shāțibī, however, makes a distinction here in reference to this participation. Although the second level is conditioned by the first level, there are two aspects to be distinguished. As far as obedience is concerned, there is no distinction among various categories of action, but insofar as reasoning is concerned, the distinction must be made between <u>cibādāt</u> and cādāt.

Considerable space is devoted to the discussion of <u><u><u></u>ada</u>, its characteristics, and its various types from the legal point of view. The question of <u><u>Kharq</u><u><u><u></u>ada</u> is also dealt with in detail.</u></u></u>

The second level of discussion of the objectives has to do with the intentions of the mukallaf.

Here Shāţibī deals with the questions of <u>niyya</u> (intent) and <u>sayniya</u> (particularity). The main theme is that the intentions of the <u>mukalla</u>f must correspond with the intentions of the lawgiver. This is the reason why Shāṯibī condemns <u>hiyal</u> (devices to evade law) and bid^ca (innovations). FOURTH BOOK: <u>Al-Àdilla Al-Shar^ciya</u> (Legal Evidences)

This book is divided into two parts:

٠.

1. The discussion of Legal Evidences in general.

This part is divided into sections:

- A) General Principles: The following are the main subjects:
 - Legal Reasoning (<u>Nazar Shar</u>ⁱ) deals only with universals, nor with particulars.
 - 2. The characteristics of Legal Evidences:

Qat^civa (Definiteness), Zanniva (probability), supplemented by other factors so as to make them definite.

- 3. These evidences do not contradict Qadāyā Uqūl (rational propositions).
- Every evidence consists of two premises; since they are definite, so also is the conclusion.
- 6. The evidences are general in application.
- The evidences are either <u>burhānī</u> (offering a logical proof) or <u>taklīfī</u> (obligatory because of command).
- 8. General rules for finding out the evidences, for undertaking what they demand and for their application.

B. Weaknesses of Legal Evidences:

In this discussion Shāțibi deals with the following:

- 1. Tashābuh (equivocations)
- 2. Naskh (abrogation)
- 3. Awamir Nawahi (commands and prohibitions)
- 4. Umūm Khuṣūṣ (general and particular)
- 5. Bayan ijmal (explanation and conciseness)
- 11. The evidences in detail

> Specific legal evidences are four: The Qur'ān, the <u>Sunna</u>, <u>limā</u>^(consensus) and <u>Qiyās</u> (analogy). Shāțibi dismisses the latter two, contending that only Qur'ān and <u>Sunna</u> are essential evidences.

A) Al-Kitāb:

The main points of this discussion are as follows:

- 1. Al-Kitāb is the whole (kulliyya) or the Sharīća.
- 2. Almost all the commands in the Qur'an are "universal".
- 3. The Qur'ān contains an explanation for each of its concise commands.
- 4. The knowledge of the historical context of the Qur'an is essential for the following reasons:
 - a) To know the mediate causes of the revelation of a command.
 - b) To establish the differences between the verses revealed in

Mecca and Medina.

The former are universals and hence regulative; the latter are particulars and subject to abrogation by the lawgiver.
- 5. Arbitrary opinion has no place in explaining a command in the Qur'an.
- B) Al-Sunna:

-17

The main topics of discussion are the following:

- 1. Definition of the Sunna
- 2. The relationship of Sunna with al-Kitab
- 3. The various kinds of <u>Sunn</u>a
- 4. The Sunna of the Companions of the Prophet.

FIFTH BOOK: litihad (Legal Reasoning)

This book is divided into five parts:

1. lįtihād:

The main points of discussion are as follows:

- Two types of <u>litihad</u>, one whose results are immutable, the other which is continuously subject-to-change.
- 2. The qualifications for ijtihad
- 3. The unity of the principle of Shari'a.
- 4. litihad and Bid'a
- 5. The measures of litihad:
 - a) <u>Sadd</u> <u>al-Dharā'i</u>^c (to block the ways possibly leading to an undesired action)
 - b) Istihsān (juristic preference against strict analogy)
 - c) The problem of Khilāf (disagreement)

11. Fatāwā (Responsa)

i. Konse

- 1. Mufti (the jurist who issues "responsa") is the deputy of the Prophet.
- A <u>fatwā</u> consists of all the three categories of <u>qawl</u> (statement), <u>fi</u>⁴
 (action) and <u>taqrīr</u> (confirmation).

Ņ

- 3. The qualifications of a mufti.
- 4. The problem of rukhsa (allowance) and fatwa.
- III. Iqtida' (Imitation)
 - The meaning and nature of "imitation", the definition of <u>muqallid</u> (one who imitates)
 - 2. muqallid and Istifta? (the action of asking for responsa)
 - 3. The problem of choosing whom to imitate.
 - 4. The conditions for the pronouncing of a fatwa.
 - 5. Imitation of actions.
 - IV. <u>Tatārud</u> (conflict) and <u>Tarjiņ</u> (preponderance)
 The problem of choosing between contradictory or conflicting evidences and its various aspects.
 - <u>Su'āl</u> <u>Jawāb</u> (question and answer)
 This part deals with the rules of debate and defence.

The following topics are discussed:

- 1. (Ilm al-Jadal (The science of disputation)
- 2 the meaning of question, division and reprehensibility of excessiveness of questions.

3. The difference between a mujtahid and a munāzir (debator).

•

)

4. The method of argument: conclusion from two premises.

A . .

APPENDIX B

A Summary-view of Al-I4tisam

This book deals with the question of Bid^ca in ten chapters.

Chapter One: The definition of bid	ter One: The definition o	of Bid'a	
------------------------------------	---------------------------	----------	--

Chapter Two: Condemnation of Bid^ca

- Chapter Three: The condemnation of <u>bid</u>^ca is general, the condemnation applies to all kinds of <u>bid</u>^ca. In this chapter Shāțibi criticizes scholars such as lk n ^CAbd al-Salām and Qarāfi who divided <u>bid</u>^ca into five categories like the five legal values. According to them some categories of <u>bid</u>^ca are not only good and recommended but also come close to obligation.
- Chapter Four: The details of the arguments of those who are in favour of bid^ca.
- Chapter Five: Two levels of bid^ca; <u>Bid^ca</u> <u>haqiqi</u>^{yy^a}(real innovation) and <u>Bid^ca</sub> <u>idafi</u>^{yy^a}(relative innovation)</u>

Chapter Six: The legal value (hukm) of bid'a

- Chapter Seven: The question of bid'a in reference to <u>ta'abbudāt</u> and <u>cādāt</u>. According to Shātibī, legally, the value of <u>bid'a</u> applies only to the former.
- Chapter Eight: <u>Bid'a</u> and <u>Maşālih</u> <u>Mursala</u> (The juristic consideration of the public interest against strict analogy) and <u>istihsān</u> (juristic preference against analogy). Are they bid'a?

Chapter Nine: <u>Bid'a</u> as heresy: the heretical sects; Schism in the Community – The problem of the one Saved (<u>nājiya</u>) Sect.

Chapter Ten: <u>Şirāt</u> <u>Mustaqīm</u> (The Right Path) and <u>Ibtidā</u>^c (Committing innovation)

- 1. Types of acts which introduce bidea
- 2. The causes of bid^ca

- a) Ignorance of the tools (adwat) for knowing the objectives of <u>Shari'a</u>. This generally means incidequate knowledge of Arabic grammar and usage.
- b) Ignorance of the objectives themselves.
- c) Reliance on the judgment of human reason alone.
- d) Submission to personal desires (Ittibā^c al-hawā).

BIBLIOGRAPHY

I - THE WRITINGS OF SHATIBI

Shātibī, Abū Ishāq. Al-Muwāfaqāt. 4 Vol. Tunis: Matba'a Dawlatiyya, 1884; 2nd print. Cairo: Salafiyya, 1923; 3rd print. Comm. 'Abd Allāh Darāz. Cairo: Mustafā Muḥammad, n.d.; Ed. M. M. 'Abd al-Ḥamiīd, Cairo: Muḥammad 'Alī, 1969.

_____<u>Al-I^ctişām</u>. (ed.) Rashīd Riḍā. Cairo: Mustafā Muḥammad. [1915].

Shātibi's Fatāwā. See Wansharisi.

≪ 4 2 3⊒7

> که د د

11 - PRIMARY SOURCES

- Aḥmad Bābā. <u>Nayl al-ibtihāi</u> / on the margin of ibn Farḥūn's <u>Al-Dībāj al-Mudhahhab</u>/. Cairo: ʿAbbās b. ʿAbd al-Salām, 1351 A.H.
- ʿĀmìdī, Sayf al-Dīn. <u>Al-Ihkām fīvsūl al-ahkām</u>. Cairo: Maṭba^ca Ma^cārif, 1914.
- Başrī, Abū'l Husayn. Al-Mu^ctamad fī 'uşūl al-fiqh. Dimashq: Al-ma^chad al-^cilmī al-Firansī, 1964.

Ghazālī. <u>Al-Mustasfā min ^cilm al-uşūl</u>. Baghdād: Muthannā, 1970.

- Hakīm, Abūʻl Hasan ʿAli b. Yūsuf. <u>Al-Dawhat al-mushtabika fi dawābit dār al-sikka</u>. Edited by Husayn Mu[·]nis. <u>Madrid: Maʿhad dirāsāt al-islāmiyya</u>, 1960.
- Ibn ^SAbbād. <u>Al-Rasā'il al-şughra</u>. Edited by Paul Nwiya. Beiruth: Imprimerie Catholique, 1958.

<u>Sharh matn al-hikam l'il-i mām Abī l-Fadl Ahmad b. 'Atā Allāh</u> al-Iskandarī. Cairo: Mustafā Afandī, 1320 A.H.

Ibn Abd al-Salām. Qawāʿid al-aḥkām fī maṣālih al-anām. Cairo: Istiqāma, n.d.

Ibn Ahmar, Ismāʿīl. Nathīr farāʾid al-jumān fi nazm fuhūl al-zamān. Edited by Muhammad Ridwān al-Dāya. Bayrūt: Dār al-Thaqāfa, 1967.

Ibn ^CArafa. <u>Hudūd fiqhiyya</u>. Comment. Muḥammad al-Raṣā^c. Tūnis: Maṭba^ca Tūnisiyya, 1350 A.H. Ibn Asim. Tuhfat al-hukkām. See Bercher.

1

-**F**

- Ibn Bațțūța. Tuhfat al-nuzzār fi gharā'ib al-amşār wa 'ajā'ib al-asfār. Vol. II. Edited by 'Awāmirī Bek and Jād al Mawlā. Cairo: Amīriyya, 1934.
- Ibn Farhun. Al-Dibaj al-mudhahhab. Cairo: Abbas b. Abd al-Salam, 1351 A.H.

Ibn Hājib. Muktasar muntahā al-usūli. Cairo: Būlāq, 1317 A.H.

Ibn Khaldūn, Kitāb al-^cibar. Vol. VII. Beirut: Maktabat al-madrasa wa dar al-Kitāb al-Lubnānī, 1959.

Muqaddima. Cairo: Amiriya, 1320 A.H.

<u>Shifā al-sā'il li tahdhīb al-masā'il</u>. Edited by Muḥammad b. Ţāvīt al-Tanjī, Ankara: Ankara University, 1958.

<u>Šifa as-sā'il litahzīb al-masā'il</u>. Edited and translated by I. A. Khalifé. Beyrouth: Imprimerie Catholique, 1958.

Ibn al-Khaţib. Asmāl al-aslām fi man būyisa qabl al-lhtilām min mulūk al-islām. Edited by Lévi-Provençal. /Histoire de l'Espagne Musulmane/. Beyruth: Dār al-Makshūf, 1956.

> <u>— Al-Ihāța fi akhbār al-gharnāța</u>. Cairo: Maţba^ca Mawsū^cāt, 1319 A.H.

_____ Khaţrat al-ţayf wa riḥlat al-shitā' wa'l-şayf. Edited by Abbādī in Mushāhadāt... see Abbādī.

Al-Katībat al-kāmina fī man laqaynāhu bi al-andalus min shu'arā' mi'at al-thāmina. Edited by Iḥsān ʿAbbās. Bayrut: Dār al-Thāqāfa, 1963. Micyār al-ikhtibār fī dhikr al-maʿāhid wa'l diyār. Edited by ʿAbbādī in Mushāhadāt...see ʿAbbādī.

_____ Mufākharāt mālaqa wa salā. Edited by ^sAbbādī in <u>Mushāhadāt</u>... See ^sAbbādi.

Ibn Maryam. Al-Bustān fī dhikr al-awliyā' wa^vulamā' bi Tilimsān. Edited by Ibn Cheneb. Algiers: Thaʿālibiyya, 1326 A.H.

Ibn Qayyim. I'lām al-muwaqqi'in. Cairo: Sa'āda, 1955.

Ibn Saʿid. Al-Mughrib fi hulā al-maghrib. Vol. I. Cairo: Dār al-Maʿārif, 1953.

- Ibn Taymiyya. Qāʿida fī'l muʿjizāt wa'l karāmāt was anwāʿ khawāriq al-ʿādāt, in Majmūʿ al -rasā ʾil wa'l-masāʾil. Vol. V. Cairo: Maṭbaʿa Manār, 1349 A.H.
- Isnawi Jamāl al-Din. Nihāyat al-suval. Printed on the margin of Ibn Amir al-Hājj. Al-Taqrir wa'l_tahbir. Cairo: Būlāq, 1317 A.H.
- Kalābādhī, Abū Bakr. <u>Al-Taʿarruf li madhhab ahl al-taṣawwuf</u>. Cairo: ^clsā al-Bābī, 1960.
- Kubrā, Najm al-Din. Fawā'iḥ al-jamāl. Edited by Fritz Meier. Weisbaden: Steiner, 1957.
- Maḥbūbī, Ṣadr al-Sharī'a. Sharḥ al-tawdīḥ wa l-tanqīḥ. Instānbul: Maṭba^ca Bosanawi, 1304 A.H.
- Malik. Al-Mudawwanat al-kubra. Vol. I. Baghdad: Muthanna, 1970.

Maqqari, Abū'l Abbās. Azhār al-riyād fi akhbār lyād. Cairo: Lajnat al-Talif, 1939.

Hamīd. Cairo: Maṭḥal-ṭīb. Edited by Muḥammad Muḥy al-Dīn ʿAbd al-Ḥamīd. Cairo: Maṭbaʿa Saʿādu, 1949.

- Murtada, Muhammad al-Zabidi. <u>Ithaf sadat al-muttadin</u>. Cairo: Matba^sa Maymaniyya, 1893.
- Nubāhi, Qādi Abū'l Hasan. <u>Al-Marqabat al-'ulya fi man yastahiqq al-qadā' wa'l-</u> futyč. Edited by Levi-Provençal. Cairo: Dār al-Kātib al-miṣri, 1948.
- Qarāfī, Shihāb al-Dīn. Al-Furūq. Cairo: Dār lhyā Kutub al- Arabiyya, 1346 A.H.

Al-Ihkām fī tamyīz al-fatāwā 'an al-ahkām wa tasarrufāt al-qādī wa'l Imām. Edited by 'Abd al-Fattāh Abū Ghadda. Halab: Matba'a Islāmiyya. 1967.

Qushayri, Abū'l Qāsim. Al-Risāla fi 'ilm al-taşawwuf. Cairo: Muhammad 'Ali, 1948.

Rāzī, Fakhr al-Dīn. Al-Maḥṣūl fī uṣūl al-fiķh. Ms. Yale University. Nemoy, A-1039 (L-643).

Salāwī, Ahmad b. Khālid. Al-Istiqṣā'fī akhbār duwal al-maghrib al-aqṣā. Cairo: Dār al-Kitāb, 1954-56.

A 13. 15 Sarrāj, Abū Naṣr. Kitāb al-luma^c fī | Taṣawwuf. Edited and commented by R.A. Nicholson. London: Luzac, 1964.

Ţ

Subkī, Tāj al-Dīn. Jam^c al-jawāmi^c. Printed in ⁶Abd al-Raḥmān Bannānī, Hāshiya... ^calā matn Jam^c al-jawāmi^c. Cairo: Mustafā Bābī, 1937.

Taftāzānī, Saʿd al-Din. Sharḥ al-tawḍiḥ wa l-tanqih. See Maḥbūbi.

Ţawfi, Najm al-Din. Sharh al-arba^cin. Printed as appendix to Mustafā Zayd, <u>Al-Maslaha fi I tashri^c al-islāmi wa Najm al-Din al-Tūfi</u>. Cairo: Dār al-Fikr al-^cArabi, 1954.

Wansharīsī, Abū'l ʿAbbās Aḥmad. <u>Al-Misyār al-mughrib wa'l-jāmis al-mufarrab an</u> fatāwā ʿulamā' ifriqiya wa'l-andalus wa'l-maghrib. Fās: n.n., 1314/1914.

III - SECONDARY SOURCES

- ^SAbbādī, Ahmad Muktar. (ed.) <u>Mushāhodāt Lisān al-Dīn Ibn al-Khatīb fī bilād</u> al-maghrib wa'l-andalus. Iskandariyya: Jāmi^sa Iskandariyya, 1958.
- Abd Allah, 'Umar. Sullam al-wuşūl li 'ilm al-uşūl. Cairo: Dar al-Ma'arif, 1956.
- Abu'l Nasr, Jamil, M. <u>A History of the Maghrib</u>. Cambridge: Cambridge Univ. Press, 1971.
- Abū Sizna, Ahmad. Madkhal al-fiqh al-Islāmi. Cairo: Matba^cat Lajnat al-Bayān al-^cArabī, 1965.
- Abū Zahra, Muḥammad. Mālik. Cairo: Maktabat al-Ānjalu al-miṣriyya, [1952].
- Aghnides, N. P. Mohammedan Theories of Finance. New York: Columbia University; Longmans, 1916.
- Anderson, J. N. D. Islamic Law in the Modern World. New York: New York University Press, 1959.
- Baghdādī, Ismā'īl Pāsha. Idaņ al-maknūn. [Anqara]: Wakālt al-Ma'ārif, 1945.
- Bercher, Léon. (Ed. Transl. Comm.) Ibn 'Âçim. <u>Al- 'Âçimiyya ou tuh'fat al-</u> <u>h'ukkâm fî nukat al-uqoûd wa'l ah'kâm</u>. Alger: Institut d'études orientales, 1958.
- Bertrand, Louis and Charles Petrie. The History of Spain. Translated by W.B. Wells. New York: Appleton, 1934.

. - -

Brockelmann, Carl. <u>Geschichte der arabischen Lit eratur</u>. Leiden: Brill, 1943–49.

____ Zweiter Supplementband. Leiden: Brill, 1938.

- Būțī, Sa'īd Ramadān. <u>D</u>awābi<u>t</u> al-maslaha fī al-Sharī'a al-Islāmiyya</u>. Damascus: Umawiyya, 1966-67.
- Cairns, Huntington. Legal Philosophy from Plato to Hegel. Baltimore: John Hopkins, 1967.
- Codera, D. F., and Zaidín. <u>Tratado de numismática arábigo-española</u>. Madrid: Murillo, 1879.
- Coulson, N. J. <u>A History of Islamic Law</u>. Edinburgh: Edinburgh University Press, 1964.
- Darāz, Muḥammad ʿAbd Allāh. Introduction to Shāṭibī's al-Muwāfaqāt. Caìro: Mustafā Muḥammad; Raḥmāniyya, n.d.
- Dawālibī, Maʿrūf. <u>Al-Madkhal ilā ʿilm uṣūl al-fiqh</u>. Beirut: Dār al-ʿiim l'il Malaʾīn, 1965.
- De Slane. Histoire des Berbères. Vol. II. / Translation of Ibn Khaldūn's Kitāb al-'ibar 7. Paris: Geuthner, 1927.
- Dozy, R.P.A. Supplément aux dictionnaires arabes. Vol. II. Paris: Maisonneuve, 1967.

- Histoire des musulmans d'Espagne. Leyde: Brill, 1932.

Etzioni, Amitai. Social Change, Sources, Patterns and Consequences. New York: Basic Books, 1964.

Fihris al-maktabät al-azhariyya. Cairo: Azhar, 1946.

- Friedmann, W. Law in a Changing Society. London: Pelican, 1964.
- Fuller, Lone L. The Morality of Law. New Haven: Yale, 1970.
- Gayangos, Pascual de. The History of the Mohammedan Dynasties in Spain. / Translation of Maggari's Nafh al-tib/. Johnson Reprint, 1964.
- Gibb, H. A. R. and H. Bowen. Islamic Society and the West. Vol. 1, Part 2. Toronto: Oxford, 1957.

Gibb, H. A. R. Mohammedanism. New York: Oxford, 1962.

Studies on the Civilization of Islam. Boston: Beacon, 1968.

Travels in Asia and Africa 1325-1354 /by/ Ibn Baţţūţa. London: Kegan

Goldziher, 1. Muslim Studies. /Transl. S. M. Stern. From <u>Muhammedanische</u> <u>Studien</u> 7. Vol. II. London: George Allen and Unwin, 1971.

______ <u>Streitschrift des Gazālīgegen die Batinijja-Sekte</u>. Leiden: Brill, 1916.

Hajawi, Muhammad b. Hasan. Al-Fikr al-sāmi fi tārikh al-fiqh al-islāmi. First and Second Volumes; Ribāţ: Idārat al-ma<ārif, 1345 A.H.; Third Vol.; Tūnis: Nahḍa, n.d.,; Fourth Vol.; Fās: Jadīda, n.d. / 1931/.

Hart, H. L. A. The Concept of Law. Oxford: Clarendon, 1965.

- Law, Liberty and Morality. New York: Vintage, 1963.

- Hazard, H. W. <u>The Numismatic History of Late Medieval North Africa</u>. New York: American Numismatics Society, 1952.
- Hourani, Albert. Arabic Thought in the Liberal Age. London: Oxford, 1962.
- Hourani, G. F. Islamic Rationalism, The Ethics of ^CAbd al-Jabbar. Oxford: Clarendon, 1971.
- Hurgronje, C. S. Selected Works of C. Snouck Hurgron, and Translated by G. H. Bousquet and J. Schacht. Leiden: Brill, 1957.
- lbn ^cĀshūr, Fadil. <u>A<lām al-fikr al-Islāmī fī tārikh al-maghrib al-^carabī</u>. Tunis: Najāh, n.d.
- Imamuddin. Political History of Spain. Dacca: Najma, 1969.
- ^sInān, Muḥammad ^cAbd Allāh. <u>Al-'Āthār al-andalusiya al-bāqiy a fī'lsbāniy a</u> <u>wa'l Burtughāl</u>. Cairo: Shirka Musāhama miṣriyya, 1956.

______Nihāyat al-Andalus. Cairo: Matbasa al-Misr, 1958.

Iqbāl, Muhammad. Reconstruction of Religious Thought in Islam. Lahore: Ashraf, 1965.

Isfahānī, Rāghib. Al-Mufradāt fī gharīb al-Qur'ān. Karachi: Tijārat Kutub, 1961.

I

- lzutsu, T. <u>God and Man in the Koran:</u> Semantics of the Koranic Weltanschauung. Tokyo: Keio Institute of Cultural and Linguistic Studies, 1964.
- Juynboll, T. W. Handbuch des İslāmischen Gesetzes. Leiden: Brill, 1910.
- Kaḥḥāla, ʿUmar Riḍā. Muʿjam al-muʾallifin. Damascus: Maktaba ʿArabiyya, 1957.
- Kerr, Malcolm H. Islamic Reform, The Political and Legal Theories of Muhammad Abduh and Rashid Ridā. California: University of California Press, 1966.
- Khallāf, ^SAbd al-Wahhāb. <u>Maşādir al-tashrīš al-Islāmi fi mā lā nassa</u> fih. Cairo: Dār al-Kutub al-Srabi, 1955.
- Khuḍri, Muḥammad. Usul al-fiqh. Cairo: Maṯbaʿa al-Istiqāma, 1938; Muṣṭafā Muḥammad, 1933.
- Lacerte, Henry. The Nature of Canon Law According to Suarez. Ottawa: University of Ottawa, 1964.
- Lammens, H. Islam, Beliefs and Institutions. London: Methuen, 1929.
- Lane, E. W. An Arabic-English Lexicon. London: William Norgate, 1863–93.
- Laoust, Henri. Contribution à une étude de la méthodolgie canonique de Taki-d-Din Ahmad b. Taimiya. Cairo: Imprimerie de l'Institut français d'archéologie orientale, 1939.

Essai sur les doctrine sociales et politiques de Taki-d-Din Ahmad b. Taimiya. Cairo: Imprimerie de l'institut français d'archéologie orientale, 1939.

Ele Califat dans la doctrine de Rashid Ridā. Beirut: Institut Français à Damas, 1938.

Lavoix, M. H. <u>Catalogue des monnaies musulmanes de la Bibliothèque Nationale</u>. Paris: Imprimerie Nationale, 1891.

3

- Levi-Provençal, E. Histoire de l'Espagne musulmane. Paris: Maisonneuve, 1950.
 - Inscriptions arabes d'espagne. Leyde: Brill, 1931.

Levy, R. The Social Structure of Islam. Cambridge: Cambridge University Press, 1957.

Linant de Bellefonds, Y. <u>Traité de droit musulman comparé</u>. Vol.1. Paris: Mouton, 1965.

1

Mahdi, Muhsin. Ibn Khaldun's Philosophy of History. Chicago: Phoenix, 1964.

438

Muqaddima fi ihyā ʿulūm al-sharīʿa. Beirut: Dār al-ʿllm, 1962.

Y.

- Maḥmūd, A. Ḥalīm. <u>Al-Madrasat al-Shādhiliyya wa</u> imāmuhā Abū'l Ḥasan al-Shādhilī. Cairo: Dār al-Kutub al-Ḥaditha, 1387 A. H.
- Makhluf, Muhammad. Shajarat al-nur al-zakiyya. Cairo: Salafiyya, 1930-31.
- Manşūr, ^SAbd al-Hafīz. <u>Fihris makhtūtāt al-maktabat al-Ahmadiyya bi Tūnis</u>. Bayrūt: Dār al-Fath, 1969.
- Mawdūdī, Abu'l A^clā. <u>The Islamic Law and Constitution</u>. Edited and Translated by K. Ahmad. Lahore: Islamic Publications, 1960.
- Monroe, J. T. Islam and The Arabs in Spanish Scholarship. Leiden: Brill, 1970.
- Morand, M. Introduction à l'étude du droit musulman algérien. Alger: J. Carbonel; Ancien Maison, 1921.
- Mūsā, M. Yūsuf, <u>Al-Madkhal li dirāsāt al-fiqh al-Islāmi</u>. Cairo: Dār al-Fikr al-Arabi, 1961.
- Nadwi, Ridwan Ali. <u>Al- Izz bin Abd al-Salām</u>. Damascus: Dār al-fikr, 1960.
- Nwiya, Paul. Ibn Abbād de Ronda. Beyrouth: Imprimerie Catholique, 1961.
- Ostrorog, Leon. The Angora Reform. London: University of London, 1927.

Rahman, F. Islam. London: Weidenfeld, 1966.

- Islamic Methodology in History. Karachi: Central Institute of Islamic Research, 1965.
- Rida, Rashid. Tarikh al-Ustadh al-Imam al-Shaykh Muhammad Abduh. Cairo: Dar al-Manar, 1350 A. H.

____ Yusr al-Islām. Cairo: Nahḍa, 1956.

Robertson, A. J. The Principles of Mohammedan Law. Rangoon: Myles Standish, 1911.

Rosenthal, E. I. J. Political Thought in Medieval Islam. Cambridge: Cambridge University Press, 1962.

Saʿīdī, ʿAbd al-Mutaʿāl. Al-Mujaddidūn fīʿˈl Islām. Cairo: Ādāb, n.d.

17

- Santillana, D. Istituzioni di dritto musulmano malichita. Roma: Istituto per l'Oriente, 1938.
- Sarkīs, Yūsuf I. Mu<jam al-matbū^sāt al-^sarabiyya wa'l mu^sarraba. Cairo: Sarkīs, 1928.

Schacht, J. Introduction to Islamic Law. Oxford: Clarendon, 1966.

The Origins of Muhammadan Jurisprudence. Oxford: Clarendon, 1959.

Seco de Lucena, Luis. Documentos Arábigo-Granadinos. Madrid: Institute des Estudios Islamicos, 1961.

Shalabi, Mustafa. Taqlil al-ahkām ^card wa tahlil li tariqat al-tahlil wa tatawwurātihā fi ^cuşūr al-ijtihād wa altaqlid. Cairo: Azhar, 1949.

Sharqāwi, E. Religion and Philosophy in the Thought of Fakhr al-Dīn al-Rāzī: The Problem of God's Existence. Unpublished Thesis Presented to McGill University, Montreal, 1970.

Smith, Margaret. Al-Ghazāli, The Mystic. London: Luzac, 1944.

- Toledano, H. The Chapter on Marriage from Sijilmāsī's Al-^cAmal al-Mutlaq: <u>A study in Moroccan Judicial Practice</u>. Unpublished Ph.D. Thesis, Columbia University, 1969.
- Tonki, Maḥmūd Ḥasan. <u>Muʿjam al-muṣannifīn</u>. Beirut: Wazankoghrāf Ṭabbāra, 1344 A.H.

Trimingham, J. Spencer. The Sufi Orders in Islam. Oxford: Clarendon, 1971.

Udovitch, Abraham L. Partnership and Profit in Medieval Islam. Princeton: Princeton University Press, 1970.

Van Ess, Josef. Die Erkenntnislehre des Adudaddin al-Ici: Übersetzung und Kommentar des ersten Buches Seiner Mawäqif. Wiesbaden: Steiner, 1966.

Van Kleffens, E. N. <u>Hispanic Law Until the End of the Middle Ages</u>. Edinburgh: University Press, 1968.

Voorhoeve, P. Handlist of Arabic Manuscripts: Library of University Leiden. Lugundi: Batavorum, 1957.

Zarqā', Mustafā Ahmad. <u>Al-Madkhal al-fiqhī al-ʿāmm</u>. Dimashq: Matba^ca Jāmiʿa, 1961.

4

Zayd, Muştafa. <u>Al-Maşlaha fi al-tashri' al-Islāmi wa Najm al-Din Ţūfi</u>. Cairo: Dar al-Fikr al-Arabi, 1954.

Zirkili, Khayr al-Din. <u>A'lām</u>. Cairo: Mu'allif, 1954.

1

III (b) - ARTICLES

- ^CAbduh, Mufti Muhammad. "Taqrir mufti al-diyār al-mişriyya fi işlāh al-mahākim al-shar ^ciyya", <u>Al-Manār</u>, 11 (1899). Pp. 577-82, 593-8, 609-12, 625-28, 648-51, 663-66, 759-764.
- Arnaldez, R. "Ibn Rushd", <u>The Encyclopaedia of Islam</u> (2nd ed.) III, 909–920. Leiden: Brill, 1971.
- Bastieva, S. M. "Ibn Khaldūn et son milieu social", Atti del Terzo Congresso di Studi Arabi, Islamici. Napoli, 1967. Pp. 133–144.
- Bel, A. "Abd al-Wāḥid al-Rashīd", The Encyclopaedia of Islam" (Ist ed.), 1, 66. Leyden: Brill, 1913.
- "Almohades", <u>The Encyclopaedia of Islam</u> (1st ed.), 314–318. Leyden: Brill, 1913.
- Berque, J. "Amal", The Encyclopaedia of Islam (2nd ed.), 1, 427-28. Leiden: Brill, 1960.
- Brunschvig, R. "Logic and Law in Classical Islam" Logic in Classical Islamic Culture. Von Grunebaum, editor. Wiesbaden: Harrassowitz, 1970. Pp. 9–20.

_____"Polémique≤médiévales autour du rite de Mālik", <u>Al-Andalus</u>, XV (1950), 377-435.

- Cahen, Claude. "Body Politic", Unity and Variety in Muslim Civilization, Von Grunebaum, editor. Chicago: University of Chicago Press, 1955. Pp. 132–158.
- Chehata, Chafik. "L'équité en tant que sources du droit hanafite", <u>Studia Islamica</u>, XXV (1966), 123–138.

Cheneb, M. "Ahmed Baba", <u>The Encyclopaedia of Islam</u> (1st ed.), 1, 191-2. Leyden: Brill, 1913. Cherbonneau, M. A. "Lettre à M. Defrémery sur Ahmad Bābā le tombouctien, auteur du Tekmilet ed-Dibădj", Journal Asiatique. 5e Série, 1 (1853), 93-100.

Coulson, N. J. "Doctrine and Practice in Islamic Law", Bulletin of School of Oriental and African Studies, XVIII/2 (1956), 211–226.

De la Granja, F. "Ibn Zamrak", The Encyclopaedia of Islam. (2nd ed.), III, 972-973. Leiden: Brill, 1971.

- Demombynes:, Gaudefroy. "Histoire des Benou'l Ahmar," Translation from Ibn Khaldūn's Kitāb al ^elbar, Journal Asiatique, 9th series, Vol. XII (Paris, 1898), 309–340; 407–462.
- Fagnan, E. "Les tabaqāt Mālikites", <u>Homenaje à D. F. Codera</u>, D. F. Saavedra, (Editor). Zaragoza: Mariano Escar, 1904. Pp. 105–113.
- Gibb, H. A. R. "Constitutional Organization", <u>Law in the Middle East</u>, M. Khadduri and H. J. Liebesny, editors. Washington: Middle East Institute, 1955. Pp. 3–27.

Goitein, S. D. "The Birth-Hour of Muslim Law?" Muslim World. L (1960), 23-29.

Goldziher, I. "Das Prinzip des Istishāb in der muhammedanischen Gesetzwissenschaft", <u>Wiener Zeitschrift für die Kunde des Morgenlandes</u>. I (1887), 128–236. <u>Summary in French by G. H. Bousquet</u>. Arabica, VII (1960), 12–15.

> "The Spanish Arabs and Islam", Muslim World. Translated by Joseph de Semogyi. LIII (1963), 5–18; 91–105; 178–184; 281–286.

- Gyekye, Kwame. "The Terms 'Prima Intentio' and 'Secunda Intentio' in Arabic Logic", Speculum, XLVI, 1 (1971), 32–38.
- Heckscher, E. F. "Mercantilism", in H. W. Spiegel, (ed.), <u>The Development of</u> Economic Thought. New York: Science Books, 1966. Pp.32-41.
- Hourani, G. F. "Two Theories of Value in Medieval Islam", <u>Muslim World</u>, L (1960) 269-278.
- Hunwick, J.O. "Ahmad Bābā and the Moroccan Invasion of the Sudan (1591)", Journal of Historical Society of Nigeria, II (3, 1962), 311-28.

"A New Source for the Biography of Ahmad Bābā al-Tinbukti" (1556-1627)", Bulletin of the School of Oriental and African Studies, XXVII (1964) 568-593.

- Idris, Roger. "Reflexions sur le Mālikism^esous les Umayyades d'Espagne", Atti del Terzo Congresso di Studi Arabi¹Islamici. Napoli, 1967. Pp.397–414.
- Juynboll, Th. W. "Hadith", Shorter Encyclopaedia of Islam. Leiden: Brill, 1961. Pp. 116–21.
- Kerr, M. H. "Rashid Rida and Legal Reform", Muslim World, L (1960), 99-108; 170-181.
- Khadduri, M. "From Religious to National Law", Modernization of the Arab World, J. J. Thompson and R. D. Reischauer, editors. Princeton: Nostrand, 1966. Pp.37-51.
- Kramers, J. H. "Droit de l'Islam et droit islamique", Analecta Orientalia, Posthumous Writings and Selected Minor Works of J. H. Kramers. Vol. II. Leiden: Brill, 1956. Pp.63–74.
- Krenkow, F. "Al-Shātibi" [Abu'l Qāsim], <u>The Encyclopaedia of Islam</u> (1st ed.), IV, 337-8. Leiden: Brill, 1934.
- Levi-Provençal, E. "Ahmad Bābā", <u>The Encyclopaedia of Islam</u> (2nd ed.), 1, 279-80. Leiden: Brill, 1960.

"Al-Andalus", <u>The Encyclopaedia of Islam</u> (2nd ed.), 1, 486-92. Leiden: Brill, 1960.

"Nașrids", <u>The Encyclopaedia of Islam</u> (1st ed.), 111, 878–81. Leiden: Brill, 1936.

"Shāțiba", <u>The Encyclopaedia of Islam</u> (1st ed.), IV, 337. Leiden: Brill, 1936.

Liebesny, H. J. "Religious Law and Westernization in the Moslem Near East", American Journal of Comparative Law, 11 (1953), 492–504.

Linant de Bellefonds, Y. "Immutabilité du droit musulman et réformes législatives en Egypte", <u>Revue international de droit comparé</u>, VII (1955), 5–34.

Lopez-Ortiz, J. "Fatwas Granadinas de los siglos XIV y XV", <u>Al-Andalus</u>, VI (1941) 73–127.

A

MacDonald, D. B. "Hizb", <u>The Encyclopaedia of Islam</u> (New Edition). III, 513-4. Leiden: Brill, 1971.

Mackeen, A. M. M. "The Early History of Sufism in the Maghrib prior to Al-Shādhili (d. 650/1256)", Journal of the American Oriental Society, XCI (1971), 398–408. Mahmaṣāni, Şubhi. "Transactions in the Shari^ca", Law in the Middle East. M. Khadduri and H. J. Liebesny, editors. Washington: Middle East Institute, 1955. Pp.179–202.

Â.,

5

Makdisi, G. "Remarks on Traditionalism in Islamic Religious History", <u>The Con-</u> flict of Traditionalism and Modernism in the Muslim Middle East. Carl Leiden, editor. Austin: University of Texas, 1966. Pp. 77–87.

Marcais, G. "Note sur les ribâts en Berbérie", Mélange Réné Basset. II. Paris: Edition Ernest Leroux, 1925. Pp.395–430.

"Ribāț", Encyclopaedia of Islam (First Edition), 111, 1150–52. Leiden: Brill, 1936.

Margoliouth, D. S. "Recent Arabic Literature", Journal of Royal Asiaric Society, London, 1916. Pp. 397–98.

Miranda, A. Huici. "The Iberian Peninsula and Sicily", <u>The Cambridge History</u> of Islam, P. M. Holt and others, editors. II. Cambridge: Cambridge University Press, 1970. Pp. 406–439.

"Gharnāța", <u>Encyclopaedia of Islam (New Edition)</u>, 11, 1012–14. Leiden: Brill, 1965.

Nakamura, Hajime. "The Indian and Buddhist Concept of Law". <u>Religious</u> <u>Pluralism and World Community</u>. E. J. Jurji, editor. Leiden: Brill, 1969. <u>Pp.131-174</u>.

Niyazi, Nadhīr. "Istişlāḥ", Urdū dā²ira ma cārif Islāmiyya, 11, 586-91. Lahore: Danishgāh Panjāb, 1965.

- Paret, Rudi. "Istiķsān and Istişlāķ", Shorter Encyclopaedia of Islam. Leiden: Brill, 1961. Pp.184–186.
- Prieto y vives, Antonio. "Indicacion del valor en las monedas Arábigo-Espa ñolas". Homenaje à D. F. Codera, D. F. Saavedra, editor. Zaragoza: Mariano Escar, 1904. Pp. 5/3-522.

Monès, Hussain. "Le role des hommes de religion dans l'histoire de L'Espagne musulmane jusqu'à la fin du Califat", <u>Studia Islamica</u>, Vol. XX (1964), 47–88.

Nadwi, R. A. "Two Great Contemporaries of Thirteenth Century A. D. Sultan al- 'ulama'al-'lzz ibn al-Salām and Ibn 'Arabi", <u>Islamic Culture</u>, XLV (1971), 193-201.

Renaud, H. P. J. "Un médecin du royaume de Granade: Muḥammad as-Ṣaqūrī", Hespéris, XXX¹ (1946), 31-64.

> "Deux Ouvrages perdus d'Ibn al-Ḫaṭīb: Identifiés dans des manuscrits de Fès – Conclusion sur Ibn al-Hatib médecin", <u>Héspéris</u>, XXX, (1946), 214-225.

Schacht, J. "Fikh", The Encyclopaedia of Islam (2nd ed.), 11, 886–91. Leiden: Brill, 1965.

> "Islamic Law", Encyclopaedia of Social Sciences, VIII, 344–49. New York: MacMillan, 1932.

> "Muhammad Abduh". Shorter Encyclopaedia of Islam. Leiden: Brill, 1961. Pp. 405–407.

"The Present State of the Studies in Islamic Law", Atti del Terzo Congresso di Studi Arabi e Islamici, Napoli, 1967. Pp.621-622.

"Theology and Law in Islam", in Theology and Law in Islam. Von Grunebaum, editor. Wiesbaden: Harrassowitz, 1971. Pp.3–23.

"Classicisme traditionalisme et ankylose dans la loi religieuse de l'Islam", Classicisme et déclin culturel dans l'histoire de l'Islam, R. Brunschvig, editor. Paris: Besson et Chantemerle, 1957. Pp. 141–166.

- Seybold, G.F. "Granada", The Encyclopaedia of Islam (1st ed.), 11, 175-7. Leiden: Brill, 1927.
- Shabāna, Muḥammad Kamāl. "Al-ḥāla al-iqtiṣādiyya bi'il-Andalus khilāl al-qarn al thāmin al-hijrī," Al-Baḥth al-ʿIlmī, III (1966), 136-150.

"Shuyūkh Ghuzāt al-maghāriba fi'l-Andalus kamā arrakha lahum Ibn al-Khatīb fi al-lhāta", Al-Baḥth al- [<]Ilmī, V (1968), 124-138.

Talbi, M. "Ibn Khaldūn", The Encyclopaedia of Islam (2nd ed.), III, 825-31. Leiden: Brill, 1971.

- Toy, C. H. "The Semetic Conception of Absolute Law", in Orientalische Studien <u>Theodor Noldeke</u>, Carl Bezold, editor. II. Gieszen: Ricker, 1906. <u>Pp. 797-804</u>.
- Turki, Abdelmagid. "Lisān al-Dīn Ibn al-Hatīb (713-76/1313-74), juriste d'après son oeuvre inédite: Mutla al-tarīga fīdamm al-wațiga [Muthlā al-țarīga fidhamm al-wathīga], Arabica, XVI (1969), 155-211.

<

Tyan, E. "Méthodologie et sources du droit en Islam", <u>Studia Islamica</u>, X (1959), 79–109.

Æ.

. <u>.</u> .

1

1

-**-**-

Vesey-Fitzgerald, S. G. "Nature and Sources of Shari^sa", Law in the Middle East". M. Khadduri and H. J. Liebesny, editors. Washington: Middle East Institute, 1955. Pp.85–112.

Von Grunebaum, G. E. "Concept and Function of Reason in Islamic Ethics", Oriens, XV (1962 1-17.

.

...



















